Universitat Politècnica de Catalunya Programa de Doctorat de Matemàtica Aplicada Departament de Matemàtiques

Elliptic problems: regularity of stable solutions and a nonlocal Weierstrass extremal field theory

by

Iñigo Urtiaga Erneta

PhD dissertation Adviser: Xavier Cabré Barcelona, May 2023

"That we are capable only of being what we are remains our unforgivable sin." G. Wolfe

Acknowledgments

First and foremost, I owe the greater part of this work to the invaluable guidance of my adviser Xavier Cabré, who showed me the many facets of researching in Mathematics. His unbounded intellectual curiosity and the remarkable clarity with which he elucidates complex ideas have been, and will continue to be, a major source of inspiration.

While the incursion of the Covid pandemic hindered my early progress, it was a great pleasure to have had the support of my collaborator Juan Carlos Felipe-Navarro. Through our association, we discovered a beautiful theorem and also found time to enjoy some lively games of $p\acute{a}del$ together.

I am also grateful to the other members of our work group: Joan de Solà-Morales, Albert Mas, Gyula Csató, Tomás Sanz-Perela, Renzo Bruera, and Olli Saari. They contributed to creating a pleasant environment to do research, and I greatly valued their counsel and kind encouragement over the years.

I owe my interest in higher Mathematics to Daniel Lasaosa. His enthusiasm during the lectures he delivered all those years ago still resonates through everything I accomplish.

My heartfelt thanks go out to the many people who assisted and encouraged me throughout my PhD studies. This journey was made the more enjoyable by my colleagues Bruno, Jaime, Mar, Román, Jose, Laura, Dídac, and Luan. My fellow *heralds* in Iruña, especially my esteemed friend Eñaut, helped me keep my sanity while the world fell apart. I am indebted to my family for their unconditional support, particularly to my parents Aurora and Mikel, and to my siblings Maddi and Jon. Last but not least, I wish to thank my beloved María for her unwavering support and precious help during these trying times.

Funding

This work has received financial support from the following:

- MINECO fellowship MDM-2014-0445-18-1 through the María de Maeztu Program for Units of Excellence in R&D
- Spanish grants MTM2017-84214-C2-1-P and PID2021-123903NB-I00 funded by MCIN/AEI/10.13039/501100011033 and by ERDF "A way of making Europe"
- Catalan project 2021 SGR 00087

Summary

This PhD dissertation deals with qualitative questions from the theory of elliptic Partial Differential Equations (PDE) and integro-differential equations. We are primarily interested in a distinguished class of solutions satisfying appropriate minimality conditions.

The first part of the thesis provides a regularity theory for stable solutions to semilinear problems involving variable coefficients. Here, stability refers to the nonnegativity of the principal eigenvalue of the linearized equation. For variational problems, this amounts to the nonnegativity of the second variation, a necessary condition for minimality. Our main achievement is to show the boundedness of stable solutions in $C^{1,1}$ domains in the optimal range of dimensions $n \leq 9$. This result is new even for the Laplacian, for which a C^3 assumption on the domain was needed.

The second part furnishes natural sufficient conditions for the minimality of critical points in a general nonlocal framework. Namely, we construct a calibration for nonlocal energy functionals, under the assumption that the critical point is embedded in a family of sub/supersolutions whose graphs produce a foliation. As a consequence, we deduce that the solution is a minimizer with respect to competitors taking values in the foliated region. Our result extends, for the first time, the classical Weierstrass extremal field theory in the Calculus of Variations to a nonlocal setting. To find a calibration for the most basic fractional functional, the Gagliardo-Sobolev seminorm, was an important open problem that we have solved.

The contents of the thesis are based off of the following articles and preprints:

[52] I. U. Erneta, Stable solutions to semilinear elliptic equations for operators with variable coefficients, Commun. Pure Appl. Anal. **22** (2023), 530–571.

[51] I. U. Erneta, *Energy estimate up to the boundary for stable solutions to semilinear elliptic problems*, preprint available at arXiv:2305.07058.

[50] I. U. Erneta, Boundary Hölder continuity of stable solutions to semilinear elliptic problems in $C^{1,1}$ domains, preprint available at arXiv:2305.07062.

[23] X. Cabré, I. U. Erneta, J.C. Felipe-Navarro, A Weierstrass extremal field theory for the fractional Laplacian, to appear in Adv. Calc. Var., preprint available at arXiv:2211.16536.

[22] X. Cabré, I. U. Erneta, J.C. Felipe-Navarro, Null-Lagrangians and Calibrations for general nonlocal functionals and an application to the viscosity theory, in preparation.

Contents

Acknowledgements	v
Summary vi	ii
Introduction	1
Stable solutions	2
Regularity of stable solutions	5
The Gelfand problem	6
Optimal dimension for the Laplacian	8
Operators with variable coefficients	1
Stable solutions to other problems	2
Minimizers and calibrations	2
The classical theory of necessary and sufficient conditions	3
Nonlocal problems	5
Results and outline of the thesis	7

I Regularity of stable solutions to semilinear elliptic problems 23

1	Inte	rior regularity	25
	1.1	Introduction	25
		1.1.1 The setting. Stability	26
		1.1.2 Main results	28
		1.1.3 Structure of the proof \ldots	29
		1.1.4 Outline of the chapter	32
	1.2	Preliminaries: Invariance under affine transformations	32
	1.3	Hessian and $W^{1,2}$ estimates $\ldots \ldots \ldots$	34
	1.4	The $W^{1,2+\gamma}$ estimate \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	44
	1.5	The weighted L^2 estimate for radial derivatives $\ldots \ldots \ldots \ldots \ldots \ldots$	46
	1.6	The radial derivative controls the function in L^1	53
	1.7	Proof of the C^{α} estimate	59
2	Ene	ergy estimate up to the boundary	65
	2.1	Introduction	65
		2.1.1 Main results	67
		2.1.2 Outline of the chapter	71
	2.2	Boundary Hessian estimates	71
	2.3	Boundary $W^{1,2+\gamma}$ estimate	83

3	Bou	ndary regularity in $C^{1,1}$ domains	89
	3.1	Introduction	89
		3.1.1 Main results	91
		3.1.3 Structure of the proof	95
		3.1.4 Outline of the chapter	97
	3.2	The boundary weighted L^2 estimate for radial derivatives	97
	3.3	In half-annuli the radial derivative controls the function in L^1	100
	3.4	Boundary C^{α} estimate	105
	3.5	Approximation and proof in $C^{1,1}$ domains $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	109
Α	Stal	oility is not equivalent to the integral inequality	117
В	A ti	ace inequality	121
С	Two	interpolation inequalities	123
D	Abs	orbing errors in larger balls	125
\mathbf{E}	App	proximating $C^{1,1}$ domains by smooth ones from the interior	127
\mathbf{F}	On	the uniqueness of stable solutions	131
тт	•		100
11	A	nonlocal Weierstrass extremal field theory	133
4	$\mathbf{A}_{4}\mathbf{V}$	Veierstrass extremal field theory for the fractional Laplacian	135_{125}
	4.1	4.1.1 The notion of calibration	136
		4.1.2 The classical theory of calibrations	137
		4.1.3 Nonlocal calibrations	140
		4.1.4 An application to monotone solutions	142
		4.1.5 Outline of the chapter \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	143
	4.2	The classical and nonlocal perimeters	143
	4.3	The theory of calibrations for local equations, and a novelty	146
	4.4	The calibration for the fractional Laplacian	153
5	Nul	-Lagrangians and calibrations for general nonlocal functionals and	101
	an a 51	Introduction	161
	0.1	511 Examples	162
		5.1.2 Calibrations and fields of extremals	163
		$5.1.3$ Main result \ldots	166
		5.1.4 An application to monotone solutions	168
		5.1.5 An application to the viscosity theory	169
		5.1.6 Outline of the chapter	171
	5.2	The calibration for general nonlocal functionals	172
	$5.2 \\ 5.3$	The calibration for general nonlocal functionals	$\begin{array}{c} 172 \\ 177 \end{array}$
	$5.2 \\ 5.3 \\ 5.4$	The calibration for general nonlocal functionals	172 177 178

	5.4.2Minimizers of fractional functionals are viscosity solutions15.4.3More general nonlocal functionals1	181 182
G	A calibration for the extension problem of the fractional Laplacian 1	185
н	Other candidates for the fractional calibration 1	187
Ι	Inimality via a comparison principle 1	89
J	The calibration for the nonlocal total variation 1	91
Bi	liography 1	195

Introduction

Our quantitative description of reality rests upon a collection of universal principles, expressed in mathematical language, from which all phenomena may be deduced. Such principles or laws often manifest in the form of equations, which are in turn satisfied by functions accounting for the state of the system under consideration.

Depending on the nature of the phenomenon being studied, different types of equations are of interest. Classically, Partial Differential Equations (PDE) have been successfully applied to model diverse phenomena arising from short-range interactions. These equations involve relations between partial derivatives of the solution. To compute such derivatives at a point, it is only necessary to know the function in a vicinity of the point in question, reflecting the *local* nature of PDEs. More recently, integro-differential equations have gained an increasing notoriety in applications. Such equations capture the effect of longrange interactions by integrating differences of the solution in the whole space. Notice that this operation takes into account the value of the solution everywhere and not just at a point. Thanks to this feature, they effectively model *nonlocal* phenomena.

Within these two broad categories of equations, *elliptic* ones hold a distinct position. This family of equations can be thought of as describing equilibrium configurations of evolution processes. As a closed system evolves in time, it reaches a steady state characterized by such an elliptic problem. These equations are typically nonlinear and solutions are by no means uniquely determined. In fact, the natural boundary (or exterior) value problems associated to such equations may admit multiple solutions or even infinitely many of them. Nevertheless, there is a special class of solutions comprising most physical situations of interest, the so called *stable solutions*. They are distinguished by the fulfillment of certain minimality conditions with respect to the problem in question (as we explain later). For instance, local minimizers of energy functionals are stable solutions of the Euler-Lagrange equations. In a sense, these solutions correspond to the observables of Nature, since they do not disappear under small perturbations of the parameters (hence their name "stable").

In this thesis, we are interested in various qualitative aspects of stable solutions to both PDEs and integro-differential equations. We focus primarily on the regularity and minimality properties of these solutions, which play crucial roles in many applications. Next, we briefly comment on our contributions to these fields.

First, we investigate the regularity of stable solutions to nonlinear elliptic PDEs. This turns out to be delicate question which usually depends on the dimension of the space, even for apparently simple semilinear problems. Our principal result shows the boundedness (and hence smoothness) of stable solutions to semilinear equations with variable coefficients, up to the optimal dimension 9. Such a result was only known for the Laplacian in C^3 domains. As a consequence of our analysis, we are able to weaken the regularity of the domain to a $C^{1,1}$ assumption, which is natural in the elliptic estimates for non-divergence form operators.

Secondly, we study sufficient conditions for minimality (and hence stability) of solutions

to equations associated to nonlinear elliptic functionals. Before our work, a theory of sufficient conditions was only available for local equations, in the form of the classical Weierstrass extremal field theory. There, the basic result states that if a solution can be embedded in a family of critical points (i.e., solutions) whose graphs produce a foliation, then it must be a minimizer with respect to competitors taking values in the foliated region. Our main result provides the analogue of this theorem in the general nonlocal framework. Namely, under the foliation assumption, we construct a *calibration* for nonlocal functionals. This calibration is a null-Lagrangian satisfying certain additional properties which lead directly to the minimality of the solution. In particular, our construction gives a calibration for the simplest nonlocal functional (the fractional Gagliardo-Sobolev seminorm corresponding to the fractional Laplacian) which was not known to exist prior to this work.

In the following, we further elaborate on the notions that we just mentioned.

Stable solutions

As mentioned above, we are interested in solutions that describe stable configurations. To better explain this concept, we consider the steady state reaction-diffusion equation

$$-\Delta u = f(u) \quad \text{in } \Omega, \tag{1}$$

where $\Omega \subset \mathbb{R}^n$ is a bounded domain and $f \colon \mathbb{R} \to \mathbb{R}$ is a smooth function. Later, we will further assume f to be nonnegative, nondecreasing, and convex.

There are several equivalent ways to define the stability of a solution to (1), as explained below. Some of these definitions are variational in nature, reflecting a minimality property of u with respect to some functional. Others are intrinsic to the linearization of (1) and can be generalized to other, non-variational situations. In the first part of the thesis, we will be particularly interested in this second case.

A first definition is based on the variational structure of (1). Notice that equation (1) is the first variation of the energy functional¹

$$\mathcal{E}_{1,F}(u) = \int_{\Omega} \left\{ \frac{1}{2} |\nabla u|^2 - F(u) \right\} \mathrm{d}x,^2 \tag{2}$$

where F is a primitive of the nonlinearity f. Computing the second variation at u

$$\frac{\mathrm{d}^2}{\mathrm{d}\varepsilon^2} \mathcal{E}_{1,F}(u+\varepsilon\xi)\Big|_{\varepsilon=0} = \int_{\Omega} \left\{ |\nabla\xi|^2 - f'(u)\xi^2 \right\} \mathrm{d}x,\tag{3}$$

where $\xi \in C_c^{\infty}(\Omega)$, we have the following:

Definition 1. A solution u of (1) is *stable* if the second variation (3) of $\mathcal{E}_{1,F}$ at u is nonnegative. Equivalently, the solution u is stable if

$$\int_{\Omega} f'(u)\xi^2 \,\mathrm{d}x \le \int_{\Omega} |\nabla\xi|^2 \,\mathrm{d}x \quad \text{for all } \xi \in C_c^{\infty}(\Omega).^3$$
(4)

¹Here, we call functionals "energies" following elliptic PDE terminology. In other contexts, to avoid ambiguity, it might be more appropriate to refer to them as *action functionals* instead.

²The subindex 1 is in contrast to the fractional parameters $s \in (0, 1)$ appearing later in the text. As mentioned below, $\mathcal{E}_{1,F}$ can be recovered as the formal limit of a family of nonlocal functionals $\mathcal{E}_{s,F}$ as $s \uparrow 1$.

 $^{^{3}}$ In the literature, solutions satisfying such a nonnegativity assumption have also been called *semistable*, while the term *stable* is reserved for the strict inequality. Instead, we will employ the terms *stable* and *strictly stable*, respectively.

Notice here that the test functions ξ vanish at the boundary $\partial\Omega$ and hence the competitors $u + \varepsilon \xi$ have the same boundary condition as u. It follows that stability is a necessary condition for u to be a minimizer with respect to small perturbations. However, being stable is not sufficient for minimality. For instance, if f is superlinear and u vanishes on $\partial\Omega$, then, taking rescalings λu as competitors, it is easy to check that $\mathcal{E}_{1,F}$ is unbounded from below. In this case, the functional does not admit global minimizers but it may have stable solutions; see [11, 24, 44, 64]. Moreover, stability does not even guarantee the local minimality of the solution, as we will see in the next paragraphs.

It is useful to note that the second variation can always be written as the quadratic form associated to the linearized equation. More precisely, integrating by parts we have

$$\frac{\mathrm{d}^2}{\mathrm{d}\varepsilon^2} \mathcal{E}_{1,F}(u+\varepsilon\xi)\Big|_{\varepsilon=0} = -\int_{\Omega} (J_u\xi)\,\xi\,\mathrm{d}x \quad \text{for all } \xi \in C_c^{\infty}(\Omega), \tag{5}$$

where J_u is the linearization of (1) at u, namely,

$$J_u\xi := \frac{\mathrm{d}}{\mathrm{d}\varepsilon} \Big(\Delta(u + \varepsilon\xi) + f(u + \varepsilon\xi) \Big) \Big|_{\varepsilon=0} = \Delta\xi + f'(u)\xi.$$

The operator J_u is also known as the *Jacobi operator* in the literature. We point out that the representation (5) holds for the second variation of general functionals, not only for the ones associated to semilinear equations.

With (5) at hand, we can give another definition of stability. Let $\mu_1[J_u, \Omega]$ denote the principal (smallest) eigenvalue of J_u in Ω with respect to homogeneous Dirichlet conditions. Here and throughout the text, we use the sign convention $J_u\xi = -\mu\xi$ for the eigenvalues of J_u . By the variational characterization of $\mu_1[J_u, \Omega]$, we have

$$\mu_1[J_u,\Omega] = \inf_{\xi \in C_c^{\infty}(\Omega), \xi \neq 0} \frac{-\int_{\Omega} (J_u\xi) \xi \, \mathrm{d}x}{\int_{\Omega} \xi^2 \, \mathrm{d}x}.$$

Hence, we see that Definition 1 above is equivalent to the following:

Definition 2. A solution u of (1) is *stable* if the principal eigenvalue of the linearized operator J_u is nonnegative, that is, if $\mu_1[J_u, \Omega] \ge 0$.

Let us come back to the minimality properties of stable solutions. Given u a stable solution of (1), we consider a principal eigenfunction ϕ_1 of J_u in Ω . Namely, ϕ_1 satisfies

$$\begin{cases} -J_u \phi_1 = \mu_1 [J_u, \Omega] \phi_1 & \text{in } \Omega \\ \phi_1 > 0 & \text{in } \Omega \\ \phi_1 = 0 & \text{on } \partial \Omega. \end{cases}$$
(6)

If $\mu_1[J_u, \Omega] = 0$ then, assuming f to be strictly convex, it is not hard to see that the competitor $u + \varepsilon \phi_1$ has less energy than u. This shows that u need not be a local minimizer, as we claimed above. On the other hand, if $\mu_1[J_u, \Omega] > 0$, then it can be shown that u is a minimizer with respect to competitors in a small tubular neighborhood around itself.⁴ When $\mu_1[J_u, \Omega] > 0$, we say that u is a *strictly stable* solution. It is worth mentioning that this last condition is equivalent to the operator J_u satisfying the maximum principle; see [9].

⁴More precisely, we have $\mathcal{E}_{1,F}(u+v) \geq \mathcal{E}_{1,F}(u)$ for all $v \in C_c^1(\Omega)$ with $\|v\|_{L^{\infty}(\Omega)} < \varepsilon$, for a small $\varepsilon > 0$.

Definition 2 offers a crucial advantage over the first one, since it applies to more general problems. The key point is that the principal eigenvalue can be defined for any linear operator, not necessarily a variational one. In particular, it allows us to consider stable solutions of equations involving non self-adjoint operators. Thus, in the first part of the thesis, our main focus will be in stable solutions to semilinear equations of the form

$$-Lu = f(u) \quad \text{in } \Omega, \tag{7}$$

where L is a linear second order elliptic operator with variable coefficients

$$L = a_{ij}(x)\partial_{ij}^2 + b_i(x)\partial_i.$$
(8)

This class of operators constitutes a natural generalization of the Laplacian. First of all, while Δ models the simplest (homogeneous, isotropic) diffusion process, the operator Lin (8) can be used to describe more complex phenomena, such as an arbitrary diffusion in the presence of advection (i.e., a flow). Studying equations of the type (7) is thus crucial in real-world applications where multiple physical phenomena act at the same time. Secondly, under a general change of coordinates, the Laplacian transforms into an operator of the form (8). Such transformations are relevant, for instance, when investigating the boundary regularity of solutions, since the curved boundary can be flattened out by a change of variables. This way, it suffices to study equations involving an operator L in a flat domain.

Notice that the linearization of (7) at u corresponds to the Jacobi operator

$$J_u = L + f'(u) = a_{ij}(x)\partial_{ij}^2 + b_i(x)\partial_i + f'(u(x)),$$

where we are still using the same notation as above. To give a more precise definition of stability for the equation (7), it will be worth recalling some fundamental properties of linear elliptic operators such as J_u .

The classic work of Berestycki, Nirenberg, and Varadhan [9] provides a theory of principal eigenvalues for elliptic operators with bounded coefficients. There, one assumes that the coefficient matrix $A(x) = (a_{ij}(x))$ is uniformly elliptic, that is, satisfying

$$c_0|p|^2 \le a_{ij}(x)p_ip_j \le C_0|p|^2 \quad \text{for all } p \in \mathbb{R}^n,$$
(9)

for some positive constants c_0 and C_0 , as well as the regularity assumptions

$$a_{ij} \in C^0(\Omega), \quad b_i \in L^\infty(\Omega),$$
(10)

$$f'(u) \in L^{\infty}(\Omega).^{5} \tag{11}$$

In [9], the authors gave the following definition for the principal eigenvalue of an operator J_u satisfying (9), (10), and (11):

$$\mu_1[J_u,\Omega] := \sup\left\{\mu \in \mathbb{R} \colon \exists \varphi > 0 \in W^{2,n}_{\text{loc}}(\Omega) \text{ satisfying } J_u\varphi + \mu\varphi \le 0 \text{ in } \Omega\right\}.$$
(12)

Moreover, without any regularity assumption on Ω , they prove the existence of a principal eigenfunction ϕ_1 in $L^{\infty}(\Omega) \cap W^{2,p}_{\text{loc}}(\Omega)$, for all $p < \infty$, satisfying (6) above, where the boundary condition must be appropriately interpreted.⁶

⁵In applications, assuming f to be locally Lipschitz, the boundedness of the zero order coefficient f'(u) will follow from the boundedness of the solution u.

⁶One first constructs a solution to $-Lu_0 = 1$ in Ω by approximation, solving the Dirichlet problem with zero boundary conditions in smooth domains that exhaust Ω and taking a converging subsequence. We say that $\phi_1 = 0$ on $\partial\Omega$ if, whenever $u_0(x_k) \to 0$ for some $x_k \in \Omega$ converging to $\partial\Omega$, then $\phi_1(x_k) \to 0$.

It is not hard to see that the variational definition of μ_1 is equivalent to (12). Essentially, μ_1 is a threshold for the existence of positive supersolutions, the principal eigenfunction being the "last" one. From this characterization follows another definition of stability:

Definition 3. A solution u of (7) is *stable* if there is a function $\varphi \in W^{2,n}_{\text{loc}}(\Omega)$ such that

$$\begin{cases} J_u \varphi \le 0 & \text{in } \Omega\\ \varphi > 0 & \text{in } \Omega. \end{cases}$$

By (12), a solution that is stable in the sense of Definition 3 is also stable for Definition 2. For the converse, it suffices to take φ equal to ϕ_1 , the principal eigenfunction of J_u above.

Although we will never use it, it is interesting to note that the function φ in Definition 3 can also be taken such that the inequality becomes an equality, i.e., with $J_u\varphi = 0$ in Ω . Indeed, if $\mu_1[J_u, \Omega] > 0$, then J_u satisfies the maximum principle and we can solve the boundary value problem $J_u\varphi = 0$ in Ω , $\varphi = c$ on $\partial\Omega$ for a positive constant c > 0; see [9]. Otherwise, if $\mu_1[J_u, \Omega] = 0$, then we simply take the principal eigenfunction.

Regularity of stable solutions

Having introduced the notion of stability in the previous section, we now turn toward the main focus of the first part of this work: the regularity of stable solutions. Given a bounded domain $\Omega \subset \mathbb{R}^n$ and a nonlinearity $f \in C^1(\mathbb{R})$, we consider the boundary value problem

$$\begin{cases} -Lu = f(u) & \text{in } \Omega\\ u = 0 & \text{on } \partial\Omega, \end{cases}$$
(13)

where L is the linear operator introduced in (8) above. We always assume that L satisfies the uniform ellipticity condition (9) and the regularity assumptions (10).

Since (13) involves a semilinear equation, the question of regularity reduces to showing the boundedness of solutions. Indeed, if u is in $L^{\infty}(\Omega)$, then f(u) is bounded and by L^{p} estimates we deduce that u is in $W^{2,p}_{loc}(\Omega)$, for all $p < \infty$. The latter estimate holds up to the boundary in domains of class $C^{1,1}$ and under the additional assumption that

$$a_{ij} \in C^0(\overline{\Omega}),\tag{14}$$

which allows us to apply the L^p estimates in [63, Theorem 9.14]. For further smoothness properties of the solution, more assumptions on the coefficients and the domain are needed.

If we do not restrict the class of solutions, then the regularity will depend on the growth of the nonlinearity. For the Laplacian $L = \Delta$ and assuming f to grow like a power

$$|f(u)| \le C(1+|u|^m),$$

if *m* is subcritical with respect to Sobolev exponent, i.e., $m < \frac{n+2}{n-2} =: m_s$, then every energy solution of (13) is bounded. By an energy solution, we mean a weak solution in $W_0^{1,2}(\Omega)$, the natural energy space for the functional $\mathcal{E}_{1,F}$ to be well defined. This boundedness result can be easily proved by the iterated L^p estimates of Crandall and Rabinowitz [44]. Moreover, using the Brezis-Kato trick [13], the same conclusion continues to hold for critical nonlinearities. However, when $n \geq 3$, the singular solution

$$U_m(x) := (m-1)\left(|x|^{-\frac{2}{m-1}} - 1\right)$$

is an energy solution in the supercritical range $m > m_s$. Notice that U_m solves (13) in $\Omega = B_1$ with $f(u) = \lambda_{n,m} (1 + \frac{1}{m-1}u)^m$, where $\lambda_{n,m} := 2\left(n - \frac{2m}{m-1}\right)$.

We are interested in a regularity theory for stable solutions that applies to *all* nonlinearities, independently of their growth. For instance, the stability of U_m amounts to

$$\frac{m}{m-1}\lambda_{n,m}\int_{B_1}\frac{\xi^2}{|x|^2}\,\mathrm{d}x \le \int_{B_1}|\nabla\xi|^2\,\mathrm{d}x \quad \text{for all } \xi \in C_c^\infty(B_1).$$
(15)

By Hardy's inequality with best constant, this is equivalent to $\frac{m}{m-1}\lambda_{n,m} \leq \frac{(n-2)^2}{4}$ and hence, in the supercritical range $m > m_s$, it can be checked that U_m is stable for $m \geq \frac{n-2\sqrt{n-1}}{n-4-2\sqrt{n-1}}$. This last condition on the exponent is equivalent to the inequality

$$n \ge 2 + 4\sqrt{1 + \frac{1}{m-1}} + 4\left(1 + \frac{1}{m-1}\right).$$
(16)

As a consequence, we deduce the existence of singular stable energy solutions for n > 10. In fact, letting $m \to \infty$ above, the solutions U_m converge to the logarithm

$$U^{\sharp}(x) := -2\log|x|, \tag{17}$$

which solves (13) with $L = \Delta$ and the exponential nonlinearity $f(u) = 2(n-2)e^u$, in the weak sense. This singular solution is always in the energy space and is stable for $n \ge 10$ (for instance, by taking the limit in (16) above).

When $\frac{n}{n-2} < m \leq m_s$, the function U_m is no longer in the energy space. Nevertheless, it still lies in a weaker Sobolev space and satisfies $f(U_m) \in L^1(B_1)$, which is enough to show that U_m solves (13) in a weak sense. What is more surprising is the fact that U_m satisfies the stability condition (15) in the smaller range $\frac{n}{n-2} < m \leq \frac{n+2\sqrt{n-1}}{n-4+2\sqrt{n-1}}$; see [14]. This last example suggests that a satisfactory regularity theory, if any, can only be expected for stable solutions in the energy class. Furthermore, even in that case, the smoothness of these solutions must depend on the dimension n of the domain. As we have seen, singularities may appear for $n \geq 10$, and it is therefore natural to ask whether this is the only case.

The Gelfand problem

Before discussing the optimality of dimension 10, we comment on a physical motivation for considering exponential nonlinearities (which led to this dimensional threshold). Such reaction terms arise naturally in combustion theory, this being the main inspiration for the first regularity results.

In the 1930's, Frank-Kamenetskii suggested the use of reaction-diffusion equations to model thermal combustion phenomena. The temperature of the combustible mixture is assumed to satisfy such an equation, with a reaction governed by the Arrhenius Law. Under the simplifying assumption that the fuel consumption time is much larger than the ignition time, the term coming from the Arrhenius Law can be well approximated by an exponential. In this theory, the solvability of the steady-state equation is used to determine whether an explosion will happen. Namely, the nonexistence of solutions is interpreted as the occurrence of an explosion, while the existence, for instance, of stable solutions would mean that the temperature evolves towards a stationary profile.

More precisely, let u denote a suitably normalized temperature of the combustible mixture. Suppose that the mixture is contained in a cylindrical vessel of cross-section $\Omega \subset \mathbb{R}^n$, with u vanishing on the boundary (meaning that the walls of the vessel are at

constant temperature). Then, after a sufficiently long time, u solves the *Gelfand problem*

$$\begin{cases} -Lu = \lambda g(u) & \text{in } \Omega \\ u > 0 & \text{in } \Omega \\ u = 0 & \text{on } \partial \Omega \end{cases}$$
(18)

with $L = \Delta$ and $g(u) = e^u$, where $\lambda > 0$ is a nondimensional parameter quantifying the relative strength of the reaction with respect to the diffusion. According to Frank-Kamenetskii, solutions to (18) are not expected to exist for sufficiently large λ ; see [57].

In the sixties, Barenblatt [60] introduced problem (18) in the mathematical literature, initiating the rigorous study of explosion phenomena.⁷ One natural way of constructing solutions to (18) is by applying the implicit function theorem at u = 0, $\lambda = 0$, solving for u in terms of λ and continuing this branch of solutions. Since the linearized operator at the trivial solution is the Laplacian, by continuity, the first portion of the branch consists of stable solutions. Another way of obtaining this same branch is by monotone iteration starting at u = 0. Here, one constructs barriers for the problem (18) to ensure that the iteration converges. Furthermore, this procedure shows that solutions are not only stable but also *minimal*, in the sense that they are the smallest positive supersolutions of (18).

Thanks to the robustness of these methods, they can also be applied to more general nonlinearities than exponential or power-type ones. To reproduce the behavior expected by Frank-Kamenetskii, it is natural to consider $g: [0, +\infty) \to \mathbb{R}$ satisfying

$$g(0) > 0$$
, g nondecreasing, convex, and superlinear at $+\infty$, (19)

where the last condition means that

$$\lim_{u \to +\infty} \frac{g(u)}{u} = +\infty$$

Under assumptions (19), the fundamental existence theory for (18) can be summarized in the following proposition, which has appeared in various forms in the literature:

Proposition 4 ([8,11,38,43]). Let $\Omega \subset \mathbb{R}^n$ be a bounded domain and let L be an operator of the form (8) with coefficients satisfying (9) and (10) in Ω . Assume that $g \in C^1(\mathbb{R})$ satisfies (19). Then, there exists a constant $\lambda^* \in (0, +\infty)$ such that:

- (i) For each $\lambda \in (0, \lambda^*)$ there is a unique minimal strong solution $u_{\lambda} \in L^{\infty}(\Omega) \cap W^{2,n}_{\text{loc}}(\Omega)$ of (18). In particular, u_{λ} is stable and we have $u_{\lambda} < u_{\lambda'}$ in Ω for $\lambda < \lambda'$.
- (ii) For $\lambda > \lambda^{\star}$ there are no strong solutions of (18) in $L^{\infty}(\Omega) \cap W^{2,n}_{\text{loc}}(\Omega)$.

Assume moreover that Ω is $C^{1,1}$ and that the coefficients of L are smooth up to the boundary.⁸ Then:

(iii) For $\lambda = \lambda^*$ there exists a unique L^1 -weak solution u^* of (18), in the following sense: $u^* \in L^1(\Omega), g(u^*) \operatorname{dist}(\cdot, \partial \Omega) \in L^1(\Omega), and, letting L^T$ denote the adjoint of L,

$$-\int_{\Omega} u^{\star} L^{T} \zeta \, \mathrm{d}x = \lambda^{\star} \int_{\Omega} g(u^{\star}) \zeta \, \mathrm{d}x$$

for all test functions $\zeta \in W^{2,n}(\Omega)$ such that $L^T \zeta \in L^{\infty}(\Omega)$ and $\zeta|_{\partial\Omega} = 0$. Moreover, u^* can be obtained as the pointwise limit $u_{\lambda} \uparrow u^*$ as $\lambda \uparrow \lambda^*$.

 $^{^7\}mathrm{Barenblatt's}$ work appeared in the volume [60] edited by Gelfand, and the latter's name became associated with the explosion problem.

⁸We will give more precise regularity assumptions in Chapter 3 below.

Recall here that, under minimal regularity assumptions on the coefficients (condition (10) above), the best solutions to (18) that we can expect are strong solutions in $L^{\infty}(\Omega) \cap W^{2,p}_{\text{loc}}(\Omega)$. By this we mean that they solve the equation almost everywhere in Ω and satisfy the boundary condition in an appropriate sense (see footnote 6). In $C^{1,1}$ domains and under the additional assumption (14), bounded solutions are in $W^{2,p}(\Omega)$, for all $p < \infty$ (and hence in $C^{1,\alpha}(\overline{\Omega})$ for all $0 < \alpha < 1$).

Proposition 4 underlines the important role played by stable solutions in the Gelfand problem. Namely, for $\lambda \in (0, \lambda^*)$ there is a distinguished stable solution u_{λ} , which is bounded (and hence smooth if the domain and the coefficients of L are smooth), while for $\lambda > \lambda^*$ there are no solutions (not even in a weak sense, as shown in [12]). As we approach the extremal parameter λ^* , the stable solutions u_{λ} converge to a function u^* in L^1 , which solves (18) in a weak sense. This is the *extremal solution* and it is also the unique solution of (18) for $\lambda = \lambda^*$, as proved by Martel [73].

A priori, the extremal solution u^* is only in L^1 and, therefore, it is natural to ask whether it is bounded. In the seventies, Joseph and Lundgren [64] considered the Gelfand problem (18) in the unit ball for exponential and power nonlinearities, with $L = \Delta$. Since positive solutions are radially symmetric in this case, the problem reduces to an ODE. Using phase plane techniques, they showed that the stable branch (and hence u^*) is always bounded when $n \leq 9$, while it can be unbounded for $n \geq 10$. In fact, for $n \geq 10$, the logarithm U^{\sharp} given by (17) above is the extremal solution for $g(u) = e^{u.9}$

As noted by Crandall and Rabinowitz [44], the boundedness of u^* in general domains could be deduced from L^{∞} a priori estimates for the stable solutions $\{u_{\lambda}\}_{0<\lambda<\lambda^*}$. In the seminal work [44], they obtained such a bound for exponential and power-type nonlinearities in the optimal range $n \leq 9$. Their result applies to variational problems involving operators in divergence form with bounded coefficients. Using appropriate test functions in the variational stability inequality, for exponential and power nonlinearities they showed that $\|u_{\lambda}\|_{L^{\infty}(\Omega)} \leq C$ for $n \leq 9$ and $\lambda \in (0, \lambda^*)$, where the constant C is independent of λ . Taking the limit as $\lambda \to \lambda^*$, it follows that u^* is bounded in this case.

Optimal dimension for the Laplacian

After the work of Crandall and Rabinowitz, no singular stable solutions were found in low dimensions, which led to the conjecture that they must all be bounded in this case. In the nineties, Brezis [11] asked whether dimension 10 could also be optimal for more general nonlinearities, emphasizing the class satisfying the natural assumptions (19). Along similar lines, Brezis and Vázquez [14] noticed that the known examples of extremal solutions were always in the energy space, independently of the dimension. Consequently, they wondered if such a phenomenon could hold for general nonlinearities as above.

Both of these questions motivated a series of papers trying to establish a priori estimates that could apply to all stable solutions, at least in low dimensions and for the model case $L = \Delta$. All these attempts over the course of a quarter of a century culminated in the recent work of Cabré, Figalli, Ros-Oton, and Serra in [24], where the two questions were answered affirmatively for the Laplacian in general C^3 regular domains. Next, we comment on the previous works leading to this last result, emphasizing the test functions used in the stability condition (4).

⁹This fact follows directly from a useful characterization of singular extremal solutions due to Brezis and Vázquez [14]. Namely, if a stable energy solution is unbounded, then it is necessarily the extremal solution.

- The first positive result for the Laplacian in smooth domains was obtained by Nedev in [78]. By choosing test functions of the form $\xi = h(u)$ in (4), with h depending on the nonlinearity g, he obtained a priori estimates for the minimal solutions u_{λ} . As a consequence, he was able to show that $u^* \in L^{\infty}(\Omega)$ for $n \leq 3$ and $u^* \in W_0^{1,2}(\Omega)$ for $n \leq 5$. Under the additional assumption that Ω is convex, he proved that the last inclusion holds in every dimension [79].
- Cabré and Capella [21] treated the radial case $(\Omega = B_1, L = \Delta)$ for general nonlinearities. They were able to reach the optimal dimension in this case, showing that $u^* \in L^{\infty}(B_1)$ for $n \leq 9$ and $u^* \in W_0^{3,2}(B_1)$ for all n. Inspired by the work of Simons on minimal surfaces [90], they considered a test function of the form $\xi = \mathbf{c}\eta$ in the stability inequality, where \mathbf{c} depends on the solution and η is a cut-off. This choice leads to the inequality

$$\int_{\Omega} (J_u \mathbf{c}) \, \mathbf{c} \eta^2 \le \int_{\Omega} \mathbf{c}^2 |\nabla \eta|^2,$$

and the goal is then to choose \mathbf{c} in a way such that $J_u \mathbf{c}$ no longer depends on the nonlinearity. Writing $u_r = \frac{x}{|x|} \cdot \nabla u$ and r = |x|, they considered $\mathbf{c} = ru_r$ and $\eta = (r^{-\alpha} - 2^{\alpha})_+$ for the exponent $\alpha < 1 + \sqrt{n-1}$. This choice allowed them to control a weighted norm of u_r , which then yields pointwise bounds for the solution. When $n \leq 9$, they obtained an L^{∞} estimate valid for all stable energy solutions of (13) and for all C^1 nonlinearities f, not only those satisfying (19). Their result gave credibility to the conjecture that 10 was indeed the lowest dimension where singularities could arise.

• Cabré [16] proved the boundedness of u^* up to $n \leq 4$ in convex domains. Using stability, he first controlled stable energy solutions in an arbitrary domain by their L^{∞} norm close to the boundary. Assuming the convexity of the domain, the method of moving planes then yields an estimate of this last quantity in terms of the L^1 norm of the solution. To obtain the first estimate, he again considered a test function of the form $\xi = \mathbf{c}\eta$ in the stability inequality, letting

$$\mathbf{c} = |\nabla u|$$
 and $\eta = h(u)$

for an appropriate h. This choice had already appeared in the works of Sternberg and Zumbrun on phase transitions [91,92] and leads to control certain geometric quantities associated to the level sets of the solution, such as their curvature. Combined with the Sobolev inequality of Michael, Simon, and Allard [2,76] on level sets, the stability of u yields the L^{∞} estimate for $n \leq 4$.

Later, Villegas [95] extended this boundedness result to all smooth domains, combining ideas of Nedev [78] and of Cabré [16]. Moreover, he showed that u^* is in $W_0^{1,2}(\Omega)$ for $n \leq 6$.

• Cabré and Ros-Oton [27] showed that u^* is bounded for $n \leq 7$ in convex domains of double revolution (invariant under rotations of the first m and last n - m variables). Here, by symmetry, the solution depends only on the radial variables $s = \sqrt{x_1^2 + \cdots + x_m^2}$ and $t = \sqrt{x_{m+1}^2 + \cdots + x_n^2}$. Similar to the radial case treated in [21], they considered test functions $\xi = \mathbf{c}\eta$, letting $\mathbf{c} = su_s$ and $\eta = s^{-\alpha}\zeta$ first (where ζ is a cut-off), and then $\mathbf{c} = tu_t$ and $\eta = t^{-\beta}\zeta$, for exponents in the range $\alpha < 1 + \sqrt{m-1}$ and $\beta < 1 + \sqrt{n-m-1}$. As in [16], this yields an L^{∞} estimate

in terms of the solution close to the boundary, which can be further bounded in convex domains. Without this convexity assumption, they were able to show that $u^* \in W_0^{1,2}(\Omega)$ for $n \leq 6$, later improved by Villegas in [95] (as commented above).

As mentioned above, the conjectures of Brezis and Brezis-Vázquez for the Laplacian were eventually solved by Cabré, Figalli, Ros-Oton, and Serra in [24], where they showed that $u^* \in L^{\infty}(\Omega)$ for $n \leq 9$ and $u^* \in W_0^{1,2}(\Omega)$ for all n, in domains of class C^3 . In that paper, the authors consider several of the test functions that we have already mentioned and they also introduce new ideas to complete the proof. Their strategy is to upgrade the proof of [21] for the radial case, which already yields an optimal result, to make it work in general domains.

The main result in [21] consists of two types of universal a priori estimates for bounded stable solutions. Namely, an energy estimate, valid in all dimensions, and a Hölder estimate in the optimal range $n \leq 9$, all in terms of the L^1 norm of the solution. We briefly discuss their proof, which is based on obtaining local interior and boundary analogues of these estimates.

The interior estimates in [24] apply to classical stable solutions of (1) in the unit ball $\Omega = B_1$ (without boundary conditions), and only need the nonnegativity of f. For the energy estimate, they choose a Sternberg-Zumbrun type test function in the stability inequality, letting $\xi = \mathbf{c}\eta$ with $\mathbf{c} = |\nabla u|$ and where η is a cut-off. Applying the divergence theorem, they are then able to control the Dirichlet energy on level surfaces, and by the coarea formula the higher integrability of the gradient follows. In contrast, the interior Hölder estimate uses a variant of the test function for the radial case in [21]. Letting

$$\mathbf{c}(x) = x \cdot \nabla u(x)$$
 and $\eta(x) = |x|^{\frac{2-n}{2}} \zeta$

for a cut-off ζ , when $n \leq 9$, the stability condition leads to a control of the weighted scale-invariant quantity $\int_{B_{\rho}} u_r^2 r^{2-n}$, with $\rho < 1/2$, by the Dirichlet energy in the annulus $B_{2\rho} \setminus B_{\rho}$. It remains to bound the full gradient in terms of the radial derivative, since the previous estimate would then lead to the algebraic decay of the weighted integral, and hence to a C^{α} bound. To show this, they argue by compactness, using that there are no nonconstant homogeneous superharmonic functions in an annulus.

On the other hand, the boundary estimates in C^3 domains require that f be nonnegative, nondecreasing, and convex. The test functions are variants of the ones used in the interior estimates, obtained by modifying \mathbf{c} so that it vanishes on the curved boundary $\partial\Omega$. Once again, the crucial point is controlling the full derivative by the radial one. Here, the superharmonicity no longer suffices to have such a property, and one must further use that u is a stable solution to a semilinear equation. The authors of [24] are only able to show this in flat domains, by compactness, which requires a subtle closedness result for stable solutions with convex nonlinearities. To obtain the Hölder estimate in curved domains, they apply a blow-up procedure, and the problem reduces to proving a Liouville theorem in half-spaces. This result follows from the stability inequality with $\mathbf{c} = x \cdot \nabla u$ combined with the estimate for u_r in half-balls.

Notice how the use of compactness arguments was crucial to obtain the optimal Hölder estimate in [24]. While these methods can be quite versatile, they do not provide a means to quantify the constants that appear in the estimates. Recently, Cabré [19,20] has found a new quantitative proof for the Laplacian in half-balls. Namely, he obtains interior estimates assuming only the nonnegativity of f, and boundary estimates in half-balls for nonnegative, nondecreasing, and convex nonlinearities. This new quantitative proof greatly simplifies the previous boundary argument. The author's approach is to control the L^1 norm of the solution by the L^1 norm of its radial derivative directly, in a quantitative way. For the interior estimates, the key is proving L^1 estimates for the Neumann problem and using the superharmonicity of the solution to carry out a comparison argument. The boundary estimates use the equation satisfied by the solution under rescalings, taking a derivative with respect to the scaling parameter and applying the stability inequality.

Another natural question concerns the optimal regularity when $n \ge 10$. This has been solved by Peng, Zhang, Zhou [81], who show that, for the Laplacian, stable energy solutions belong to the Morrey space $M^{m_n,4+\frac{2}{m_n-2}}(\Omega)$ with $m_n = \frac{2(n-2\sqrt{n-1}-2)}{n-2\sqrt{n-1}-4}$.¹⁰

Operators with variable coefficients

While the works mentioned above dealt with the Laplacian in sufficiently smooth domains, the same optimal regularity question can be asked in more general scenarios. Some natural extensions include substituting the Laplacian by an operator with variable coefficients, or weakening the regularity of the domain Ω in the boundary value problem. We will obtain an optimal regularity result for operators in non-divergence form in $C^{1,1}$ domains, as we explain later after first describing the previously known literature.

Concerning operators with variable coefficients, the only optimal results apply to particular nonlinearities. For (self-adjoint) operators in divergence form, Crandall and Rabinowitz [44] showed the boundedness up to dimension 9 for exponential and power-type nonlinearities. This result was extended by Cowan and Ghoussoub [40] to non self-adjoint operators of the form $L = \Delta + b_i(x)\partial_i$ in smooth domains, with $b_i \in C^{\infty}(\overline{\Omega})$. In this setting, a variational stability inequality is no longer available and one must solely rely on the nonnegativity of the principal eigenvalue $\mu_1[J_u, \Omega]$. To surmount this difficulty, the authors in [40] apply a general Hardy inequality by Cowan [39] to the principal eigenfunction, resulting in an alternative integral inequality. Subsequently, the test functions of [44] lead to the boundedness of u^* in the optimal range. The Hardy inequality by Cowan uses a Hodge-type decomposition of the vector field $b(x) = (b_i(x))$, which requires it to be smooth. We will get a more general result for bounded b_i , as explained in the last section of this Introduction.

We now discuss the regularity of the domain. Before the optimal dimension for the Laplacian $(n \leq 9)$ was reached in [24], all previous works assumed the smoothness of $\partial\Omega$ but did not investigate the possibility of weakening this assumption. On the other hand, the local analysis of [24] around the boundary works precisely in C^3 regular domains.

To prove the Hölder estimate for classical stable solutions in [24], the authors consider a portion of the boundary that is close to flat in C^3 norm. Under this assumption, when applying the blow-up procedure, the domain converges to a half-space and all error terms in their estimates vanish. Moreover, the C^3 regularity assumption ensures that classical stable solutions are in the Sobolev space $W^{3,p}(\Omega)$, for all $p < \infty$. The existence of third weak derivatives is crucial for the estimates not to depend on the nonlinearity. Here, when computing the linearized operator J_u acting on the test functions $\mathbf{c} = |\nabla u|$ and $\mathbf{c} = x \cdot \nabla u$, the nonlinearity cancels out, but this can only happen if u has third derivatives.

Instead of using a blow-up procedure, we will study curved boundaries by flattening them out locally. Under such a transformation, the Laplacian becomes an operator with variables coefficients of the form (8). Our finer estimates for these operators in flat domains

¹⁰Here, we say that u belongs to the Morrey space $M^{m,\beta}(\Omega)$ if $\sup_{y \in \Omega, \rho > 0} \rho^{\beta-n} \int_{\Omega \cap B_{\rho}(y)} |u|^m < \infty$.

will be applicable to domains that are close to flat in C^2 norm. Moreover, by an approximation argument, such estimates will continue to hold in $C^{1,1}$ regular domains. This is a natural assumption from the point of view of the regularity theory for non-divergence form operators.

Stable solutions to other problems

The regularity of stable solutions can also be studied in other frameworks. Here, we briefly mention some natural extensions obtained by modifying the nonlinearity or the operator in the equation.

An important class of nonlinearities consists in functions blowing up at a point. The Gelfand problem in this case models the behavior of MEMS devices. For a positive, increasing, and convex nonlinearity $f: [0, 1) \rightarrow (0, \infty)$ blowing up at 1, stable solutions u are always bounded by 1, but are only regular if u < 1. Here, the optimal dimension for the regularity of stable solutions depends on the blow-up rate of f, and is open in certain general classes; see the monograph [53].

Quasilinear operators have also been considered in the literature. Here, the operator L in the equation -Lu = f(u) is replaced by a quasilinear one. When L is the p-Laplacian, the optimal dimension depends on p. Namely, explicit examples of singular stable energy solutions (in $W^{1,p}$) have only been found for $n \ge p + \frac{4p}{p-1}$. For p > 2, Cabré, Miraglio, and Sanchón [25] have shown that this range is optimal. When 1 , they have also proved the boundedness of stable solutions for <math>n < 5p, but it is not known whether this can be improved. Mellet and Vovelle [75] have treated the case when L is the mean curvature operator. Here, smoothness does not follow from the boundedness of the solution (in fact, stable solutions are always bounded) and the goal is to show that the gradient is bounded. In [75] the authors show that, in the radial case, stable solutions are regular independently of the dimension. Whether this result holds in general domains is an open problem.

Finally, another interesting scenario arises when substituting the differential operator L by a nonlocal one. The model case here is the fractional Laplacian $(-\Delta)^s$ (on which we comment more in the next section below), and the regularity of stable solutions depends on the fractional parameter $s \in (0, 1)$. For this operator, the optimal dimension is believed to be the one given by the exponential nonlinearity in [83]. However, for general nonlinearities, this dimension has not been reached, not even in the radial case; see [29, 85, 86] and the references therein.

Minimizers and calibrations

Up until this point we have focused on the stability of solutions, a necessary condition for minimality. Now, we turn our attention to minimizers of energy functionals. In the second part of this dissertation, we will be particularly interested in the theory of sufficient conditions for nonlocal functionals. Namely, given a solution to a variational equation, we would like to identify auxiliary conditions guaranteeing its minimality. While such a theory has been known for a long time for classical local problems, before this work there were almost no results in the nonlocal setting.

First, we give a brief summary of the classical theory of necessary and sufficient conditions. For this part, we mostly follow the celebrated monograph of Giaquinta and Hildebrandt [61]. Then, we move onto nonlocal energy functionals, focusing on the fractional Gagliardo-Sobolev seminorm.

The classical theory of necessary and sufficient conditions

One of the greatest discoveries in Physics was the realization that the laws of nature could be deduced from variational principles. An early example is Fermat's principle of least time in optics, whereby the path of a light ray traveling between two points is determined by minimizing the traveling time. Such principles motivated the development of the Calculus of Variations, the mathematical discipline dealing with the minimization of functionals.

A great part of the classical theory deals with general energy functionals of the form

$$\mathcal{E}_{\mathrm{L}}(w) = \int_{\Omega} G_{\mathrm{L}}(x, w(x), \nabla w(x)) \,\mathrm{d}x.^{11}$$

Here, we assume that the Lagrangian function $G_{\rm L} = G_{\rm L}(x, \lambda, q)$ is smooth and that the functional $\mathcal{E}_{\rm L} : \mathcal{A}_{\rm L} \to \mathbb{R}$ acts on a class $\mathcal{A}_{\rm L}$ of scalar functions $w : \overline{\Omega} \to \mathbb{R}$ in $C^1(\overline{\Omega})$, defined in some bounded domain $\Omega \subset \mathbb{R}^n$.¹² We say that $u \in \mathcal{A}_{\rm L}$ is a minimizer of $\mathcal{E}_{\rm L}$ if

 $\mathcal{E}_{\mathrm{L}}(u) \leq \mathcal{E}_{\mathrm{L}}(w)$ for all $w \in \mathcal{A}_{\mathrm{L}}$ with w = u on $\partial \Omega$.

That is, we always consider minimizers with respect to competitors satisfying the same Dirichlet boundary conditions.

To compute minimizers, the first step is to find necessary conditions for minimality. This is achieved by testing the minimality condition with suitable competitors. Given $u \in \mathcal{A}_{L}$ a minimizer of \mathcal{E}_{L} , we consider competitors of the form $u + \varepsilon \xi$ with $\xi \in C_{c}^{\infty}(\Omega)$ and where $\varepsilon > 0$ is sufficiently small so that $u + \varepsilon \xi \in \mathcal{A}_{L}$ (which is satisfied in most applications). By minimality of u, the first variation of \mathcal{E}_{L} at u (i.e., the derivative of $\varepsilon \mapsto \mathcal{E}_{L}(u + \varepsilon \xi)$ at $\varepsilon = 0$) must vanish. Since ξ is arbitrary, after a simple computation and integrating by parts, it follows that u satisfies the Euler-Lagrange equation

$$-\operatorname{div}\left(\partial_q G_{\mathrm{L}}(x, u, \nabla u)\right) + \partial_\lambda G_{\mathrm{L}}(x, u, \nabla u) = 0.$$

Solutions to the Euler-Lagrange equation are called *critical points* or *extremals* of the functional \mathcal{E}_{L} .

As discussed above, another necessary condition follows from taking the second variation of $\mathcal{E}_{\rm L}$ at u, which must be nonnegative and leads to the stability of the critical point. Choosing test functions of the form $\xi = \varepsilon \eta(\cdot/\varepsilon)$ in the stability inequality and letting $\varepsilon \to 0$ leads to Legendre's necessary condition

$$\partial_{q_i,q_j}^2 G_{\mathrm{L}}(x,u(x),\nabla u(x))p_ip_j \ge 0 \quad \text{for all } p \in \mathbb{R}^n \text{ and } x \in \Omega.$$

Notice that the linearized Euler-Lagrange equation at u is the Jacobi operator

$$J_{u}\xi = \operatorname{div}\left(\partial_{q,q}^{2}G_{\mathrm{L}}(x,u,\nabla u)\nabla\xi\right) + \left\{\operatorname{div}\left(\partial_{q,\lambda}^{2}G_{\mathrm{L}}(x,u,\nabla u)\right) - \partial_{\lambda,\lambda}^{2}G_{\mathrm{L}}(x,u,\nabla u)\right\}\xi.$$

When u is a minimizer, it follows that the principal part of J_u is nonnegative definite, that is, J_u is an elliptic operator in this case.

Until the XIX century, it was believed that extremals of elliptic energy functionals were necessarily minimizers. Weierstrass disproved this belief by producing examples of critical

¹¹The subindices L and N will be used throughout the work to denote "local" and "nonlocal" objects, respectively.

¹²Here it is worth mentioning that, historically, vector valued functions of one variable were also crucial in the development of the Calculus of Variations. Minimization problems in that case leads to the study of systems of ODEs, while in this thesis we focus on scalar PDEs.

points which did not minimize the corresponding functional. The claim is only true for convex Lagrangians, that is, such that the joint function $(\lambda, q) \mapsto G_{L}(x, \lambda, q)$ is convex. However, this convexity assumption is too restrictive in most applications, where only ellipticity is available. These observations motivate the search for sufficient conditions.

Jacobi developed a theory of sufficient conditions for *weak* minimizers, namely, local minimizers in a C^1 neighborhood. More precisely, $u \in \mathcal{A}_L$ is a weak minimizer of \mathcal{E}_L if

$$\mathcal{E}_{\mathrm{L}}(u) \leq \mathcal{E}_{\mathrm{L}}(u+\xi) \quad \text{for all } \xi \in C_{c}^{\infty}(\Omega) \text{ with } \|\xi\|_{C^{1}(\overline{\Omega})} < \varepsilon_{\varepsilon}$$

for some small $\varepsilon > 0$. Arguing as above, we have that weak minimizers satisfy Legendre's condition. Under a uniform ellipticity assumption on J_u (the *strict Legendre condition*), Jacobi proved that strictly stable solutions are weak minimizers. To see this, by Taylor's expansion, it suffices to show that the second variation at $u + \xi$ is positive. Since G_L and all its derivatives are continuous with respect to C^1 perturbations, the positivity at $u + \xi$ follows from the one at u, up to errors of order $\|\nabla \xi\|_{L^2(\Omega)}^2$. These errors can then be controlled using Gårding's inequality, which shows the claim.

As we just saw, Jacobi's sufficient condition (i.e., the strict stability of u) yields local minimality in a very small neighborhood of functions whose slopes are close to those of u. In applications, however, a stronger and more precise notion of minimality is often needed. Following this direction, Weierstrass studied the class of *strong* minimizers: local minimizers with respect to the C^0 topology. Thus, $u \in \mathcal{A}_L$ is a strong minimizer of \mathcal{E}_L if there is a $\varepsilon > 0$ such that

$$\mathcal{E}_{\mathrm{L}}(u) \leq \mathcal{E}_{\mathrm{L}}(u+\xi) \quad \text{ for all } \xi \in C_{c}^{\infty}(\Omega) \text{ with } \|\xi\|_{C^{0}(\overline{\Omega})} < \varepsilon.$$

First, Weierstrass obtained a necessary condition for strong minimizers. For this, taking perturbations of the form $\xi(x) = \psi_{\varepsilon}(x_n)\zeta(x'/\varepsilon)$ (up to translations and rotations), where $x = (x', x_n) \in \mathbb{R}^{n-1} \times \mathbb{R}$, ζ is a cut-off, and

$$\psi_{\varepsilon}(x_n) := \begin{cases} (x_n + \varepsilon^2)_+ & \text{for } x_n \le 0\\ \varepsilon(\varepsilon - x_n)_+ & \text{for } x_n > 0, \end{cases}$$

and letting $\varepsilon \to 0$ in the strong minimality condition, it follows that

$$G_{\mathrm{L}}(x, u(x), \nabla u(x) + p) \ge G_{\mathrm{L}}(x, u(x), \nabla u(x)) + \partial_{q_i} G_{\mathrm{L}}(x, u(x), \nabla u(x)) p_i$$

for all $p \in \mathbb{R}^n$ and $x \in \Omega$. This is the Weierstrass necessary condition, which corresponds to a stronger notion of ellipticity for J_u than Legendre's condition.

Later, Weierstrass found a sufficient condition which gives minimality in a more precise region of space, not just in a small C^0 neighborhood. It states that if a solution can be embedded in a family of extremals, whose graphs produce a foliation, then it is a minimizer in the region foliated by these critical points. In the literature, this is known as the Weierstrass extremal field theory.

We now explain this necessary condition in more detail. Let $\{u^t\}_{t\in I}$, where $I \subset \mathbb{R}$ is an interval, be a one-parameter family of critical points whose graphs give a foliation of a certain region $\mathcal{G}_{\mathrm{L}} \subset \overline{\Omega} \times \mathbb{R}$. In particular, the graphs of u^t do not touch. Such a family is called a *field of extremals*,¹³ a nomenclature originating in scalar ODE, where the slopes of the u^t would define a vector field in \mathcal{G}_{L} . The extremal field theory asserts the following: if every critical point $\{u^t\}_{t\in I}$ satisfies the Weierstrass necessary condition, then each u^t is a

 $^{^{13}}$ The term *extremal field* is also often used in the literature, but we find it ambiguous.

minimizer with respect to competitors $w \in \mathcal{A}_{L}$, taking values in the foliated region \mathcal{G}_{L} , and satisfying $w = u^{t}$ on $\partial \Omega$. In particular, the field of extremals is made up of minimizers.

This minimality result can be proved directly by constructing an auxiliary functional $C_{\rm L}: \mathcal{A}_{\rm L} \to \mathbb{R}$, defined next. Here, the foliation property of the field is crucial. Using that for each point $(x, \lambda) \in \mathcal{G}_{\rm L} \subset \overline{\Omega} \times \mathbb{R}$ there is exactly one $t = t(x, \lambda) \in I$ such that $u^t(x) = \lambda$, we define $\mathcal{C}_{\rm L}$ by

$$\mathcal{C}_{\mathrm{L}}(w) := \int_{\Omega} \left\{ G_{\mathrm{L}}(x, u^{t}(x), \nabla u^{t}(x)) + \partial_{q} G_{\mathrm{L}}(x, u^{t}(x), \nabla u^{t}(x)) \cdot (\nabla w - \nabla u^{t}) \right\} \Big|_{t=t(x, w(x))} \, \mathrm{d}x.$$

Notice that $C_{\rm L}$ is obtained by letting $p = \nabla w(x) - \nabla u^t(x) \big|_{t=t(x,w(x))}$ in the Weierstrass necessary condition satisfied by $u^{t(x,w(x))}$ and then integrating in $x \in \Omega$. It follows immediately that $C_{\rm L}$ touches $\mathcal{E}_{\rm L}$ by below at each extremal u^t . Moreover, using that $\{u^t\}_{t\in I}$ are critical points, it can be shown that $C_{\rm L}$ is also a null-Lagrangian, in the sense that it only depends on the boundary values of the functions. Hence, we have $C_{\rm L}(w) = C_{\rm L}(\tilde{w})$ for all $w, \tilde{w} \in \mathcal{A}_{\rm L}$ such that $w = \tilde{w}$ on $\partial\Omega$. From these properties, it is now easy to deduce the minimality of each u^t .

Functionals such as C_{L} are called *calibrations* in the Calculus of Variations, and they can be used in many situations to conclude that certain solutions are minimizers. The strategy is to embed such a critical point u in a family of extremals $\{u^t\}_{t\in I}$, so that $u = u^{t_0}$ for some $t_0 \in I$. More generally, assuming that $t \mapsto u^t$ is increasing in $t \in I$, to prove the minimality of u^{t_0} it suffices that the functions u^t above u^{t_0} (with $t \ge t_0$) are supersolutions of the Euler-Lagrange equations, while the ones below (with $t \le t_0$) are subsolutions. Such approaches have found important applications in the theory of minimal surfaces; see [10, 45, 46, 71].

We conclude this section by mentioning that Jacobi's sufficient condition above yields the existence of an extremal field in a small C^0 neighborhood of the solution. Namely, assuming the strict Legendre condition, if u is strictly stable, then one can construct extremals of the form $u^t = u + \xi^t$ by the Banach fixed point theorem, with ξ^t small. To conclude their minimality in the foliated neighborhood, the Weierstrass necessary condition is still needed.

Nonlocal problems

Recently, there has been an increasing interest in nonlocal problems. Classical functionals such as \mathcal{E}_{L} cannot capture the long range interactions present in many relevant situations. Minimizing the energy functionals associated to these processes leads to nonlocal Euler-Lagrange equations. Nonlocality refers to the fact that evaluating these equations at a point requires knowledge of the solution everywhere in space.

One of the simplest energy functionals modeling nonlocal phenomena is given by

$$\mathcal{E}_{s,F}(w) = \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|w(x) - w(y)|^2}{|x - y|^{n + 2s}} \,\mathrm{d}x \,\mathrm{d}y - \int_{\Omega} F(w(x)) \,\mathrm{d}x,$$

where $s \in (0, 1)$, $c_{n,s}$ is a positive normalizing constant, $F \in C^1(\mathbb{R})$, and

$$Q(\Omega) = (\mathbb{R}^n \times \mathbb{R}^n) \setminus (\Omega^c \times \Omega^c)$$

for a bounded domain $\Omega \subset \mathbb{R}^n$, with $\Omega^c = \mathbb{R}^n \setminus \Omega$. The functional $\mathcal{E}_{s,F}$ is the "fractional" version of $\mathcal{E}_{1,F}$ considered in (2) above. It is obtained substituting the Dirichlet energy by

the Gagliardo seminorm [58]. Moreover, one can formally recover $\mathcal{E}_{1,F}$ from $\mathcal{E}_{s,F}$ by letting $s \uparrow 1$; see [47].

Notice that here, in the definition of $\mathcal{E}_{s,F}$, we are considering functions $w \colon \mathbb{R}^n \to \mathbb{R}$ taking values in the whole space, not just in the domain Ω . Thus, the functional accounts for interactions between points that can be very far apart from each other. In this context, it is natural to consider competitors having the same value in the exterior $\Omega^c = \mathbb{R}^n \setminus \Omega$. This is one reason why we do not include $\Omega^c \times \Omega^c$ interactions in $\mathcal{E}_{s,F}$, the other being that such an integral need not be finite, even for smooth and bounded functions. Hence, the exterior datum becomes a nonlocal analogue of the boundary conditions from classical local problems.

Minimizers of $\mathcal{E}_{s,F}$ satisfy the Euler-Lagrange equation

$$(-\Delta)^s u = f(u) \quad \text{in } \Omega,$$

where $(-\Delta)^s$ is the fractional Laplacian defined by

$$(-\Delta)^s u(x) = c_{n,s} \operatorname{P.V.} \int_{\mathbb{R}^n} \frac{u(x) - u(y)}{|x - y|^{n+2s}} \,\mathrm{d}y.$$

Here, P.V. indicates that the integral must be understood in the principal value sense. Once again, in order to compute the fractional Laplacian of u at a point, we need to know the values of the function in the whole \mathbb{R}^n . The normalizing constant $c_{n,s}$ is determined by taking the Fourier transform \mathcal{F} and imposing that

$$\mathcal{F}[(-\Delta)^s u](z) = |z|^{2s} \mathcal{F}[u](z) \text{ for all } z \in \mathbb{R}^n.$$

In particular, $(-\Delta)^s$ converges to the ordinary Laplacian $-\Delta$ as $s \uparrow 1$, in a suitable sense.

While being a nonlocal operator, the fractional Laplacian can also be studied via the Caffarelli-Silvestre extension [33], an auxiliary local problem in an extended n + 1 dimensional space. There, writing $(x, y) \in \mathbb{R}^n \times (0, \infty) =: \mathbb{R}^{n+1}_+$, one considers $U: \overline{\mathbb{R}^{n+1}_+} \to \mathbb{R}$ to be the solution of the degenerate elliptic problem

$$\begin{cases} \operatorname{div} (y^{1-2s} \nabla U) = 0 & \operatorname{in} \mathbb{R}^{n+1}_+ \\ U = u & \operatorname{on} \partial \mathbb{R}^{n+1}_+ = \mathbb{R}^n. \end{cases}$$

It can then be shown that $(-\Delta)^s u(x) = -\lim_{y\to 0^+} y^{1-2s} \partial_y U(x, y)$, allowing us to recover the fractional Laplacian of u from its harmonic s-extension U. Notice that the semilinear equation in Ω above becomes a Neumann reaction problem for the extension.

A theory of necessary of sufficient conditions as the one described in the previous section does not seem to be available for nonlocal functionals at the moment. Such results would help us to better understand nonlocal variational problems and would also equip us with a valuable toolkit to effectively attack these problems. For instance, in the last section of this Introduction, we give some nontrivial applications of having sufficient conditions.

Concerning necessary conditions, even the notion of ellipticity has not been properly defined in that case. The fractional Laplacian is admitted to be an elliptic operator in some sense, since its Fourier symbol is invertible away from zero (a concept of ellipticity from the theory of pseudo-differential operators) and because it satisfies a maximum principle. While the former condition is only applicable to pseudo-differential operators, the latter condition continues to hold for more general nonlocal operators satisfying some monotonicity assumption, which can be taken as a nonlocal notion of ellipticity (as explained below in the last section of this Introduction). On the other hand, sufficient conditions for minimality had only been found in the context of nonlocal minimal surfaces. There, one considers functionals \mathcal{P}_{N} acting on sets $E \subset \mathbb{R}^{n}$ instead of functions, and which describe a nonlocal analogue of the perimeter of the surface ∂E . These functionals are of the form

$$\mathcal{P}_{\mathrm{N}}(E) = \frac{1}{2} \iint_{Q(\Omega)} |\mathbb{1}_{E}(x) - \mathbb{1}_{E}(y)| K(x-y) \,\mathrm{d}x \,\mathrm{d}y,$$

where K is a singular, nonnegative, symmetric kernel. We note in particular that quadratic functionals of the form $\mathcal{E}_{s,F}$ are not covered by this setting. For nonlocal perimeters, the only results are by Cabré [18] and Pagliari [80]. In [18], the author built a calibration for \mathcal{P}_{N} in the presence of a field of extremals by nonlocal minimal surfaces (i.e., critical points of \mathcal{P}_{N}). This construction suggested that a nonlocal version of the Weierstrass extremal field theory could exist for other functionals, but the method of proof could not be generalized to that framework. As we will see in the last section below, the present dissertation will confirm the validity of such a conjecture. Lastly, [80] gave a calibration for the nonlocal total variation functional

$$\mathcal{E}_{\mathrm{NTV}}(w) := \frac{1}{2} \iint_{Q(\Omega)} |w(x) - w(y)| K(x-y) \, \mathrm{d}x \, \mathrm{d}y,$$

(which can be obtained as the integral of the \mathcal{P}_N on level sets) for the particular case of characteristic functions of the half-space. As an application, the author of [80] deduced that halfspaces minimize \mathcal{P}_N . While this is the first nonlocal calibration (for the types of functionals that we are interested in) that has appeared in the literature, the proof does not use fields of extremals and only works for the characteristic function of a half-space. Thus, it was not known how it could be generalized to other situations.

Finally, we would like to point out that it is not clear a priori how to use the local extension problem to obtain sufficient conditions for $\mathcal{E}_{s,F}$. As shown in [32], minimality properties of $\mathcal{E}_{s,F}$ can be related to those of a functional acting on functions defined in the extended space \mathbb{R}^{n+1}_+ . However, applying the local theory to functionals in \mathbb{R}^{n+1}_+ leads to intricate conditions that, at first glance, cannot be written on \mathbb{R}^n .

Results and outline of the thesis

The thesis is divided in two parts. Part I is devoted to the regularity of stable solutions, while Part II deals with necessary conditions for minimality in a nonlocal setting. First, in Part I, we develop a regularity theory for stable solutions to semilinear elliptic equations with variable coefficients. It consists of Chapters 1–3 and Appendices A–F. Later, in Part II, we present a nonlocal Weierstrass extremal field theory. This portion includes Chapters 4–5 and Appendices G-J.

Part I: Regularity of stable solutions

Below, L denotes a linear elliptic operator in non-divergence form, given by (8), that is,

$$L = a_{ij}(x)\partial_{ij} + b_i(x)\partial_i.$$

As before, we assume that L is uniformly elliptic, i.e., satisfying (9) above or

$$c_0|p|^2 \le a_{ij}(x)p_ip_j \le C_0|p|^2 \quad \text{for all } p \in \mathbb{R}^n,$$
(20)

recalled here for convenience. Our regularity theory requires some additional assumptions on the coefficients beyond the basic hypothesis (10). Namely, we always assume that

$$a_{ij} \in C^{0,1}(\overline{\Omega}), \quad b_i \in L^{\infty}(\Omega).$$
 (21)

For the boundary regularity, we will also need the interior continuity assumption

$$b_i \in L^{\infty}(\Omega) \cap C^0(\Omega).$$
(22)

Chapter 1 provides interior estimates for stable solutions under the sole assumption that the nonlinearity is nonnegative. Here and in the following, whenever we write C = C(...), we mean that the constant C depends only on the quantities between the parentheses. The main result of that chapter is the following:

Theorem 5 (Theorem 1.1.1). Let L satisfy conditions (20) and (21) in $\Omega = B_1$, and let $f \in C^1(\mathbb{R})$ be nonnegative.

Let $u \in W^{3,p}(B_1)$, for some p > n, be a stable solution of -Lu = f(u) in B_1 . Then

$$\|\nabla u\|_{L^{2+\gamma}(B_{1/2})} \le C \|u\|_{L^1(B_1)},$$

where $\gamma = \gamma(n) > 0$ and $C = C(n, c_0, C_0, \|\nabla a_{ij}\|_{L^{\infty}(B_1)}, \|b_i\|_{L^{\infty}(B_1)})$. In addition,

$$||u||_{C^{\alpha}(\overline{B_{1/2}})} \le C||u||_{L^{1}(B_{1})} \quad \text{if } n \le 9,$$

where $\alpha = \alpha(n, c_0, C_0) > 0$ and $C = C(n, c_0, C_0, \|\nabla a_{ij}\|_{L^{\infty}(B_1)}, \|b_i\|_{L^{\infty}(B_1)}).$

This theorem gives two types of a priori estimates: an energy estimate in every dimension and a Hölder estimate in the optimal range $n \leq 9$. Our bounds depend only on the ellipticity constants, as well as the Lipschitz norm of a_{ij} and the L^{∞} norm of b_i . In particular, the constants do not depend on the nonlinearity. Previously, such a result was only known to hold for the Laplacian; see [24].

Notice that u is assumed to have three derivatives. This is needed to have a cancellation which removes the nonlinearity in the stability condition, ensuring that the estimates do not depend on f. Assuming the coefficient b_i to be smooth, one can show the existence of third derivatives for bounded strong solutions. However, in our boundary regularity result below, thanks to an approximation argument, we will not need this additional assumption.

In order to establish Theorem 5, we test the stability condition with variants of the test functions in [24]. Since our problem is not variational, the characterization of stability via the second variation is not available, and we must obtain an integral inequality from the pointwise condition in Definition 3. A first approach to the computations leads to bounds depending on the C^2 norm of a_{ij} and the C^1 norm of b_i . The main difficulty in the proofs is to obtain the claimed $C^{0,1}$ and L^{∞} dependence, which is crucial for the subsequent boundary estimates to hold in $C^{1,1}$ domains. This forces us to treat certain error integrals involving the Hessian of the solution, which are controlled via a Sternberg-Zumbrun type inequality (see Theorem 1.1.2).

Moreover, thanks to a device of [19] for the Laplacian, all our estimates are quantifiable. Instead, the proof in [24] used contradiction-compactness arguments that did not yield a quantitative control of the constants. Adapting the method from [19] to our setting adds some technical difficulties. In essence, the problem reduces to obtaining estimates for a Neumann problem, which we only know how to achieve by a Moser iteration. In Chapter 2, we prove an energy estimate up to the boundary in half-balls, valid in every dimension. This result is the boundary analogue of the energy estimate in Theorem 5, and requires the nonlinearity f to be nonnegative and nondecreasing. Its proof uses the interior estimates from Chapter 1 together with a delicate Sternberg-Zumbrun estimate up to the boundary in flat domains (Theorem 2.1.4 below). For this, we again use a variant of the test functions in [24]. The dependence on the constants is through the same coefficient norms appearing in Theorem 5. This chapter serves as an intermediate step between the interior estimates and the curved boundary estimates.

Finally, in Chapter 3, we prove boundary estimates in curved domains of class $C^{1,1}$. These estimates are the main conclusion of Part I, and can be summarized in the following:

Theorem 6 (Theorem 3.1.1). Let $\Omega \subset \mathbb{R}^n$ be a bounded $C^{1,1}$ domain. Let L satisfy conditions (9), (21), and (22) in Ω , and let $f \in C^1(\mathbb{R})$ be nonnegative, nondecreasing, and convex.

Let $u \in C^0(\overline{\Omega}) \cap W^{2,n}_{\text{loc}}(\Omega)$ be a stable solution of -Lu = f(u) in Ω , with u = 0 on $\partial\Omega$. Then,

 $\|\nabla u\|_{L^{2+\gamma}(\Omega)} \le C \|u\|_{L^1(\Omega)},$

where $\gamma = \gamma(n) > 0$ and $C = C(\Omega, c_0, C_0, \|\nabla a_{ij}\|_{L^{\infty}(\Omega)}, \|b_i\|_{L^{\infty}(\Omega)})$. In addition,

$$||u||_{C^{\alpha}(\overline{\Omega})} \le C||u||_{L^{1}(\Omega)} \quad \text{if } n \le 9,$$

where $\alpha = \alpha(n, c_0, C_0) > 0$ and $C = C(\Omega, c_0, C_0, \|\nabla a_{ij}\|_{L^{\infty}(\Omega)}, \|b_i\|_{L^{\infty}(\Omega)}).$

Similarly to the interior bounds above, Theorem 6 establishes boundary energy and Hölder estimates in the optimal range of dimensions. These are new even for the Laplacian in [24], a result which required a C^3 regularity assumption on the domain. Moreover, by adapting some ideas of [19] for the Laplacian in half-balls, we give the first quantitative proof in curved domains. It is worth mentioning that in both [24], [19], and in Theorem 6, the nonlinearity is required to be nonnegative, nondecreasing, and convex.

Our proof uses all previous results from Chapters 1 and 2. Here, the strategy is to flatten the boundary locally and then choosing appropriate test functions in the alternative integral stability inequality obtained in Chapter 1.

Notice that our result applies to strong stable solutions. In fact, since we are only assuming conditions (21) and (22) on the coefficients, our solutions need not have third derivatives, while the a priori estimates from the preceding chapters required this assumption. To circumvent this issue, we carry out an approximation argument. Namely, we consider a smooth exhaustion Ω_k of our $C^{1,1}$ domain, and we construct regular stable solutions u_k to an auxiliary problem in each Ω_k . These solutions approximate the original solution in some sense. Applying our estimates to the sequence u_k and passing to the limit then yields the result. For this method to work in $C^{1,1}$ domains, it is crucial that the bounds depend only on the $C^{0,1}$ and L^{∞} norms of a_{ij} and b_i , respectively. Indeed, when flattening the boundary by a change of variables, the solution in the new coordinates satisfies an equation of the same form but with different coefficients. Namely, the new a_{ij} coefficients involve first derivatives of the flattening map, while the b_i contain second derivatives of the same map. Thus, the stated dependence on the norms corresponds to a $C^{1,1}$ domain.

We conclude Part I with Appendices A–F, which provide certain accessory results required by the proofs of the main theorems above. In Appendix A we show that the stability condition is not equivalent to the integral inequality obtained in Chapter 1. Appendix B contains an elementary proof of the Sobolev trace inequality in the ball, which is needed in Chapter 1 to carry out the Moser iteration leading to quantitative interior estimates. In Appendix C we recall some useful interpolation inequalities of Cabré [19,20]. Appendix D serves the purpose of recalling a celebrated lemma by Simon [89] for absorbing errors in large balls. In Appendix E, we explain how to approximate $C^{1,1}$ domains from the interior by smooth sets satisfying uniform bounds, an important fact in the boundary regularity theory of Chapter 3. Finally, Appendix F is devoted to proving the uniqueness of stable solutions for convex nonlinearities, an auxiliary result needed in our approximation argument.

Part II: A nonlocal Weierstrass extremal field theory

In Chapter 4, we extend the classical Weierstrass extremal field theory to the fractional setting. As explained above, here the central theme is the notion of "field", namely, a one-parameter family of functions producing a foliation. If the field is made of extremals (i.e., solutions), then one expects to be able to construct a calibration functional. The existence of such an object proves the minimality of each extremal among competitors taking values in the foliated region.

Given an interval $I \subset \mathbb{R}$ (not necessarily open or bounded), we say that a family $\{u^t\}_{t\in I}$ of functions $u^t \colon \mathbb{R}^n \to \mathbb{R}$ is a C^2 field in \mathbb{R}^n if the map $(x, t) \mapsto u^t(x)$ is C^2 in $\mathbb{R}^n \times I$ and for each $x \in \mathbb{R}^n$ the function $t \mapsto u^t(x)$ is increasing in I. In particular, the graphs of $\{u^t\}_{t\in I}$ produce a foliation of a region $\mathcal{G} \subset \mathbb{R}^n \times \mathbb{R}$ and we may define the parameter $t = t(x, \lambda)$ as above (i.e., as the unique $t \in I$ such that $u^t(x) = \lambda$).

Our main result of Chapter 4 is the construction of a calibration for the functional $\mathcal{E}_{s,F}$, the Gagliardo-Sobolev seminorm with a potential term, introduced above. Recall that, given a bounded domain $\Omega \subset \mathbb{R}^n$, we write $Q(\Omega) = (\mathbb{R}^n \times \mathbb{R}^n) \setminus (\Omega^c \times \Omega^c)$, where $\Omega^c = \mathbb{R}^n \setminus \Omega$. The existence of a calibration for $\mathcal{E}_{s,F}$ was an important open problem that we have solved in the following:

Theorem 7 (Theorem 4.1.3). Let $I \subset \mathbb{R}$ be an interval, $\Omega \subset \mathbb{R}^n$ a bounded domain, and $s \in (0,1)$. Let $\{u^t\}_{t \in I}$ be a C^2 field in \mathbb{R}^n satisfying

$$|u^t(x)| + |\partial_t u^t(x)| \le C$$
 for all $x \in \mathbb{R}^n$ and $t \in I$,

for some constant C. Consider the admissible functions

$$\mathcal{A}_s = \{ w \in C^0(\mathbb{R}^n) \cap L^\infty(\mathbb{R}^n) : \operatorname{graph} w \subset \mathcal{G} \},\$$

where

$$\mathcal{G} = \{ (x, \lambda) \in \mathbb{R}^n \times \mathbb{R} \colon \lambda = u^t(x) \text{ for some } t \in I \}.$$

Given $t_0 \in I$ and $F \in C^1(\mathbb{R})$, let $\mathcal{C}_{s,F}$ be the functional

$$\begin{aligned} \mathcal{C}_{s,F}(w) &:= c_{n,s} \operatorname{P.V.} \iint_{Q(\Omega)} \int_{u^{t_0}(x)}^{w(x)} \frac{u^t(x) - u^t(y)}{|x - y|^{n + 2s}} \Big|_{t = t(x,\lambda)} \mathrm{d}\lambda \, \mathrm{d}x \, \mathrm{d}y - \int_{\Omega} F(w(x)) \, \mathrm{d}x \\ &+ \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|u^{t_0}(x) - u^{t_0}(y)|^2}{|x - y|^{n + 2s}} \, \mathrm{d}x \, \mathrm{d}y \end{aligned}$$

defined for $w \in \mathcal{A}_s$, where $c_{n,s}$ is the positive constant in the definition of $\mathcal{E}_{s,F}$.

If $\{u^t\}_{t\in I}$ is a field of extremals, that is, if

$$(-\Delta)^s u^t = F'(u^t) \quad in \ \Omega \quad for \ all \ t \in I,$$

then $\mathcal{C}_{s,F}$ is a calibration for $\mathcal{E}_{s,F}$ and u^{t_0} . More precisely, $\mathcal{C}_{s,F}$ satisfies the three properties

 $(\mathcal{C}1) \ \mathcal{C}_{s,F}(u^{t_0}) = \mathcal{E}_{s,F}(u^{t_0}).$

(C2) $\mathcal{C}_{s,F}(w) \leq \mathcal{E}_{s,F}(w)$ for all $w \in \mathcal{A}_s$ such that $w \equiv u^{t_0}$ in Ω^c .

(C3) $\mathcal{C}_{s,F}(w) = \mathcal{C}_{s,F}(\widetilde{w})$ for all $w, \widetilde{w} \in \mathcal{A}$ with $w \equiv \widetilde{w} \equiv u^{t_0}$ in Ω^c .

As a consequence, the function u^{t_0} minimizes $\mathcal{E}_{s,F}$ among functions w in \mathcal{A}_s such that $w \equiv u^{t_0}$ in Ω^c .

In fact, Theorem 7 is a simplified version of our main result in Theorem 4.1.3 below. There, in addition to fields of extremals, we consider families of super and subsolutions, which suffice to prove the minimality of u^{t_0} . The regularity of the field can also be

Recall that, prior to our result above, a nonlocal calibration had only been found in the setting of nonlocal minimal surfaces, in [18,80]. Before our work, a calibration or null-Lagrangian structure for fractional quadratic functionals was not known. Thus, it was not even clear whether a calibration such as $C_{s,F}$ could exist.

We first succeeded in constructing many functionals satisfying properties (C1) and (C2) above, but the null-Lagrangian property (C3) was either not satisfied or seemed too difficult to prove. Coming back to the work [18] on the nonlocal perimeter, we noticed that the calibration there could be written as a sum of two terms: one depending on the nonlocal mean curvature (the first variation of the perimeter) and another depending only on the exterior datum. Inspired by this observation, we searched for such a structure in the classical local theory but, to our surprise, it was never mentioned in the literature. Eventually, we were able to show that the classical calibration $C_{\rm L}$ could be written in terms of the Euler-Lagrange equation and the Neumann condition satisfied by the field. In particular, in a field of extremals, such an expression depends only on the values of the function on the boundary. This discovery allowed us to guess a natural calibration candidate for $\mathcal{E}_{s,F}$, obtained by simply considering the fractional Euler-Lagrange and Neumann operators in the previous identity. By construction, the candidate satisfied properties (C1) and (C3), and we later proved that it also satisfied (C2).

Finally, in Chapter 5, we extend the fractional construction from Chapter 4 to more general nonlocal elliptic energy functionals. For this, the idea is to carry out the process explained in the previous paragraph, identifying a natural notion of ellipticity. Our main result applies to energy functionals of the form

$$\mathcal{E}_{\mathrm{N}}(w) = \frac{1}{2} \iint_{Q(\Omega)} G_{\mathrm{N}}(x, y, w(x), w(y)) \,\mathrm{d}x \,\mathrm{d}y,$$

where $G_N = G_N(x, y, a, b)$ is a function with $G_N(x, y, a, b) = G_N(y, x, b, a)$ and satisfying the ellipticity condition

$$\partial_{ab}^2 G_{\rm N}(x, y, a, b) \le 0. \tag{23}$$

Here, we are only interested in the most general class of Lagrangians for which our proof above works and, hence, we do not make any growth and regularity assumptions on the Lagrangian G_N . In that sense, our next result is only formal, but can be made fully rigorous (as Theorem 7 above) for appropriate classes of Lagrangians, which would be modeled after each particular nonlinear problem. This is what we mean by "sufficiently regular for G_N " in the statement below. We have the following:

Theorem 8. Let $I \subset \mathbb{R}$ be an interval and let $\Omega \subset \mathbb{R}^n$ be a bounded domain. Given a smooth function $G_N = G_N(x, y, a, b)$, with $G_N(x, y, a, b) = G_N(y, x, b, a)$, and satisfying the ellipticity condition (23), let $\{u^t\}_{t \in I}$ be a field in \mathbb{R}^n which is sufficiently regular for G_N .

Given $t_0 \in I$, let \mathcal{C}_N be the functional

$$\mathcal{C}_{\mathrm{N}}(w) := \iint_{Q(\Omega)} \int_{u^{t_0}(x)}^{w(x)} \left\{ \partial_a G_{\mathrm{N}}(x, y, u^t(x), u^t(y)) \right\} \Big|_{t=t(x,\lambda)} \mathrm{d}\lambda \, \mathrm{d}x \, \mathrm{d}y + \mathcal{E}_{\mathrm{N}}(u^{t_0})$$

defined in a set \mathcal{A}_N of sufficiently regular admissible functions $w \colon \mathbb{R}^n \to \mathbb{R}$ (for G_N) satisfying graph $w \subset \mathcal{G}$, where

$$\mathcal{G} = \{ (x, \lambda) \in \mathbb{R}^n \times \mathbb{R} \colon \lambda = u^t(x) \quad \text{for some } t \in I \}.$$

If $\{u^t\}_{t\in I}$ is a field of extremals, that is, if

$$\int_{\mathbb{R}^n} \partial_a G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \, \mathrm{d}y = 0 \quad \text{for all} \quad x \in \Omega \quad \text{and} \quad t \in I,$$

then C_N is a calibration for \mathcal{E}_N and u^{t_0} . As a consequence, the function u^{t_0} minimizes \mathcal{E}_N among functions w in \mathcal{A}_N such that $w \equiv u^{t_0}$ in Ω^c .

As a first application of Theorem 8, we prove that monotone solutions are minimizers. While such a result was known for functionals with an existence and regularity theorem for minimizers, our calibration proof does not need to assume the existence of a minimizer a priori. This allows us to prove the minimality of monotone solutions for more general functionals. Such a question was motivated by the fractional version of a conjecture of De Giorgi conjecture on the symmetry of monotone solutions to the Allen-Cahn equation, see the survey [36].

As a second application, we show that minimizers of \mathcal{E}_{N} are viscosity solutions. This is perhaps surprising, since here minimality is assumed instead of being concluded. Previous proofs [5,66,87] of such a result needed a weak comparison principle for the Euler-Lagrange equation, restricting the class of functionals where they could be applied. However, a calibration argument will yield the viscosity property in a more general setting, without the need of a comparison principle.

Part II includes the following appendices expanding on the theory of nonlocal calibrations. In Appendix G, following Chapter 4, we apply the local theory to the Caffarelli-Silvestre extension to give a calibration for the fractional Laplacian in the extended space. Appendix H enumerates other natural candidates for a calibration associated to the fractional Laplacian; some of them are shown not to be calibrations. In Appendix I we relate the ellipticity condition for the general Lagrangian in Chapter 5 to a strong comparison principle for nonlocal nonlinear operators. Finally, in Appendix J we apply our calibration from Chapter 5 to the nonlocal total variation, relating it with the calibration for the nonlocal perimeter constructed in [18].

Part I

Regularity of stable solutions to semilinear elliptic problems
Chapter 1

Interior regularity

In this chapter, we extend the interior estimates for stable solutions in [24] to operators with variable coefficients. We show that stable solutions to the semilinear elliptic equation $a_{ij}(x)u_{ij} + b_i(x)u_i + f(u) = 0$ are Hölder continuous in the optimal range of dimensions $n \leq 9$. Our bounds are independent of the nonlinearity $f \in C^1$, which we assume to be nonnegative.

The main achievement of our work is to make the constants in our estimates depend on the $C^{0,1}$ norm of a_{ij} and the L^{∞} norm of b_i , instead of their C^2 and C^1 norms, respectively, which arise in a first approach to the computations.

1.1 Introduction

Let $\Omega \subset \mathbb{R}^n$ be a bounded domain and $f \colon \mathbb{R} \to \mathbb{R}$ a C^1 function. We consider stable solutions (defined below) $u \colon \overline{\Omega} \to \mathbb{R}$ to the semilinear equation

$$-Lu = f(u) \quad \text{in } \Omega, \tag{1.1.1}$$

where L is a second order linear elliptic differential operator of the form

$$Lu = a_{ij}(x)u_{ij} + b_i(x)u_i, \quad a_{ij}(x) = a_{ji}(x).$$
(1.1.2)

For simplicity, throughout this chapter we assume that the coefficients are smooth up to the boundary, i.e., we have $a_{ij} \in C^{\infty}(\overline{\Omega})$ and $b_i \in C^{\infty}(\overline{\Omega})$. However, our results continue to hold for merely Lipschitz a_{ij} and bounded b_i .

The purpose of this chapter is to extend the recent results of Cabré, Figalli, Ros-Oton, and Serra in [24] and of Cabré in [20] for the Laplacian to the above operators with variable coefficients. In [24], the authors solved a long-standing conjecture concerning the regularity of stable solutions to semilinear problems. They showed that stable solutions are bounded (and hence smooth) in dimension $n \leq 9$. This result is optimal, since there are examples of singular stable solutions for $n \geq 10$.

In the papers [20, 24], the authors obtain universal a priori estimates that do not depend on the nonlinearity f. They prove interior regularity bounds, assuming $f \ge 0$, and boundary regularity estimates on C^3 domains, assuming $f \ge 0$, $f' \ge 0$, and $f'' \ge 0$. The boundary result applies only to solutions vanishing on the boundary.

The first main interest in extending these results to operators with variable coefficients (besides possible future applications to nonlinear problems) is to simplify the boundary regularity arguments, even for the Laplacian. Indeed, starting from a curved boundary, the proof of regularity in [24] requires a delicate blow-up and Liouville theorem argument, which is needed in order to apply a result of theirs only available on a flat boundary. In addition, this proof is by contradiction-compactness and does not allow to quantify the constants in the estimates. Recently in [20], Cabré has given a quantitative proof of this result for the Laplacian in the case of a flat boundary. On the other hand, a curved boundary can be flattened out by a change of variables. Note now that, in the new coordinates, the Laplacian is written as an operator of the form (1.1.2). It is therefore natural to establish quantitative a priori estimates for our family of equations in the half-ball. By extending the techniques in [20] to operators with variable coefficients, we will avoid the intricate blow-up and Liouville theorem result form [24] as well as the compactness part.

An important feature of our estimates is that they depend only on the ellipticity constants and on the norms $\|\nabla a_{ij}\|_{L^{\infty}}$ and $\|b_i\|_{L^{\infty}}$ of the coefficients. The main difficulty in our proofs will be to obtain this dependence instead of on $\|a_{ij}\|_{C^2}$ and $\|b_i\|_{C^1}$, which are the norms that appear naturally in a first approach to the computations. This will be especially relevant for boundary regularity (treated in the next chapters), since it will allow us to relax the C^3 regularity requirement of the domain in [24]. The key point here is that, as mentioned before, flattening the boundary transforms the Laplacian into an operator of the form (1.1.2), with coefficients a_{ij} and b_i involving first and second derivatives of the boundary surface, respectively. Thus, a $C^{0,1}$, C^0 bound of the coefficients a_{ij} , b_i would correspond to a C^2 bound of the boundary. In fact, the result holds in weaker domains, as shown in Chapters 2 and 3.

Moreover, our methods will also be useful in a future work where we treat the case of the Laplacian with non-homogeneous boundary conditions. Recall that the previous papers [24] and [20] require strongly that the solutions vanish on the boundary. Flattening the boundary, we will be able to reduce the problem to an equation on the half-space for an operator of the form (1.1.2) with zero boundary conditions and an additional source term.

The study of the regularity of stable solutions was initiated in the seventies by Crandall and Rabinowitz in [44]. There, they showed the boundedness of stable solutions when $n \leq 9$ for exponential and power-type nonlinearities. Their work was motivated by problems in combustion [60], commonly known as "Gelfand-type problems"; for more information on these problems, we refer the reader to the monograph of Dupaigne [48] (see also the Introduction or Chapter 3 of this thesis). Later, in the mid-nineties, Brezis [11] asked for an extension of this regularity result to a larger class of nonlinearities. The boundedness of stable solutions was proven by Nedev [78] for $n \leq 3$, and by Cabré [16] for n = 4. The optimal dimension $n \leq 9$ remained open until it was finally reached by Cabré, Figalli, Ros-Oton, and Serra in [24].

1.1.1 The setting. Stability.

We are interested in the class of stable solutions to the semilinear equation (1.1.1). Assume that the domain $\Omega \subset \mathbb{R}^n$ is smooth. We say that u is a *stable* solution of (1.1.1) if there exists a function $\varphi \in C^0(\overline{\Omega}) \cap C^2(\Omega)$ such that

$$\begin{cases} -L\varphi \geq f'(u)\varphi & \text{in }\Omega, \\ \varphi > 0 & \text{in }\Omega, \\ \varphi = 0 & \text{on }\partial\Omega. \end{cases}$$
(1.1.3)

Equivalently, a solution is stable when the principal Dirichlet eigenvalue of the linearized equation is nonnegative; see [9]. We denote the linearized equation at u by

$$J_u \varphi = L \varphi + f'(u) \varphi, \qquad (1.1.4)$$

also known as the Jacobi operator.

Since the coefficients of L are smooth, our operator (1.1.2) can be written in divergence form as

$$\mathcal{L}u = \partial_i \left(a_{ij}(x)\partial_j u \right) + d_i(x)\partial_i u \tag{1.1.5}$$

for certain appropriate coefficients d_i . Now, recall that our bounds for non-divergence operators depend on the norms $||a_{ij}||_{C^1}$ and $||b_i||_{C^0}$ of the coefficients. As a consequence, our results continue to hold for every divergence-form operator \mathcal{L} as in (1.1.5), with constants depending on $||a_{ij}||_{C^1}$ and $||d_i||_{C^0}$ instead.

We assume that the symmetric coefficient matrix $A(x) = (a_{ij}(x))$ is uniformly elliptic, i.e., there are positive constants c_0 , C_0 such that

$$c_0|p|^2 \le a_{ij}(x)p_ip_j \le C_0|p|^2$$
 for all $p \in \mathbb{R}^n$.

This condition will be written as $c_0 \leq A(x) \leq C_0$. In particular, the matrix A(x) is positive definite and defines a norm

$$|p|_{A(x)} := (a_{ij}(x)p_ip_j)^{1/2}$$

on vectors $p \in \mathbb{R}^n$.

For variational equations $-\partial_i (a_{ij}(x)\partial_j u) = f(u)$ stability is equivalent to the nonnegativity of the second variation of the associated energy functional. This provides the useful integral inequality

$$\int_{\Omega} f'(u)\xi^2 \,\mathrm{d}x \le \int_{\Omega} |\nabla\xi|^2_{A(x)} \,\mathrm{d}x,\tag{1.1.6}$$

satisfied by all test functions $\xi \in C_c^1(\Omega)$. A key strategy to derive a priori estimates in that setting is to choose appropriate test functions in (1.1.6). When chosen correctly in terms of the Jacobi operator, the test functions allow to get rid of the nonlinearity within the proofs. This is what is done for the Laplacian in [24].

Since our operator L does not have variational structure, (1.1.6) is not available. Nevertheless, we are able to exploit the pointwise stability condition (1.1.3) for φ to obtain a convenient integral inequality which does not involve the function φ . We will use it as replacement of (1.1.6) in our non-variational setting. To derive the integral inequality, we first write the operator L in divergence form as in (1.1.5) with $d_i(x) = \hat{b}_i(x)$, where \hat{b} is the vector field given by

$$\hat{b}_i(x) = b_i(x) - \partial_k a_{ki}(x), \qquad (1.1.7)$$

hence $Lu = \partial_i (a_{ij}(x)u_j) + \hat{b}_i(x)u_i$. Now, for a test function $\xi \in C_c^1(\Omega)$, multiply (1.1.3) by ξ^2/φ and integrate by parts to obtain

$$\int_{\Omega} f'(u)\xi^2 \, \mathrm{d}x \le \int_{\Omega} \left(A(x)\nabla\varphi \cdot \nabla\left(\frac{\xi^2}{\varphi}\right) - \widehat{b}(x) \cdot \frac{\xi^2}{\varphi}\nabla\varphi \right) \, \mathrm{d}x$$
$$= \int_{\Omega} \left(-|\xi\nabla\log\varphi|^2_{A(x)} + 2A(x)\xi\nabla\log\varphi \cdot \nabla\xi - \xi\,\widehat{b}(x) \cdot \xi\,\nabla\log\varphi \right) \, \mathrm{d}x.$$

Using that

$$2A(x)\xi\nabla\log\varphi\cdot\left(\nabla\xi-\tfrac{1}{2}\xi A^{-1}(x)\widehat{b}(x)\right)-|\xi\nabla\log\varphi|_{A(x)}^2\leq\left|\nabla\xi-\tfrac{1}{2}\xi A^{-1}(x)\widehat{b}(x)\right|_{A(x)}^2,$$

we deduce

$$\int_{\Omega} f'(u)\xi^2 \,\mathrm{d}x \le \int_{\Omega} \left| \nabla \xi - \frac{1}{2}\xi A^{-1}(x)\widehat{b}(x) \right|_{A(x)}^2 \,\mathrm{d}x \quad \text{for all } \xi \in C_c^1(\Omega). \tag{1.1.8}$$

We remark that, in general, (1.1.8) is not equivalent to the stability condition (1.1.3). The main reason is that our equation -Lu = f(u) is not variational due to the presence of \hat{b} when written in the divergence form (1.1.5). In Appendix A, we give an example of the non-equivalence, and, at the same time, we characterize the drifts \hat{b} for which the equivalence holds.

1.1.2 Main results

This chapter concerns the interior regularity of stable solutions. Boundary regularity results will be treated in Chapters 2 and 3. Therefore, it suffices to consider stable solutions to -Lu = f(u) in the unit ball B_1 . A constant depending only on n, c_0 , and C_0 will be called *universal*, a terminology that we use throughout the work.

The following is our main result of this chapter, which provides interior a priori estimates for stable solutions: a Hölder bound when $n \leq 9$, and a $W^{1,2+\gamma}$ estimate in every dimension. The only requirement for the nonlinearity is $f \geq 0$, as in [24]. An important accomplishment in our estimates is that they involve the norms $\|\nabla a_{ij}\|_{L^{\infty}}$ and $\|b_i\|_{L^{\infty}}$, while a first approach to the problem leads to computations including second derivatives of a_{ij} and first derivatives of b_i . On the other hand, our bounds are independent of f. Here and throughout the work, when we write $C = C(\ldots)$ we mean that the constant Cdepends only on the quantities inside the parentheses.

Theorem 1.1.1. Let $u \in C^{\infty}(\overline{B}_1)$ be a stable solution of -Lu = f(u) in $B_1 \subset \mathbb{R}^n$, for some nonnegative function $f \in C^1(\mathbb{R})$.

Then

$$\|\nabla u\|_{L^{2+\gamma}(B_{1/2})} \le C \|u\|_{L^{1}(B_{1})}, \tag{1.1.9}$$

where $\gamma = \gamma(n) > 0$ and $C = C(n, c_0, C_0, \|\nabla a_{ij}\|_{L^{\infty}(B_1)}, \|b_i\|_{L^{\infty}(B_1)})$. In addition,

$$\|u\|_{C^{\alpha}(\overline{B}_{1/2})} \le C \|u\|_{L^{1}(B_{1})} \quad \text{if } n \le 9,$$
(1.1.10)

where $\alpha = \alpha(n, c_0, C_0) > 0$ and $C = C(n, c_0, C_0, \|\nabla a_{ij}\|_{L^{\infty}(B_1)}, \|b_i\|_{L^{\infty}(B_1)}).$

For applications, it may be useful to point out that the result only needs a_{ij} to be Lipschitz and b_i to be bounded. Our direct computations within the proofs assume $a_{ij} \in C^1$ and $b_i \in C^1$ in order to evaluate certain identities pointwise. However, we only need these to be meaningful in a weaker sense; see Remarks 1.3.1 and 1.5.2. Similarly, we only need u to be C^2 and to have weak third derivatives. These last conditions seem to require more regularity of the drift b; see Remark 1.3.1. When we treat the boundary regularity in Chapters 2 and 3, we will make weaker, more precise, regularity assumptions on the data.

The proof of Theorem 1.1.1 will rely on our second main result, Theorem 1.1.2 below, and its consequences. It consists of two types of Hessian estimates. The first one, (1.1.11), is an extension of the geometric stability condition due to Sternberg and Zumbrun [91] to operators with variable coefficients. The second one, (1.1.13)-(1.1.14), controls the L^1 norm of the "Hessian times the gradient", $|D^2u||\nabla u|$, in balls and annuli by the L^2 norm squared of the gradient whenever the lower order coefficients are small. **Theorem 1.1.2.** Let $u \in C^{\infty}(\overline{B}_1)$ be a stable solution of -Lu = f(u) in B_1 , for some function $f \in C^1(\mathbb{R})$. Assume that

$$||DA||_{L^{\infty}(B_1)} + ||b||_{L^{\infty}(B_1)} \le \varepsilon$$

for some $\varepsilon > 0$. Then

$$\int_{B_1} \mathcal{A}^2 \eta^2 \,\mathrm{d}x \le \int_{B_1} |\nabla u|^2_{A(0)} \left(|\nabla \eta|^2_{A(x)} + C\varepsilon |\nabla (\eta^2)| + C\varepsilon^2 \eta^2 \right) \,\mathrm{d}x + C\varepsilon \int_{B_1} |D^2 u| |\nabla u| \eta^2 \,\mathrm{d}x$$

$$(1.1.11)$$

for all $\eta \in C_c^{\infty}(B_1)$, where C is a universal constant and

$$\mathcal{A} := \begin{cases} \left(\operatorname{tr}(A(x)D^2 u A(0)D^2 u) - |\nabla u|_{A(0)}^{-2}|D^2 u A(0)\nabla u|_{A(x)}^2 \right)^{1/2} & \text{if } \nabla u \neq 0 \\ 0 & \text{if } \nabla u = 0. \end{cases}$$
(1.1.12)

Assume in addition that $f \geq 0$. If $\varepsilon \leq \varepsilon_0$, then

$$\| |\nabla u| D^2 u \|_{L^1(B_{3/4})} \le C \| \nabla u \|_{L^2(B_1)}^2$$
(1.1.13)

and

$$\|\nabla u\| D^2 u\|_{L^1(B_{1/2} \setminus B_{1/4})} \le C \|\nabla u\|_{L^2(B_1 \setminus B_{1/8})}^2, \tag{1.1.14}$$

where $\varepsilon_0 > 0$ and C are universal constants.

Our first inequality (1.1.11) generalizes the Sternberg-Zumbrun estimate for the Laplacian, which corresponds to the case $\varepsilon = 0$ and A(0) = I. The peculiar form of the function \mathcal{A} in (1.1.12) (the coefficients of A are evaluated both at x and 0) will guarantee that the direct computations within our proofs give dependence on the norm $||a_{ij}||_{C^1}$, instead of $||a_{ij}||_{C^2}$ for other choices of \mathcal{A} . In this direction, it is worth noting that the classical Sternberg-Zumbrun result and the function \mathcal{A} have a Riemannian analogue (found by Farina, Sire, and Valdinoci [56]) which can be related to our Euclidean setting with variable coefficients. The estimate from the Riemannian framework leads to bounds depending on $||a_{ij}||_{C^2}$. We elaborate on these topics further in Remarks 1.3.2 and 1.3.3.

The "Hessian times the gradient" estimates (1.1.13)-(1.1.14) rely on the inequality (1.1.11) with sufficiently small errors ε , and will require the assumption $f \ge 0$. While the bound on annuli (1.1.14) can be deduced from the one in balls (1.1.13) by a standard scaling and covering argument, we include it in the statement since it will be crucial in the proof of the Hölder estimate in Theorem 1.1.1.

1.1.3 Structure of the proof

By a scaling and covering argument, it suffices to obtain the a priori estimates from Theorem 1.1.1 in small balls. There, the problem can be written as an equation in the unit ball involving an operator L that is close to the Laplacian, i.e., whose coefficients satisfy A(0) = I and $\|\nabla a_{ij}\|_{L^{\infty}(B_1)} + \|b_i\|_{L^{\infty}(B_1)} \leq \varepsilon$, with ε small. We explain this in more detail in Section 1.2 below.

The key estimates leading to Theorem 1.1.1 are contained in Propositions 1.1.3, 1.1.4, and 1.1.5 below. Our proofs are all quantitative as in the paper [20] and avoid the compactness argument from the previous work [24]. The proofs of the first two propositions

use the Hessian estimates of Theorem 1.1.2 above. In particular, this forces us to prove the Sternberg-Zumbrun inequality before the crucial weighted L^2 estimate for the radial derivative (Proposition 1.1.4). It is worth noting that, for the Laplacian, these two results are independent from each other (and hence can be obtained in any order, as in the works [24] and [20]), while this is no longer the case for operators with variable coefficients.

In the first proposition, we control the L^2 norm of the gradient by the L^1 norm of the solution under a smallness condition on the coefficients, namely, when the error ε is sufficiently small. This is a direct consequence of Theorem 1.1.2 and the interpolation inequalities of Cabré in [20]. We recall these inequalities in Appendix C.

Proposition 1.1.3. Let $u \in C^{\infty}(\overline{B}_1)$ be a stable solution of -Lu = f(u) in B_1 , for some nonnegative function $f \in C^1(\mathbb{R})$. Assume that

$$||DA||_{L^{\infty}(B_1)} + ||b||_{L^{\infty}(B_1)} \le \varepsilon$$

for some $\varepsilon > 0$. If $\varepsilon \leq \varepsilon_0$, then

$$\|\nabla u\|_{L^2(B_{1/2})} \le C \|u\|_{L^1(B_1)},\tag{1.1.15}$$

where $\varepsilon_0 > 0$ and C are universal constants.

The second proposition is a weighted L^2 estimate for the radial derivative in a ball by the full gradient in an annulus. It is here that we need $n \leq 9$. Again, we will assume that the coefficient error ε is small and that the nonlinearity is nonnegative $f \geq 0$. Here and throughout the paper we use the notation

$$r = |x|, \quad u_r = \frac{x}{|x|} \cdot \nabla u$$

for the radial derivative.

Proposition 1.1.4. Let $u \in C^{\infty}(\overline{B}_1)$ be a stable solution of -Lu = f(u) in B_1 , for some nonnegative function $f \in C^1(\mathbb{R})$. Assume that

$$||DA||_{L^{\infty}(B_1)} + ||b||_{L^{\infty}(B_1)} \le \varepsilon$$

for some $\varepsilon > 0$.

If $3 \le n \le 9$ and $\varepsilon \le \varepsilon_0$, then

$$\int_{B_{\rho}} r^{2-n} u_r^2 \, \mathrm{d}x \le C \int_{B_{2\rho} \setminus B_{\rho}} r^{2-n} |\nabla u|^2 \, \mathrm{d}x + C\varepsilon \int_{B_{4\rho}} r^{3-n} |\nabla u|^2 \, \mathrm{d}x \tag{1.1.16}$$

for all $\rho \leq 1/4$, where $\varepsilon_0 > 0$ and C are universal constants.

Notice that this result requires $n \geq 3$. However, adding superfluous variables to the solution, we can also use it when $n \leq 2$.

Our inequality (1.1.16) in Proposition 1.1.4 is an analogue of Lemma 2.1 in [24], where the authors obtain a similar bound for the Laplacian ($\varepsilon = 0$ and A(0) = I) without the nonnegativity assumption on f. Recall that this assumption is needed in the Hessian estimates (1.1.13)-(1.1.14) in Theorem 1.1.2 above, which will allow us to treat a weighted $|D^2u||\nabla u|$ error term which does not appear for the Laplacian. We will be able to control this error by writing it as an infinite sum on dyadic annuli, pulling the weight out of the integral in each annulus, and applying the bound (1.1.14). Finally in the third proposition we show that, under the assumption that A(0) = I, (generalized) superharmonic functions are controlled by the radial derivative plus an error involving the full gradient in L^1 . This is an extension of Lemma 4.1 in Cabré [20] to operators with variable coefficients.

Proposition 1.1.5. Let $u \in C^{\infty}(\overline{B}_1)$ be superharmonic in the sense that $Lu \leq 0$ in B_1 . Assume that

$$A(0) = I$$
 and $||DA||_{L^{\infty}(B_1)} + ||b||_{L^{\infty}(B_1)} \le \varepsilon$

for some $\varepsilon > 0$.

Then there exists a constant t, which depends on u, such that

 $||u - t||_{L^{1}(B_{1} \setminus B_{1/8})} \le C ||u_{r}||_{L^{1}(B_{1} \setminus B_{1/8})} + C\varepsilon ||\nabla u||_{L^{1}(B_{1})},$

where C is a constant depending only on n and c_0 .

Our proof of Proposition 1.1.5 is by comparison with harmonic functions. The required L^1 estimates for harmonic functions will follow by duality from the L^{∞} bounds of a Neumann problem. Such L^{∞} estimates are the most technical part of the argument, where we use a Moser iteration in the spirit of Winkert [96] to deduce the uniform bounds.

The $W^{1,2+\gamma}$ result (1.1.9) in Theorem 1.1.1 will follow from the Hessian estimate on balls (1.1.13) in Theorem 1.1.2 together with Proposition 1.1.3. To show it, we first control the L^2 norm of the gradient uniformly on level sets by the Dirichlet integral in a ball, and hence by the L^1 norm of the solution. A device from [24] will then allow us to deduce the higher integrability.

To prove the Hölder estimate (1.1.10) in Theorem 1.1.1, we will show that the scaleinvariant weighted integral $\int_{B_{\rho}} r^{2-n} |\nabla u|^2$ decays algebraically. In the previous works [24] and [20], the authors proved the decay of the weighted radial derivative instead. They could later deduce the C^{α} estimate by either averaging or applying a version of Morrey's embedding for radial derivatives. Here we will obtain the decay of the full gradient directly for the first time. For this, combining Propositions 1.1.3 and 1.1.5, we are able to bound the full gradient by the radial derivative on annuli in L^2 . This, together with the dyadic decomposition explained above, allows us to control the weighted integral of the gradient by that of the radial derivative (up to gradient errors). Now, Proposition 1.1.4 will yield a control of the weighted integral of the gradient in the ball by the same quantity on an annulus. A standard iteration then leads to the decay.

Our integral stability inequality (1.1.8) will be crucial in the proofs of both Theorem 1.1.2 and Proposition 1.1.4. These will follow from (1.1.8), with $\Omega = B_1$, by choosing appropriate test functions in terms of the Jacobi operator, as we explain next. Taking a test function of the form $\xi = \mathbf{c}\eta$, where \mathbf{c} and η are smooth and $\operatorname{supp} \eta \subset B_1$, the integrand on the right-hand side of (1.1.8) becomes

$$\begin{aligned} \left| \nabla \xi - \frac{1}{2} \xi A^{-1}(x) \widehat{b}(x) \right|_{A(x)}^2 &= \left| \eta \nabla \mathbf{c} + \mathbf{c} \nabla \eta - \frac{1}{2} \mathbf{c} \eta A^{-1}(x) \widehat{b}(x) \right|_{A(x)}^2 \\ &= \left| \eta \nabla \mathbf{c} \right|_{A(x)}^2 + 2A(x) \eta \nabla \mathbf{c} \cdot \mathbf{c} \nabla \eta - \eta^2 \mathbf{c} \, \widehat{b}(x) \cdot \nabla \mathbf{c} + \mathbf{c}^2 \left| \nabla \eta - \frac{1}{2} \eta A^{-1}(x) \widehat{b}(x) \right|_{A(x)}^2. \end{aligned}$$

$$(1.1.17)$$

Integrating in B_1 , the first term $\int_{B_1} |\eta \nabla \mathbf{c}|^2_{A(x)} dx$ in (1.1.17) can be integrated by parts as

$$\int_{B_1} |\eta \nabla \mathbf{c}|^2_{A(x)} \, \mathrm{d}x = \int_{B_1} \left(-\operatorname{div} \left(A(x) \nabla \mathbf{c} \right) \, \mathbf{c} \eta^2 - 2A(x) \eta \nabla \mathbf{c} \cdot \mathbf{c} \nabla \eta \right) \, \mathrm{d}x$$

and hence, by (1.1.17), (1.1.8), and rearranging terms, it follows that

$$\int_{B_1} (J_u \mathbf{c}) \, \mathbf{c} \eta^2 \, \mathrm{d}x \le \int_{B_1} \mathbf{c}^2 \left| \nabla \eta - \frac{1}{2} \eta A^{-1}(x) \widehat{b}(x) \right|_{A(x)}^2 \mathrm{d}x. \tag{1.1.18}$$

The key idea now is to choose the **c** function in a way such that $J_u \mathbf{c}$ becomes independent of the nonlinearity. This will yield universal a priori estimates for stable solutions. Since usolves the equation -Lu = f(u), taking a derivative, we have that $f'(u)\nabla u = -\nabla Lu$ and hence $J_u\nabla u = L\nabla u - \nabla Lu$ no longer involves f. This computation suggests that we choose **c** as a function of the gradient of u. Thus, to prove the estimate for \mathcal{A} in Theorem 1.1.2, we will make the choice

$$\mathbf{c}(x) = |\nabla u(x)|_{A(0)} = \left(a_{ij}(0)u_iu_j\right)^{1/2}$$

On the other hand, the weighted L^2 bound in Proposition 1.1.4 will also require to choose the auxiliary function η above carefully. The test functions leading to this estimate are

$$\mathbf{c}(x) = x \cdot \nabla u = r u_r$$
 and $\eta(x) = |x|_{A^{-1}(0)}^{\frac{2-n}{2}} \zeta$,

where $\zeta \in C_c^{\infty}(B_1)$ is a cut-off.

We note that our test functions above are the ones used in the paper [24] under the linear transformation $x \mapsto A^{1/2}(0)x$, where $A^{1/2}(0)$ is the positive square root of the matrix A(0). These seem to be the simplest functions leading to a priori estimates in the variable coefficients framework. Moreover, thanks to the particular form of these functions, all direct computations within our proofs will only involve first derivatives of the coefficients A and b, while other choices of functions require two derivatives of A. A suitable integration by parts will yield bounds in terms of the norms $||A||_{C^1}$ and $||b||_{C^0}$, as in the results mentioned above.

1.1.4 Outline of the chapter

In Section 1.2 we briefly comment on the invariance of stability under affine transformations. Section 1.3 is devoted to proving Theorem 1.1.2 and Proposition 1.1.3. In Section 1.4 we prove the $W^{1,2+\gamma}$ bound (1.1.9) from Theorem 1.1.1. Section 1.5 contains the proof of Proposition 1.1.4. In Section 1.6 we prove Proposition 1.1.5. Finally, in Section 1.7 we prove the Hölder bound (1.1.10) in Theorem 1.1.1.

1.2 Preliminaries: Invariance under affine transformations

To prove Theorem 1.1.1, we will analyze the semilinear equation -Lu = f(u) in small balls. Since the class of stable solutions is invariant under affine transformations, the question reduces to studying an equation in the unit ball involving an operator that is close to the Laplacian. After proving the necessary estimates in this setting, the theorem will follow from a scaling and covering argument. It is worth mentioning that the nonnegativity of the nonlinearity, which is required in our main results, is preserved under these transformations.

We now explain this invariance with more detail in different particular situations. First we study the equation under translations and scalings. These simple yet important transformations will be used several times throughout the paper. Secondly, we consider the equation under general linear transformations. These allow us to reduce ourselves to the case where the coefficient matrix is the identity at the origin. Notice that this is only required in Proposition 1.1.5, but will be crucial in the proof of the C^{α} bound (1.1.10) in Theorem 1.1.1 given in Section 1.7.

As mentioned in the Introduction, the bounds in our a priori estimates depend only on the ellipticity constants c_0 and C_0 and on the quantity

$$\|DA\|_{L^{\infty}(B_1)} + \|b\|_{L^{\infty}(B_1)} \le \varepsilon \tag{1.2.1}$$

involving the coefficients. As we will see now, the two norms in (1.2.1) have the same scaling. It is therefore natural to state our results in terms of this quantity.

(i) **Translation and scale invariance**. If u is a stable solution of -Lu = f(u) in a ball $B_{\rho}(y)$, then the function $u^{y,\rho} := u(y + \rho \cdot)$ is a solution of $-L^{y,\rho}u^{y,\rho} = \rho^2 f(u^{y,\rho})$ in B_1 , where $L^{y,\rho}$ is the linear operator

$$L^{y,\rho}v = \operatorname{tr}\left(A^{y,\rho}(x)D^2v\right) + b^{y,\rho}(x)\cdot\nabla v$$

with coefficients

$$A^{y,\rho}(x) = A(y+\rho x) \quad \text{and} \quad b^{y,\rho}(x) = \rho \, b(y+\rho x).$$

The stability condition (1.1.3) in $B_{\rho}(y)$ becomes $-L^{y,\rho}\varphi^{y,\rho} \leq \rho^2 f'(u^{y,\rho})\varphi^{y,\rho}$ in B_1 , where $\varphi^{y,\rho} = \varphi(y+\rho \cdot)$ satisfies the assumptions in (1.1.3), and hence $u^{y,\rho}$ is stable. Since the coefficients satisfy the bounds

$$||DA^{y,\rho}||_{L^{\infty}(B_1)} \le \rho ||DA||_{L^{\infty}(B_{\rho}(y))}$$
 and $||b^{y,\rho}||_{L^{\infty}(B_1)} \le \rho ||b||_{L^{\infty}(B_{\rho}(y))}$

whenever $B_{\rho}(y) \subset B_R$ for some R > 0 and L is defined in this larger ball, we have

$$\|DA^{y,\rho}\|_{L^{\infty}(B_1)} + \|b^{y,\rho}\|_{L^{\infty}(B_1)} \le \rho \left(\|DA\|_{L^{\infty}(B_R)} + \|b\|_{L^{\infty}(B_R)}\right), \qquad (1.2.2)$$

which can be made small for ρ small. In particular, assuming that L is close to the Laplacian as in the statements of our propositions and by (1.2.2), we deduce the following property: if $\|DA\|_{L^{\infty}(B_R)} + \|b\|_{L^{\infty}(B_R)} \leq \varepsilon$, then

$$\|DA^{y,\rho}\|_{L^{\infty}(B_1)} + \|b^{y,\rho}\|_{L^{\infty}(B_1)} \le \rho\varepsilon \quad \text{for all } B_{\rho}(y) \subset B_R.$$

This elementary observation will be used throughout the paper.

(ii) **Invariance under linear transformations**. Given a symmetric positive definite matrix $M \in \mathbb{R}^{n \times n}$, if u is a stable solution of -Lu = f(u) in the unit ball B_1 , then the function $u^M := u(M \cdot)$ is a solution of $-L^M u^M = f(u^M)$ in $M^{-1}(B_1)$, where L^M is the operator

$$L^{M}v = \operatorname{tr}(A^{M}(x)D^{2}v) + b^{M}(x) \cdot \nabla v,$$

with coefficients

$$A^{M}(x) = M^{-1}A(Mx)M^{-1}$$
 and $b^{M}(x) = M^{-1}b(Mx).$

As above, the stability condition (1.1.3) in B_1 becomes $-L^M \varphi^M \leq f'(u^M) \varphi^M$ in $M^{-1}(B_1)$, where $\varphi^M = \varphi(M \cdot)$, and hence u^M is a stable solution.

If M satisfies $\sqrt{c_0} \leq M \leq \sqrt{C_0}$, then the new coefficients A^M are uniformly elliptic with $c_0/C_0 \leq A^M(x) \leq C_0/c_0$. Moreover, we have the bounds

$$|DA^{M}(x)| \le \frac{\sqrt{C_{0}}}{c_{0}}|DA(Mx)|$$
 and $|b^{M}(x)| \le \frac{1}{\sqrt{c_{0}}}|b(Mx)|,$

and taking the supremum in $x \in M^{-1}(B_1)$, using that $B_{1/\sqrt{C_0}} \subset M^{-1}(B_1)$, we deduce

$$\|DA^{M}\|_{L^{\infty}(B_{1/\sqrt{C_{0}}})} + \|b^{M}\|_{L^{\infty}(B_{1/\sqrt{C_{0}}})} \le \frac{\sqrt{C_{0}}}{c_{0}} \left(\|DA\|_{L^{\infty}(B_{1})} + \|b\|_{L^{\infty}(B_{1})}\right).$$
(1.2.3)

In particular, taking $M = A^{1/2}(0)$ as the unique positive square root of A(0), we see that $u^{A^{1/2}(0)}$ solves an elliptic equation in the ball $B_{1/\sqrt{C_0}}$ with coefficients $(A^{A^{1/2}(0)})(x)$ satisfying $(A^{A^{1/2}(0)})(0) = I$, i.e., equal to the identity at the origin. By the monotonicity of the principal eigenvalue with respect to the domain, it follows that $u^{A^{1/2}(0)}$ is also a stable solution in this ball.

It is now easy to combine these transformations with the ones given in the first part (i). For each ball $B_{\rho}(y) \subset B_1$, the function $\tilde{u} = u\left(y + \frac{\rho}{\sqrt{C_0}}A^{1/2}(y)\cdot\right)$ is a stable solution of an elliptic equation $-\tilde{L}\tilde{u} = \tilde{f}(\tilde{u})$ in B_1 . Here, \tilde{f} is the nonlinearity $\tilde{f} = \frac{\rho^2}{C_0}f$, while \tilde{L} is an operator of the form (1.1.2) with coefficients

$$\widetilde{A}(x) = A^{-1/2}(y) A \left(y + \frac{\rho}{\sqrt{C_0}} A^{1/2}(y) x \right) A^{-1/2}(y)$$

and

$$\widetilde{b}(x) = \frac{\rho}{\sqrt{C_0}} A^{-1/2}(y) b \left(y + \frac{\rho}{\sqrt{C_0}} A^{1/2}(y) x \right).$$

Notice that the matrix $\widetilde{A}(x)$ is uniformly elliptic with $c_0/C_0 \leq \widetilde{A}(x) \leq C_0/c_0$ and is equal to the identity at the origin. Furthermore, combining (1.2.3) and (1.2.2), the coefficients can be bounded by

$$\|D\widetilde{A}\|_{L^{\infty}(B_{1})} + \|\widetilde{b}\|_{L^{\infty}(B_{1})} \le \frac{\rho}{c_{0}} \left(\|DA\|_{L^{\infty}(B_{1})} + \|b\|_{L^{\infty}(B_{1})}\right)$$

As mentioned above, this observation will be important in the proof of the Hölder estimate 1.1.10 in Section 1.7 below.

1.3 Hessian and $W^{1,2}$ estimates

The goal of this section is to prove Theorem 1.1.2 and the energy estimate Proposition 1.1.3.

Recall the function $\mathcal{A}: \overline{B}_1 \to \mathbb{R}$ introduced in (1.1.12) in the statement of Theorem 1.1.2. This function can also be written as

$$\mathcal{A} = \begin{cases} \left(\|A^{1/2}(x)D^2 u A^{1/2}(0)\|_{\mathrm{HS}}^2 - |A^{1/2}(x)D^2 u A^{1/2}(0)\mathbf{n}(x)|^2 \right)^{1/2} & \text{if } \nabla u \neq 0\\ 0 & \text{if } \nabla u = 0, \end{cases}$$
(1.3.1)

where $\|\cdot\|_{\text{HS}}$ denotes the Euclidean Hilbert-Schmidt norm for matrices¹ and $\mathbf{n}(x)$ is the unit vector field $\mathbf{n} \colon \overline{B}_1 \cap \{\nabla u \neq 0\} \to \mathbb{R}$ given by

$$\mathbf{n}(x) := |\nabla u|_{A(0)}^{-1} A^{1/2}(0) \nabla u(x).$$
(1.3.2)

¹Recall that, for a matrix $M \in \mathbb{R}^{n \times n}$, this norm squared is $||M||_{\text{HS}}^2 = \text{tr}(M^T M) = \sum_{i,j=1}^n M_{ij}^2$.

The equivalence between the expressions (1.1.12) and (1.3.1) follows from the identities

$$||A^{1/2}(x)D^2uA^{1/2}(0)||_{\mathrm{HS}}^2 = \mathrm{tr}\left(A(x)D^2uA(0)D^2u\right)$$

and

$$|A^{1/2}(x)D^2uA^{1/2}(0)\mathbf{n}(x)|^2 = |\nabla u|_{A(0)}^{-2}|D^2uA(0)\nabla u|_{A(x)}^2,$$

which are easy to check.

We start by proving the bound (1.1.11) in Theorem 1.1.2, which is a generalization of the geometric stability inequality due to Sternberg and Zumbrun [91] for stable solutions to $-\Delta u = f(u)$. For this, we will test the integral stability inequality (1.1.18) with the function

$$\mathbf{c}(x) = |\nabla u|_{A(0)}$$

and a cut-off η . The proof of the remaining estimates in Theorem 1.1.2 will rely on this preliminary inequality.

Two comments are in order. First, with this choice of \mathbf{c} , our result requires an appropriate integration by parts to allow dependence of the bounds on only $||b||_{C^0}$. Secondly, after the proof, in Remarks 1.3.2 and 1.3.3 we will comment on alternative choices of \mathbf{c} and of the function \mathcal{A} .

We originally took $|\nabla u|_{A(x)}$ as our **c** function, a choice that required the regularity $A \in C^2$ and $b \in C^1$ when computing $J_u \mathbf{c}$ in the stability inequality (1.1.18). With that choice, a further integration by parts was needed to obtain bounds depending only on $||A||_{C^1}$. Instead, our function $|\nabla u|_{A(0)}$ only needs $A \in C^1$ and $b \in C^1$. Moreover, the proof with our choice is easier and we only need an integration by parts to get rid of the first derivatives of b. Note that the function $|\nabla u|_{A(x)}$ is motivated by geometric considerations and had already appeared in the Riemannian analogue of the Sternberg-Zumbrun estimates, as explained in Remark 1.3.3.

Proof of (1.1.11) in Theorem 1.1.2. Since $|\nabla u|_{A(0)}$ is not necessarily smooth when $\nabla u = 0$, we consider the smooth function

$$\mathbf{c}_{\delta} := \sqrt{|\nabla u|_{A(0)}^2 + \delta^2}$$

instead. We will apply the integral stability inequality (1.1.18) with $\mathbf{c} = \mathbf{c}_{\delta}$. In the end we will let $\delta \to 0$, which will yield the claim. Throughout this proof, the letter C denotes a generic universal constant.

By the stability inequality (1.1.18), we have the upper bound

$$\int_{B_1} \mathbf{c}_{\delta} J_u \mathbf{c}_{\delta} \eta^2 \, \mathrm{d}x \leq \int_{B_1} \left(|\nabla u|_{A(0)}^2 + \delta^2 \right) \left| \nabla \eta - \frac{1}{2} \eta A^{-1}(x) \widehat{b}(x) \right|_{A(x)}^2 \, \mathrm{d}x \\
\leq \int_{B_1} \left(|\nabla u|_{A(0)}^2 + \delta^2 \right) \left(|\nabla \eta|_{A(x)}^2 + C\varepsilon |\nabla (\eta^2)| + C\varepsilon^2 \eta^2 \right) \, \mathrm{d}x,$$
(1.3.3)

where in the last line we have expanded the quadratic expression and applied the bounds of the coefficients.

We will bound the expression $\mathbf{c}_{\delta} J_u \mathbf{c}_{\delta} = \mathbf{c}_{\delta} L \mathbf{c}_{\delta} + f'(u) \mathbf{c}_{\delta}^2$ from below. Since

$$\partial_i \mathbf{c}_\delta = \mathbf{c}_\delta^{-1} \, u_{ik} a_{kl}(0) u_l$$

and

$$\partial_{ij}^{2} \mathbf{c}_{\delta} = \mathbf{c}_{\delta}^{-1} u_{ijk} a_{kl}(0) u_{l} + \mathbf{c}_{\delta}^{-1} u_{ik} a_{kl}(0) u_{jl} - \mathbf{c}_{\delta}^{-3} u_{ik} a_{kl}(0) u_{l} u_{jp} a_{pq}(0) u_{q}$$

we deduce

$$\mathbf{c}_{\delta} a_{ij}(x) \partial_{ij}^{2} \mathbf{c}_{\delta} = a_{kl}(0) u_{l} a_{ij}(x) u_{ijk} + \|A^{1/2}(x) D^{2} u A^{1/2}(0)\|_{\mathrm{HS}}^{2} - \frac{|\nabla u|_{A(0)}^{2}}{|\nabla u|_{A(0)}^{2} + \delta^{2}} |A^{1/2}(x) D^{2} u A^{1/2}(0) \mathbf{n}|^{2}$$
(1.3.4)
$$\geq a_{kl}(0) u_{l} a_{ij}(x) u_{ijk} + \mathcal{A}^{2}$$

and

$$\mathbf{c}_{\delta} b_i(x) \partial_i \mathbf{c}_{\delta} = a_{kl}(0) u_l b_i(x) u_{ik}. \tag{1.3.5}$$

Adding (1.3.4) and (1.3.5), we obtain

$$\mathbf{c}_{\delta} J_u \mathbf{c}_{\delta} \ge a_{kl}(0) u_l L u_k + \mathcal{A}^2 + f'(u) \mathbf{c}_{\delta}^2.$$
(1.3.6)

Differentiating the equation -Lu = f(u) in the direction of $A(0)\nabla u$, we have

$$A(0)\nabla(Lu) \cdot \nabla u = -f'(u)|\nabla u|^{2}_{A(0)}.$$
(1.3.7)

The first term on the right-hand side of (1.3.6) can be written in terms of this derivative as

$$a_{kl}(0)u_l Lu_k = A(0)\nabla(Lu) \cdot \nabla u - a_{kl}(0)u_l \partial_k a_{ij}(x)u_{ij} - a_{kl}(0)u_l \partial_k b_i(x)u_i,$$

hence, by (1.3.7) and the coefficient estimates, we can bound this expression from below as

$$a_{kl}(0)u_l Lu_k \ge -f'(u)|\nabla u|^2_{A(0)} - C\varepsilon |D^2 u||\nabla u| - a_{kl}(0)u_l \partial_k b_i(x)u_i.$$
(1.3.8)

Applying (1.3.8) in (1.3.6), since $\mathbf{c}_{\delta}^2 - |\nabla u|_{A(0)}^2 = \delta^2$, we obtain

$$\mathbf{c}_{\delta} J_{u} \mathbf{c}_{\delta} \ge \mathcal{A}^{2} + \delta^{2} f'(u) - C\varepsilon |D^{2}u| |\nabla u| - a_{kl}(0) u_{l} \partial_{k} b_{i}(x) u_{i}.$$
(1.3.9)

Multiplying (1.3.9) by η^2 and integrating, the last term $-\int_{B_1} a_{kl}(0)u_l\partial_k b_i(x)u_i\eta^2 dx$ can be integrated by parts as

$$\left| -\int_{B_1} a_{kl}(0) u_l \partial_k b_i(x) u_i \eta^2 \, \mathrm{d}x \right| = \left| \int_{B_1} b_i(x) \partial_k(a_{kl}(0) u_l u_i \eta^2) \, \mathrm{d}x \right|$$
$$\leq C\varepsilon \int_{B_1} |D^2 u| |\nabla u| \eta^2 \, \mathrm{d}x + C\varepsilon \int_{B_1} |\nabla u|^2 |\nabla (\eta^2)| \, \mathrm{d}x.$$
(1.3.10)

Combining (1.3.9), (1.3.10), and (1.3.3), rearranging terms, we obtain

$$\int_{B_1} \left(\mathcal{A}^2 + \delta f'(u) \right) \eta^2 \, \mathrm{d}x$$

$$\leq \int_{B_1} \left(|\nabla u|_{A(0)}^2 + \delta^2 \right) \left(|\nabla \eta|_{A(x)}^2 + C\varepsilon |\nabla (\eta^2)| + C\varepsilon^2 \eta^2 \right) \, \mathrm{d}x$$

$$+ C\varepsilon \int_{B_1} |D^2 u| |\nabla u| \eta^2 \, \mathrm{d}x,$$

and letting $\delta \to 0$ yields the claim.

Several remarks are in order:

Remark 1.3.1. In (1.3.7) above we took a derivative of the equation in the direction $A(0)\nabla u$ to get rid of the dependence on the nonlinearity. Instead, we could have multiplied the equation by the test function $\xi = \operatorname{div} (A(0)\nabla u \eta^2)$ and integrated by parts. Notice that this avoids having to take any derivatives of b, since the term involving it can be bounded directly.

In the argument above, we need u to have three (weak) derivatives, otherwise we cannot compute $L\mathbf{c}$ (or rather $L\mathbf{c}_{\delta}$). In [24], the authors only need to assume $u \in C^2(B_1)$ to deduce the analogue estimate for the Laplacian, since this already gives $u \in W_{\text{loc}}^{3,p}$ for all $p < \infty$. Indeed, differentiating the equation, $-\Delta u_i = f'(u)u_i \in L_{\text{loc}}^p$ and by L^p estimates they deduce $u_i \in W_{\text{loc}}^{2,p}$, hence $u \in W_{\text{loc}}^{3,p}$ for all $p < \infty$. This fact allows them to carry out a similar argument to the one explained above.²

For an operator with variable coefficients L, the regularity of the solution depends on that of the coefficients. Assuming $a_{ij} \in C^{0,1}(B_1)$, $b_i \in L^{\infty}(B_1)$, and u bounded, applying L^p estimates to the equation $-Lu = f(u) \in L^p_{loc}$, we deduce $u \in W^{2,p}_{loc}$ for all $p < \infty$ (and hence in $C^{1,\alpha}$ for all $0 < \alpha < 1$). Now, for u to be in $C^2 \cap W^{3,p}_{loc}$ we need more regularity on the drift b. To see this, taking a derivative of the equation we have $-Lu_k =$ $\partial_k a_{ij}(x)u_{ij} + \partial_k b_i(x)u_i + f'(u)u_k$ and the right hand side is in L^p for $\partial_k a_{ij} \in L^{\infty}$ (i.e., a_{ij} Lipschitz) and $\partial_k b_i \in L^p$. In particular, if $b \in W^{1,p}(B_1)$ with p > n, we deduce $u \in C^2(B_1) \cap W^{3,p}_{loc}(B_1)$. This is somewhat surprising, since our estimates do not involve any derivatives of b.

Remark 1.3.2. The following comments concern the form of the function \mathcal{A} in our a priori estimate (1.1.11) in Theorem 1.1.2. Recall that \mathcal{A} quantifies a part of the "mixed", non-symmetric matrix $A^{1/2}(x)D^2uA^{1/2}(0)$, which includes both the variable coefficients $A^{1/2}(x)$ and the constants $A^{1/2}(0)$. We are led naturally to it from the choice of test function $\mathbf{c} = |\nabla u|_{A(0)}$ in the integral stability inequality, which is the function used by Cabré, Figalli, Ros-Oton, and Serra in [24] after a linear transformation.

We could have also given an estimate for the function

$$\mathcal{A}_{x} := \left(\|A^{1/2}(x)D^{2}uA^{1/2}(x)\|_{\mathrm{HS}}^{2} - \left|A^{1/2}(x)D^{2}uA^{1/2}(x)\mathbf{n}_{A}(x)\right|^{2} \right)^{1/2}$$
(1.3.11)

involving the symmetric matrix $A^{1/2}(x)D^2uA^{1/2}(x)$, where \mathbf{n}_A is the vector field $\mathbf{n}_A(x) := |\nabla u|_{A(x)}^{-1}A^{1/2}(x)\nabla u$, by choosing the test function $\mathbf{c} = |\nabla u|_{A(x)}$ instead. However, in this case, the proof of the analogue of Theorem 1.1.2 is more involved. This is why we prefer our choice of \mathcal{A} . On the other hand, the choice $\mathbf{c} = |\nabla u|_{A(x)}$ is related to an existing Riemannian version of the Sternberg and Zumbrun inequality, which we explain next in Remark 1.3.3.

This discussion leads to the question of whether a similar estimate exists for \mathcal{A}_0 , the natural part of the simpler symmetric matrix $A^{1/2}(0)D^2uA^{1/2}(0)$, which only involves the constant coefficients $A^{1/2}(0)$. There does not seem to be a direct way to derive such an estimate from the stability inequality, since it is not clear which **c** function could lead to it. Nevertheless, when the parameter ε is small, thanks to (1.3.15) below, it can be shown that \mathcal{A}_0 is comparable to \mathcal{A} . Hence, for ε small, we can deduce the desired bound for \mathcal{A}_0 from our result (1.1.11) for \mathcal{A} . We will need this fact in the proof of the Hessian estimates in Theorem 1.1.2, as explained below.

²In fact, they are able to deduce the estimate without computing $L\mathbf{c}$ directly, but they still need to have three derivatives of the solution; see the proof of Lemma 2.1 in [24].

Remark 1.3.3. Our result (1.1.11) is related to a Riemannian analogue of the Sternberg and Zumbrun estimate found by Farina, Sire, and Valdinoci in [56]. It states that stable solutions to the equation $-\Delta_{\text{LB}}u = f(u)$ in a Riemannian manifold (M, g), where Δ_{LB} is the Laplace-Beltrami operator, satisfy the inequality

$$\int_{M} \mathcal{A}_{\text{Riem}}^{2} \eta^{2} + \int_{M} \text{Ric}(\nabla_{g} u, \nabla_{g} u) \eta^{2} \leq \int_{M} |\nabla_{g} u|_{g}^{2} |\nabla_{g} \eta|_{g}^{2}.$$
 (1.3.12)

Here, $\mathcal{A}_{\text{Riem}}$ (given by (1.3.13)) is a Riemannian analogue of the function \mathcal{A} in Theorem 1.1.2, Ric denotes the Ricci tensor, and all the norms, gradients, and integrals are intrinsic to the metric g.

When expressed in coordinates, these Riemannian quantities fit within our Euclidean setting with variable coefficients. For instance, the operator Δ_{LB} can be written in coordinates as $Lu = \text{div} (A(x)\nabla u) + \hat{b}(x) \cdot \nabla u$. Here $A(x) = (a_{ij}(x)) = (g^{ij}(x))$ is the inverse of the metric and $\hat{b}_i(x) = \frac{1}{2}g^{ij}(x)\partial_j \log |g|$ involves the volume density $|g| = \text{det}(g_{ij}(x))$. Moreover, with our notation for matrices, the function $\mathcal{A}_{\text{Riem}}$ in (1.3.12) can be written locally in $\{\nabla_g u \neq 0\}$ as

$$\mathcal{A}_{\text{Riem}} = \left(\|A^{1/2}(x)HuA^{1/2}(x)\|_{\text{HS}}^2 - \left|A^{1/2}(x)HuA^{1/2}(x)\mathbf{n}_A(x)\right|^2 \right)^{1/2}, \quad (1.3.13)$$

where $\mathbf{n}_A(x) = |\nabla u|_{A(x)}^{-1} A^{1/2}(x) \nabla u$ has appeared in the definition (1.3.11) of \mathcal{A}_x in Remark 1.3.2 and $Hu = ((Hu)_{ij})$ is the Riemannian Hessian matrix given by $(Hu)_{ij} = u_{ij} - \Gamma_{ij}^k u_k$, where Γ_{ij}^k are the Christoffel symbols of the metric.

By this identification of Δ_{LB} with L, applying the Riemannian result in [56], collecting all lower order terms, and estimating the derivatives of the metric, we are led to an a priori bound for the function \mathcal{A}_x in (1.3.11) which involves errors of the same type as in (1.1.11). Due to the presence of the Ricci tensor in (1.3.12), this estimate derived from the Riemannian inequality (1.3.12) depends on the norm $||\mathcal{A}||_{C^2(\overline{B}_1)}$, i.e., it requires two derivatives of the metric. Nevertheless, integrating the unwanted coefficient derivatives by parts as we did in our proof of (1.1.11), we could deduce an estimate depending only on $||\mathcal{A}||_{C^1(\overline{B}_1)}$.

The authors in [56] obtain (1.3.12) by choosing the test function $\mathbf{c} = |\nabla_{\mathbf{g}}\mathbf{u}|_{\mathbf{g}}$ in their stability inequality. In our coordinates, this function reads as $\mathbf{c}(x) = |\nabla u|_{A(x)}$. As explained in Remark 1.3.2, this choice of \mathbf{c} and our integral stability inequality (1.1.18) lead to a similar estimate for \mathcal{A}_x by using the ideas from the proof of (1.1.11) above.

We emphasize that both approaches (the Riemannian one and ours) give an estimate for \mathcal{A}_x which contains an error term involving the product $|D^2u||\nabla u|$. This error arises from the interaction between the second and first order terms in the Riemannian Hessian Hu when squaring $\mathcal{A}_{\text{Riem}}$, and thus squaring $Hu = D^2u - \Gamma \nabla u = D^2u + O(\varepsilon |\nabla u|)$.

Next, we prove the "Hessian times the gradient" estimates (1.1.13) and (1.1.14) in Theorem 1.1.2. For this, we will need to consider the auxiliary function

$$\mathcal{A}_{0} := \begin{cases} \left(\|A^{1/2}(0)D^{2}uA^{1/2}(0)\|_{\mathrm{HS}}^{2} - |A^{1/2}(0)D^{2}uA^{1/2}(0)\mathbf{n}(x)|^{2} \right)^{1/2} & \text{if } \nabla u \neq 0 \\ 0 & \text{if } \nabla u = 0, \end{cases}$$
(1.3.14)

where $\mathbf{n}(x) = |\nabla u|_{A^{-1}(0)}^{-1} A^{1/2}(0) \nabla u$ is again the vector field in the definition of \mathcal{A} in (1.3.1). Notice that (1.3.14) is precisely the definition of \mathcal{A} in (1.3.1) with the matrix $A^{1/2}(x)$ replaced by $A^{1/2}(0)$; see Remark 1.3.2. The greatest advantage of the function \mathcal{A}_0 over \mathcal{A} is the symmetry of the matrix $A^{1/2}(0)D^2uA^{1/2}(0)$ in the definition above. This will allow us to bound the Hessian of the solution by \mathcal{A}_0 , with the exception of the $\mathbf{n} \otimes \mathbf{n}$ component, which can be treated separately thanks to the nonnegativity assumption on the nonlinearity.

We will also need the a priori estimate (1.1.11) proved above, which gives a bound for the L^2 norm of the function \mathcal{A} . In the proof below, for $||D\mathcal{A}||_{L^{\infty}(B_1)} \leq \varepsilon$, we will see that

$$|\mathcal{A}^2 - \mathcal{A}_0^2| \le C\varepsilon |x|\mathcal{A}_0^2 \quad \text{in } B_1, \tag{1.3.15}$$

where C is a universal constant. In particular, for ε small, the functions are comparable and (1.1.11) allows us to bound the L^2 norm of \mathcal{A}_0 as well.

Proof of (1.1.13) and (1.1.14) in Theorem 1.1.2. Throughout the proof, C denotes a generic universal constant. The proof is divided into four steps.

Step 1: We prove that

$$|D^2 u| \le -C \operatorname{tr} \left(A(0) D^2 u \right) + C \mathcal{A}_0 + C \varepsilon |x| |D^2 u| + C \varepsilon |\nabla u| \quad a.e. \text{ in } B_1, \qquad (1.3.16)$$

where C > 0 is universal.

First we bound the full Hessian of u almost everywhere by the function \mathcal{A}_0 and the $\mathbf{n} \otimes \mathbf{n}$ component of the matrix $A^{1/2}(0)D^2uA^{1/2}(0)$. If $\nabla u(x) \neq 0$, then, extending $\mathbf{n}(x)$ to an orthonormal basis of \mathbb{R}^n , it is easy to see³ that

$$\|A^{1/2}(0)D^2 u A^{1/2}(0)\|_{\mathrm{HS}}^2 \le 2\mathcal{A}_0^2 + \left| (A^{1/2}(0)D^2 u A^{1/2}(0))\mathbf{n}(x) \cdot \mathbf{n}(x) \right|^2.$$
(1.3.17)

Moreover, by Stampacchia's result, $|D^2u| = 0$ a.e. in $\nabla u = 0$ (see [69, Theorem 6.19]), and the inequality (1.3.17) holds almost everywhere in B_1 . By uniform ellipticity we also have $|D^2u| \leq C|A^{1/2}(0)D^2uA^{1/2}(0)| \leq C||A^{1/2}(0)D^2uA^{1/2}(0)||_{\text{HS}}$ and hence

$$|D^{2}u| \leq C\mathcal{A}_{0} + C |(A^{1/2}(0)D^{2}uA^{1/2}(0))\mathbf{n}(x)\cdot\mathbf{n}(x)| \quad \text{a.e. in } B_{1}.$$
(1.3.18)

Next we use that the nonlinearity is nonnegative to bound the $\mathbf{n} \otimes \mathbf{n}$ component $(A^{1/2}(0)D^2uA^{1/2}(0))\mathbf{n}(x) \cdot \mathbf{n}(x)$ in (1.3.18) in terms of the equation, the function \mathcal{A}_0 , and lower order terms.

Since $0 \ge -f(u) = Lu = tr(A(x)D^2u) + b(x) \cdot \nabla u$, we have

$$\begin{aligned} \left| \operatorname{tr}(A(x)D^{2}u) \right| &= \left| Lu - b(x) \cdot \nabla u \right| \\ &\leq -Lu + \left| b(x) \cdot \nabla u \right| = -\operatorname{tr}(A(x)D^{2}u) - b(x) \cdot \nabla u + \left| b(x) \cdot \nabla u \right| \quad (1.3.19) \\ &\leq -\operatorname{tr}(A(x)D^{2}u) + 2\varepsilon |\nabla u|. \end{aligned}$$

By the mean value theorem we have $|A(x) - A(0)| \le \varepsilon |x|$, and hence by (1.3.19)

$$\begin{aligned} \left| \operatorname{tr}(A(0)D^{2}u) \right| &\leq \left| \operatorname{tr}(A(x)D^{2}u) \right| + C\varepsilon |x| |D^{2}u| \\ &\leq -\operatorname{tr}(A(x)D^{2}u) + C\varepsilon |x| |D^{2}u| + C\varepsilon |\nabla u| \\ &\leq -\operatorname{tr}(A(0)D^{2}u) + C\varepsilon |x| |D^{2}u| + C\varepsilon |\nabla u|. \end{aligned}$$
(1.3.20)

This follows immediately from the fact that, for any symmetric matrix $M \in \mathbb{R}^{n \times n}$, we have $||M||_{\text{HS}}^2 = \sum_{i,j=1}^{n-1} M_{ij}^2 + 2\sum_{i=1}^{n-1} M_{in}^2 + M_{nn}^2$ and $||M||_{\text{HS}}^2 - |Me_n|^2 = \sum_{i,j=1}^{n-1} M_{ij}^2 + \sum_{i=1}^{n-1} M_{in}^2$.

By the same argument to deduce (1.3.17) above, it is easy to see that⁴

$$\left|A^{1/2}(0)D^2 u A^{1/2}(0) - \left[(A^{1/2}(0)D^2 u A^{1/2}(0))\mathbf{n} \cdot \mathbf{n}\right]\mathbf{n} \otimes \mathbf{n}\right| \le C\mathcal{A}_0 \quad \text{a.e. in } B_1. \quad (1.3.21)$$

Writing the $\mathbf{n} \otimes \mathbf{n}$ component of $A^{1/2}(0)D^2uA^{1/2}(0)$ as

$$(A^{1/2}(0)D^2uA^{1/2}(0))\mathbf{n} \cdot \mathbf{n}$$

= tr $\left(A^{1/2}(0)D^2uA^{1/2}(0)\right)$
- tr $\left(A^{1/2}(0)D^2uA^{1/2}(0) - \left[(A^{1/2}(0)D^2uA^{1/2}(0))\mathbf{n} \cdot \mathbf{n}\right]\mathbf{n} \otimes \mathbf{n}\right),$

from (1.3.20) and (1.3.21), it follows that

$$\begin{aligned} \left| (A^{1/2}(0)D^2 u A^{1/2}(0)) \mathbf{n} \cdot \mathbf{n} \right| &\leq \left| \operatorname{tr} \left(A^{1/2}(0)D^2 u A^{1/2}(0) \right) \right| + C\mathcal{A}_0 \\ &\leq -\operatorname{tr}(A(0)D^2 u) + C\mathcal{A}_0 + C\varepsilon |x| |D^2 u| + C\varepsilon |\nabla u| \end{aligned} \tag{1.3.22}$$

a.e. in B_1 . Combining (1.3.22) and (1.3.18) yields the claimed inequality (1.3.16).

Step 2: We prove that there is a universal $\varepsilon_0 > 0$ such that, if $\varepsilon \leq \varepsilon_0$, then

$$\int_{B_1} |D^2 u| |\nabla u| \eta^2 \, \mathrm{d}x \le C \int_{B_1} \mathcal{A} |\nabla u| \, \eta^2 \, \mathrm{d}x + C \int_{B_1} |\nabla u|^2 \left(|\nabla(\eta^2)| + \varepsilon \eta^2 \right) \, \mathrm{d}x$$

for all $\eta \in C_c^{\infty}(B_1)$, where C is universal.

By uniform ellipticity, it suffices to estimate the integral $\int_{B_1} |D^2 u| |\nabla u|_{A(0)} \eta^2 dx$. Multiplying (1.3.16) in Step 1 by $|\nabla u|_{A(0)} \eta^2$ and integrating in B_1 , by uniform ellipticity we have

$$\int_{B_{1}} |D^{2}u| |\nabla u|_{A(0)} \eta^{2} dx
\leq -C \int_{B_{1}} |\nabla u|_{A(0)} \operatorname{tr} \left(A(0) D^{2}u \right) \eta^{2} dx + C \int_{B_{1}} \mathcal{A}_{0} |\nabla u|_{A(0)} \eta^{2} dx
+ C \varepsilon \int_{B_{1}} |x| |D^{2}u| |\nabla u|_{A(0)} \eta^{2} dx + C \varepsilon \int_{B_{1}} |\nabla u|^{2} \eta^{2} dx.$$
(1.3.23)

The only delicate term in the right-hand side of (1.3.23) is the first one, which can be treated as follows.

We write the product $-|\nabla u|_{A(0)} \operatorname{tr}(A(0)D^2u)$ in $\{\nabla u \neq 0\}$ as

$$-2|\nabla u|_{A(0)} \operatorname{tr} (A(0)D^2 u) = -|\nabla u|_{A(0)} \operatorname{tr} (A(0)D^2 u) - \operatorname{div} (|\nabla u|_{A(0)}A(0)\nabla u) + \nabla |\nabla u|_{A(0)} \cdot A(0)\nabla u.$$
(1.3.24)

Since

$$\begin{aligned} \nabla |\nabla u|_{A(0)} \cdot A(0) \nabla u &= |\nabla u|_{A(0)}^{-1} D^2 u A(0) \nabla u \cdot A(0) \nabla u \\ &= |\nabla u|_{A(0)} (A^{1/2}(0) D^2 u A^{1/2}(0)) \mathbf{n} \cdot \mathbf{n}, \end{aligned}$$

⁴Follows from the fact that, for any symmetric matrix $M \in \mathbb{R}^{n \times n}$, we have $|M| \leq n ||M||_{\text{HS}}$ and $||M - M_{nn}e_n \otimes e_n||_{\text{HS}}^2 \leq 2 \left(||M||_{\text{HS}}^2 - |Me_n|^2 \right)$.

by (1.3.24) and using that **n** is unitary, it follows that

$$-2|\nabla u|_{A(0)} \operatorname{tr} \left(A(0)D^{2}u \right)$$

= $-|\nabla u|_{A(0)} \operatorname{tr} \left(A^{1/2}(0)D^{2}uA^{1/2}(0) - \left[\left(A^{1/2}(0)D^{2}uA^{1/2}(0) \right) \mathbf{n} \cdot \mathbf{n} \right] \mathbf{n} \otimes \mathbf{n} \right)$ (1.3.25)
 $-\operatorname{div} \left(|\nabla u|_{A(0)}A(0)\nabla u \right)$

a.e. in B_1 . By the bound (1.3.21) in the proof of Step 1 above, it follows that

$$\left| \operatorname{tr} \left(A^{1/2}(0) D^2 u A^{1/2}(0) - \left[\left(A^{1/2}(0) D^2 u A^{1/2}(0) \right) \mathbf{n} \cdot \mathbf{n} \right] \mathbf{n} \otimes \mathbf{n} \right) \right| \le C \mathcal{A}_0$$

a.e. in B_1 , and hence from (1.3.25) we deduce

$$-2|\nabla u|_{A(0)}\operatorname{tr}(A(0)D^{2}u) \leq -\operatorname{div}(|\nabla u|_{A(0)}A(0)\nabla u) + C\mathcal{A}_{0}|\nabla u|_{A(0)} \quad \text{a.e. in } B_{1}. \quad (1.3.26)$$

Substituting (1.3.26) in (1.3.23) leads to

$$\begin{split} \int_{B_1} |D^2 u| |\nabla u|_{A(0)} \eta^2 \, \mathrm{d}x \\ &\leq -C \int_{B_1} \operatorname{div} \left(|\nabla u|_{A(0)} A(0) \nabla u \right) \eta^2 \, \mathrm{d}x + C \int_{B_1} \mathcal{A}_0 |\nabla u|_{A(0)} \eta^2 \, \mathrm{d}x \\ &+ C\varepsilon \int_{B_1} |x| |D^2 u| |\nabla u|_{A(0)} \eta^2 \, \mathrm{d}x + C\varepsilon \int_{B_1} |\nabla u|^2 \eta^2 \, \mathrm{d}x, \end{split}$$

and integrating by parts the divergence term, we obtain the inequality

$$\int_{B_1} |D^2 u| |\nabla u|_{A(0)} \eta^2 \, \mathrm{d}x \le C \int_{B_1} |\nabla u|^2 \left(|\nabla \eta^2| + \varepsilon \eta^2 \right) \, \mathrm{d}x + C \int_{B_1} \mathcal{A}_0 |\nabla u|_{A(0)} \, \eta^2 \, \mathrm{d}x + C \varepsilon \int_{B_1} |x| |D^2 u| |\nabla u|_{A(0)} \, \eta^2 \, \mathrm{d}x.$$

$$(1.3.27)$$

Since $|x| \leq 1$ in B_1 , choosing $\varepsilon_0 > 0$ universal small such that $C\varepsilon_0 = 1/2$, we can absorb the "Hessian times the gradient" error in (1.3.27) into the left-hand side to obtain

$$\int_{B_1} |D^2 u| |\nabla u|_{A(0)} \eta^2 \, \mathrm{d}x \le C \int_{B_1} |\nabla u|^2 \, \left(|\nabla \eta^2| + \varepsilon \eta^2 \right) \, \mathrm{d}x + C \int_{B_1} \mathcal{A}_0 |\nabla u|_{A(0)} \, \eta^2 \, \mathrm{d}x.$$
(1.3.28)

To conclude the argument, let us show that \mathcal{A} and \mathcal{A}_0 are comparable for ε small. Letting E(x) = A(x) - A(0) and $M(x) = D^2 u(x) A^{1/2}(0)$, it is easy to check that

$$\mathcal{A}^2 = \mathcal{A}_0^2 + \operatorname{tr}\left(M(x)^T E(x) M(x)\right) - (M(x)^T E(x) M(x)) \mathbf{n} \cdot \mathbf{n} \quad \text{in } \{\nabla u \neq 0\},\$$

and for $x \in \{\nabla u \neq 0\}$, extending $\mathbf{n} = \mathbf{n}(x)$ to an orthonormal basis $\mathbf{e}_1, \ldots, \mathbf{e}_n = \mathbf{n}$ of \mathbb{R}^n , we can rewrite this identity as

$$\mathcal{A}^{2} = \mathcal{A}_{0}^{2} + \sum_{i=1}^{n-1} E(x)M(x)\mathbf{e}_{i} \cdot M(x)\mathbf{e}_{i} \quad \text{in } \{\nabla u \neq 0\}.$$
(1.3.29)

By the mean value theorem we can bound the error by $|E(x)| \leq \varepsilon |x|$, and hence, by uniform ellipticity,

$$\left|\sum_{i=1}^{n-1} E(x)M(x)\mathbf{e}_{i} \cdot M(x)\mathbf{e}_{i}\right| \le \varepsilon |x| \sum_{i=1}^{n-1} |M(x)\mathbf{e}_{i}|^{2} \le C\varepsilon |x| \sum_{i=1}^{n-1} |M(x)\mathbf{e}_{i}|_{A(0)}^{2}.$$
 (1.3.30)

Since $|M(x)\mathbf{e}_i|_{A(0)}^2 = |A^{1/2}(0)D^2u(x)A^{1/2}(0)\mathbf{e}_i|$, by (1.3.21) above, the sum in right-hand side of (1.3.30) can be further bounded by

$$\sum_{i=1}^{n-1} |M(x)\mathbf{e}_i|_{A(0)}^2 \le C\mathcal{A}_0 \quad \text{a.e. in } B_1.$$
(1.3.31)

Combining (1.3.30) and (1.3.31), from (1.3.29) we conclude that

$$(1 - C\varepsilon |x|) \mathcal{A}_0^2 \le \mathcal{A}^2 \le (1 + C\varepsilon |x|) \mathcal{A}_0^2 \quad \text{in } B_1,$$

which was the inequality (1.3.15) mentioned before the proof. Choosing ε_0 smaller if necessary, we may assume that $\mathcal{A}_0 \leq 2\mathcal{A}$, which applied in (1.3.27) yields the claim.

Step 3: We prove that, if $\varepsilon \leq \varepsilon_0$, with $\varepsilon_0 > 0$ as in Step 1, then

$$\int_{B_1} \mathcal{A}^2 \eta^2 \, \mathrm{d}x \le C \int_{B_1} |\nabla u|^2 \left(|\nabla \eta|^2 + \varepsilon^2 \eta^2 \right) \, \mathrm{d}x$$

for all $\eta \in C_c^{\infty}(B_1)$, where C is a universal constant.

Combining (1.1.11) in Theorem 1.1.2 and Step 2, we have

$$\int_{B_1} \mathcal{A}^2 \eta^2 \,\mathrm{d}x \le C\varepsilon \int_{B_1} \mathcal{A} |\nabla u| \eta^2 \,\mathrm{d}x + \int_{B_1} |\nabla u|_{A(0)}^2 |\nabla \eta|_{A(x)}^2 \,\mathrm{d}x + C\varepsilon \int_{B_1} |\nabla u|^2 (|\nabla (\eta^2)| + \varepsilon \eta^2) \,\mathrm{d}x.$$
(1.3.32)

By Young's inequality, the first term on the right-hand side of (1.3.32) can be bounded by

$$C\varepsilon \int_{B_1} \mathcal{A} |\nabla u| \eta^2 \, \mathrm{d}x \le \frac{1}{2} \int_{B_1} \mathcal{A}^2 \eta^2 \, \mathrm{d}x + C\varepsilon^2 \int_{B_1} |\nabla u|^2 \eta^2 \, \mathrm{d}x,$$

and the $\mathcal{A}^2 \eta^2$ integral can be absorbed into the left-hand side. By uniform ellipticity and the bound $\varepsilon |\nabla(\eta^2)| \leq |\nabla \eta|^2 + \varepsilon^2 \eta^2$, we deduce the claim.

Step 4: Conclusion.

Combining Steps 2 and 3, for $\varepsilon \leq \varepsilon_0$ as above and by Cauchy-Schwarz, we obtain

$$\int_{B_1} |D^2 u| |\nabla u| \eta^2 \, \mathrm{d}x$$

$$\leq C \left(\int_{B_1} \mathcal{A}^2 \eta^2 \, \mathrm{d}x \right)^{1/2} \left(\int_{B_1} |\nabla u|^2 \eta^2 \, \mathrm{d}x \right)^{1/2} + C \int_{B_1} |\nabla u|^2 (|\nabla (\eta^2)| + \varepsilon \eta^2) \, \mathrm{d}x$$

$$\leq C \left(\int_{B_1} |\nabla u|^2 (|\nabla \eta|^2 + \varepsilon^2 \eta^2) \, \mathrm{d}x \right)^{1/2} \left(\int_{B_1} |\nabla u|^2 \eta^2 \, \mathrm{d}x \right)^{1/2}$$

$$+ C \int_{B_1} |\nabla u|^2 (|\nabla (\eta^2)| + \varepsilon \eta^2) \, \mathrm{d}x.$$
(1.3.33)

The inequalities (1.1.13) and (1.1.14) in Theorem 1.1.2 follow easily from (1.3.33) by choosing appropriate cut-off functions and using that ε is bounded by a universal constant ε_0 . Choosing $\eta \in C_c^{\infty}(B_1)$ such that $\eta = 1$ in $B_{3/4}$ and $0 \leq \eta \leq 1$ in B_1 leads to the estimate in balls (1.1.13). The second estimate in annuli (1.1.14) follows by choosing $\eta \in C_c^{\infty}(B_1 \setminus \overline{B}_{1/8})$ with $\eta = 1$ in $B_{1/2} \setminus B_{1/4}$ and $0 \leq \eta \leq 1$ in B_1 .

Remark 1.3.4. By the proof above, we can also deduce an interior a priori estimate for the L^1 norm of the Hessian. For this, assuming ε to be small, recalling that $\mathcal{A}_0 \leq C\mathcal{A}$ by (1.3.15), and absorbing the Hessian term in Step 1, we obtain

$$|D^2 u| \leq -C \operatorname{div} (A(0)\nabla u) + C\mathcal{A} + C|\nabla u|$$
 a.e. in B_1 .

Multiplying this inequality by a cut-off function and integrating by parts, using the bound for \mathcal{A} in Step 3 and applying Cauchy-Schwarz, we deduce an estimate for the L^1 norm of the Hessian in terms of the L^2 norm of the gradient in a larger ball.

We conclude this section by proving Proposition 1.1.3. To show that the L^2 norm of the gradient is controlled by the L^1 norm of the function in a larger ball, we use the interpolation inequalities of Cabré [20] combined with the Hessian estimates from Theorem 1.1.2. The errors in larger balls can then be absorbed thanks to a well-known lemma of Simon [89]. We recall the interpolation inequalities of Cabré in Appendix C and Simon's lemma in Appendix D.

Proof of Proposition 1.1.3. We cover $B_{1/2}$ (except for a set of measure zero) with a family of disjoint open cubes Q_j of the same side-length and small enough so that $Q_j \subset B_{3/4}$. The side-length and the number of cubes depend only on n. Combining the interpolation inequalities of Propositions C.1 and C.3, rescaled from the unit cube to Q_j , with $\tilde{\delta} = \delta^{3/2}$ for a given $\delta \in (0, 1)$, we obtain

$$\int_{Q_j} |\nabla u|^2 dx \le C\delta \int_{Q_j} |D^2 u| |\nabla u| \, dx + C\delta \int_{Q_j} |\nabla u|^2 dx + C\delta^{-2-\frac{3n}{2}} \left(\int_{Q_j} |u| \, dx \right)^2.$$

Since $Q_j \subset B_{3/4}$, applying (1.1.13) from Theorem 1.1.2, for $\varepsilon \leq \varepsilon_0$ we have

$$\int_{Q_j} |\nabla u|^2 dx \le C\delta \int_{B_1} |\nabla u|^2 dx + C\delta^{-2-\frac{3n}{2}} \left(\int_{B_1} |u| \, dx \right)^2.$$

Adding up these inequalities, we obtain

$$\|\nabla u\|_{L^{2}(B_{1/2})}^{2} \leq C\delta \|\nabla u\|_{L^{2}(B_{1})}^{2} + C\delta^{-2-\frac{3n}{2}} \|u\|_{L^{1}(B_{1})}^{2} \quad \text{for } \delta \in (0,1) \text{ and } \varepsilon \leq \varepsilon_{0}.$$
(1.3.34)

As explained in Section 1.2, for $B_{\rho}(y) \subset B_1$, the function $u^{y,\rho} := u(y + \rho \cdot)$ is a stable solution to a semilinear equation with coefficients $A^{y,\rho} = A(y + \rho \cdot)$ and $b^{y,\rho} = \rho b(y + \rho \cdot)$. In particular, since $\rho \leq 1$, for $\varepsilon \leq \varepsilon_0$ we have that

$$\|DA^{y,\rho}\|_{L^{\infty}(B_1)} + \|b^{y,\rho}\|_{L^{\infty}(B_1)} \le \rho \varepsilon \le \varepsilon_0,$$

and we can apply (1.3.34) to $u^{y,\rho}$, which yields

$$\rho^{n+2} \int_{B_{\rho/2}(y)} |\nabla u|^2 \, dx \le C\delta\rho^{n+2} \int_{B_{\rho}(y)} |\nabla u|^2 \, dx + C\delta^{-2-\frac{3n}{2}} \left(\int_{B_{\rho}(y)} |u| \, dx \right)^2 \\ \le C\delta\rho^{n+2} \int_{B_{\rho}(y)} |\nabla u|^2 \, dx + C\delta^{-2-\frac{3n}{2}} \left(\int_{B_1} |u| \, dx \right)^2.$$

By Lemma D.1 with $\sigma(B) := \|\nabla u\|_{L^2(B)}^2$, the claim follows.

1.4 The $W^{1,2+\gamma}$ estimate

Here we prove the higher integrability estimate (1.1.9) in Theorem 1.1.1. The strategy of proof is the same as for the Laplacian in [24]. First we bound the $L^{2+\gamma}$ norm in terms of the L^2 norm of the gradient when the coefficients are small. This will follow from a uniform estimate of the Dirichlet norm on level sets, which relies on the Hessian estimates in Theorem 1.1.2.

Lemma 1.4.1. Let $u \in C^{\infty}(\overline{B}_1)$ be a stable solution of -Lu = f(u) in B_1 , for some nonnegative function $f \in C^1(\mathbb{R})$. Assume that

$$||DA||_{L^{\infty}(B_1)} + ||b||_{L^{\infty}(B_1)} \le \varepsilon$$

for some $\varepsilon > 0$. If $\varepsilon \leq \varepsilon_0$, then

$$\|\nabla u\|_{L^{2+\gamma}(B_{1/2})} \le C \|\nabla u\|_{L^{2}(B_{1})}$$

where $\gamma > 0$ depends only on n, and $\varepsilon_0 > 0$ and C are universal constants.

Proof. The proof is divided in two steps.

Step 1: We prove that, if $\varepsilon \leq \varepsilon_0$, then for a.e. $t \in \mathbb{R}$ we have

$$\int_{\{u=t\}\cap B_{1/2}} |\nabla u|^2 \, \mathrm{d}\mathcal{H}^{n-1} \le C \|\nabla u\|_{L^2(B_1)}^2,$$

where $\varepsilon_0 > 0$ and C are universal.

Since $|\operatorname{div}(|\nabla u|\nabla u)| \leq C|D^2u||\nabla u|$, by (1.1.13) in Theorem 1.1.2, for $\varepsilon \leq \varepsilon_0$ we have

$$\left\| \operatorname{div} \left(|\nabla u| \nabla u \right) \right\|_{L^1(B_{3/4})} \le C \|\nabla u\|_{L^2(B_1)}^2.$$
(1.4.1)

Consider a cut-off function $\eta \in C_c^{\infty}(B_{3/4})$ with $\eta = 1$ in $B_{1/2}$ and $0 \leq \eta \leq 1$. By the divergence theorem, for a.e. $t \in \mathbb{R}$ we have

$$\int_{\{u=t\}\cap B_{1/2}} |\nabla u|^2 \,\mathrm{d}\mathcal{H}^{n-1} \leq \int_{\{u=t\}\cap B_1\cap\{\nabla u\neq 0\}} |\nabla u|^2 \eta^2 \,\mathrm{d}\mathcal{H}^{n-1}$$
$$= -\int_{\{u>t\}\cap B_1\cap\{\nabla u\neq 0\}} \operatorname{div}\left(|\nabla u|\nabla u\,\eta^2\right) \,\mathrm{d}x$$
$$\leq \int_{B_1} |\nabla u|^2 |\nabla(\eta^2)| \,\mathrm{d}x + \int_{B_1} \left|\operatorname{div}\left(|\nabla u|\nabla u\right)\right| \eta^2 \,\mathrm{d}x$$

and applying (1.4.1) we obtain the claim.

Step 2: Conclusion.

Let

$$v := \frac{u - (u)_{B_1}}{\|\nabla u\|_{L^2(B_1)}},$$

where $(u)_{B_1} := \frac{1}{|B_1|} \int_{B_1} u \, dx$. In particular $\|\nabla v\|_{L^2(B_1)} = 1$ and by the Sobolev-Poincaré inequality, for some dimensional p > 2, we have

$$\left(\int_{B_1} |v|^p \,\mathrm{d}x\right)^{\frac{1}{p}} \le C \left(\int_{B_1} |\nabla v|^2 \,\mathrm{d}x\right)^{\frac{1}{2}} = C.$$
(1.4.2)

By the coarea formula and (1.4.2), we have

$$\int_{\mathbb{R}} \mathrm{d}t \int_{\{v=t\} \cap \{|\nabla v| \neq 0\}} |t|^p |\nabla v|^{-1} \,\mathrm{d}\mathcal{H}^{n-1} = \int_{B_1 \cap \{|\nabla v| \neq 0\}} |v|^p \,\mathrm{d}x \le C.$$
(1.4.3)

Since p > 2, we can choose dimensional constants q > 1 and $\theta \in (0, 1/3)$ such that $p/q = (1 - \theta)/\theta$. We define

$$h(t) := \max\{1, |t|\}.$$

Using the coarea formula and the Hölder inequality (note that $p\theta - q(1-\theta) = 0$), we obtain

$$\begin{split} \int_{B_{1/2}} |\nabla v|^{3-3\theta} \, \mathrm{d}x &= \int_{\mathbb{R}} \mathrm{d}t \int_{\{v=t\} \cap B_{1/2} \cap \{|\nabla v| \neq 0\}} h(t)^{p\theta - q(1-\theta)} |\nabla v|^{-\theta + 2(1-\theta)} \, \mathrm{d}x \\ &\leq \left(\int_{\mathbb{R}} \mathrm{d}t \int_{\{v=t\} \cap B_{1/2} \cap \{|\nabla v| \neq 0\}} h(t)^{p} |\nabla v|^{-1} \, \mathrm{d}\mathcal{H}^{n-1} \right)^{\theta} \cdot \\ &\cdot \left(\int_{\mathbb{R}} \mathrm{d}t \int_{\{v=t\} \cap B_{1/2}} h(t)^{-q} |\nabla v|^{2} \, \mathrm{d}\mathcal{H}^{n-1} \right)^{1-\theta}. \end{split}$$

Thanks to (1.4.3) and the definition of h(t), we have

$$\int_{\mathbb{R}} \mathrm{d}t \int_{\{v=t\}\cap B_1\cap\{|\nabla v|\neq 0\}} h(t)^p |\nabla v|^{-1} \,\mathrm{d}\mathcal{H}^{n-1}$$

$$\leq \int_{-1}^1 \mathrm{d}t \int_{\{v=t\}\cap B_1\cap\{|\nabla v|\neq 0\}} |\nabla v|^{-1} \,\mathrm{d}\mathcal{H}^{n-1} + C \leq |B_1| + C \leq C.$$

Since q > 1, it follows that $\int_{\mathbb{R}} h(t)^{-q} dt$ is finite and by Step 1, for $\varepsilon \leq \varepsilon_0$, we have

$$\int_{\mathbb{R}} \mathrm{d}t \, h(t)^{-q} \int_{\{v=t\} \cap B_{1/2}} |\nabla v|^2 \, \mathrm{d}\mathcal{H}^{n-1} \le C.$$

Finally, we obtain

$$\int_{B_{1/2}} |\nabla v|^{3-3\theta} \,\mathrm{d}x \le C$$

which gives the claim, since $\nabla v \equiv \nabla u / \|\nabla u\|_{L^2(B_1)}$.

To deduce the $L^{2+\gamma}$ estimate (1.1.9) in Theorem 1.1.1, we will combine Proposition 1.1.3 with Lemma 1.4.1, and apply a scaling and covering argument.

Proof of (1.1.9) in Theorem 1.1.1. Combining Proposition 1.1.3 and Lemma 1.4.1, applied to $u(\cdot/2)$, we deduce that there is a universal $\varepsilon_0 > 0$ such that, if $\varepsilon \leq \varepsilon_0$, then

$$\|\nabla u\|_{L^{2+\gamma}(B_{1/4})} \le C \|u\|_{L^{1}(B_{1})}, \tag{1.4.4}$$

where $\gamma > 0$ depends only on n, and C is universal.

Now (1.1.9) will follow easily from (1.4.4) by a scaling and covering argument. Let $\rho \in (0, 1)$ to be chosen later. We cover the ball $B_{1/2}$ by a finite number of balls $B_{\rho/4}(y_j)$ with $B_{\rho}(y_j) \subset B_1$. The number balls depends only on n and ρ . As explained in Section 1.2, the functions $u^{y_j,\rho} = u(y_j + \rho \cdot)$ are stable solutions to a semilinear equation with coefficients

 $A^{y_j,\rho} = A(y_j + \rho \cdot)$ and $b^{y_j,\rho} = \rho b(y_j + \rho \cdot)$. Choosing ρ small enough so that $\rho(||DA||_{L^{\infty}(B_1)} + ||b||_{L^{\infty}(B_1)} \leq \varepsilon) \leq \varepsilon_0$, it follows that

$$||DA^{y_j,\rho}||_{L^{\infty}(B_1)} + ||b^{y_j,\rho}||_{L^{\infty}(B_1)} \le \varepsilon_0,$$

and we can apply (1.4.4) to each $u^{y_j,\rho}$, which yields

$$\|\nabla u\|_{L^{2+\gamma}(B_{1/2})} \le \sum_{j} \|\nabla u\|_{L^{2+\gamma}(B_{\rho/4}(y_j))} \le C \sum_{j} \|u\|_{L^1(B_{\rho}(y_j))} \le C \|u\|_{L^1(B_1)},$$

for some C depending only on n, c_0 , C_0 , and ρ . Since ρ depends only on $\|DA\|_{L^{\infty}(B_1)}$, $\|b\|_{L^{\infty}(B_1)}$, and ε_0 , which is universal, this concludes the proof.

1.5 The weighted L^2 estimate for radial derivatives

Our goal in this section is to prove Proposition 1.1.4, where we bound the weighted L^2 norm of the radial derivative in balls by the L^2 norm of the full gradient in annuli. To prove the estimate, we will first apply the integral stability inequality with the test functions

$$\mathbf{c}(x) = x \cdot \nabla u$$
 and $\eta = |x|_{A^{-1}(0)}^{\frac{2-n}{2}} \zeta$,

where ζ is a cut-off. Our choice will yield the desired bound with an additional error term involving a weighted integral of the "Hessian times the gradient", which we will be able to treat thanks to the a priori estimates on annuli from Theorem 1.1.2.

We start by choosing $\mathbf{c} = x \cdot \nabla u$ and a generic test function η in the integral stability inequality:

Lemma 1.5.1. Let $u \in C^{\infty}(\overline{B}_1)$ be a stable solution of -Lu = f(u) in B_1 , for some function $f \in C^1(\mathbb{R})$. Assume that

$$||DA||_{L^{\infty}(B_1)} + ||b||_{L^{\infty}(B_1)} \le \varepsilon$$

for some $\varepsilon > 0$.

Then

$$\begin{split} &\int_{B_1} |\nabla u|^2_{A(x)} \Big((n-2)\eta^2 + x \cdot \nabla(\eta^2) \Big) \,\mathrm{d}x \\ &+ \int_{B_1} \Big(-2(x \cdot \nabla u)A(x)\nabla u \cdot \nabla(\eta^2) - |x \cdot \nabla u|^2 |\nabla \eta|^2_{A(x)} \Big) \,\mathrm{d}x \\ &\leq C\varepsilon \int_{B_1} |D^2 u| |\nabla u| |x|^2 \eta^2 \,\mathrm{d}x \\ &+ C\varepsilon \int_{B_1} |\nabla u|^2 \Big(|x|^2 |\nabla(\eta^2)| + \left(|x| + |x|^2\varepsilon \right) \eta^2 \Big) \,\mathrm{d}x \end{split}$$

for all $\eta \in C_c^{\infty}(B_1)$, where C is a universal constant.

Proof. Throughout the proof, C denotes a generic universal constant. Testing the integral stability inequality (1.1.18) with η and $\mathbf{c} = x \cdot \nabla u$, we deduce

$$\int_{B_1} (x \cdot \nabla u) J_u(x \cdot \nabla u) \eta^2 \,\mathrm{d}x \le \int_{B_1} |x \cdot \nabla u|^2 \left| \nabla \eta - \frac{1}{2} \eta A^{-1}(x) \widehat{b}(x) \right|_{A(x)}^2 \,\mathrm{d}x. \tag{1.5.1}$$

The quadratic term on the right-hand side of (1.5.1) can be bounded by

$$\left|\nabla\eta - \frac{1}{2}\eta A^{-1}(x)\widehat{b}(x)\right|_{A(x)}^2 \le |\nabla\eta|_{A(x)}^2 + C\varepsilon|\nabla(\eta^2)| + C\varepsilon^2\eta^2,$$

and hence

$$\int_{B_1} (x \cdot \nabla u) J_u(x \cdot \nabla u) \eta^2 \, \mathrm{d}x \le \int_{B_1} |x \cdot \nabla u|^2 |\nabla \eta|^2_{A(x)} \, \mathrm{d}x + C\varepsilon \int_{B_1} |\nabla u|^2 |x|^2 \left(|\nabla (\eta^2)| + \varepsilon \eta^2 \right) \, \mathrm{d}x.$$
(1.5.2)

To compute the Jacobi operator $J_u(x \cdot \nabla u) = L(x \cdot \nabla u) + f'(u)(x \cdot \nabla u)$, we differentiate the equation -Lu = f(u) in the direction of x, which yields

$$-x \cdot \nabla(Lu) = f'(u) \left(x \cdot \nabla u\right), \qquad (1.5.3)$$

and hence

$$J_u(x \cdot \nabla u) = L (x \cdot \nabla u) - x \cdot \nabla (Lu)$$

= $2a_{ij}(x)u_{ij} - x_k \partial_k a_{ij}(x)u_{ij} + b_i(x)u_i - x_j \partial_j b_i(x)u_i.$ (1.5.4)

From (1.5.4), by the coefficient bounds, it follows that

$$(x \cdot \nabla u) J_u(x \cdot \nabla u) \ge 2x_k u_k a_{ij}(x) u_{ij} - x_k u_k x_j \partial_j b_i(x) u_i - C\varepsilon |x|^2 |D^2 u| |\nabla u| - C\varepsilon |x| |\nabla u|^2.$$
(1.5.5)

The idea now is to integrate by parts to get rid of the highest order terms on the left-hand side of (1.5.2). For this we must rewrite the term $2x_ku_k a_{ij}(x)u_{ij}$ in (1.5.5) in divergence form. By the chain rule, we have

$$x_k u_k a_{ij}(x) u_{ij} = \partial_i \left(x_k u_k a_{ij}(x) u_j \right) - x_k u_k \partial_i a_{ij}(x) u_j - a_{ij}(x) u_i u_j - x_k a_{ij}(x) u_{ik} u_j.$$
(1.5.6)

Using that $a_{ij}(x) = a_{ji}(x)$, the last term in (1.5.6) can be written as

$$x_{k}a_{ij}(x)u_{ik}u_{j} = \frac{1}{2}\partial_{k}\left(a_{ij}(x)u_{i}u_{j}x_{k}\right) - \frac{n}{2}a_{ij}(x)u_{i}u_{j} - \frac{1}{2}x_{k}\partial_{k}a_{ij}(x)u_{i}u_{j},$$

and hence

$$2x_{k}u_{k}a_{ij}(x)u_{ij} = \partial_{i}\left(2x_{k}u_{k}a_{ij}(x)u_{j} - a_{jk}(x)u_{j}u_{k}x_{i}\right) + (n-2)a_{ij}(x)u_{i}u_{j}$$

$$- 2x_{k}u_{k}\partial_{i}a_{ij}(x)u_{j} + x_{k}\partial_{k}a_{ij}(x)u_{i}u_{j}$$

$$\geq \operatorname{div}\left(2(x \cdot \nabla u)A(x)\nabla u - |\nabla u|^{2}_{A(x)}x\right) + (n-2)|\nabla u|^{2}_{A(x)}$$

$$- C\varepsilon|x||\nabla u|^{2},$$

$$(1.5.7)$$

where in the last inequality we have used the estimates for the coefficients. Combining (1.5.7) and (1.5.5), we obtain

$$(x \cdot \nabla u) J_u(x \cdot \nabla u)$$

$$\geq \operatorname{div} \left(2(x \cdot \nabla u) A(x) \nabla u - |\nabla u|^2_{A(x)} x \right) + (n-2) |\nabla u|^2_{A(x)}$$

$$- x_k u_k x_j \partial_j b_i(x) u_i - C \varepsilon |x|^2 |D^2 u| |\nabla u| - C \varepsilon |x| |\nabla u|^2.$$
(1.5.8)

Multiplying (1.5.8) by η^2 and integrating, the term $-\int_{B_1} x_k u_k x_j \partial_j b_i(x) u_i \eta^2 dx$ can be integrated by parts and estimated by

$$\left| -\int_{B_1} x_k u_k x_j \partial_j b_i(x) u_i \eta^2 \, \mathrm{d}x \right| = \left| \int_{B_1} b_i(x) \partial_j(x_k u_k x_j u_i \eta^2) \, \mathrm{d}x \right|$$

$$\leq C \varepsilon \int_{B_1} |D^2 u| |\nabla u| |x|^2 \eta^2 \, \mathrm{d}x \qquad (1.5.9)$$

$$+ C \varepsilon \int_{B_1} |\nabla u|^2 \left(|x|^2 |\nabla(\eta^2)| + |x|\eta^2 \right) \, \mathrm{d}x.$$

Substituting (1.5.8) in the inequality (1.5.2), rearranging terms and by the error bound (1.5.9), it follows that

$$\int_{B_{1}} \left((n-2) |\nabla u|^{2}_{A(x)} \eta^{2} - |x \cdot \nabla u|^{2} |\nabla \eta|^{2}_{A(x)} \right) dx
+ \int_{B_{1}} \operatorname{div} \left(2(x \cdot \nabla u) A(x) \nabla u - |\nabla u|^{2}_{A(x)} x \right) \eta^{2} dx
\leq C \varepsilon \int_{B_{1}} |D^{2} u| |\nabla u| |x|^{2} \eta^{2} dx + C \varepsilon \int_{B_{1}} |\nabla u|^{2} \left(|x|^{2} |\nabla (\eta^{2})| + (|x| + \varepsilon |x|^{2}) \eta^{2} \right) dx.$$
(1.5.10)

Integrating by parts the divergence term on the left-hand side of (1.5.10) yields the claim.

Remark 1.5.2. In (1.5.3) we took a derivative of the equation in the x direction to get rid of the dependence on the nonlinearity. Instead, we could have multiplied the equation by the test function $\xi = \operatorname{div} (x (x \cdot \nabla u)\eta^2)$ and integrated by parts. Thanks to this, we avoid having to take any derivatives of b, since the term involving it can be bounded directly. Notice also that we need u to have three derivatives to be able to compute $L\mathbf{c}$. This is the same phenomenon as in the proof of Theorem 1.1.2; see the discussion in Remark 1.3.1.

Remark 1.5.3. Since $|A(x) - A(0)| \leq C\varepsilon |x|$, the inequality in Lemma 1.5.1 also holds if we replace A(x) by the constant matrix A(0) and we add an additional error term $C\varepsilon \int_{B_1} |\nabla u|^2 |x|^3 |\nabla \eta|^2 dx$ on the right-hand side. For future use, the final estimate involving A(0) instead of A(x) reads as

$$\int_{B_{1}} |\nabla u|^{2}_{A(0)} \left((n-2)\eta^{2} + x \cdot \nabla(\eta^{2}) \right) \mathrm{d}x
+ \int_{B_{1}} \left(-2(x \cdot \nabla u)A(0)\nabla u \cdot \nabla(\eta^{2}) - |x \cdot \nabla u|^{2}|\nabla \eta|^{2}_{A(0)} \right) \mathrm{d}x
\leq C\varepsilon \int_{B_{1}} |D^{2}u||\nabla u||x|^{2}\eta^{2} \mathrm{d}x
+ C\varepsilon \int_{B_{1}} |\nabla u|^{2} \left(|x|^{3}|\nabla \eta|^{2} + (|x| + \varepsilon|x|^{2})\eta^{2} \right) \mathrm{d}x,$$
(1.5.11)

where we have used that $|x|^2 |\nabla(\eta^2)| \le |x|^3 |\nabla\eta|^2 + |x|\eta^2$.

Next, we choose the singular test function $\eta = |x|_{A^{-1}(0)}^{-a/2} \zeta$ in Lemma 1.5.1, where the exponent $a \ge 0$ will satisfy $a \le n-2$ when $n \le 9$. Recall our notation for the modulus of the position vector and the radial derivative

$$r = |x|$$
 and $u_r = \frac{x}{|x|} \cdot \nabla u.$

Lemma 1.5.4. Let $u \in C^{\infty}(\overline{B}_1)$ be a stable solution of -Lu = f(u) in B_1 , for some function $f \in C^1(\mathbb{R})$. Assume that

$$||DA||_{L^{\infty}(B_1)} + ||b||_{L^{\infty}(B_1)} \le \varepsilon$$

for some $\varepsilon > 0$. If

$$0 \le a \le \min\{10, n\} - 2, \tag{1.5.12}$$

then

$$(n-2-a)\int_{B_{\rho}}r^{-a}|\nabla u|^{2} dx + \frac{a(8-a)}{4}\int_{B_{\rho}}r^{-a}u_{r}^{2} dx$$

$$\leq C\int_{B_{2\rho}\setminus B_{\rho}}r^{-a}|\nabla u|^{2} dx + C\varepsilon\int_{B_{2\rho}}r^{2-a}|D^{2}u||\nabla u| dx \qquad (1.5.13)$$

$$+ C\varepsilon\int_{B_{2\rho}}\left(r^{1-a}+\varepsilon r^{2-a}\right)|\nabla u|^{2} dx.$$

for all $\rho \leq 1/2$, where C is a universal constant.

Proof. By approximation, the inequality in Lemma 1.5.1 is valid for Lipschitz test functions $\eta \in C_c^{0,1}(B_1)$. Moreover, this inequality also holds for the singular test function

$$\eta := |x|_{A^{-1}(0)}^{-a/2}\zeta,$$

where $\zeta \in C_c^{0,1}(B_1)$ is a cut-off. To see this, for $\delta > 0$ consider the $C_c^{0,1}$ approximation

$$\eta_{\delta} = \min\{|x|_{A^{-1}(0)}^{-a/2}, \delta^{-a/2}\}\zeta$$

and apply dominated convergence to take the limit as $\delta \to 0$.

By Remark 1.5.3, it suffices to compute the left-hand side of the inequality in Lemma 1.5.1 with A(0) in place of A(x). Since

$$\nabla(\eta^2) = -a\zeta^2 |x|_{A^{-1}(0)}^{-(a+2)} A^{-1}(0)x + |x|_{A^{-1}(0)}^{-a} \nabla(\zeta^2),$$

the first integrand in (1.5.11) is equal to

$$\begin{aligned} |\nabla u|^{2}_{A(0)} \Big((n-2)\eta^{2} + x \cdot \nabla(\eta^{2}) \Big) \\ &= (n-2-a)|x|^{-a}_{A^{-1}(0)} |\nabla u|^{2}_{A(0)} \zeta^{2} + |x|^{-a}_{A^{-1}(0)} |\nabla u|^{2}_{A(0)} (x \cdot \nabla(\zeta^{2})). \end{aligned}$$
(1.5.14)

Moreover, since

$$|\nabla \eta|_{A(0)}^2 = \frac{a^2}{4} |x|_{A^{-1}(0)}^{-(a+2)} \zeta^2 - \frac{a}{2} |x|_{A^{-1}(0)}^{-(a+2)} \left(x \cdot \nabla(\zeta^2) \right) + |x|_{A^{-1}(0)}^{-a} |\nabla \zeta|_{A(0)}^2,$$

the second integrand is

$$-2(x \cdot \nabla u)A(0)\nabla u \cdot \nabla(\eta^{2}) - |x \cdot \nabla u|^{2}|\nabla \eta|_{A(0)}^{2}$$

$$= \frac{a(8-a)}{4} |x|_{A^{-1}(0)}^{-(a+2)}|x \cdot \nabla u|^{2}\zeta^{2} - 2|x|_{A^{-1}(0)}^{-a}(x \cdot \nabla u)A(0)\nabla u \cdot \nabla(\zeta^{2}) \qquad (1.5.15)$$

$$- |x \cdot \nabla u|^{2}|x|_{A^{-1}(0)}^{-a}|\nabla \zeta|_{A(0)}^{2} + \frac{a}{2}|x \cdot \nabla u|^{2}|x|_{A^{-1}(0)}^{-a-2}(x \cdot \nabla(\zeta^{2})).$$

From the identities (1.5.14) and (1.5.15), by (1.5.11), it follows that

$$(n-2-a)\int_{B_{1}}|x|_{A^{-1}(0)}^{-a}|\nabla u|_{A(0)}^{2}\zeta^{2} dx + \frac{a(8-a)}{4}\int_{B_{1}}|x|_{A^{-1}(0)}^{-a-2}|x \cdot \nabla u|^{2}\zeta^{2} dx$$

$$\leq C\int_{B_{1}}\left(r^{2-a}|\nabla \zeta|^{2} + r^{1-a}|\nabla (\zeta^{2})|\right)|\nabla u|^{2} dx + C\varepsilon\int_{B_{1}}r^{2-a}|D^{2}u||\nabla u|\zeta^{2} dx$$

$$+ C\varepsilon\int_{B_{1}}\left(r^{3-a}|\nabla \zeta|^{2} + r^{2-a}|\nabla (\zeta^{2})|\right)|\nabla u|^{2} dx$$

$$+ C\varepsilon\int_{B_{1}}\left(r^{1-a} + \varepsilon r^{2-a}\right)|\nabla u|^{2}\zeta^{2} dx,$$
(1.5.16)

for some universal constant C, where we have controlled the remainder terms thanks to the uniform ellipticity and the fact that a is bounded by a dimensional constant.

For $0 < \rho \leq 1/2$ as in the statement, we consider a Lipschitz function ζ , with $0 \leq \zeta \leq 1$, such that $\zeta|_{B_{\rho}} = 1$, supp $\zeta \subset \overline{B}_{2\rho}$, and $|\nabla \zeta| \leq C/\rho$. Plugging this cutoff function in (1.5.16), using that r is comparable with ρ inside supp $\nabla \zeta \subset \overline{B}_{2\rho} \setminus B_{\rho}$, we deduce that

$$(n-2-a) \int_{B_{1}} |x|_{A^{-1}(0)}^{-a} |\nabla u|_{A(0)}^{2} \zeta^{2} dx + \frac{a(8-a)}{4} \int_{B_{1}} |x|_{A^{-1}(0)}^{-a-2} |x \cdot \nabla u|^{2} \zeta^{2} dx$$

$$\leq C \int_{B_{2\rho} \setminus B_{\rho}} r^{-a} |\nabla u|^{2} dx + C\varepsilon \int_{B_{2\rho}} r^{2-a} |D^{2}u| |\nabla u| dx \qquad (1.5.17)$$

$$+ C\varepsilon \int_{B_{2\rho}} (r^{1-a} + \varepsilon r^{2-a}) |\nabla u|^{2} dx.$$

Since *a* is in the range (1.5.12), the constants in the left-hand side of (1.5.17) are nonnegative. Moreover, by uniform ellipticity we have $|x|_{A^{-1}(0)} \leq c_0^{-1/2}|x|$ and $|\nabla u|_{A(0)}^2 \geq c_0|\nabla u|^2$, hence, since $\zeta|_{B_{\rho}} = 1$ and $\zeta \geq 0$, it follows that

$$\lambda^{a/2+1} \left((n-2-a) \int_{B_{\rho}} r^{-a} |\nabla u|^2 \, \mathrm{d}x + \frac{a(8-a)}{4} \int_{B_{\rho}} r^{-a} u_r^2 \, \mathrm{d}x \right)$$

$$\leq (n-2-a) \int_{B_1} |x|_{A^{-1}(0)}^{-a} |\nabla u|_{A(0)}^2 \zeta^2 \, \mathrm{d}x + \frac{a(8-a)}{4} \int_{B_1} |x|_{A^{-1}(0)}^{-a-2} |x \cdot \nabla u|^2 \zeta^2 \, \mathrm{d}x.$$
(1.5.18)

Using (1.5.18) in (1.5.17) and multiplying by $\lambda^{-a/2-1}$ now yields the claim.

We can finally prove Proposition 1.1.4. For this, we will apply Lemma 1.5.4 with the exponent a = n - 2. The key point in the proof will be to control the weighted L^1 norm of $|D^2u||\nabla u|$ in the right-hand side of (1.5.13) by a weighted L^2 norm of the gradient. We obtain this bound by writing the integral as an infinite sum on dyadic annuli and by using that the weight in each annulus can be pulled out of the integral. This allows us to apply the non-weighted a priori estimate for the "Hessian times the gradient" (1.1.14) from Theorem 1.1.2.

Proof of Proposition 1.1.4. Since $3 \le n \le 9$, we have that $\min\{10, n\} - 2 = n - 2$ and we

may choose the exponent a = n - 2 in Lemma 1.5.4, which yields the inequality

$$\frac{(n-2)(10-n)}{4} \int_{B_{\rho}} r^{2-n} u_r^2 \,\mathrm{d}x$$

$$\leq C \int_{B_{2\rho} \setminus B_{\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x + C\varepsilon \int_{B_{2\rho}} r^{4-n} |D^2 u| |\nabla u| \,\mathrm{d}x$$

$$+ C\varepsilon \int_{B_{2\rho}} (r^{3-n} + \varepsilon r^{4-n}) |\nabla u|^2 \,\mathrm{d}x.$$
(1.5.19)

To bound the Hessian term $\int_{B_{2\rho}} r^{4-n} |D^2 u| |\nabla u| dx$ on the right-hand side of (1.5.19), we will apply the a priori estimate on annuli (1.1.14) from Theorem 1.1.2 at different scales.

Let $r_j := 2^{-j}$ with $j \ge 0$. As explained in Section 1.2, the functions $u(r_j \cdot)$ are stable solutions to a semilinear equation with coefficients $A(r_j \cdot)$ and $r_j b(r_j \cdot)$. In particular, since $\|DA\|_{C^0(\overline{B}_1)} + \|b\|_{C^0(\overline{B}_1)} \le \varepsilon$, we also have $\|DA(r_j \cdot)\|_{L^{\infty}(B_1)} + \|r_j b(r_j \cdot)\|_{L^{\infty}(B_1)} \le \varepsilon r_j \le \varepsilon$. Hence, by (1.1.14) in Theorem 1.1.2 applied to $u(r_j \cdot)$, there is a universal $\varepsilon_0 > 0$ with the following property: if $\varepsilon \le \varepsilon_0$, then

$$\int_{B_{r_{j+1}} \setminus B_{r_{j+2}}} |D^2 u| |\nabla u| \, \mathrm{d}x \le C r_j^{-1} \int_{B_{r_j} \setminus B_{r_{j+3}}} |\nabla u|^2 \, \mathrm{d}x \quad \text{for all } j \ge 0, \tag{1.5.20}$$

where C is a universal constant.

Writing the weighted integral as an infinite sum on annuli, we have

$$\int_{B_{1/2}} r^{4-n} |D^2 u| |\nabla u| \, \mathrm{d}x = \sum_{j=0}^{\infty} \int_{B_{r_{j+1}} \setminus B_{r_{j+2}}} r^{4-n} |D^2 u| |\nabla u| \, \mathrm{d}x$$

$$\leq C \sum_{j=0}^{\infty} r_j^{4-n} \int_{B_{r_{j+1}} \setminus B_{r_{j+2}}} |D^2 u| |\nabla u| \, \mathrm{d}x,$$
(1.5.21)

where in the last line we have used that $r^{4-n} \leq Cr_j^{4-n}$ in $B_{r_{j+1}} \setminus B_{r_{j+2}}$, with C universal. Multiplying (1.5.20) by r_j^{4-n} and summing in j, the right-hand side in (1.5.21) can be bounded by

$$\sum_{j=0}^{\infty} r_j^{4-n} \int_{B_{r_{j+1}} \setminus B_{r_{j+2}}} |D^2 u| |\nabla u| \, \mathrm{d}x \le C \sum_{j=0}^{\infty} r_j^{3-n} \int_{B_{r_j} \setminus B_{r_{j+3}}} |\nabla u|^2 \, \mathrm{d}x$$
$$\le C \sum_{j=0}^{\infty} \int_{B_{r_j} \setminus B_{r_{j+3}}} r^{3-n} |\nabla u|^2 \, \mathrm{d}x \qquad (1.5.22)$$
$$\le C \int_{B_1} r^{3-n} |\nabla u|^2 \, \mathrm{d}x.$$

Combining (1.5.21) and (1.5.22), we deduce

$$\int_{B_{1/2}} r^{4-n} |D^2 u| |\nabla u| \, \mathrm{d}x \le C \int_{B_1} r^{3-n} |\nabla u|^2 \, \mathrm{d}x, \tag{1.5.23}$$

where C is universal. Applying (1.5.23) to the stable solutions $u(4\rho \cdot)$, there is a universal $\varepsilon_0 > 0$ with the following property: if $\varepsilon \leq \varepsilon_0$, then

$$\int_{B_{2\rho}} r^{4-n} |D^2 u| |\nabla u| \, \mathrm{d}x \le C \int_{B_{4\rho}} r^{3-n} |\nabla u|^2 \, \mathrm{d}x \quad \text{for all } \rho \le 1/4, \tag{1.5.24}$$

where C is a universal constant.

Applying (1.5.24) in (1.5.19), we deduce the key estimate

$$\frac{(n-2)(10-n)}{4} \int_{B_{\rho}} r^{2-n} u_r^2 \,\mathrm{d}x$$

$$\leq C \int_{B_{2\rho} \setminus B_{\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x + C\varepsilon \int_{B_{4\rho}} r^{3-n} |\nabla u|^2 \,\mathrm{d}x$$
for $\rho \leq 1/4$ and $\varepsilon \leq \varepsilon_0$,
$$(1.5.25)$$

where we have additionally bounded the last integrand in (1.5.19) by $\varepsilon r^{4-n} \leq Cr^{3-n}$. Finally, since (n-2)(10-n) > 0, absorbing this constant on the right-hand side of (1.5.25) yields the claim.

Remark 1.5.5. Our proof in Section 1.7 of the C^{α} bound (1.1.10) from Theorem 1.1.1 controls the weighted integral $\int_{B_{\rho}} r^{2-n} |\nabla u|^2 dx$. It will require a delicate estimate proven in Section 1.6. As a consequence, we will also obtain a bound for the less singular error terms $\int_{B_{\rho}} r^{3-n} |\nabla u|^2 dx$. Here we point out that this last quantity can be estimated directly from our previous Lemma 1.5.4, without the use of Section 1.6. This is done as follows.

Letting a = n-3 in Lemma 1.5.4, since $a(8-a) = (n-3)(11-n) \ge 0$ for $3 \le n \le 11$, we can drop the radial term in (1.5.13) and the left-hand side becomes $\int_{B_{\rho}} r^{3-n} |\nabla u|^2 dx$. The right-hand side now includes an error term $\varepsilon \int_{B_{2\rho}} r^{5-n} |D^2 u| |\nabla u| dx$, which can be bounded by $\varepsilon \int_{B_{4\rho}} r^{4-n} |\nabla u|^2 dx$ for $\varepsilon \le \varepsilon_0$ as in the proof of Proposition 1.1.4. Hence, we obtain

$$\int_{B_{\rho}} r^{3-n} |\nabla u|^2 \, \mathrm{d}x \le C \int_{B_{4\rho} \setminus B_{\rho}} r^{3-n} |\nabla u|^2 \, \mathrm{d}x + C\varepsilon \int_{B_{\rho}} r^{4-n} |\nabla u|^2 \, \mathrm{d}x$$

for $\rho \leq 1/4$. Making ε_0 smaller if necessary, since $r^{4-n} \leq r^{3-n}$ in B_{ρ} , we can absorb the rightmost term into the left-hand side. This leads to an expression which can be hole-filled, and by a standard iteration argument it is easy to deduce the decay estimate

$$\int_{B_{\rho}} r^{3-n} |\nabla u|^2 \,\mathrm{d}x \le C \rho^{2\alpha} \|\nabla u\|_{L^2(B_1)}^2 \quad \text{for all } \rho \le 1/4, \tag{1.5.26}$$

where $\alpha > 0$ and C are universal constants. We note that, adding superfluous variables, the decay (1.5.26) is satisfied when $n \leq 11$.

In general, this strategy allows to prove the decay of weighted integrals of the form $\int_{B_{\rho}} r^{-a} |\nabla u|^2 dx$ for exponents a < n-2 and $a \leq 8$. As an application, we could extend the optimal regularity result of Peng, Zhang, and Zhou [81] for stable solutions in dimensions $n \geq 10$ to our setting of variable coefficients. The key point in [81] is to find an a a priori estimate of the form $\rho^{-a_n} \int_{B_{\rho}} |\nabla u|^2 dx \leq C ||\nabla u||^2_{L^2(B_1)}$, where $a_n = 2(1 + \sqrt{n-1})$ is a critical exponent. When considering variable coefficients, one has to deal with error terms $\varepsilon \int_{B_{\rho}} r^{1-a_n} |\nabla u|^2 dx$. Our method above gives the decay of these error terms whenever $10 \leq n \leq 13$. For this, we choose $a = a_n - 1$ in (1.5.13), and notice that $a_n - 1 < n - 2$ and $a_n - 1 \leq 8$ in this dimension range. The case $n \geq 14$ can be treated similarly using that $a = a_n - 1 \geq 8$ and $u_r^2 \leq |\nabla u|^2$ in (1.5.13).

Remark 1.5.6. The key estimate in the proof of Proposition 1.1.4 is an inequality for weighted integrals,

$$\int_{B_{2\rho}} r^{4-n} |D^2 u| |\nabla u| \, \mathrm{d}x \le C \int_{B_{4\rho}} r^{3-n} |\nabla u|^2 \, \mathrm{d}x, \tag{1.5.27}$$

which has been proved decomposing the integral in dyadic annuli. There is a way to prove a weaker inequality than (1.5.27), namely,

$$\int_{B_{2\rho}} r^{4-n} |D^2 u| |\nabla u| \, \mathrm{d}x \le C \left(\int_{B_{4\rho}} r^{2-n} |\nabla u|^2 \, \mathrm{d}x \right)^{1/2} \left(\int_{B_{4\rho}} r^{4-n} |\nabla u|^2 \, \mathrm{d}x \right)^{1/2} + C \int_{B_{4\rho}} r^{3-n} |\nabla u|^2 \, \mathrm{d}x,$$
(1.5.28)

which avoids the use of annuli and which involves a very recent test function of Cabré [17] and Peng, Zhang, and Zhou [82].

To obtain (1.5.28), one uses the inequality (1.3.33) from the proof of Theorem 1.1.2 with the singular test function $\eta = r^{-n/2}\zeta$, where $\zeta \in C_c^{\infty}(B_1)$ is a cut-off. It is worth noting that this inequality relies on the Sternberg-Zumbrun estimate for $\int_{B_1} \mathcal{A}^2 \eta^2 dx$, which comes from choosing the test function $\xi = |\nabla u| \eta = |\nabla u| r^{-n/2}\zeta$ in the integral stability inequality (1.1.8). A function of this form had already appeared in [17], where the author considered $\xi = |\nabla u| r^{-(n+\varepsilon)/2}\zeta$ to prove the boundedness of stable solutions for unsigned nonlinearities when $n \leq 4$. Interestingly, our choice $\xi = |\nabla u| r^{-n/2}$ coincides with the test function used by the authors in [82], where they obtained an a priori Hölder estimate for stable solutions when $n \leq 5$. While their strategy involves integrating by parts an expression that is already quadratic in the gradient, expressing the new "coupled" Hessian errors $D^2 u \nabla u$ as the uncoupled expression $\Delta u \nabla u$ and an error term related to \mathcal{A}_0 , we do not integrate by parts.

By Young's inequality, the coefficient in front of $\int_{B_{4\rho}} r^{2-n} |\nabla u|^2 dx$, the leading order term in (1.5.28), can be made arbitrarily small. This error in a larger ball can still be treated with our methods to yield the Hölder estimate from Theorem 1.1.1.

1.6 The radial derivative controls the function in L^1

The goal of this section is to prove Proposition 1.1.5, where we control the L^1 norm of (generalized) superharmonic functions $Lu \leq 0$ by the L^1 norm of the radial derivative on annuli. By a comparison argument, it will suffice to derive the analogue L^1 estimate on spheres for harmonic functions Lv = 0, which are obtained by duality from the L^{∞} estimates of a Neumann problem.

Let $g \in C^{\infty}(\partial B_1)$, and consider the Neumann problem in divergence form

$$\begin{cases} \operatorname{div}(A(x)\nabla\varphi) = 0 & \operatorname{in} B_1\\ A(x)\nabla\varphi \cdot \nu = g & \operatorname{on} \partial B_1, \end{cases}$$
(1.6.1)

which admits solutions if and only if $\int_{\partial B_1} g \, d\mathcal{H}^{n-1} = 0$. Recall that the solutions of (1.6.1) are unique up to addition of constants. We will derive an L^{∞} a priori estimate for the zero mean solutions of (1.6.1) in terms of the conormal derivative

$$N\varphi := A(x)\nabla\varphi \cdot \frac{x}{|x|}.$$
(1.6.2)

This is achieved by a Moser iteration based on the following Sobolev trace inequality

$$\|u\|_{L^{2^{\star}}(\partial B_{1})}^{2} \leq C\left(\|\nabla u\|_{L^{2}(B_{1})}^{2} + \|u\|_{L^{2}(\partial B_{1})}^{2}\right)$$
(1.6.3)

for $u \in W^{1,2}(B_1)$, where C depends only on n and

$$2^* := \frac{n-1}{n-2}2\tag{1.6.4}$$

is the Sobolev trace exponent. We give a short proof of this inequality in Appendix B.

Our proof by Moser iteration is inspired by the one of Winkert in [96], where he obtains L^{∞} estimates for general quasilinear Neumann problems. While the author employs certain technical interpolation and trace inequalities from the theory of Besov and Lizorkin-Triebel spaces, we only need the basic trace inequality (1.6.3), for which we give an elementary proof.

Lemma 1.6.1. Let φ be the unique solution of (1.6.1) with $\int_{B_1} \varphi \, dx = 0$. Then

$$\|\varphi\|_{L^{\infty}(B_1)} \le C \|g\|_{L^{\infty}(\partial B_1)},$$

where C depends only on n and c_0 .

Proof. Dividing φ by the norm $||g||_{L^{\infty}(\partial B_1)}$, we may assume $||g||_{L^{\infty}(\partial B_1)} = 1$. By the maximum principle, it suffices to bound the L^{∞} norm of φ on the sphere ∂B_1 . In this proof, C always denotes a generic constant depending only on n and c_0 .

First, we obtain a basic energy estimate. Multiplying the equation (1.6.1) by φ and integrating by parts, we have

$$\int_{B_1} |\nabla \varphi|^2_{A(x)} \,\mathrm{d}x = \int_{\partial B_1} g\varphi \,\mathrm{d}\mathcal{H}^{n-1}.$$
(1.6.5)

Combining the standard trace inequality $||u||_{L^2(\partial B_1)}^2 \leq C(||\nabla u||_{L^2(B_1)}^2 + ||u||_{L^2(B_2)}^2)$ with the Poincaré inequality in the ball (recall that $\int_{B_1} \varphi = 0$), we also have

$$\|\varphi\|_{L^2(\partial B_1)} \le C \|\nabla\varphi\|_{L^2(B_1)}.$$
 (1.6.6)

Hence, applying Cauchy-Schwarz in (1.6.5), by (1.6.6) we obtain

$$\int_{B_1} |\nabla \varphi|^2_{A(x)} \,\mathrm{d}x \le \|g\|_{L^2(\partial B_1)} \|\varphi\|_{L^2(\partial B_1)} \le C \|g\|_{L^2(\partial B_1)} \|\nabla \varphi\|_{L^2(B_1)}.$$
(1.6.7)

By uniform ellipticity and the bound $||g||_{L^{\infty}(B_1)} = 1$, from (1.6.7) it follows that

$$\|\nabla\varphi\|_{L^2(B_1)} \le C. \tag{1.6.8}$$

Therefore, by (1.6.3), (1.6.6), and (1.6.8) we deduce the a priori estimate

$$\|\varphi\|_{L^{2^*}(\partial B_1)} \le C. \tag{1.6.9}$$

Next, we derive an L^{∞} bound for the positive part of the solution by Moser iteration. Let $m \geq 2$. Multiplying the equation by the power $(\varphi^+)^{m-1}$ and integrating by parts

$$(m-1)\int_{B_1} (\varphi^+)^{m-2} |\nabla \varphi^+|^2_{A(x)} \,\mathrm{d}x = \int_{\partial B_1} (\varphi^+)^{m-1} g \,\mathrm{d}\mathcal{H}^{n-1}.$$
 (1.6.10)

Since

$$|\nabla(\varphi^+)^{\frac{m}{2}}|^2_{A(x)} = \frac{m^2}{4}(\varphi^+)^{m-2}|\nabla(\varphi^+)|^2_{A(x)},$$

by (1.6.10) and the uniform ellipticity, using that $\frac{m}{m-1} \leq 2$ for $m \geq 2$, we have

$$\int_{B_1} |\nabla(\varphi^+)^{\frac{m}{2}}|^2 \,\mathrm{d}x \le Cm \int_{\partial B_1} (\varphi^+)^{m-1} \,\mathrm{d}\mathcal{H}^{n-1}.$$
(1.6.11)

Adding the integral $\int_{\partial B_1} (\varphi^+)^m \, \mathrm{d}\mathcal{H}^{n-1}$ to both sides of (1.6.11), we have

$$\|\nabla(\varphi^+)^{\frac{m}{2}}\|_{L^2(B_1)}^2 + \|(\varphi^+)^{\frac{m}{2}}\|_{L^2(\partial B_1)}^2 \le Cm\|(\varphi^+)^{m-1}\|_{L^1(\partial B_1)} + \|(\varphi^+)^{\frac{m}{2}}\|_{L^2(\partial B_1)}^2$$

and applying the Sobolev trace inequality (1.6.3) on the left-hand side yields

$$\|\varphi^{+}\|_{L^{\frac{2^{\star}}{2}m}(\partial B_{1})}^{m} \leq Cm \|(\varphi^{+})^{m-1}\|_{L^{1}(\partial B_{1})} + C \|\varphi^{+}\|_{L^{m}(\partial B_{1})}^{m}.$$
 (1.6.12)

By Hölder and since $m \ge 2$, the L^{m-1} norm in (1.6.12) can be bounded by

$$\|(\varphi^{+})^{m-1}\|_{L^{1}(\partial B_{1})} \leq |\partial B_{1}|^{\frac{1}{m}} \|\varphi^{+}\|_{L^{m}(\partial B_{1})}^{m-1} \leq C \|\varphi^{+}\|_{L^{m}(\partial B_{1})}^{m-1},$$

and hence

$$\|\varphi^{+}\|_{L^{\frac{2^{\star}}{2}m}(\partial B_{1})}^{m} \leq Cm \|\varphi^{+}\|_{L^{m}(\partial B_{1})}^{m-1} + C \|\varphi^{+}\|_{L^{m}(\partial B_{1})}^{m}.$$
 (1.6.13)

Since $\|\varphi^+\|_{L^m(\partial B_1)}^{m-1} \le \max\{1, \|\varphi^+\|_{L^m(\partial B_1)}\}^m$, from (1.6.13) it follows that

$$\|\varphi^+\|_{L^{\frac{2^{\star}}{2}m}(\partial B_1)} \le C^{\frac{1}{m}} m^{\frac{1}{m}} \max\{1, \|\varphi^+\|_{L^m(\partial B_1)}\}.$$
(1.6.14)

We wish to iterate (1.6.14). Let $m_0 := 2^*$ and, for $k \in \mathbb{N}$, let

$$m_k := \left(\frac{2^\star}{2}\right)^k m_0$$

By (1.6.14) and the definition of m_k , we have

$$\begin{aligned} \|\varphi^{+}\|_{L^{m_{k}}(\partial B_{1})} &\leq C^{\frac{1}{m_{k-1}}} m_{k-1}^{\frac{1}{m_{k-1}}} \max\{1, \|\varphi^{+}\|_{L^{m_{k-1}}(\partial B_{1})}\} \\ &= (Cm_{0})^{\frac{1}{m_{0}}\left(\frac{2}{2^{\star}}\right)^{k-1}} \left(\frac{2^{\star}}{2}\right)^{\frac{k-1}{m_{0}}\left(\frac{2}{2^{\star}}\right)^{k-1}} \max\{1, \|\varphi^{+}\|_{L^{m_{k-1}}(\partial B_{1})}\}. \end{aligned}$$
(1.6.15)

We have exactly one of the following three cases:

• <u>Case 1</u>:

$$\|\varphi^+\|_{L^{m_k}(\partial B_1)} \le 1 \tag{1.6.16}$$

• <u>Case 2</u>: there is an $l \in \{1, 2, \dots, k-1\}$ such that

$$\|\varphi^+\|_{L^{m_{k+1-i}}(\partial B_1)} > 1 \quad \text{for } 1 \le i \le l, \quad \text{and} \quad \|\varphi^+\|_{L^{m_{k-l}}(\partial B_1)} \le 1.$$
 (1.6.17)

• <u>Case 3</u>:

$$\|\varphi^+\|_{L^{m_{k+1-i}}(\partial B_1)} > 1 \quad \text{for } 1 \le i \le k.$$
 (1.6.18)

Case 1 already yields a uniform bound for $\|\varphi^+\|_{L^{m_k}(B_1)}$. If Case 2 holds then, iterating (1.6.15) l-1 times, we arrive at

$$\|\varphi^{+}\|_{L^{m_{k}}(\partial B_{1})} \leq (Cm_{0})^{\frac{1}{m_{0}}\sum_{j=k-l}^{k-1}\left(\frac{2}{2^{\star}}\right)^{j}} \left(\frac{2^{\star}}{2}\right)^{\frac{1}{m_{0}}\sum_{j=k-l}^{k-1}j\left(\frac{2}{2^{\star}}\right)^{j}}.$$
 (1.6.19)

The right-hand side of (1.6.19) is nondecreasing in l for (say) $C \ge 1$, which we can always assume. Finally, if Case 3 holds then, iterating (1.6.15), we obtain

$$\|\varphi^{+}\|_{L^{m_{k}}(\partial B_{1})} \leq (Cm_{0})^{\frac{1}{m_{0}}\sum_{j=0}^{k-1}\left(\frac{2}{2^{\star}}\right)^{j}} \left(\frac{2^{\star}}{2}\right)^{\frac{1}{m_{0}}\sum_{j=0}^{k-1}j\left(\frac{2}{2^{\star}}\right)^{j}} \|\varphi^{+}\|_{L^{m_{0}}(\partial B_{1})}.$$
(1.6.20)

By the monotonicity of (1.6.19) in l and using the a priori estimate (1.6.9) for $\|\varphi\|_{L^{m_0}(\partial B_1)}$ in (1.6.20), we see that in all three cases above we have

$$\|\varphi^+\|_{L^{m_k}(\partial B_1)} \le C(Cm_0)^{\frac{1}{m_0}\sum_{j=0}^{k-1}\left(\frac{2}{2^\star}\right)^j} \left(\frac{2^\star}{2}\right)^{\frac{1}{m_0}\sum_{j=0}^{k-1}j\left(\frac{2}{2^\star}\right)^j}$$

and since the exponent on the right-hand side is uniformly bounded, we deduce

$$\|\varphi^+\|_{L^{m_k}(\partial B_1)} \le C. \tag{1.6.21}$$

Taking the limit as $k \to \infty$ in (1.6.21) now yields

$$\|\varphi^+\|_{L^{\infty}(\partial B_1)} \le C,$$

which is the desired L^{∞} estimate for the positive part of the solutions. The same argument gives an a priori estimate for the negative part φ^- and yields the claim.

By duality, from the L^{∞} estimate in Lemma 1.6.1 we deduce an L^1 bound for the elliptic problem with a source:

Lemma 1.6.2. Given $h \in C^{\infty}(\overline{B_1})$, let $v \in C^{\infty}(\overline{B_1})$ satisfy

$$\operatorname{div}(A(x)\nabla v) + h(x) = 0 \quad in \ B_1.$$

Then

$$\|v - t\|_{L^{1}(\partial B_{1})} \leq C \|Nv\|_{L^{1}(\partial B_{1})} + C \|h\|_{L^{1}(B_{1})},$$

where $t := \inf\{\bar{t} : |\{v > \bar{t}\} \cap \partial B_{1}| \leq |\partial B_{1}|/2\}$ and C depends only on n and c_{0} .

Proof. Replacing v by v-t we may assume that t = 0, therefore $|\{v > 0\} \cap \partial B_1| \leq |\partial B_1|/2$ and $|\{v < 0\} \cap \partial B_1| \leq |\partial B_1|/2$. The function $\operatorname{sgn}(v) = v/|v|$ in $v \neq 0$ can then be extended to $\{v = 0\} \cap \partial B_1$, taking values ± 1 and in such a way that $\int_{\partial B_1} \operatorname{sgn}(v) d\mathcal{H}^{n-1} = 0$. In particular, $|v| = v \operatorname{sgn}(v)$ on ∂B_1 .

We define the convolutions on ∂B_1

$$g_k := \operatorname{sgn}(v) \star \eta_k,$$

where $\{\eta_k\}$ is a sequence of smooth mollifiers on ∂B_1 . We have $g \in C^{\infty}(\partial B_1)$, $|g_k| \leq 1$, and $\int_{\partial B_1} g_k \, \mathrm{d}\mathcal{H}^{n-1} = 0$ since $\operatorname{sgn}(v)$ has zero average on ∂B_1 . Moreover, it holds that

$$\int_{\partial B_1} |v| \, \mathrm{d}\mathcal{H}^{n-1} = \lim_k \int_{\partial B_1} v g_k \, \mathrm{d}\mathcal{H}^{n-1}.$$
(1.6.22)

Since g_k has zero average on ∂B_1 , we can uniquely solve the Neumann problem

$$\begin{cases} \operatorname{div}(A(x)\nabla\varphi_k) = 0 & \text{in } B_1 \\ N\varphi_k = g_k & \text{on } \partial B_1 \end{cases}$$

imposing additionally that $\int_{B_1} \varphi_k \, dx = 0$. By Lemma 1.6.1, we deduce

$$\|\varphi_k\|_{L^{\infty}(\partial B_1)} \le C,\tag{1.6.23}$$

where C depends only on n and c_0 . Notice that, integrating by parts, we have

$$\int_{\partial B_1} v g_k \, \mathrm{d}\mathcal{H}^{n-1} = \int_{\partial B_1} v N \varphi_k \, \mathrm{d}\mathcal{H}^{n-1}$$
$$= \int_{\partial B_1} (Nv) \varphi_k \, \mathrm{d}\mathcal{H}^{n-1} - \int_{B_1} \operatorname{div}(A(x)\nabla v) \varphi_k \, \mathrm{d}x$$
$$= \int_{\partial B_1} (Nv) \, \varphi_k \, \mathrm{d}\mathcal{H}^{n-1} + \int_{B_1} h \, \varphi_k \, \mathrm{d}x,$$

where in the last equality we have used the equation satisfied by v. Hence, by (1.6.23)

$$\left| \int_{\partial B_1} v g_k \, \mathrm{d}\mathcal{H}^{n-1} \right| \le C \int_{\partial B_1} |Nv| \, \mathrm{d}\mathcal{H}^{n-1} + C \int_{B_1} |h| \, \mathrm{d}x,$$

and the claim follows by (1.6.22).

We now use the previous estimates on spheres to obtain L^1 bounds on annuli for a divergence-form operator with drift. The drift term will be treated as a source, which will appear as an error in the right-hand side of the estimate. If the coefficient matrix A(x) is close to the identity, then the conormal derivative Nu is close to the radial derivative u_r . Hence, we will obtain Proposition 1.1.5 as a corollary of the following:

Proposition 1.6.3. Let $u \in C^{\infty}(\overline{B}_1)$ be a supersolution $\mathcal{L}u \leq 0$ in B_1 , where \mathcal{L} is the operator $\mathcal{L}u = \operatorname{div}(A(x)\nabla u) + d(x) \cdot \nabla u$. Assume that

$$\|d\|_{C^0(\overline{B}_1)} \le \varepsilon$$

for some $\varepsilon > 0$.

Then there exists a constant t, which depends on u, such that

$$||u - t||_{L^{1}(B_{1} \setminus B_{1/8})} \le C ||u_{r}||_{L^{1}(B_{1} \setminus B_{1/8})} + C ||Nu||_{L^{1}(B_{1} \setminus B_{1/8})} + C\varepsilon ||\nabla u||_{L^{1}(B_{1})},$$

where C is a constant depending only on n and c_0 .

Proof. Since $||Nu||_{L^1(B_1\setminus B_{1/8})} = \int_{1/8}^1 \mathrm{d}r \int_{\partial B_r} |Nu| \,\mathrm{d}\mathcal{H}^{n-1}$, by the mean value theorem

$$\|Nu\|_{L^{1}(B_{1}\setminus B_{1/8})} = \frac{7}{8} \int_{\partial B_{\rho}} |Nu| \, \mathrm{d}\mathcal{H}^{n-1}$$
(1.6.24)

for some $\rho \in [1/8, 1]$. Let v be the unique solution of the boundary value problem

$$\begin{cases} \operatorname{div}(A(x)\nabla v) + d(x) \cdot \nabla u = 0 & \text{in } B_{\rho} \\ v = u & \text{on } \partial B_{\rho} \end{cases}$$

Since $\operatorname{div}(A(x)\nabla(u-v)) \leq 0$ in B_{ρ} , by the comparison principle $u \geq v$ in B_{ρ} . Moreover, using that u = v on ∂B_{ρ} , we deduce that $Nu \leq Nv$ on ∂B_{ρ} . In particular, this gives

$$(Nv)^- \le (Nu)^- \text{ on } \partial B_{\rho}.$$
 (1.6.25)

Notice also that, integrating the equation $\operatorname{div}(A(x)\nabla v) = -d(x) \cdot \nabla u$ in B_{ρ} , by the divergence theorem we have

$$\left| \int_{\partial B_{\rho}} Nv \, \mathrm{d}\mathcal{H}^{n-1} \right| = \left| \int_{B_{\rho}} \operatorname{div} \left(A(x) \nabla v \right) \, \mathrm{d}x \right| = \left| - \int_{B_{\rho}} d(x) \cdot \nabla u \, \mathrm{d}x \right| \le \varepsilon \| \nabla u \|_{L^{1}(B_{\rho})}$$

and since $\int_{\partial B_{\rho}} Nv \, \mathrm{d}\mathcal{H}^{n-1} = \|(Nv)^+\|_{L^1(B_{\rho})} - \|(Nv)^-\|_{L^1(B_{\rho})}$, we deduce

$$\|(Nv)^+\|_{L^1(B_{\rho})} \le \|(Nv)^-\|_{L^1(B_{\rho})} + \varepsilon \|\nabla u\|_{L^1(B_{\rho})}.$$
(1.6.26)

Using (1.6.26) and (1.6.25) we obtain

$$\|Nv\|_{L^{1}(\partial B_{\rho})} = \|(Nv)^{-}\|_{L^{1}(\partial B_{\rho})} + \|(Nv)^{+}\|_{L^{1}(\partial B_{\rho})}$$

$$\leq 2\|(Nv)^{-}\|_{L^{1}(\partial B_{\rho})} + \varepsilon\|\nabla u\|_{L^{1}(B_{\rho})}$$

$$\leq 2\|Nu\|_{L^{1}(\partial B_{\rho})} + \varepsilon\|\nabla u\|_{L^{1}(B_{\rho})}.$$
(1.6.27)

Applying Lemma 1.6.2 with coefficients $A(\rho \cdot)$ and source $h(x) = \rho^2 d(\rho x) \cdot \nabla u(\rho x)$ to the function $v(\rho \cdot)$ yields the estimate

$$\|v - t\|_{L^{1}(\partial B_{\rho})} \le C\rho \|Nv\|_{L^{1}(\partial B_{\rho})} + C\rho\varepsilon \|\nabla u\|_{L^{1}(B_{\rho})}.$$
 (1.6.28)

Since u - t = v - t on ∂B_{ρ} , combining (1.6.28) and (1.6.27), we obtain

$$||u-t||_{L^1(\partial B_\rho)} \le C\rho ||Nu||_{L^1(\partial B_\rho)} + C\rho\varepsilon ||\nabla u||_{L^1(B_\rho)},$$

and since $\rho \in [1/8, 1]$, by (1.6.24), we deduce that

$$\|u - t\|_{L^{1}(\partial B_{\rho})} \le C \|Nu\|_{L^{1}(B_{1} \setminus B_{1/8})} + C\varepsilon \|\nabla u\|_{L^{1}(B_{1})}.$$
(1.6.29)

To conclude the proof, it suffices to show that

$$||u - t||_{L^{1}(B_{1} \setminus B_{1/8})} \le C ||u - t||_{L^{1}(\partial B_{\rho})} + C ||u_{r}||_{L^{1}(B_{1} \setminus B_{1/8})}.$$
(1.6.30)

Since $(u-t)(s\sigma) = (u-t)(\rho\sigma) - \int_s^{\rho} u_r(r\sigma) dr$ for every $s \in (1/8, 1)$ and $\sigma \in \partial B_1$, we have

$$s^{n-1}|(u-t)(s\sigma)| \le 8^{n-1}\rho^{n-1}|(u-t)(\rho\sigma)| + 8^{n-1}\int_{1/8}^{1} r^{n-1}|u_r(r\sigma)|\,dr$$

Integrating in $\sigma \in \partial B_1$, and then in $s \in (1/8, 1)$, we deduce (1.6.30). Combining (1.6.29) and (1.6.30) yields the claim.

Proof of Proposition 1.1.5. We consider the operator \mathcal{L} with $d(x) = \hat{b}(x)$ given by (1.1.7), so that $\mathcal{L}u = Lu = \operatorname{div}(A(x)\nabla u) + \hat{b}(x) \cdot \nabla u$. Since A(0) = I, writing the conormal derivative (1.6.2) as $Nu = u_r + (A(x) - I)\nabla u \cdot \frac{x}{|x|}$ and by the mean value theorem, we have $|Nu| \leq |u_r| + C\varepsilon |\nabla u|$ in B_1 . Applying Proposition 1.6.3 now, the conormal term on the right-hand side of the estimate can be bounded by $C||u_r||_{L^1(B_1\setminus B_{1/8})} + C\varepsilon ||\nabla u||_{L^1(B_1)}$, hence the claim.

1.7 Proof of the C^{α} **estimate**

This section is devoted to proving the Hölder regularity estimate (1.1.10) in Theorem 1.1.1. The main goal will be to show that the scale-invariant weighted integral $\int_{B_{\rho}} r^{2-n} |\nabla u|^2$ decays like a power $\rho^{2\alpha}$, since this implies a C^{α} bound of the original function. We will show this property under the additional assumption that the operator L is close to the Laplacian, i.e., assuming A(0) = I and $\|DA\|_{L^{\infty}(B_1)} + \|b\|_{L^{\infty}(B_1)} \leq \varepsilon$ with ε sufficiently small. An affine transformation will then lead to an estimate that is valid for all operators, with bounds depending on the norms of the coefficients.

The key idea is to write the weighted integral of the gradient as an infinite sum on dyadic annuli, pulling out the weights, and applying Propositions 1.1.3 and 1.1.5 in each annulus. This allows to control the weighted L^2 norm of the gradient by a weighted L^2 norm of the radial derivative. Once we have this bound, Proposition 1.1.4 will lead directly to the decay by a standard iteration argument. This will yield a bound of the C^{α} norm in terms of the L^2 norm of the gradient, which can be controlled by the L^1 norm of the solutions thanks to Proposition 1.1.3.

Proof of the Hölder estimate (1.1.10) in Theorem 1.1.1. We may assume that $3 \leq n \leq 9$. Indeed, when n = 2, we recover the estimate by applying Theorem 1.1.1 to the function $\tilde{u}(x_1, x_2, x_3) := u(x_1, x_2)$, which is a stable solution to the elliptic equation $L\tilde{u} + c_0\tilde{u}_{x_3x_3} = f(\tilde{u})$ in $B_1 \subset \mathbb{R}^3$. Similarly, when n = 1, one considers the function $\tilde{u}(x_1, x_2, x_3) := u(x_1)$.

Throughout the proof, C denotes a generic universal constant unless stated otherwise. The proof is divided in three steps.

Step 1: Under the assumption that

$$A(0) = I$$
 and $||DA||_{L^{\infty}(B_1)} + ||b||_{L^{\infty}(B_1)} \le \varepsilon$,

we prove that it $\varepsilon \leq \varepsilon_0$, then

$$\int_{B_{\rho}} r^{2-n} |\nabla u|^2 \, \mathrm{d}x \le C \|\nabla u\|_{L^2(B_1)}^2 \rho^{2\alpha} \quad \text{for all } \rho \le 1/8, \tag{1.7.1}$$

where $\alpha > 0$, $\varepsilon_0 > 0$, and C are universal constants.

As explained before, we will write the weighted Dirichlet integral as an infinite sum on dyadic annuli, similarly to what we did for the weighted Hessian estimates in the proof of Proposition 1.1.4. We treat the case $\rho = 1/2$ first, and then apply the scaling of the problem.

Let $r_j := 2^{-j}$ with $j \ge 0$. We have

$$\int_{B_{1/2}} r^{2-n} |\nabla u|^2 \, \mathrm{d}x = \sum_{j=0}^{\infty} \int_{B_{r_{j+1}} \setminus B_{r_{j+2}}} r^{2-n} |\nabla u|^2 \, \mathrm{d}x$$

$$\leq C \sum_{j=0}^{\infty} r_j^{2-n} \int_{B_{r_{j+1}} \setminus B_{r_{j+2}}} |\nabla u|^2 \, \mathrm{d}x.$$
(1.7.2)

We want to apply Proposition 1.1.3 on annuli to control the Dirichlet integrals in (1.7.2) by the L^1 norm of the solution, and then Proposition 1.1.5 to obtain bounds in terms of the radial derivative.

We cover the annulus $B_{1/2} \setminus B_{1/4}$ by a finite number of balls $B_{d/2}(y_j)$, where d = d(n) is small enough so that $B_d(y_j) \subset B_1 \setminus B_{1/8}$. The number of balls depends only on n.

As explained in Section 1.2, the functions $u(y_j + d \cdot)$ are stable solutions to a semilinear equation with coefficients $A(y_j + d \cdot)$ and $d b(y_j + d \cdot)$. Applying Proposition 1.1.3 to each $u(y_j + d \cdot)$, there is a universal $\varepsilon_0 > 0$ such that, for $\varepsilon \leq \varepsilon_0$, we have

$$\begin{aligned} |\nabla u||_{L^{2}(B_{1/2}\setminus B_{1/4})}^{2} &\leq \sum_{j} \|\nabla u\|_{L^{2}(B_{d/2}(y_{j}))}^{2} \leq C \sum_{j} \|u\|_{L^{1}(B_{2d}(y_{j}))}^{2} \\ &\leq C \|u\|_{L^{1}(B_{1}\setminus B_{1/8})}^{2}. \end{aligned}$$
(1.7.3)

For each $t \in \mathbb{R}$, the function u - t is a stable solution to $-L\tilde{u} = f(\tilde{u} + t)$ in B_1 . Hence, by (1.7.3), it follows that

$$\|\nabla u\|_{L^2(B_{1/2}\setminus B_{1/4})} \le C\|u-t\|_{L^1(B_1\setminus B_{1/8})} \quad \text{for all } t \in \mathbb{R} \text{ and } \varepsilon \le \varepsilon_0.$$

$$(1.7.4)$$

Since A(0) = I, we can choose t in (1.7.4) to be the constant in the conclusion of Proposition 1.1.5, and by this result we deduce

$$\|\nabla u\|_{L^{2}(B_{1/2}\setminus B_{1/4})} \le C\|u_{r}\|_{L^{1}(B_{1}\setminus B_{1/8})} + C\varepsilon\|\nabla u\|_{L^{1}(B_{1})} \quad \text{for all } \varepsilon \le \varepsilon_{0}.$$
(1.7.5)

Squaring (1.7.5) and by Cauchy-Schwarz, we also have the weaker

$$\|\nabla u\|_{L^{2}(B_{1/2}\setminus B_{1/4})}^{2} \leq C \|u_{r}\|_{L^{2}(B_{1}\setminus B_{1/8})}^{2} + C\varepsilon^{2} \|\nabla u\|_{L^{2}(B_{1})}^{2} \quad \text{for all } \varepsilon \leq \varepsilon_{0}.$$
(1.7.6)

Now we apply (1.7.6) to the rescaled functions $u(r_j \cdot)$, which gives (see the comments in (i) in Section 1.2)

$$\int_{B_{r_{j+1}}\setminus B_{r_{j+2}}} |\nabla u|^2 \,\mathrm{d}x \le C \int_{B_{r_j}\setminus B_{r_{j+3}}} u_r^2 \,\mathrm{d}x + C\varepsilon^2 r_j^2 \int_{B_{r_j}} |\nabla u|^2 \,\mathrm{d}x \quad \text{for all } \varepsilon \le \varepsilon_0.$$
(1.7.7)

Hence, multiplying (1.7.7) by r_j^{2-n} and summing in j

$$\sum_{j=0}^{\infty} r_j^{2-n} \int_{B_{r_j+1} \setminus B_{r_{j+2}}} |\nabla u|^2 dx$$

$$\leq C \sum_{j=0}^{\infty} r_j^{2-n} \int_{B_{r_j} \setminus B_{r_{j+3}}} u_r^2 dx + C\varepsilon^2 \sum_{j=0}^{\infty} r_j^{4-n} \int_{B_{r_j}} |\nabla u|^2 dx \qquad (1.7.8)$$

$$\leq C \sum_{j=0}^{\infty} \int_{B_{r_j} \setminus B_{r_{j+3}}} r^{2-n} u_r^2 dx + C\varepsilon^2 \sum_{j=0}^{\infty} r_j \int_{B_{r_j}} r^{3-n} |\nabla u|^2 dx \quad \text{for all } \varepsilon \leq \varepsilon_0,$$

where in the last line we have used that $r_j^{3-n} \leq r^{3-n}$ in B_{r_j} for $n \geq 3$. Since $r_j = 2^{-j}$, splitting the annuli into $B_{r_j} \setminus B_{r_{j+3}} = (B_{r_j} \setminus B_{r_{j+1}}) \cup (B_{r_{j+1}} \setminus B_{r_{j+2}}) \cup (B_{r_{j+2}} \setminus B_{r_{j+3}})$, we see that the first integral in the right-hand side of (1.7.8) is bounded by

$$\sum_{j=0}^{\infty} \int_{B_{r_j} \setminus B_{r_{j+3}}} r^{2-n} u_r^2 \le 3 \int_{B_1} r^{2-n} u_r^2,$$

while the second can be bounded by

$$\sum_{j=0}^{\infty} r_j \int_{B_{r_j}} r^{3-n} |\nabla u|^2 \le \left(\sum_{j=0}^{\infty} r_j\right) \int_{B_1} r^{3-n} |\nabla u|^2 = 2 \int_{B_1} r^{3-n} |\nabla u|^2.$$
From this, it follows that

$$\sum_{j=0}^{\infty} r_j^{2-n} \int_{B_{r_{j+1}} \setminus B_{r_{j+2}}} |\nabla u|^2 dx$$

$$\leq C \int_{B_1} r^{2-n} u_r^2 dx + C\varepsilon^2 \int_{B_1} r^{3-n} |\nabla u|^2 dx \quad \text{for all } \varepsilon \leq \varepsilon_0.$$
(1.7.9)

Combining (1.7.2) and (1.7.9) now yields

$$\int_{B_{1/2}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C \int_{B_1} r^{2-n} u_r^2 \,\mathrm{d}x + C\varepsilon^2 \int_{B_1} r^{3-n} |\nabla u|^2 \,\mathrm{d}x \quad \text{for all } \varepsilon \le \varepsilon_0, \quad (1.7.10)$$

and applying (1.7.10) to rescaled functions $u(2\rho)$, we deduce

$$\int_{B_{\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C \int_{B_{2\rho}} r^{2-n} u_r^2 \,\mathrm{d}x + C\varepsilon^2 \rho \int_{B_{2\rho}} r^{3-n} |\nabla u|^2 \,\mathrm{d}x$$
(1.7.11)
for all $\rho \le 1/2$ and $\varepsilon \le \varepsilon_0$.

Next, we apply the radial estimate (1.1.16) from Proposition 1.1.4 (with 2ρ) to bound the right-hand side of (1.7.11), which gives

$$\int_{B_{\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C \int_{B_{4\rho} \setminus B_{2\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x + C\varepsilon (1+\varepsilon\rho) \int_{B_{8\rho}} r^{3-n} |\nabla u|^2 \,\mathrm{d}x$$

for all $\rho \le 1/8$ and $\varepsilon \le \varepsilon_0$.
(1.7.12)

Hence, using that the bounds $\rho \leq 1/8$ and $\varepsilon \leq \varepsilon_0$ are universal, splitting the last integral into $B_{8\rho} = (B_{8\rho} \setminus B_{\rho}) \cup B_{\rho}$, and by $r^{3-n} \leq r^{2-n}$, from (1.7.12) we deduce

$$\int_{B_{\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C \int_{B_{8\rho} \setminus B_{\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x + C\varepsilon \int_{B_{\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x$$
(1.7.13)
for all $\rho \le 1/8$ and $\varepsilon \le \varepsilon_0$.

Taking $\varepsilon_0 > 0$ universal smaller if necessary, we can absorb the last integral into the left-hand side and obtain

$$\int_{B_{\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C \int_{B_{8\rho} \setminus B_{\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \quad \text{for all } \rho \le 1/8 \text{ and } \varepsilon \le \varepsilon_0.$$
(1.7.14)

Hole-filling (1.7.14), we also have

$$\int_{B_{\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le \theta \int_{B_{8\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \quad \text{for all } \rho \le 1/8 \text{ and } \varepsilon \le \varepsilon_0, \tag{1.7.15}$$

where $\theta = \frac{C}{1+C} \in (0,1)$ is universal. Iterating (1.7.15), for $8^{-(k+1)} < \rho \le 8^{-k}$ we deduce

$$\int_{B_{\rho}} r^{2-n} |\nabla u|^2 \, \mathrm{d}x \le \theta^k \int_{B_{8^{k_{\rho}}}} r^{2-n} |\nabla u|^2 \, \mathrm{d}x \le \frac{1}{\theta} \rho^{2\alpha} \int_{B_1} r^{2-n} |\nabla u|^2 \, \mathrm{d}x,$$

where $\alpha = -\frac{1}{2}\log_8 \theta > 0$, and hence

$$\int_{B_{\rho}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C\rho^{2\alpha} \int_{B_1} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \quad \text{for all } \rho \le 1/8 \text{ and } \varepsilon \le \varepsilon_0.$$
(1.7.16)

Finally, we can estimate the integral in the right-hand side of (1.7.16) by splitting $B_1 = (B_1 \setminus B_{1/8}) \cup B_{1/8}$ and applying (1.7.14) with $\rho = 1/8$ to bound the term in the annulus. This yields the claim.

Step 2: Assuming

$$\|DA\|_{L^{\infty}(B_1)} + \|b\|_{L^{\infty}(B_1)} \le \varepsilon,$$

we prove that if $\varepsilon \leq \varepsilon_0$, then

$$\|u\|_{C^{\alpha}(\overline{B}_{\theta})} \le C \|u\|_{L^{1}(B_{1})}, \tag{1.7.17}$$

where $\alpha > 0$, $\theta > 0$, $\varepsilon_0 > 0$, and C are universal.

As explained in Section 1.2, for each ball $B_d(y) \subset B_1$, by uniform ellipticity, the function $u^{y,d}(x) := u(y + \frac{d}{\sqrt{C_0}}A^{1/2}(y)x)$ is a stable solution of an equation in B_1 with coefficients

$$A^{y,d}(x) := A^{-1/2}(y)A\left(y + \frac{d}{\sqrt{C_0}}A^{1/2}(y)x\right)A^{-1/2}(y)$$

and

$$b^{y,d}(x) := \frac{d}{\sqrt{C_0}} A^{-1/2}(y) b \Big(y + \frac{d}{\sqrt{C_0}} A^{1/2}(y) x \Big).$$

Notice that the matrix $A^{y,d}$ satisfies $A^{y,d}(0) = I$ and the coefficients can be bounded by $\|DA^{y,d}\|_{L^{\infty}(B_1)} + \|b^{y,d}\|_{L^{\infty}(B_1)} \leq Cd(\|DA\|_{L^{\infty}(B_1)} + \|b\|_{L^{\infty}(B_1)}) \leq \varepsilon$. Choosing d > 0universal sufficiently small so that $Cd \leq 1$, we have

$$||DA^{y,d}||_{L^{\infty}(B_1)} + ||b^{y,d}||_{L^{\infty}(B_1)} \le \varepsilon$$
 for all $y \in B_{1-d}$

Hence, for $\varepsilon \leq \varepsilon_0$ with the $\varepsilon_0 > 0$ from Step 1, by (1.7.1) we deduce

$$\int_{B_{\rho}} r^{2-n} |\nabla u^{y,d}|^2 \, \mathrm{d}x \le C \|\nabla u^{y,d}\|_{L^2(B_1)}^2 \rho^{2\alpha} \quad \text{for } y \in B_{1-d} \text{ and } \rho \le 1/8,$$

and since $\int_{B_{\rho}} r^{2-n} |\nabla u^{y,d}|^2 \, \mathrm{d}x \ge \rho^{2-n} \int_{B_{\rho}} |\nabla u^{y,d}|^2 \, \mathrm{d}x$, we also have

$$\int_{B_{\rho}} |\nabla u^{y,d}|^2 \,\mathrm{d}x \le C \|\nabla u^{y,d}\|_{L^2(B_1)}^2 \rho^{2\alpha+n-2} \quad \text{for } y \in B_{1-d} \text{ and } \rho \le 1/8.$$
(1.7.18)

For the remaining part of the proof of Step 2, we assume that $\varepsilon \leq \varepsilon_0$.

Now we express (1.7.18) in terms of the original function u. By the change of variables $z = y + \frac{d}{\sqrt{C_0}} A^{1/2}(y) x$ and by uniform ellipticity, using that $B_{\sqrt{C_0}\rho} \subset A^{1/2}(y)(B_\rho)$, we have

$$\begin{split} \int_{B_{\rho}} |\nabla u^{y,d}|^2 \, \mathrm{d}x &= \frac{d^{2-n}}{C_0^{1-n/2}} \det(A(y))^{-1/2} \int_{y+\frac{d}{\sqrt{C_0}} A^{1/2}(y)(B_{\rho})} |\nabla u|^2_{A(y)} \, \mathrm{d}z \\ &\ge c \, d^{2-n} \int_{B_{d\sqrt{\frac{c_0}{C_0}}\rho}(y)} |\nabla u|^2 \, \mathrm{d}z \end{split}$$

for some universal c > 0. Similarly, we also have $\|\nabla u^{y,d}\| \leq Cd^{2-n} \|\nabla u\|_{L^2(B_1)}$ and, therefore, from (1.7.18) we deduce

$$\int_{B_{d\sqrt{\frac{c_0}{C_0}\rho}}(y)} |\nabla u|^2 \, \mathrm{d}z \le C \|\nabla u\|_{L^2(B_1)}^2 \rho^{n-2+2\alpha} \quad \text{for } y \in B_{1-d} \text{ and } \rho \le 1/8.$$
(1.7.19)

Dividing ρ by $d\sqrt{\frac{c_0}{C_0}}$ in (1.7.19) and letting $\theta := \frac{d}{16}\sqrt{\frac{c_0}{C_0}}$, since d is universal, we obtain

$$\int_{B_{\rho}(y)} |\nabla u|^2 \, \mathrm{d}z \le C \|\nabla u\|_{L^2(B_1)}^2 \rho^{n-2+2\alpha} \quad \text{for } y \in B_{1-d} \text{ and } \rho \le 2\theta,$$

and by Cauchy-Schwarz we also have the weaker

$$\int_{B_{\rho}(y)} |\nabla u| \, \mathrm{d}z \le C \|\nabla u\|_{L^{2}(B_{1})} \rho^{n-1+\alpha} \quad \text{for } y \in B_{1-d} \text{ and } \rho \le 2\theta.$$
(1.7.20)

Taking d smaller if necessary, we may assume that $B_{2\theta} \subset B_{1-d}$. Hence, from (1.7.20) it follows that

$$\int_{B_{\rho}(y)} |\nabla u| \,\mathrm{d}z \le C \|\nabla u\|_{L^2(B_1)} \rho^{n-1+\alpha} \quad \text{for all balls } B_{\rho}(y) \subset B_{2\theta}. \tag{1.7.21}$$

Applying [63, Theorem 7.19] with $\Omega = B_{2\theta}$, we deduce the Hölder estimate

$$\|u\|_{C^{\alpha}(B_{2\theta})} \le C \|\nabla u\|_{L^{2}(B_{1})}.$$
(1.7.22)

To obtain the final bound (1.7.17) in terms of the L^1 norm, apply (1.7.22) to the rescaled function $u(\cdot/2)$ first, and then Proposition 1.1.3 (taking ε_0 smaller if necessary).

Step 3: Conclusion. Scaling and covering argument.

We cover $B_{1/2}$ by balls $B_{\theta\rho}(y_j)$, where θ is the universal constant in Step 2 above and ρ is small so that $B_{\rho}(y_j) \subset B_1$. The number of balls depends only on n, ρ , and $\theta = \theta(n, c_0, C_0)$. We choose ρ smaller still so that

$$\left(\|DA\|_{L^{\infty}(B_{1})}+\|b\|_{L^{\infty}(B_{1})}\right)\rho \leq \varepsilon_{0},\tag{1.7.23}$$

with $\varepsilon_0 > 0$ the universal constant in Step 2. Thus $\rho = \rho(n, c_0, C_0, \|DA\|_{L^{\infty}(B_1)}, \|b\|_{L^{\infty}(B_1)})$. The functions $u(y_j + \rho \cdot)$ are stable solutions of an elliptic equation with coefficients $A^{y_j,\rho} = A(y_j + \rho \cdot)$ and $b^{y_j,\rho} = \rho b(y_j + \rho \cdot)$. Since $B_{\rho}(y_j) \subset B_1$ and by (1.7.23), the coefficients satisfy the bounds

$$\|DA^{y_j,\rho}\|_{L^{\infty}(B_1)} + \|b^{y_j,\rho}\|_{L^{\infty}(B_1)} \le \left(\|DA\|_{L^{\infty}(B_{\rho}(y_j))} + \|b\|_{L^{\infty}(B_{\rho}(y_j))}\right)\rho \le \varepsilon_0,$$

therefore, we can apply Step 2 to deduce

$$\|u\|_{C^{\alpha}(\overline{B}_{1/2})} \leq \sum_{j} \|u\|_{C^{\alpha}(\overline{B}_{\theta\rho}(y_{j}))} \leq C \sum_{j} \|u\|_{L^{1}(B_{\rho}(y_{j}))} \leq C \|u\|_{L^{1}(B_{1})},$$

where $C = C(n, c_0, C_0, ||DA||_{L^{\infty}(B_1)}, ||b||_{L^{\infty}(B_1)})$. This concludes the proof of the theorem.

Chapter 2

Energy estimate up to the boundary

In this chapter, we obtain a universal energy estimate up to the boundary for stable solutions of semilinear equations with variable coefficients. Namely, we consider solutions to -Lu = f(u), where L is a linear uniformly elliptic operator and f is C^1 , such that the linearized equation -L - f'(u) has nonnegative principal eigenvalue. Our main result is an estimate for the $L^{2+\gamma}$ norm of the gradient of stable solutions vanishing on the flat part of a half-ball, for *any* nonnegative and nondecreasing f. This bound only requires the elliptic coefficients to be Lipschitz. As a consequence, our estimate continues to hold in general $C^{1,1}$ domains if we further assume the nonlinearity f to be convex. This result is new even for the Laplacian, for which a C^3 regularity assumption on the domain was needed.

2.1 Introduction

Given a bounded domain $\Omega \subset \mathbb{R}^n$ and a function $f \in C^1(\mathbb{R})$, we consider stable solutions $u: \overline{\Omega} \to \mathbb{R}$ to the semilinear boundary value problem

$$\begin{cases} -Lu = f(u) & \text{in } \Omega\\ u = 0 & \text{on } \partial\Omega. \end{cases}$$
(2.1.1)

Throughout the text, L denotes a uniformly elliptic operator of the form

$$L = a_{ij}(x)\partial_{ij} + b_i(x)\partial_i, \quad a_{ij}(x) = a_{ji}(x).$$

$$(2.1.2)$$

A solution u of (2.1.1) is called *stable* if the principal eigenvalue (with respect to Dirichlet conditions) of the linearized equation $J_u := L + f'(u)$ is nonnegative.¹ When the problem is variational, this amounts to the nonnegativity of the second variation, a necessary condition for the minimality of u.

The goal of the present article is to obtain a universal energy estimate for stable solutions to (2.1.1) in the spirit of the pioneering work of Cabré, Figalli, Ros-Oton, and Serra [24] for the Laplacian. In [24], the authors proved two types of a priori bounds for classical stable solutions when $L = \Delta$. Namely, a control of the $L^{2+\gamma}$ norm of the gradient (for some $\gamma > 0$) by the L^1 norm of the function, valid in all dimensions, and an estimate of the Hölder norm of the solution when $n \leq 9$. The latter result is optimal, since there are examples of singular (unbounded) stable solutions in dimensions $n \geq 10$. A notable feature of these estimates is that they do not depend on the nonlinearity, which is assumed to be nonnegative, nondecreasing, and convex. Thanks to this, the paper [24]

¹Here we adopt the sign convention $J_u \varphi = -\mu \varphi$ for the eigenvalues μ of J_u .

answered positively two long-standing open questions of Brezis and Vázquez [14] and of Brezis [11] concerning the regularity of extremal solutions (which are L^1 limits of classical stable solutions), recalled briefly below.

Here we will be interested in extending the $L^{2+\gamma}$ energy estimate to operators with variable coefficients as in (2.1.2). Our main achievement is to make the constants in our bounds depend on the $C^{0,1}$ norm of a_{ij} and the L^{∞} norm of b_i , this being the major difficulty in our proofs. As a consequence, we will obtain a global estimate in $C^{1,1}$ domains. This result is new even when L is the Laplacian, as [24] required a C^3 regularity assumption on the domain. For this, starting from a curved boundary, we flatten it out locally by a change of variables. In the new coordinates, our solution is still a stable solution to an equation of the form (2.1.1), where the new operator L now involves the derivatives of the flattening map. More precisely, the new coefficients a_{ij} depend on the differential of this map, while b_i additionally depend on its Hessian. It follows that the $C^{0,1}$ and L^{∞} regularity of the coefficients corresponds to a $C^{1,1}$ domain. In particular, it will suffice to prove a priori estimates in half-balls with the stated dependence on the coefficients.

Furthermore, when $n \leq 9$, our energy bound (as well as the auxiliary Hessian estimates in Theorem 2.1.4 below) will be crucial to establish Hölder estimates up to the boundary in $C^{1,1}$ domains. We tackle this issue in the next chapter, where we extend the optimal C^{α} bounds of [24] to equations with coefficients. The previous work [24] relied on delicate contradiction-compactness arguments which do not allow to quantify the constants in the estimates. Here, thanks to a new device of Cabré [19] for the Laplacian in flat domains we will be able to give a direct, quantitative proof of all our estimates in [50].

The study of the regularity of stable solutions to (2.1.1) was originally motivated by problems in combustion theory. In that setting, the interest lies in positive, nondecreasing, convex, and superlinear nonlinearities f accounting for the reaction of a combustible mixture. It is also natural to consider a multiple λf of the nonlinearity, where $\lambda > 0$ is a nondimensional parameter measuring the relative strength of the reaction with respect to the processes modeled by L. Applying the implicit function theorem at $\lambda = 0$ and by the properties of f, one obtains a branch of positive classical stable solutions $\{u_{\lambda}\}_{0<\lambda<\lambda^{*}}$ of $-Lu_{\lambda} = \lambda f(u_{\lambda})$ in Ω , $u_{\lambda} = 0$ on $\partial\Omega$, where $0 < \lambda^{*} < \infty$ is the maximal threshold for the existence of classical solutions to this problem. Moreover, by maximum principle, $\lambda \mapsto u_{\lambda}$ is increasing in $(0, \lambda^{*})$ and converges in L^{1} to a weak (distributional) solution u^{*} , the so called *extremal solution*; see, for instance [11,44,48].

By construction, the extremal solution u^* is a priori only in L^1 and can be unbounded. In [14], Brezis and Vázquez gave a characterization of singular (unbounded) extremal solutions in the energy space $W_0^{1,2}(\Omega)$ when L is the Laplacian. Their result led them to ask whether extremal solutions are necessarily in this space; see [14, Problem 1]. This question has been addressed in various works, always considering the model operator $L = \Delta$. The first result in this direction was obtained by Nedev [78], who showed the validity of the claim for $n \leq 5$. Later, assuming Ω to be convex (or, more generally, "bean shaped"), he was able to extend this result to all dimensions in an unpublished preprint [79] (which is recalled and proven again in [28]). Then, Cabré and Capella studied radial stable solutions in $\Omega = B_1$, showing that $u^* \in W^{3,2}(B_1)$ in this case. After that, Cabré and Ros-Oton [27] proved the claim for $n \leq 6$ in domains of double revolution, and Villegas [95] obtained the same result in general smooth domains. Recently, Cabré, Figalli, Ros-Oton, and Serra [24] settled the conjecture, showing that $u^* \in W_0^{1,2+\gamma}(\Omega)$ in all dimensions, where $\gamma > 0$ depends only on n, and Ω is of C^3 class. For this, as mentioned above, they proved a universal energy estimate for smooth stable solutions. Then, they applied it to the functions $\{u_\lambda\}_{0<\lambda>^*}$ and passed to the limit as $\lambda \to \lambda^*$.

For further regularity properties of u^* , the dimension of the space plays a critical role. Notice that, by the linear theory, the smoothness of u^* follows from its boundedness. When $n \ge 10$, explicit unbounded extremal solutions had been known for a long time, while no such examples were found in lower dimensions. In [11, Open problem 1], Brezis asked whether the extremal solution was always bounded in the latter case. His question prompted a series of works trying to establish L^{∞} bounds for classical stable solutions in the range $n \le 9$. Recently, in the breakthrough paper [24] mentioned above, the question was answered positively for the Laplacian in C^3 domains. For more information on that problem, see the references in [24, 50] or, for instance, in Chapters 1 and 3 of this thesis.

2.1.1 Main results

We assume that the coefficient matrix $A(x) = (a_{ij}(x))$ is uniformly elliptic in Ω , that is, there are positive constants c_0 , C_0 such that

$$c_0|p|^2 \le a_{ij}(x)p_ip_j \le C_0|p|^2 \quad \text{for all } p \in \mathbb{R}^n.$$
(2.1.3)

Our global theorem in $C^{1,1}$ domains requires the assumption

$$a_{ij} \in C^{0,1}(\overline{\Omega}), \quad b_i \in L^{\infty}(\Omega) \cap C^0(\Omega),$$

$$(2.1.4)$$

For our local results in half-balls, we further need the auxiliary condition

$$b_i \in C^0(\overline{\Omega}). \tag{2.1.5}$$

We will be able to remove (2.1.5) by an approximation argument, as explained in Remark 2.1.6.

Since we always assume $a_{ij} \in C^{0,1}(\overline{\Omega}) = W^{1,\infty}(\Omega)$, we can write L in divergence form

$$Lu = \operatorname{div} \left(A(x)\nabla u \right) + \widehat{b}(x) \cdot \nabla u, \qquad (2.1.6)$$

where $\widehat{b}(x) = (\widehat{b}_i(x))$ is the vector field given by

$$\widehat{b}_i(x) = b_i(x) - \partial_k a_{ki}(x).$$
(2.1.7)

Notice that \hat{b}_i is in $L^{\infty}(\Omega)$ by assumption (2.1.4).

Having specified the regularity of the coefficients, we can give a more precise definition of stable solution. Assuming (2.1.3) and (2.1.4), we consider the class of *strong solutions* to (2.1.1), that is, functions $u \in C^0(\overline{\Omega}) \cap W^{2,n}_{\text{loc}}(\Omega)$ such that -Lu = f(u) a.e. in Ω and u = 0 on $\partial\Omega$. As commented above, a strong solution u of (2.1.1) is *stable* if the principal eigenvalue of the linearized equation at u is nonnegative. Equivalently (see [9]), the solution u is stable if there exists a function $\varphi \in W^{2,n}_{\text{loc}}(\Omega)$ such that

$$\begin{cases} J_u \varphi \le 0 & \text{a.e. in } \Omega, \\ \varphi > 0 & \text{in } \Omega, \end{cases}$$
(2.1.8)

where, recall, $J_u = L + f'(u)$ denotes the Jacobi operator (the linearization) at u. We would like to point out that the notion of stable solution refers only to the equation satisfied by u and not to its boundary value. Our energy estimate in $C^{1,1}$ domains will apply to strong stable solutions as above. In a sense, these functions are the natural replacement of the classical solutions for the Laplacian in C^3 domains considered in [24]. Notice that, since $f(u) \in L^{\infty}(\Omega)$, by L^p estimates in $C^{1,1}$ domains (see [63, Theorem 9.13]), strong solutions belong to $W^{2,p}(\Omega)$ for all $p < \infty$. For further regularity properties, more assumptions on the coefficients and the domain are needed. In fact, our a priori estimates in half-balls below require our solutions to have third weak derivatives, but we will be able to remove this assumption by an approximation argument; see Remark 2.1.6.

We now state the main result of this chapter, an energy estimate up to the boundary in flat domains. For $\rho > 0$, we denote the half-ball of radius ρ centered at 0 by

$$B_{\rho}^{+} := \{x_n > 0\} \cap B_{\rho},$$

where $B_{\rho} = \{ |x| < \rho \} \subset \mathbb{R}^n$ is the full-ball. We also write

$$\partial^0 B_\rho^+ = \{x_n = 0\} \cap \partial B_\rho^+.$$

for the lower boundary of B_{ρ}^+ . In the results below, $C = C(\ldots)$ denotes a constant C depending only on the quantities appearing inside the parentheses. We have the following:

Theorem 2.1.1. Let L satisfy conditions (2.1.3), (2.1.4), and (2.1.5) in $\Omega = B_1^+ \subset \mathbb{R}^n$. Assume that $f \in C^1(\mathbb{R})$ is nonnegative and nondecreasing.

Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a nonnegative stable solution to -Lu = f(u) in B_1^+ , with u = 0 on $\partial^0 B_1^+$.

Then

$$\|\nabla u\|_{L^{2+\gamma}(B^+_{1/2})} \le C \|u\|_{L^1(B^+_{1})},$$

where $\gamma = \gamma(n) > 0$ and $C = C(n, c_0, C_0, \|\nabla a_{ij}\|_{L^{\infty}(B_1^+)}, \|b_i\|_{L^{\infty}(B_1^+)}).$

Remark 2.1.2. Note that we are further assuming $u \in W^{3,p}(B_1^+)$ with p > n. In particular, by Sobolev embedding, u is in $C^2(\overline{B_1^+})$ and the solution is classical.² We need third weak derivatives in order to have a cancellation which removes the nonlinearity in the stability condition. This step is crucial for our bounds to be independent of f.

Remark 2.1.3. The continuity $b_i \in C^0(\overline{B_1^+})$ up to the boundary (assumption (2.1.5) above) will allow us to control these coefficients on certain surface integrals over $\partial^0 B_1^+$ arising in the proof. Assuming only $b_i \in L^{\infty}(B_1^+)$ does not suffice for such estimates on surfaces.

To prove Theorem 2.1.1, the stability condition (2.1.8) will come into play through a useful integral inequality that has already appeared in the previous chapter and was introduced in our work [52]. Recall that since the coefficient matrix $A(x) = (a_{ij}(x))$ is positive definite, it gives rise to a norm

$$|p|_{A(x)} := (a_{ij}(x)p_ip_j)^{1/2} \quad \text{for } p \in \mathbb{R}^n.$$

$$u(x', x_n) = -10u(x', -x_n) + 160u(x', -\frac{x_n}{2}) - 405u(x', -\frac{x_n}{3}) + 256u(x', -\frac{x_n}{4})$$

for $x_n < 0$ and $x' \in \mathbb{R}^{n-1}$, which is in $W^{3,p}(B_1) \subset C^2(\overline{B_1})$.

²For the embedding in half-balls, just apply the usual Sobolev embedding in the full ball to a third order reflection of u, for instance, letting

In [52], we showed that if u is stable, then

$$\int_{\Omega} f'(u)\xi^2 \,\mathrm{d}x \le \int_{\Omega} \left| \nabla \xi - \frac{1}{2}\xi A^{-1}(x)\widehat{b}(x) \right|_{A(x)}^2 \,\mathrm{d}x \quad \text{for all } \xi \in C_c^{\infty}(\Omega), \tag{2.1.9}$$

where $\hat{b}(x)$ is the vector field introduced in (2.1.7) above. As shown in Chapter 1 above, (2.1.9) follows from the pointwise inequality in (2.1.8) multiplying by ξ^2/φ , integrating by parts, and completing squares. We often refer to (2.1.9) as the "integral stability inequality" to distinguish it from the pointwise condition (2.1.8) above. Moreover, we would like to point out that the inequality (2.1.9) is not equivalent to our stability condition (2.1.8) in general; see [52] or Chapter 1 above.

A fundamental ingredient in the proof of Theorem 2.1.1 will be to control the Hessian of a stable solution in half-balls. The following boundary Hessian estimates can be interpreted as a generalization of a geometric stability condition due to Sternberg and Zumbrun [91]. As above, throughout this chapter, a constant depending only on n, c_0 , and C_0 will be called *universal*.

Theorem 2.1.4. Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a nonnegative stable solution of -Lu = f(u) in B_1^+ , with u = 0 on $\partial^0 B_1^+$. Assume that $f \in C^1(\mathbb{R})$ is nonnegative. Assume that L satisfies conditions (2.1.3), (2.1.4), and (2.1.5) in $\Omega = B_1^+$, and that

$$\|DA\|_{L^{\infty}(B_{1}^{+})} + \|b\|_{L^{\infty}(B_{1}^{+})} \le \varepsilon$$

for some $\varepsilon > 0$. Then

$$\begin{split} \int_{B_1^+} \mathcal{A}^2 \eta^2 \, \mathrm{d}x &\leq C \int_{B_1^+} |\nabla u|^2 \left(|\nabla \eta|^2 + |D^2(\eta^2)| + \varepsilon |\nabla(\eta^2)| + \varepsilon^2 \eta^2 \right) \mathrm{d}x \\ &+ C \int_{B_1^+} |D^2 u| |\nabla u| \left(|\nabla(\eta^2)| + \varepsilon \eta^2 \right) \mathrm{d}x \\ &+ C \int_{\partial^0 B_1^+} |\nabla u|^2 (|\nabla(\eta^2)| + \varepsilon \eta^2) \, \mathrm{d}\mathcal{H}^{n-1} \end{split}$$
(2.1.10)

for all $\eta \in C_c^{\infty}(B_1)$, where C is a universal constant and

$$\mathcal{A} := \begin{cases} \left(\operatorname{tr}(A(x)D^2 u A(0)D^2 u) - |\nabla u|_{A(0)}^{-2}|D^2 u A(0)\nabla u|_{A(x)}^2 \right)^{1/2} & \text{if } \nabla u \neq 0 \\ 0 & \text{if } \nabla u = 0. \end{cases}$$
(2.1.11)

Assume moreover that f is nondecreasing and $\varepsilon \leq \varepsilon_0$. Then

$$\|\nabla u\|_{L^2(\partial^0 B^+_{2/3})} \le C \|\nabla u\|_{L^2(B^+_1)}, \tag{2.1.12}$$

$$\||\nabla u|D^2 u\|_{L^1(B^+_{4/7})} \le C \|\nabla u\|_{L^2(B^+_1)}^2, \tag{2.1.13}$$

$$\|\mathcal{A}\|_{L^{2}(B_{1/2}^{+})} \leq C \|\nabla u\|_{L^{2}(B_{1}^{+})}, \qquad (2.1.14)$$

and

$$\|D^2 u\|_{L^1(B^+_{4/7})} \le C \|\nabla u\|_{L^2(B^+_1)}, \tag{2.1.15}$$

where $\varepsilon_0 > 0$ and C are universal constants.

To prove the first bound (2.1.10) in Theorem 2.1.4, we will exploit the integral stability inequality (2.1.9) by choosing appropriate test functions. Letting $\xi = \mathbf{c}\eta$ in (2.1.9) with $\Omega = B_1^+$, where \mathbf{c} , η are smooth functions satisfying $\mathbf{c} = 0$ on $\partial^0 B_1^+$ and $\operatorname{supp} \eta \subset B_1$, if we integrate by parts, then (2.1.9) becomes

$$\int_{B_1^+} \mathbf{c} J_u \mathbf{c} \, \eta^2 \, \mathrm{d}x \le \int_{B_1^+} \mathbf{c}^2 \left| \nabla \eta - \frac{1}{2} \eta A^{-1}(x) b(x) \right|_{A(x)}^2 \mathrm{d}x.$$
(2.1.16)

In order to obtain universal estimates, the crucial point will be to choose \mathbf{c} in such a way that the Jacobi operator $J_u \mathbf{c}$ in the left-hand side of (2.1.16) becomes independent of the nonlinearity. Thus, in the proof of (2.1.10), our choice will be a smooth approximation of

$$\mathbf{c}(x) = |\nabla u(x)|_{A(0)} - \mathbf{N} \cdot \nabla u(x)$$

for an appropriate constant vector field $\mathbf{N} \colon \mathbb{R}^n_+ \to \mathbb{R}^n$ (given by (2.2.3) in Section 2.2 below). Here, we need $f \geq 0$ to make sure that such an approximation of \mathbf{c} vanishes on $\partial^0 B_1^+$, but otherwise is a technical assumption in this step.

Under a smallness assumption on the coefficients ($\varepsilon \leq \varepsilon_0$), the function \mathcal{A} in (2.1.11) controls part of the Hessian of u (as explained in [52] or in Section 2.2 below). We can further bound the full Hessian by assuming that the equation has a sign $-Lu = f(u) \geq 0$. For the the final form of the Hessian estimates in (2.1.13), (2.1.14), and (2.1.15), we need to control the third term in the right-hand side of (2.1.10), which is a surface integral and arises at every integration by parts. To control such an integral requires both the monotonicity of f and the stability of u, while the previous works [20,24] only needed the condition on f. The reason for this is an additional Hessian error which does not appear for the Laplacian on C^3 domains when trying to control the boundary integral.

Once Theorem 2.1.4 is available, our main result, Theorem 2.1.1, will follow directly by the ideas of [19, 24] combined with a scaling and covering argument.

To conclude this section, we state our energy estimate in general domains of $C^{1,1}$ class. Approximating (2.1.1) by stable solutions to smoother problems (as explained next in Remark 2.1.6), flattening the boundary, the result will follow easily from Theorem 2.1.1 and by the interior estimates obtained in Chapter 1. This argument requires the convexity of f to ensure that the approximating sequence of stable solutions converges to the original one. The same procedure can be used to obtain Hölder estimates up to the boundary in $C^{1,1}$ domains, which has been carried out in our work [50] and we describe in detail in Chapter 3 below. Since the ideas are very similar, we defer the complete proof of Theorem 2.1.5 below to Chapter 3, where we implement the approximation and flattening argument in full detail. Here, we just give indications in Remark 2.1.6, after the theorem.

Theorem 2.1.5 (Theorem 3.1.1). Let $\Omega \subset \mathbb{R}^n$ be a bounded domain of class $C^{1,1}$ and let L satisfy conditions (2.1.3) and (2.1.4) in Ω . Assume that $f \in C^1(\mathbb{R})$ is nonnegative, nondecreasing, and convex.

Let $u \in \tilde{C}^{0}(\overline{\Omega}) \cap W^{2,n}_{\text{loc}}(\Omega)$ be a nonnegative stable solution of -Lu = f(u) in Ω , with u = 0 on $\partial\Omega$.

Then

 $\|\nabla u\|_{L^{2+\gamma}(\Omega)} \le C \|u\|_{L^{1}(\Omega)},$ where $\gamma = \gamma(n) > 0$ and $C = C(\Omega, n, c_{0}, C_{0}, \|\nabla a_{ij}\|_{L^{\infty}(\Omega)}, \|b_{i}\|_{L^{\infty}(\Omega)}).$

Remark 2.1.6. As mentioned above, our energy estimate in $C^{1,1}$ domains will follow from Theorem 2.1.1 but, unlike this result, it does not require third derivatives of the solution or

assumption (2.1.5) (the continuity of b_i up to the boundary). To achieve this, we consider an exhaustion of Ω by smooth sets Ω_k . Using u as a barrier, by monotone iteration, we construct strong stable solutions u_k to a semilinear equation $-L_k u_k = f_k(u_k)$ in Ω_k with smoother coefficients.

Flattening the boundary $\partial\Omega_k$, we obtain solutions in the half-ball, where we would like to apply Theorem 2.1.1. For this, we need to ensure the existence of third weak derivatives in L^p for these solutions, which is guaranteed if the new coefficients $(b_i^k)_k$ are sufficiently regular.³ The interior continuity of b_i (assumption (2.1.4) above) will make sure that $b_i^k \to b_i$ locally uniformly in Ω , which is essential to show that u is a barrier.

Finally, we need the convexity of f for u_k to converge to the original solution u and not to some other limit. The deeper reason behind this is that stable solutions with convex nonlinearities are unique; see [48,50] or Appendix F in Chapter 3 below. For C^3 domains and smooth coefficients, we do not need the approximation procedure and we could give the analogue of Theorem 2.1.5 without the convexity assumption on f.

2.1.2 Outline of the chapter

Section 2.2 is devoted to the proof of Theorem 2.1.4 containing the Hessian estimates up to the boundary. In Section 2.3 we prove Theorem 2.1.1, the energy estimate in half-balls.

2.2 Boundary Hessian estimates

Recall the function $\mathcal{A}: \overline{B}_1 \to \mathbb{R}$ given by (2.1.11) in the statement of Theorem 2.1.4 (and introduced in (1.3.1) in the previous chapter). It can also be written as

$$\mathcal{A} = \left(\|A^{1/2}(x)D^2 u A^{1/2}(0)\|_{\mathrm{HS}}^2 - |A^{1/2}(x)D^2 u A^{1/2}(0)\mathbf{n}(x)|^2 \right)^{1/2} \quad \text{in } \{\nabla u \neq 0\}, \quad (2.2.1)$$

where $\|\cdot\|_{\text{HS}}$ denotes the Euclidean Hilbert-Schmidt norm for matrices and $\mathbf{n}(x)$ is the unit vector field $\mathbf{n}: (B_1^+ \cup \partial^0 B_1^+) \cap \{\nabla u \neq 0\} \to \mathbb{R}$ given by

$$\mathbf{n}(x) := |\nabla u|_{A(0)}^{-1} A^{1/2}(0) \nabla u(x).$$
(2.2.2)

Here we are following the notation for the Hessian estimates in Chapter 1 above.

First we prove the bound (2.1.10) for \mathcal{A} in Theorem 2.1.4. This is an analogue of the Sternberg-Zumbrun geometric estimate up to the boundary. For this, it is convenient to define the constant vector field

$$\mathbf{N} := |e_n|_{A(0)}^{-1} A(0) e_n, \qquad \mathbf{N}_i := (a_{nn}(0))^{-1/2} a_{in}(0).$$
(2.2.3)

Notice that **N** has unit norm with respect to the scalar product defined by the inverse matrix $A^{-1}(0)$, i.e., $|\mathbf{N}|_{A^{-1}(0)} = 1$. Moreover, since u is nonnegative and u = 0 on $\partial^0 B_1^+$, we have the identity

$$\nabla u|_{A(0)} = \mathbf{N} \cdot \nabla u \quad \text{on } \partial^0 B_1^+.$$
(2.2.4)

³For instance, suppose that $a_{ij} \in C^{\overline{0,1}}(\overline{B_1^+})$ and $b_i \in W^{1,p}(B_1^+)$ for some p > n, and let $u \in W^{2,p}(B_1^+)$ be a strong solution to -Lu = f(u) in B_1^+ , u = 0 on $\partial^0 B_1^+$. Since $f(u) \in L^{\infty}(B_1^+)$, by Calderón-Zygmund estimates (see [63, Theorem 9.13]) we have $u \in W^{2,q}_{\text{loc}}(B_1^+ \cup \partial^0 B_1^+)$ for all $q < \infty$. Formally taking tangential derivatives, for $k = 1, \ldots, n-1$ we obtain $-Lu_k = f'(u)u_k + \partial_k a_{ij}(x)u_{ij} + \partial_k b_i(x)u_i \in L^p_{\text{loc}}(B_1^+ \cup \partial^0 B_1^+)$ and $u_k = 0$ on $\partial^0 B_1^+$, hence, again by L^p estimates, we deduce $u \in W^{3,p}_{\text{loc}}(B_1^+ \cup \partial^0 B_1^+)$. It remains to check that the weak derivative u_{nnn} exists and lies in $L^p_{\text{loc}}(B_1^+ \cup \partial^0 B_1^+)$, but this follows easily from the equation.

The vector field \mathbf{N} will also be useful later when controlling the Dirichlet energy on the boundary.

Proof of (2.1.10) *in Theorem 2.1.4.* We test the stability inequality (2.1.16) with a variant of

$$\mathbf{c} := |\nabla u|_{A(0)} - \mathbf{N} \cdot \nabla u,$$

where $\mathbf{N} = (\mathbf{N}_i)$ is the constant vector field defined in (2.2.3) above. Since $|\nabla u|_{A(0)}$ is not necessarily smooth when $\nabla u = 0$, following [24], we take a convex $C^{1,1}$ regularization of the modulus $|\cdot|_{A(0)}$ instead. For each small $\delta > 0$, we define

$$\phi_{\delta}(z) := |z|_{A(0)} \mathbb{1}_{\{|z|_{A(0)} > \delta\}} + \left(\frac{\delta}{2} + \frac{|z|_{A(0)}^2}{2\delta}\right) \mathbb{1}_{\{|z|_{A(0)} < \delta\}}.$$
(2.2.5)

Given that u is nonnegative and superharmonic (in the sense that $Lu = -f(u) \leq 0$), unless $u \equiv 0$ (in which case there is nothing to prove), by the Hopf lemma and uniform ellipticity we have $|\nabla u|_{A(0)} \geq c > 0$ on $\partial^0 B_1^+ \cap \operatorname{supp} \eta$, for some constant c. Hence, for $\delta > 0$ sufficiently small we have

$$\phi_{\delta}(\nabla u) = |\nabla u|_{A(0)}$$
 in a neighborhood of $\partial^0 B_1^+ \cap \operatorname{supp} \eta$ inside B_1^+ . (2.2.6)

Choosing $\delta > 0$ small enough such that (2.2.6) holds, we let

$$\mathbf{c}_{\delta} := \phi_{\delta}(\nabla u) - \mathbf{N} \cdot \nabla u.$$

Since \mathbf{c}_{δ} vanishes on $\partial^0 B_1^+$, this is a valid test function in the stability inequality (2.1.16).

We can write the Jacobi operator acting on \mathbf{c}_{δ} as the sum of three terms

$$\begin{aligned} \mathbf{c}_{\delta} J_{u} \mathbf{c}_{\delta} &= \mathbf{c}_{\delta} (L \mathbf{c}_{\delta} + f'(u) \mathbf{c}_{\delta}) \\ &= \phi_{\delta} (\nabla u) J_{u} \left[\phi_{\delta} (\nabla u) \right] - \mathbf{c}_{\delta} J_{u} \left[\mathbf{N} \cdot \nabla u \right] - (\mathbf{N} \cdot \nabla u) J_{u} \left[\phi_{\delta} (\nabla u) \right]. \end{aligned}$$

Multiplying this identity by η^2 and integrating in B_1^+ yields the left-hand side of (2.1.16), i.e.,

$$\int_{B_1^+} \mathbf{c}_{\delta} J_u \, \mathbf{c}_{\delta} \eta^2 \, \mathrm{d}x = \int_{B_1^+} \phi_{\delta}(\nabla u) J_u \left[\phi_{\delta}(\nabla u)\right] \eta^2 \, \mathrm{d}x - \int_{B_1^+} \mathbf{c}_{\delta} J_u \left[\mathbf{N} \cdot \nabla u\right] \eta^2 \, \mathrm{d}x - \int_{B_1^+} (\mathbf{N} \cdot \nabla u) J_u \left[\phi_{\delta}(\nabla u)\right] \eta^2 \, \mathrm{d}x.$$

$$(2.2.7)$$

We now study each of the three terms in (2.2.7) separately.

First term. We prove that

$$\int_{B_1^+} \phi_{\delta}(\nabla u) J_u \left[\phi_{\delta}(\nabla u)\right] \eta^2 dx$$

$$\geq \int_{B_1^+} \mathcal{A}^2 \mathbb{1}_{\{|\nabla u|_{A(0)} > \delta\}} \eta^2 dx - C\delta \int_{B_1^+} |f'(u)| (|\nabla u| + \delta) \eta^2 dx$$

$$- C\varepsilon \int_{B_1^+} (|\nabla u| + \delta) \left(|D^2 u| \eta^2 + |\nabla u| |\nabla(\eta^2)| \right) dx - C\varepsilon \int_{\partial^0 B_1^+} |\nabla u|^2 \eta^2 d\mathcal{H}^{n-1}.$$
(2.2.8)

Recalling that $J_u = L + f'(u)$, we start by computing $L[\phi_{\delta}(\nabla u)]$ first. Here, since $u \in W^{3,p}(B_1^+)$ with p > n, by Sobolev embedding $u \in C^2(\overline{B_1^+})$ and, moreover, the Hessian D^2u is differentiable a.e. in B_1^+ (for instance, see [55]). Thus we have

$$L\left[\phi_{\delta}(\nabla u)\right] = a_{ij}(x)\partial_{ij}^{2}\left[\phi_{\delta}(\nabla u)\right] + b_{i}(x)\partial_{i}\left[\phi_{\delta}(\nabla u)\right]$$

= $a_{ij}(x)\partial_{z_{k}}\phi_{\delta}(\nabla u)u_{ijk} + a_{ij}(x)\partial_{z_{k}z_{l}}^{2}\phi_{\delta}(\nabla u)u_{jk}u_{il} + b_{i}(x)\partial_{z_{k}}\phi_{\delta}(\nabla u)u_{ik}$ (2.2.9)

a.e. in B_1^+ . By the convexity of ϕ_{δ} and its definition (2.2.5), it is easy to check that

$$\phi_{\delta}(\nabla u)a_{ij}(x)\partial_{z_k z_l}^2 \phi_{\delta}(\nabla u)u_{jk}u_{il} \ge \mathcal{A}^2 \,\mathbb{1}_{\{|\nabla u|_{A(0)} > \delta\}}.$$
(2.2.10)

Therefore, multiplying (2.2.9) by $\phi_{\delta}(\nabla u)\eta^2$, using (2.2.10), and integrating we obtain

$$\int_{B_1^+} \phi_{\delta}(\nabla u) L\left[\phi_{\delta}(\nabla u)\right] \eta^2 dx$$

$$\geq \int_{B_1^+} a_{ij}(x) \nabla u_{ij} \cdot \nabla \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2 dx + \int_{B_1^+} \mathcal{A}^2 \mathbb{1}_{\{|\nabla u|_{A(0)} > \delta\}} \eta^2 dx \qquad (2.2.11)$$

$$+ \int_{B_1^+} b_i(x) \partial_{z_k} \phi_{\delta}(\nabla u) u_{ik} \phi_{\delta}(\nabla u) \eta^2 dx.$$

Next, we treat the zero order term $f'(u)\phi_{\delta}(\nabla u)$ in the linearization $J_u[\phi_{\delta}(\nabla u)]$. By direct computation $|\phi_{\delta}(\nabla u) - \nabla \phi_{\delta}(\nabla u) \cdot \nabla u| \leq \delta$ and hence

$$\int_{B_1^+} f'(u)\phi_{\delta}(\nabla u)^2 \eta^2 \,\mathrm{d}x \ge \int_{B_1^+} f'(u)\nabla u \cdot \nabla \phi_{\delta}(\nabla u)\phi_{\delta}(\nabla u)\eta^2 \,\mathrm{d}x - \delta \int_{B_1^+} |f'(u)|\phi_{\delta}(\nabla u)\eta^2 \,\mathrm{d}x.$$
(2.2.12)

Using the equation, we integrate by parts the first term in the right-hand side of (2.2.12) as

$$\int_{B_1^+} f'(u) \nabla u \cdot \nabla \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2 \, \mathrm{d}x = \int_{B_1^+} \nabla [f(u)] \cdot \nabla \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2 \, \mathrm{d}x$$

$$= \int_{B_1^+} Lu \, \mathrm{div} \left(\nabla \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2 \right) \, \mathrm{d}x - \int_{\partial^0 B_1^+} f(u) \partial_{z_n} \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2 \, \mathrm{d}\mathcal{H}^{n-1}.$$
(2.2.13)

Moreover, undoing the integration by parts in

$$\begin{split} \int_{B_1^+} a_{ij}(x) u_{ij} \operatorname{div} \left(\nabla \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2 \right) \mathrm{d}x \\ &= - \int_{B_1^+} \nabla [a_{ij}(x) u_{ij}] \cdot \nabla \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2 \mathrm{d}x \\ &+ \int_{\partial^0 B_1^+} a_{ij}(x) u_{ij} \partial_{z_n} \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2 \mathrm{d}\mathcal{H}^{n-1}, \end{split}$$

substituting in (2.2.13) and using that -Lu = f(u) up to $\overline{B_1^+}$ by continuity, we deduce

$$\int_{B_1^+} f'(u) \nabla u \cdot \nabla \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2 dx$$

$$= -\int_{B_1^+} \nabla [a_{ij}(x)u_{ij}] \cdot \nabla \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2 dx + \int_{B_1^+} b_i(x)u_i \operatorname{div} \left(\nabla \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2\right) dx$$

$$+ \int_{\partial^0 B_1^+} b_i(x)u_i \partial_{z_n} \phi_{\delta}(\nabla u) \phi_{\delta}(\nabla u) \eta^2 d\mathcal{H}^{n-1}.$$
(2.2.14)

Finally, combining (2.2.11), (2.2.12), and (2.2.14), we obtain

$$\begin{split} \int_{B_1^+} \phi_{\delta}(\nabla u) J_u \left[\phi_{\delta}(\nabla u)\right] \eta^2 \, \mathrm{d}x &= \int_{B_1^+} \phi_{\delta}(\nabla u) L \left[\phi_{\delta}(\nabla u)\right] \eta^2 \, \mathrm{d}x + \int_{B_1^+} f'(u) \phi_{\delta}(\nabla u)^2 \eta^2 \, \mathrm{d}x \\ &\geq \int_{B_1^+} \mathcal{A}^2 \, \mathbb{1}_{\{|\nabla u|_{A(0)} > \delta\}} \eta^2 \, \mathrm{d}x - \delta \int_{B_1^+} |f'(u)| \phi_{\delta}(\nabla u) \eta^2 \, \mathrm{d}x \\ &- \int_{B_1^+} \nabla a_{ij}(x) u_{ij} \cdot \nabla \phi_{\delta}(\nabla u) \, \phi_{\delta}(\nabla u) \eta^2 \, \mathrm{d}x \\ &+ \int_{B_1^+} b_i(x) \left\{ u_i \operatorname{div} \left(\nabla \phi_{\delta}(\nabla u) \, \phi_{\delta}(\nabla u) \eta^2 \right) + \partial_{z_k} \phi_{\delta}(\nabla u) u_{ik} \phi_{\delta}(\nabla u) \eta^2 \right\} \, \mathrm{d}x \\ &+ \int_{\partial^0 B_1^+} b_i(x) u_i |e_n|_{A(0)} |\nabla u|_{A(0)} \eta^2 \, \mathrm{d}\mathcal{H}^{n-1}, \end{split}$$

$$(2.2.15)$$

where in the boundary term we have used (2.2.6) and (2.2.4) to write

$$\partial_{z_n}\phi_{\delta}(\nabla u)\phi_{\delta}(\nabla u) = e_n \cdot A(0)\nabla u = |e_n|_{A(0)}|\nabla u|_{A(0)} \quad \text{on } \partial^0 B_1^+.$$

The claim now follows from (2.2.15) by applying the uniform ellipticity, the coefficient bounds $\|\nabla a_{ij}\|_{L^{\infty}} + \|b_i\|_{C^0} \leq \varepsilon$, and

$$\phi_{\delta}(\nabla u) \le C\left(|\nabla u| + \delta\right),\tag{2.2.16}$$

$$|\nabla\phi_{\delta}(\nabla u)| + \phi_{\delta}(\nabla u)|D^{2}\phi_{\delta}(\nabla u)| \le C, \qquad (2.2.17)$$

where C are universal constants (and hence independent of δ).

Second term. We prove that

$$\left| \int_{B_1^+} \mathbf{c}_{\delta} J_u \left[\mathbf{N} \cdot \nabla u \right] \eta^2 \, \mathrm{d}x \right|$$

$$\leq C \varepsilon \int_{B_1^+} \left(|\nabla u| + \delta \right) \left(|D^2 u| \eta^2 + |\nabla u| |\nabla (\eta^2)| \right) \, \mathrm{d}x.$$
(2.2.18)

Computing, we have

$$L[\mathbf{N} \cdot \nabla u] = \mathbf{N}_k a_{ij}(x) u_{ijk} + \mathbf{N}_k b_i(x) u_{ik} \quad \text{a.e. in } B_1^+.$$
(2.2.19)

Since $\mathbf{c}_{\delta} = 0$ on $\partial^0 B_1^+$, integrating by parts and using the equation

$$\int_{B_1^+} f'(u) \left(\mathbf{N} \cdot \nabla u \right) \, \mathbf{c}_{\delta} \eta^2 \, \mathrm{d}x = \int_{B_1^+} \mathbf{N} \cdot \nabla [f(u)] \, \mathbf{c}_{\delta} \eta^2 \, \mathrm{d}x = \int_{B_1^+} (Lu) \, \mathbf{N} \cdot \nabla \left(\mathbf{c}_{\delta} \eta^2 \right) \, \mathrm{d}x$$
$$= -\int_{B_1^+} \mathbf{N} \cdot \nabla [a_{ij}(x)u_{ij}] \, \mathbf{c}_{\delta} \eta^2 \, \mathrm{d}x + \int_{B_1^+} b_i(x)u_i \, \mathbf{N} \cdot \nabla \left(\mathbf{c}_{\delta} \eta^2 \right) \, \mathrm{d}x,$$
(2.2.20)

where in the last line we have integrated by parts again. Combining (2.2.19) and (2.2.20)

$$\int_{B_1^+} \mathbf{c}_{\delta} J_u \left[\mathbf{N} \cdot \nabla u \right] \eta^2 \, \mathrm{d}x$$

= $-\int_{B_1^+} \mathbf{N} \cdot \nabla a_{ij}(x) u_{ij} \, \mathbf{c}_{\delta} \eta^2 \, \mathrm{d}x + \int_{B_1^+} \mathbf{N}_k b_i(x) u_{ik} \, \mathbf{c}_{\delta} \eta^2 + \int_{B_1^+} b_i(x) u_i \, \mathbf{N} \cdot \nabla \left(\mathbf{c}_{\delta} \eta^2 \right) \, \mathrm{d}x.$
(2.2.21)

The claim follows from (2.2.21) by applying (2.2.16), (2.2.17), and the coefficient bounds.

Third term. We prove that

$$\left| \int_{B_1^+} (\mathbf{N} \cdot \nabla u) J_u \left[\phi_{\delta}(\nabla u) \right] \eta^2 \, \mathrm{d}x \right|$$

$$\leq C \int_{B_1^+} (|\nabla u| + \delta) \left(|D^2 u| \{ |\nabla(\eta^2)| + \varepsilon \eta^2 \} + \varepsilon |\nabla u| |\nabla(\eta^2)| + |\nabla u| |D^2(\eta^2)| \right) \, \mathrm{d}x \quad (2.2.22)$$

$$+ C \int_{\partial^0 B_1^+} |\nabla u|^2 \left(|\nabla(\eta^2)| + \varepsilon \eta^2 \right) \, \mathrm{d}\mathcal{H}^{n-1}.$$

By definition, we have

$$(\mathbf{N} \cdot \nabla u) J_u \left[\phi_\delta(\nabla u) \right] = (\mathbf{N} \cdot \nabla u) L[\phi_\delta(\nabla u)] + \phi_\delta(\nabla u) f'(u) (\mathbf{N} \cdot \nabla u).$$
(2.2.23)

The idea is to integrate the first term in (2.2.23), $\int_{B_1^+} (\mathbf{N} \cdot \nabla u) L[\phi_{\delta}(\nabla u)] \eta^2 dx$, by parts to get the linearized equation acting on the directional derivative $\mathbf{N} \cdot \nabla u$ instead of on the modulus $\phi_{\delta}(\nabla u)$. It will then be easy to bound the remaining terms as in Step 2 above.

We write the operator in divergence form $Lu = \operatorname{div}(A(x)\nabla u) + \widehat{b}(x) \cdot \nabla u$ as in (2.1.6). Integrating by parts twice in $\int_{B_1^+} (\mathbf{N} \cdot \nabla u) \operatorname{div}(A(x)\nabla [\phi_{\delta}(\nabla u)]) \eta^2 dx$, we have

$$\int_{B_{1}^{+}} (\mathbf{N} \cdot \nabla u) \operatorname{div} \left(A(x) \nabla \left[\phi_{\delta}(\nabla u) \right] \right) \eta^{2} dx
= \int_{B_{1}^{+}} \phi_{\delta}(\nabla u) \operatorname{div} \left(A(x) \nabla (\mathbf{N} \cdot \nabla u) \right) \eta^{2} dx
+ \int_{B_{1}^{+}} \phi_{\delta}(\nabla u) \left(2A(x) \nabla (\mathbf{N} \cdot \nabla u) \cdot \nabla (\eta^{2}) + (\mathbf{N} \cdot \nabla u) \operatorname{div} \left\{ A(x) \nabla (\eta^{2}) \right\} \right) dx
+ \int_{\partial^{0} B_{1}^{+}} \left(\phi_{\delta}(\nabla u) A(x) \nabla \left\{ (\mathbf{N} \cdot \nabla u) \eta^{2} \right\} \cdot e_{n} - (\mathbf{N} \cdot \nabla u) A(x) \nabla \left[\phi_{\delta}(\nabla u) \right] \cdot e_{n} \eta^{2} \right) d\mathcal{H}^{n-1}.$$
(2.2.24)

Since u is nonnegative and u = 0 on $\partial^0 B_1^+$, we have $\nabla u = |\nabla u|_{A(0)} |e_n|_{A(0)}^{-1} e_n$ and hence, using (2.2.4) and (2.2.6), the boundary integrand in (2.2.24) can be written simply as

$$\phi_{\delta}(\nabla u)A(x)\nabla\left\{(\mathbf{N}\cdot\nabla u)\eta^{2}\right\}\cdot e_{n} - (\mathbf{N}\cdot\nabla u)A(x)\nabla\left[\phi_{\delta}(\nabla u)\right]\cdot e_{n}\eta^{2} = |\nabla u|_{A(0)}^{2}A(x)\nabla(\eta^{2})\cdot e_{n}.$$
(2.2.25)

Combining (2.2.24) and (2.2.25), we deduce

$$\begin{split} &\int_{B_{1}^{+}} (\mathbf{N} \cdot \nabla u) L[\phi_{\delta}(\nabla u)] \eta^{2} \, \mathrm{d}x \\ &= \int_{B_{1}^{+}} (\mathbf{N} \cdot \nabla u) \mathrm{div} \left(A(x) \nabla \left[\phi_{\delta}(\nabla u) \right] \right) \eta^{2} \, \mathrm{d}x + \int_{B_{1}^{+}} (\mathbf{N} \cdot \nabla u) \left(\widehat{b}(x) \cdot \nabla \left[\phi_{\delta}(\nabla u) \right] \right) \eta^{2} \, \mathrm{d}x \\ &= \int_{B_{1}^{+}} \phi_{\delta}(\nabla u) L[\mathbf{N} \cdot \nabla u] \eta^{2} \, \mathrm{d}x \\ &+ \int_{B_{1}^{+}} \phi_{\delta}(\nabla u) \left(2A(x) \nabla (\mathbf{N} \cdot \nabla u) \cdot \nabla (\eta^{2}) + (\mathbf{N} \cdot \nabla u) \operatorname{div} \left\{ A(x) \nabla (\eta^{2}) \right\} \right) \mathrm{d}x \\ &+ \int_{B_{1}^{+}} \left(-\phi_{\delta}(\nabla u) \, \widehat{b}(x) \cdot \nabla (\mathbf{N} \cdot \nabla u) + (\mathbf{N} \cdot \nabla u) \, \widehat{b}(x) \cdot \nabla \left[\phi_{\delta}(\nabla u) \right] \right) \eta^{2} \, \mathrm{d}x \\ &+ \int_{\partial^{0} B_{1}^{+}} |\nabla u|_{A(0)}^{2} A(x) \nabla (\eta^{2}) \cdot e_{n} \, \mathrm{d}\mathcal{H}^{n-1}. \end{split}$$

$$(2.2.26)$$

We now treat the second term in (2.2.23). Integrating by parts twice as in the proof of Step 2 (this time including boundary terms) and using the equation, it follows that

$$\int_{B_1^+} f'(u) \left(\mathbf{N} \cdot \nabla u \right) \phi_{\delta}(\nabla u) \eta^2 \, \mathrm{d}x = \int_{B_1^+} \mathbf{N} \cdot \nabla[f(u)] \phi_{\delta}(\nabla u) \eta^2 \, \mathrm{d}x$$

$$= \int_{B_1^+} (Lu) \, \mathbf{N} \cdot \nabla \left(\phi_{\delta}(\nabla u) \eta^2 \right) \, \mathrm{d}x - \int_{\partial^0 B_1^+} f(u) |e_n|_{A(0)} \phi_{\delta}(\nabla u) \eta^2 \, \mathrm{d}\mathcal{H}^{n-1}$$

$$= -\int_{B_1^+} \mathbf{N} \cdot \nabla[a_{ij}(x)u_{ij}] \phi_{\delta}(\nabla u) \eta^2 \, \mathrm{d}x + \int_{B_1^+} b_i(x)u_i \, \mathbf{N} \cdot \nabla \left(\phi_{\delta}(\nabla u) \eta^2 \right) \, \mathrm{d}x$$

$$+ \int_{\partial^0 B_1^+} b_i(x)u_i |e_n|_{A(0)} \phi_{\delta}(\nabla u) \eta^2 \, \mathrm{d}\mathcal{H}^{n-1}.$$
(2.2.27)

Finally, summing (2.2.26) and (2.2.27), we obtain

$$\begin{split} \int_{B_{1}^{+}} (\mathbf{N} \cdot \nabla u) J_{u}[\phi_{\delta}(\nabla u)] \eta^{2} dx \\ &= \int_{B_{1}^{+}} \left(L[\mathbf{N} \cdot \nabla u] - \mathbf{N} \cdot \nabla [a_{ij}(x)u_{ij}] \right) \phi_{\delta}(\nabla u) \eta^{2} dx \\ &+ \int_{B_{1}^{+}} \phi_{\delta}(\nabla u) \left(2A(x) \nabla (\mathbf{N} \cdot \nabla u) \cdot \nabla (\eta^{2}) + (\mathbf{N} \cdot \nabla u) \operatorname{div} \left\{ A(x) \nabla (\eta^{2}) \right\} \right) dx \\ &+ \int_{B_{1}^{+}} \widehat{b}(x) \cdot \left((\mathbf{N} \cdot \nabla u) \nabla [\phi_{\delta}(\nabla u)] - \phi_{\delta}(\nabla u) \nabla (\mathbf{N} \cdot \nabla u) \right) \eta^{2} dx \\ &+ \int_{B_{1}^{+}} b_{i}(x) u_{i} \mathbf{N} \cdot \nabla (\phi_{\delta}(\nabla u) \eta^{2}) dx \\ &+ \int_{\partial^{0} B_{1}^{+}} \left(|\nabla u|^{2}_{A(0)} A(x) \nabla (\eta^{2}) \cdot e_{n} + b_{i}(x) u_{i} |e_{n}|_{A(0)} \phi_{\delta}(\nabla u) \eta^{2} \right) d\mathcal{H}^{n-1}. \end{split}$$

Noticing that $L[\mathbf{N} \cdot \nabla u] - \mathbf{N} \cdot \nabla [a_{ij}(x)u_{ij}] = -\mathbf{N} \cdot \nabla a_{ij}(x)u_{ij} + \mathbf{N}_k b_i(x)u_{ik}$, every term in the right-hand side of (2.2.28) can be bounded as claimed in (2.2.22). For this, apply

the uniform ellipticity, the coefficient bounds $\|\nabla a_{ij}\|_{L^{\infty}} + \|b_i\|_{C^0} + \|\widehat{b}_i\|_{L^{\infty}} \leq 2\varepsilon$, and the estimates (2.2.16) and (2.2.17).

Conclusion. Applying the three estimates (2.2.8), (2.2.18), and (2.2.22) in (2.2.7) yields the lower bound

$$\begin{split} \int_{B_1^+} \mathbf{c}_{\delta} J_u \mathbf{c}_{\delta} \eta^2 \, \mathrm{d}x \\ &\geq \int_{B_1^+ \cap \{|\nabla u|_{A(0)} > \delta\}} \mathcal{A}^2 \eta^2 \, \mathrm{d}x - C\delta \int_{B_1^+} |f'(u)| (|\nabla u| + \delta) \eta^2 \, \mathrm{d}x \\ &\quad - C \int_{B_1^+} (|\nabla u| + \delta) \left(|D^2 u| \{ |\nabla(\eta^2)| + \varepsilon \eta^2 \} + \varepsilon |\nabla u| |\nabla(\eta^2)| + |\nabla u| |D^2(\eta^2)| \right) \mathrm{d}x \\ &\quad - C \int_{\partial^0 B_1^+} |\nabla u|^2 (|\nabla(\eta^2)| + \varepsilon \eta^2) \, \mathrm{d}\mathcal{H}^{n-1}. \end{split}$$

$$(2.2.29)$$

By the integral stability inequality (2.1.16) with $\mathbf{c} = \mathbf{c}_{\delta}$, we also have the upper bound

$$\int_{B_1^+} \mathbf{c}_{\delta} J_u \mathbf{c}_{\delta} \eta^2 \, \mathrm{d}x \le \int_{B_1^+} \phi_{\delta} (\nabla u)^2 |\nabla \eta - \frac{1}{2} \eta A^{-1}(x) \widehat{b}(x)|_{A(x)}^2 \, \mathrm{d}x$$

$$\le C \int_{B_1^+} (|\nabla u| + \delta)^2 \left(|\nabla \eta|^2 + \varepsilon^2 \eta^2 \right) \, \mathrm{d}x.$$
(2.2.30)

Hence, combining (2.2.29) and (2.2.30) and taking the limit as $\delta \to 0$, we deduce the claim

In order to prove the remaining estimates in Theorem 2.1.4, we need to control the righthand side of (2.1.10). For this, next we prove two basic Hessian estimates for (generalized) superharmonic functions. We essentially follow the proof of Theorem 1.1.2 above, but now including boundary terms.

Lemma 2.2.1. Let $u \in C^2(\overline{B_1^+})$ be superharmonic in the sense that $Lu \leq 0$ in B_1^+ , where L satisfies conditions (2.1.3) and (2.1.4) in $\Omega = B_1^+$. Assume that

$$\|DA\|_{L^{\infty}(B_{1}^{+})} + \|b\|_{L^{\infty}(B_{1}^{+})} \le \varepsilon$$

for some $\varepsilon > 0$.

Then, there exists a universal $\varepsilon_0 > 0$ with the following property: if $\varepsilon \leq \varepsilon_0$, then, for all $\zeta \in C_c^{0,1}(B_1)$ with $\zeta \geq 0$, we have

$$\int_{B_1^+} |D^2 u| \zeta \,\mathrm{d}x \le C \int_{B_1^+} |\nabla u| \,\left(|\nabla \zeta| + \varepsilon \zeta \right) \mathrm{d}x + C \int_{B_1^+} \mathcal{A} \,\zeta \,\mathrm{d}x + C \int_{\partial^0 B_1^+} |\nabla u| \,\zeta \,\mathrm{d}x \quad (2.2.31)$$

and

$$\int_{B_1^+} |D^2 u| |\nabla u| \zeta \, \mathrm{d}x \le C \int_{B_1^+} |\nabla u|^2 \left(|\nabla \zeta| + \varepsilon \zeta \right) \mathrm{d}x + C \int_{B_1^+} \mathcal{A} |\nabla u| \zeta \, \mathrm{d}x + C \int_{\partial^0 B_1^+} |\nabla u|^2 \zeta \, \mathrm{d}\mathcal{H}^{n-1},$$

$$(2.2.32)$$

where C is a universal constant.

Proof. Consider the auxiliary function

$$\mathcal{A}_{0} := \begin{cases} \left(\|A^{1/2}(0)D^{2}uA^{1/2}(0)\|_{\mathrm{HS}}^{2} - |A^{1/2}(0)D^{2}uA^{1/2}(0)\mathbf{n}(x)|^{2} \right)^{1/2} & \text{if } \nabla u \neq 0\\ 0 & \text{if } \nabla u = 0, \end{cases}$$

where the vector field $\mathbf{n}(x)$ has been introduced in (2.2.2) in the definition of \mathcal{A} in (2.2.1). Using that $\|DA\|_{L^{\infty}(B_1^+)} \leq \varepsilon$, it is easy to show (see [52] or Step 2 in the proof of Theorem 1.1.2 above) that

$$|\mathcal{A}^2 - \mathcal{A}_0^2| \le C\varepsilon |x|\mathcal{A}_0^2 \quad \text{in } B_1^+, \qquad (2.2.33)$$

where C always denotes a universal constant. In particular, the functions \mathcal{A} and \mathcal{A}_0 are comparable for ε small. Using that $Lu \leq 0$, following [52], it is not hard to show that

$$|D^2 u| \le -C \operatorname{tr} \left(A(0) D^2 u \right) + C \mathcal{A}_0 + C \varepsilon |x| |D^2 u| + C \varepsilon |\nabla u| \quad \text{a.e. in } B_1^+.$$
 (2.2.34)

First we prove the Hessian bound (2.2.31). Multiplying (2.2.34) by ζ and integrating in B_1^+

$$\int_{B_1^+} |D^2 u| \zeta \, \mathrm{d}x \leq -C \int_{B_1^+} \operatorname{tr} \left(A(0) D^2 u \right) \zeta \, \mathrm{d}x + C \int_{B_1^+} \mathcal{A}_0 \zeta \, \mathrm{d}x + C \varepsilon \int_{B_1^+} |x| |D^2 u| \zeta \, \mathrm{d}x + C \varepsilon \int_{B_1^+} |\nabla u| \zeta \, \mathrm{d}x.$$

$$(2.2.35)$$

Integrating by parts, we have

$$-\int_{B_1^+} \operatorname{tr} \left(A(0) D^2 u \right) \zeta \, \mathrm{d}x = \int_{B_1^+} A(0) \nabla u \cdot \nabla \zeta \, \mathrm{d}x - \int_{\partial^0 B_1^+} A(0) \nabla u \cdot e_n \zeta \, \mathrm{d}\mathcal{H}^{n-1},$$

and substituting in (2.2.35), by uniform ellipticity,

$$\int_{B_1^+} |D^2 u| \zeta \, \mathrm{d}x \leq C \int_{B_1^+} |\nabla u| |\nabla \zeta| \, \mathrm{d}x + C \int_{B_1^+} \mathcal{A}_0 \zeta \, \mathrm{d}x + C \int_{\partial^0 B_1^+} |\nabla u| \zeta \, \mathrm{d}x \\
+ C \varepsilon \int_{B_1^+} |x| |D^2 u| \zeta \, \mathrm{d}x + C \varepsilon \int_{B_1^+} |\nabla u| \zeta \, \mathrm{d}x.$$
(2.2.36)

Choosing $\varepsilon_0 > 0$ universal sufficiently small, we can absorb the Hessian term in the righthand side of (2.2.36), and by (2.2.33) (taking ε_0 smaller) we deduce the first claim.

For the second estimate (2.2.32), multiplying (2.2.34) by $|\nabla u|_{A(0)}\zeta$ and integrating in B_1^+

$$\begin{split} \int_{B_{1}^{+}} |D^{2}u| |\nabla u|_{A(0)} \zeta \, \mathrm{d}x &\leq -C \int_{B_{1}^{+}} |\nabla u|_{A(0)} \mathrm{tr} \left(A(0) D^{2}u \right) \zeta \, \mathrm{d}x + C \int_{B_{1}^{+}} \mathcal{A}_{0} |\nabla u|_{A(0)} \zeta \, \mathrm{d}x \\ &+ C \varepsilon \int_{B_{1}^{+}} |x| |D^{2}u| |\nabla u|_{A(0)} \zeta \, \mathrm{d}x + C \varepsilon \int_{B_{1}^{+}} |\nabla u|^{2} \zeta \, \mathrm{d}x. \end{split}$$

$$(2.2.37)$$

The first integrand in the right-hand side of (2.2.37) can be bounded by

$$-|\nabla u|_{A(0)} \operatorname{tr} \left(A(0) D^2 u \right) \le -\frac{1}{2} \operatorname{div} \left(|\nabla u|_{A(0)} A(0) \nabla u \right) + C \mathcal{A}_0 |\nabla u|_{A(0)} \quad \text{a.e. in } B_1^+.$$
(2.2.38)

Substituting (2.2.38) in (2.2.37) leads to

$$\begin{split} \int_{B_1^+} |D^2 u| |\nabla u|_{A(0)} \zeta \, \mathrm{d}x &\leq -C \int_{B_1^+} \operatorname{div} \left(|\nabla u|_{A(0)} A(0) \nabla u \right) \zeta \, \mathrm{d}x + C \int_{B_1^+} \mathcal{A}_0 |\nabla u|_{A(0)} \zeta \, \mathrm{d}x \\ &+ C \varepsilon \int_{B_1^+} |x| |D^2 u| |\nabla u|_{A(0)} \zeta \, \mathrm{d}x + C \varepsilon \int_{B_1^+} |\nabla u|^2 \zeta \, \mathrm{d}x, \end{split}$$

and integrating by parts the divergence term, we obtain the inequality

$$\int_{B_{1}^{+}} |D^{2}u| |\nabla u|_{A(0)} \zeta \, \mathrm{d}x \leq C \int_{B_{1}^{+}} |\nabla u|^{2} \left(|\nabla \zeta| + \varepsilon \zeta \right) \mathrm{d}x + C \int_{B_{1}^{+}} \mathcal{A}_{0} |\nabla u|_{A(0)} \zeta \, \mathrm{d}x \\
+ C \varepsilon \int_{B_{1}^{+}} |x| |D^{2}u| |\nabla u|_{A(0)} \zeta \, \mathrm{d}x + C \int_{\partial^{0} B_{1}^{+}} |\nabla u|^{2} \zeta \, \mathrm{d}\mathcal{H}^{n-1}.$$
(2.2.39)

Once again, choosing $\varepsilon_0 > 0$ universal small, we can absorb the "Hessian times the gradient" error in (2.2.39) into the left-hand side, and by (2.2.33) we deduce the second claim.

Thanks to Lemma 2.2.1, we can get rid of the Hessian terms appearing in the right-hand side of the first inequality (2.1.10) in Theorem 2.1.4:

Lemma 2.2.2. Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a nonnegative stable solution of -Lu = f(u) in B_1^+ , with u = 0 on $\partial^0 B_1^+$. Assume that $f \in C^1(\mathbb{R})$ is nonnegative. Assume that L satisfies conditions (2.1.3), (2.1.4), and (2.1.5) in $\Omega = B_1^+$, and that

$$\|DA\|_{L^{\infty}(B_{1}^{+})} + \|b\|_{L^{\infty}(B_{1}^{+})} \le \varepsilon$$

for some $\varepsilon > 0$. If $\varepsilon \leq \varepsilon_0$, then

$$\int_{B_{8/9}^+} \mathcal{A}^2 \, \mathrm{d}x \le C \int_{B_1^+} |\nabla u|^2 \, \mathrm{d}x + C \int_{\partial^0 B_1^+} |\nabla u|^2 \, \mathrm{d}\mathcal{H}^{n-1}$$

,

where $\varepsilon_0 > 0$ and C are universal constants.

Proof. Let $\varepsilon_0 > 0$ be the universal constant in the conclusion of Lemma 2.2.1. Applying (2.2.32) in Lemma 2.2.1 with $\zeta = |\nabla(\eta^2)| + \varepsilon \eta^2 \in C_c^{0,1}(B_1)$ yields

$$\begin{split} \int_{B_1^+} |D^2 u| |\nabla u| (|\nabla(\eta^2)| + \varepsilon \eta^2) \, \mathrm{d}x \\ &\leq C \int_{B_1^+} |\nabla u|^2 \left(|D^2(\eta^2)| + \varepsilon |\nabla(\eta^2)| + \varepsilon^2 \eta^2 \right) \, \mathrm{d}x + C \int_{B_1^+} \mathcal{A} |\nabla u| \left(|\nabla(\eta^2)| + \varepsilon \eta^2 \right) \, \mathrm{d}x \\ &\quad + C \int_{\partial^0 B_1^+} |\nabla u|^2 (|\nabla(\eta^2)| + \varepsilon \eta^2) \, \mathrm{d}\mathcal{H}^{n-1}. \end{split}$$

$$(2.2.40)$$

Since $|\nabla(\eta^2)| + \varepsilon \eta^2 = |\eta| (2|\nabla \eta| + \varepsilon |\eta|)$, by Cauchy-Schwarz, the second term in (2.2.40) can be bounded by

$$\int_{B_1^+} \mathcal{A}|\nabla u| \left(|\nabla(\eta^2)| + \varepsilon \eta^2\right) \mathrm{d}x \le C \left(\int_{B_1^+} \mathcal{A}^2 \eta^2 \,\mathrm{d}x\right)^{1/2} \left(\int_{B_1^+} |\nabla u|^2 \left(|\nabla \eta|^2 + \varepsilon^2 \eta^2\right) \,\mathrm{d}x\right)^{1/2}.$$
(2.2.41)

Hence, applying (2.2.40) and (2.2.41) to the Hessian errors in the right-hand side of (2.1.10) in Theorem 2.1.4, we obtain

$$\begin{split} \int_{B_{1}^{+}} \mathcal{A}^{2} \eta^{2} \, \mathrm{d}x &\leq C \left(\int_{B_{1}^{+}} \mathcal{A}^{2} \eta^{2} \, \mathrm{d}x \right)^{1/2} \left(\int_{B_{1}^{+}} |\nabla u|^{2} \left(|\nabla \eta|^{2} + \varepsilon^{2} \eta^{2} \right) \, \mathrm{d}x \right)^{1/2} \\ &+ C \int_{B_{1}^{+}} |\nabla u|^{2} \left(|\nabla \eta|^{2} + |D^{2}(\eta^{2})| + \varepsilon |\nabla(\eta^{2})| + \varepsilon^{2} \eta^{2} \right) \, \mathrm{d}x \qquad (2.2.42) \\ &+ C \int_{\partial^{0} B_{1}^{+}} |\nabla u|^{2} (|\nabla(\eta^{2})| + \varepsilon \eta^{2}) \, \mathrm{d}\mathcal{H}^{n-1}. \end{split}$$

Therefore, by Young's inequality, we can absorb the $\int_{B_1^+} \mathcal{A}^2 \eta^2 dx$ term in (2.2.42) into the left-hand side. Choosing $\eta \in C_c^{\infty}(B_1)$ with $0 \leq \eta \leq 1$ in B_1 and $\eta = 1$ in $B_{8/9}$, by the universal bound $\varepsilon \leq \varepsilon_0$, we deduce the claim.

Thanks to the preliminary lemmas above, we are now in position to conclude the proof of Theorem 2.1.4:

Proof of the boundary estimates (2.1.12), (2.1.13), (2.1.14), and (2.1.15) in Theorem 2.1.4. Once we obtain the boundary gradient estimate 2.1.12, the remaining inequalities (2.1.13), (2.1.14), and (2.1.15) will follow easily from Lemmas 2.2.1 and 2.2.2.

To control the gradient on the boundary, we proceed in two steps. First we employ the Pohozaev trick to bound the L^2 norm of ∇u on the lower boundary by the Dirichlet energy up to Hessian errors. Secondly, we use Lemmas 2.2.1 and 2.2.2 to control these Hessian errors and apply Simon's lemma (recalled in Appendix D).

Step 1. We prove that

$$\|\nabla u\|_{L^2(\partial^0 B^+_{2/3})}^2 \le C(1+\varepsilon) \|\nabla u\|_{L^2(B^+_{7/9})}^2 + C\varepsilon \||D^2 u| |\nabla u|\|_{L^1(B^+_{7/9})},$$

where C is a universal constant.

Let $\eta \in C_c^{\infty}(B_{7/9})$. Integrating by parts, by the properties of u and the vector field **N** defined in (2.2.3), it is easy to check that

$$|e_n|_{A(0)} \int_{\partial^0 B_1^+} |\nabla u|_{A(0)}^2 \, \mathrm{d}\mathcal{H}^{n-1}$$

$$= \int_{B_1^+} \mathrm{div} \left(|\nabla u|_{A(0)}^2 \mathbf{N} - 2(\mathbf{N} \cdot \nabla u) A(0) \nabla u \right) \eta^2 \, \mathrm{d}x \qquad (2.2.43)$$

$$+ \int_{B_1^+} \left(|\nabla u|_{A(0)}^2 \mathbf{N} - 2(\mathbf{N} \cdot \nabla u) A(0) \nabla u \right) \cdot \nabla(\eta^2) \, \mathrm{d}x.$$

The divergence term in (2.2.43) can be written as

$$div \left(|\nabla u|^2_{A(0)} \mathbf{N} - 2(\mathbf{N} \cdot \nabla u) A(0) \nabla u \right) = -2(\mathbf{N} \cdot \nabla u) tr(A(0) D^2 u)$$

= $-2(\mathbf{N} \cdot \nabla u) Lu + 2(\mathbf{N} \cdot \nabla u)(b(x) \cdot \nabla u) + 2(\mathbf{N} \cdot \nabla u) tr(\{A(x) - A(0)\} D^2 u)$
 $\leq -2(\mathbf{N} \cdot \nabla u) Lu + C\varepsilon |\nabla u|^2 + C\varepsilon |x| |D^2 u| |\nabla u|,$

where in the last line we have used the bounds $||b||_{L^{\infty}(B_1^+)} \leq \varepsilon$ and $|A(x) - A(0)| \leq \varepsilon |x|$ for $x \in B_1^+$. It follows that

$$\begin{aligned} |e_{n}|_{A(0)} \int_{\partial^{0}B_{1}^{+}} |\nabla u|_{A(0)}^{2} \, \mathrm{d}\mathcal{H}^{n-1} \\ &\leq -2 \int_{B_{1}^{+}} (\mathbf{N} \cdot \nabla u) L u \, \eta^{2} \, \mathrm{d}x + C \int_{B_{1}^{+}} |\nabla u|^{2} \big(|\nabla(\eta^{2})| + \varepsilon \eta^{2} \big) \, \mathrm{d}x \\ &+ C \varepsilon \int_{B_{1}^{+}} |x| |D^{2}u| |\nabla u| \eta^{2} \, \mathrm{d}x \end{aligned}$$
(2.2.44)

and, thus, it remains to control the term $-2 \int_{B_1^+} (\mathbf{N} \cdot \nabla u) Lu \eta^2 dx$ in (2.2.44).

Since -Lu = f(u) in B_1^+ , the primitive $F(t) := \int_0^t f(s) \, ds$ of f satisfies

$$\mathbf{N} \cdot \nabla[F(u)] = (\mathbf{N} \cdot \nabla u)f(u) = -(\mathbf{N} \cdot \nabla u)Lu,$$

and the first term on the right hand side of (2.2.44) can be integrated by parts as

$$-\int_{B_1^+} (\mathbf{N} \cdot \nabla u) Lu \,\eta^2 \,\mathrm{d}x = \int_{B_1^+} \mathbf{N} \cdot \nabla [F(u)] \eta^2 \,\mathrm{d}x = -\int_{B_1^+} F(u) \left(\mathbf{N} \cdot \nabla (\eta^2)\right) \mathrm{d}x. \quad (2.2.45)$$

By the monotonicity of f, since u and f are nonnegative, we have $|F(u)| \leq uf(u) = -uLu$. Hence, writing L in divergence form $Lu = \operatorname{div}(A(x)\nabla u) + \widehat{b}(x) \cdot \nabla u$ as in (2.1.6), by the coefficient bound $\|\widehat{b}\|_{L^{\infty}(B_1^+)} \leq C\varepsilon$ we deduce

$$|F(u)| \le -u \operatorname{div}(A(x)\nabla u) + C\varepsilon u |\nabla u|.$$
(2.2.46)

Using (2.2.46), we estimate the right-hand side of (2.2.45) by

$$\left| -\int_{B_1^+} F(u) \left(\mathbf{N} \cdot \nabla(\eta^2) \right) \mathrm{d}x \right| \leq -C \int_{B_1^+} u \operatorname{div} \left(A(x) \nabla u \right) |\nabla(\eta^2)| \, \mathrm{d}x + C\varepsilon \int_{B_1^+} u |\nabla u| |\nabla(\eta^2)| \, \mathrm{d}x,$$

$$(2.2.47)$$

and since $|\nabla(\eta^2)|$ is Lipschitz, the divergence term in (2.2.47) can be integrated by parts as

$$-\int_{B_1^+} u \operatorname{div} \left(A(x) \nabla u \right) |\nabla(\eta^2)| \, \mathrm{d}x = \int_{B_1^+} |\nabla u|_{A(x)}^2 |\nabla(\eta^2)| \, \mathrm{d}x + \int_{B_1^+} u \, A(x) \nabla u \cdot \nabla |\nabla(\eta^2)| \, \mathrm{d}x.$$
(2.2.48)

Therefore, combining (2.2.45), (2.2.47), and (2.2.48), we deduce

$$-\int_{B_{1}^{+}} (\mathbf{N} \cdot \nabla u) L u \,\eta^{2} \,\mathrm{d}x \leq \int_{B_{1}^{+}} |\nabla u|_{A(x)}^{2} |\nabla (\eta^{2})| \,\mathrm{d}x + C \int_{B_{1}^{+}} u |\nabla u| \left(|D^{2}(\eta^{2})| + \varepsilon |\nabla (\eta^{2})| \right) \,\mathrm{d}x.$$
(2.2.49)

Moreover, we can bound the last term in (2.2.49) by Cauchy-Schwarz and the Poincaré inequality (valid since u = 0 on $\partial^0 B_1^+$) as

$$\int_{B_{1}^{+}} u |\nabla u| \left(|D^{2}(\eta^{2})| + \varepsilon |\nabla(\eta^{2})| \right) dx
\leq C \left(\int_{B_{7/9}^{+}} |\nabla u|^{2} dx \right)^{1/2} \left(\int_{B_{1}^{+}} |\nabla u|^{2} \left(|D^{2}(\eta^{2})| + \varepsilon |\nabla(\eta^{2})| \right)^{2} dx \right)^{1/2}.$$
(2.2.50)

Applying the bounds (2.2.49) and (2.2.50) in (2.2.44), by uniform ellipticity, we obtain

$$\int_{\partial^{0}B_{1}^{+}} |\nabla u|^{2} \eta^{2} \, \mathrm{d}\mathcal{H}^{n-1} \\
\leq C \left(\int_{B_{7/9}^{+}} |\nabla u|^{2} \, \mathrm{d}x \right)^{1/2} \left(\int_{B_{1}^{+}} |\nabla u|^{2} \left(|D^{2}(\eta^{2})| + \varepsilon |\nabla(\eta^{2})| \right)^{2} \, \mathrm{d}x \right)^{1/2} \\
+ C \int_{B_{1}^{+}} |\nabla u|^{2} \left(|\nabla(\eta^{2})| + \varepsilon \eta^{2} \right) \, \mathrm{d}x + C\varepsilon \int_{B_{1}^{+}} |x|| D^{2} u ||\nabla u| \eta^{2} \, \mathrm{d}x.$$
(2.2.51)

Finally, choosing $\eta \in C_c^{\infty}(B_{7/9})$ in (2.2.51) satisfying $\eta = 1$ in $B_{2/3}$ and $0 \le \eta \le 1$ in $B_{7/9}$, we deduce

$$\int_{\partial^0 B_{2/3}^+} |\nabla u|^2 \, \mathrm{d}\mathcal{H}^{n-1} \le C \, (1+\varepsilon) \int_{B_{7/9}^+} |\nabla u|^2 \, \mathrm{d}x + C\varepsilon \int_{B_{7/9}^+} |x| |D^2 u| |\nabla u| \, \mathrm{d}x,$$

which yields the claim.

Step 2. Conclusion.

Let $\varepsilon_0 > 0$ be the universal constant in the conclusion of Lemma 2.2.1. Applying this result with a cut-off $\zeta \in C_c^1(B_{8/9})$ such that $0 \leq \zeta \leq 1$ and $\zeta = 1$ in $B_{7/9}$, if $\varepsilon \leq \varepsilon_0$, then

$$\||D^{2}u| |\nabla u|\|_{L^{1}(B^{+}_{7/9})} \leq C \|\nabla u\|_{L^{2}(B^{+}_{8/9})}^{2} + C \|\nabla u\|_{L^{2}(\partial^{0}B^{+}_{8/9})}^{2} + C \|\mathcal{A}|\nabla u|\|_{L^{1}(B^{+}_{8/9})}.$$
 (2.2.52)

Hence, applying Cauchy-Schwarz in (2.2.52) and by Lemma 2.2.2, we deduce

$$||D^{2}u| |\nabla u||_{L^{1}(B^{+}_{7/9})} \leq C ||\nabla u||_{L^{2}(B^{+}_{1})}^{2} + C ||\nabla u||_{L^{2}(\partial^{0}B^{+}_{1})}^{2}.$$
(2.2.53)

Let $\delta > 0$. Using (2.2.53) in Step 1 above, letting $\varepsilon_{\delta} := \min\{\varepsilon_0, \delta/C\}$, we obtain

$$\|\nabla u\|_{L^{2}(\partial^{0}B^{+}_{2/3})}^{2} \leq \delta \|\nabla u\|_{L^{2}(\partial^{0}B^{+}_{1})}^{2} + C\|\nabla u\|_{L^{2}(B^{+}_{1})}^{2} \quad \text{for } \varepsilon \leq \varepsilon_{\delta}.$$
(2.2.54)

Hence, by translation and rescaling of (2.2.54), for all $y \in \partial^0 B_1^+$ and $\rho > 0$ such that $B_{\rho}^+(y) \subset B_1^+$, we have

$$\rho \int_{\partial^0 B^+_{2\rho/3}(y)} |\nabla u|^2 \, \mathrm{d}\mathcal{H}^{n-1} \leq \delta\rho \int_{\partial^0 B^+_{\rho}(y)} |\nabla u|^2 \, \mathrm{d}\mathcal{H}^{n-1} + C \int_{B^+_{\rho}(y)} |\nabla u|^2 \, \mathrm{d}x \\
\leq \delta\rho \int_{\partial^0 B^+_{\rho}(y)} |\nabla u|^2 \, \mathrm{d}\mathcal{H}^{n-1} + C \int_{B^+_{1}} |\nabla u|^2 \, \mathrm{d}x \quad \text{for } \varepsilon \leq \varepsilon_{\delta}.$$
(2.2.55)

Since $y \in \partial^0 B_1^+$, we have y = (y', 0) for some $y' \in \mathbb{R}^{n-1}$, and the lower boundary $\partial^0 B_{\rho}^+(y)$ is simply the (n-1)-dimensional ball $B'_{\rho}(y') := \{x \in \mathbb{R}^{n-1} : |x - y'| < \rho\} \subset \mathbb{R}^{n-1} = \partial^0 \mathbb{R}^n$. By (2.2.55), we can apply the Simon lemma to the subadditive quantity

$$B' \mapsto \int_{B'} |\nabla u|^2 \,\mathrm{d}\mathcal{H}^{n-1}$$

on balls $B' \subset B'_1 \subset \mathbb{R}^{n-1} = \partial^0 \mathbb{R}^n$ to deduce the bound

$$\int_{\partial^0 B_{2/3}^+} |\nabla u|^2 \, \mathrm{d}\mathcal{H}^{n-1} \le C \int_{B_1^+} |\nabla u|^2 \, \mathrm{d}x \quad \text{ for } \varepsilon \le \varepsilon_\delta, \tag{2.2.56}$$

for some universal $\delta > 0$. In particular, we may take ε_0 universal equal to ε_{δ} and this concludes the proof of (2.1.12).

Finally, to deduce the remaining Hessian estimates we proceed as in the proof of (2.2.53). To prove (2.1.13), we apply (2.2.32) from Lemma 2.2.1 with a cut-off function $\zeta \in C_c^1(B_{16/27})$ such that $0 \leq \zeta \leq 1$ and $\zeta = 1$ in $B_{4/7=16/28} \subset B_{16/27}$, and by Cauchy-Schwarz

$$\begin{aligned} \| |D^{2}u| |\nabla u| \|_{L^{1}(B^{+}_{4/7})} &\leq C \|\nabla u\|_{L^{2}(B^{+}_{16/27})}^{2} + C \|\nabla u\|_{L^{2}(\partial^{0}B^{+}_{16/27})}^{2} + C \|\mathcal{A}\|_{L^{2}(B^{+}_{16/27})}^{2} \\ &\leq C \|\nabla u\|_{L^{2}(B^{+}_{2/3})}^{2} + C \|\nabla u\|_{L^{2}(\partial^{0}B^{+}_{2/3})}^{2}, \end{aligned}$$

$$(2.2.57)$$

where in the last line we have used Lemma 2.2.2 applied to the rescaled function $u(\frac{2}{3}\cdot)$. Applying (2.2.56) to (2.2.57) now leads to (2.1.13).

Now, the bound (2.1.14) is easily obtained combining Lemma 2.2.2 with the boundary estimate (2.1.12). The final estimate (2.1.15) follows from Lemma 2.2.1 and the above. \Box

2.3 Boundary $W^{1,2+\gamma}$ estimate

First we control the Dirichlet energy by the L^1 norm of the solution under a smallness condition on the coefficients. This follows from Theorem 2.1.4 and the interpolation inequalities of Cabré in [19] (recalled in Appendix C below).

Lemma 2.3.1. Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a nonnegative stable solution of -Lu = f(u) in B_1^+ , with u = 0 on $\partial^0 B_1^+$. Assume that $f \in C^1(\mathbb{R})$ is nonnegative and nondecreasing. Assume that L satisfies conditions (2.1.3), (2.1.4), and (2.1.5) in $\Omega = B_1^+$, and

$$\|DA\|_{L^{\infty}(B_{1}^{+})} + \|b\|_{L^{\infty}(B_{1}^{+})} \le \varepsilon$$

for some $\varepsilon > 0$. If $\varepsilon \leq \varepsilon_0$, then

$$\|\nabla u\|_{L^2(B_{1/2}^+)} \le C \|u\|_{L^1(B_1^+)},$$

where $\varepsilon_0 > 0$ and C are universal constants.

Proof. We cover $B_{1/2}^+$ (except for a set of measure zero) with a family of disjoint open cubes $Q_j \subset \mathbb{R}^n_+$ of the same side-length and small enough so that $Q_j \subset B_{4/7}^+$. The side-length and the number of cubes depend only on n. Combining the interpolation inequalities of Proposition C.1 (with p = 2) and Proposition C.3, rescaled from the unit cube to Q_j , with $\tilde{\delta} = \delta^{3/2}$ for a given $\delta \in (0, 1)$, we have

$$\int_{Q_j} |\nabla u|^2 dx \le C\delta \int_{Q_j} |D^2 u| |\nabla u| \, dx + C\delta \int_{Q_j} |\nabla u|^2 dx + C\delta^{-2 - \frac{3n}{2}} \left(\int_{Q_j} |u| \, dx \right)^2.$$

Since $Q_j \subset B_{4/7}^+$, applying (2.1.13) from Theorem 2.1.4, for $\varepsilon \leq \varepsilon_0$ we deduce

$$\int_{Q_j} |\nabla u|^2 dx \le C\delta \int_{B_1^+} |\nabla u|^2 dx + C\delta^{-2-\frac{3n}{2}} \left(\int_{B_1^+} |u| \, dx \right)^2.$$

Adding up these inequalities, we obtain

$$\|\nabla u\|_{L^{2}(B^{+}_{1/2})}^{2} \leq C\delta \|\nabla u\|_{L^{2}(B^{+}_{1})}^{2} + C\delta^{-2-\frac{3n}{2}} \|u\|_{L^{1}(B^{+}_{1})}^{2} \quad \text{for } \delta \in (0,1) \text{ and } \varepsilon \leq \varepsilon_{0}.$$
 (2.3.1)

For $B^+_{\rho}(y) \subset B^+_1$ with $y \in \partial^0 B^+_1$, the function $u^{y,\rho} := u(y + \rho \cdot)$ is a stable solution to a semilinear equation with coefficients $A^{y,\rho} = A(y + \rho \cdot)$ and $b^{y,\rho} = \rho b(y + \rho \cdot)$. In particular, since $\rho \leq 1$, for $\varepsilon \leq \varepsilon_0$ we have

$$\|DA^{y,\rho}\|_{L^{\infty}(B_1^+)} + \|b^{y,\rho}\|_{L^{\infty}(B_1^+)} \le \rho \varepsilon \le \varepsilon_0,$$

and we may apply (2.3.1) to $u^{y,\rho}$, which yields

$$\rho^{n+2} \int_{B^+_{\rho/2}(y)} |\nabla u|^2 \, dx \le C\delta\rho^{n+2} \int_{B^+_{\rho}(y)} |\nabla u|^2 \, dx + C\delta^{-2-\frac{3n}{2}} \left(\int_{B^+_{\rho}(y)} |u| \, dx \right)^2,$$

hence

$$\rho^{n+2} \int_{B_{\rho/2}^+(y)} |\nabla u|^2 \, dx \leq C \delta \rho^{n+2} \int_{B_{\rho}^+(y)} |\nabla u|^2 \, dx + C \delta^{-2-\frac{3n}{2}} \|u\|_{L^1(B_1^+)}^2$$
for all $B_{\rho}^+(y) \subset B_1^+$ with $y \in \partial^0 B_1^+$ and $\delta \in (0,1)$.
$$(2.3.2)$$

To deduce the desired bound, we must combine (2.3.2) with the following interior estimates derived in [52, Proposition 1.3]:

$$\rho^{n+2} \int_{B_{\rho/2}(y)} |\nabla u|^2 \,\mathrm{d}x \le C \|u\|_{L^1(B_1^+)}^2 \quad \text{for all } B_{\rho}(y) \subset B_1^+.$$
(2.3.3)

We now claim that for all balls $B_{\rho}(y) \subset B_1$ (not necessarily contained in B_1^+) and every $\delta \in (0, 1)$, we have

$$\rho^{n+2} \int_{\partial \mathbb{R}^n_+ \cap B_{\rho/2}(y)} |\nabla u|^2 \, dx \leq C \delta \rho^{n+2} \int_{\partial \mathbb{R}^n_+ \cap B_{\rho}(y)} |\nabla u|^2 \, dx + C \delta^{-2 - \frac{3n}{2}} \|u\|^2_{L^1(B^+_1)}.$$
(2.3.4)

This is achieved by a simple covering argument. The key observation is that $\mathbb{R}^n_+ \cap B_{\rho/2}(y)$ can be covered by a dimensional number of balls $\{B_{\rho/16}(y_i)\}_i$ and $\{B_{3\rho/16}(z_j)\}_j$, where y_i are such that $B_{\rho/8}(y_i) \subset \mathbb{R}^n_+ \cap B_{\rho}(y) \subset B_1^+$ are interior balls, while $z_j \in \partial \mathbb{R}^n_+$ satisfy $B^+_{3\rho/8}(z_j) \subset \mathbb{R}^n_+ \cap B_{\rho}(y) \subset B_1^+$. Applying (2.3.3) to the interior balls and (2.3.2) to the boundary balls, it is not hard to deduce (2.3.4). For more details, we refer the reader to the proof of Lemma 8.2 in [20].

By (2.3.4), applying Simon's lemma to the subadditive quantity $B \mapsto \|\nabla u\|_{L^2(\mathbb{R}^n_+ \cap B)}^2$ now yields the claim.

Following ideas from [24], the higher integrability estimate in Theorem 2.1.1 will now be a direct consequence of the Hessian estimates in Theorem 2.1.4 and of Lemma 2.3.1.

Proof of Theorem 2.1.1. There are three steps in our proof. First, by the divergence theorem and Theorem 2.1.4, we control the surface integral of $|\nabla u|^2$ on every level set of u by the Dirichlet energy. Secondly, using coarea formula, Hölder, and Sobolev inequality, we will bound the $L^{2+\gamma}$ norm of the gradient by the L^2 norm. Finally, Lemma 2.3.1 will yield the final estimate in terms of the L^1 norm of the solution. All these bounds are shown under a smallness condition on the coefficients which is removed in the last step.

Step 1: We prove that, if $\varepsilon \leq \varepsilon_0$, then for a.e. $t \in \mathbb{R}$ we have

$$\int_{\{u=t\}\cap B_{1/2}} |\nabla u|^2 \, \mathrm{d}\mathcal{H}^{n-1} \le C \|\nabla u\|_{L^2(B_1)}^2,$$

where $\varepsilon_0 > 0$ and C are universal.

Since $|\operatorname{div}(|\nabla u|\nabla u)| \leq C|D^2u||\nabla u|$, by (2.1.13) in Theorem 2.1.4, for $\varepsilon \leq \varepsilon_0$ we have

$$\left\| \operatorname{div} \left(|\nabla u| \nabla u \right) \right\|_{L^1(B^+_{4/7})} \le C \|\nabla u\|_{L^2(B^+_1)}^2.$$
(2.3.5)

Consider a cut-off function $\eta \in C_c^{\infty}(B_{4/7})$ with $\eta = 1$ in $B_{1/2}$ and $0 \leq \eta \leq 1$. By the divergence theorem, for a.e. $t \in \mathbb{R}$ we have

$$\begin{split} & \int_{\{u=t\}\cap B_{1/2}^+} |\nabla u|^2 \, \mathrm{d}\mathcal{H}^{n-1} \\ & \leq \int_{\{u=t\}\cap B_1^+\cap\{\nabla u\neq 0\}} |\nabla u|^2 \eta^2 \, \mathrm{d}\mathcal{H}^{n-1} \\ & = -\int_{\{u>t\}\cap B_1^+\cap\{\nabla u\neq 0\}} \operatorname{div} \left(|\nabla u|\nabla u\,\eta^2\right) \, \mathrm{d}x - \int_{\{u>t\}\cap\partial^0 B_1^+\cap\{\nabla u\neq 0\}} |\nabla u|^2 \, \eta^2 \, \mathrm{d}x \\ & \leq \int_{B_{4/7}^+} |\nabla u|^2 |\nabla(\eta^2)| \, \mathrm{d}x + \int_{B_{4/7}^+} \left|\operatorname{div} \left(|\nabla u|\nabla u\right) \right| \eta^2 \, \mathrm{d}x \end{split}$$

and (2.3.5) now yields the claim

Step 2: We prove that, if $\varepsilon \leq \varepsilon_0$, then

$$\|\nabla u\|_{L^{2+\gamma}(B_{1/2}^+)} \le C \|\nabla u\|_{L^2(B_1^+)},$$

where $\gamma > 0$ is dimensional and $\varepsilon_0 > 0$ and C are universal constants.

Multiplying by a constant, we may assume that $\|\nabla u\|_{L^2(B_1^+)} = 1$.

Letting $h(t) = \max\{1, t\}$, by the Sobolev embedding for functions vanishing on $\partial^0 B_1^+$,

$$\int_{\mathbb{R}^{+}} \mathrm{d}t \int_{\{u=t\} \cap B_{1}^{+} \cap \{|\nabla u| \neq 0\}} \mathrm{d}\mathcal{H}^{n-1}h(t)^{p} |\nabla u|^{-1}$$

$$\leq |B_{1}^{+} \cap \{u < 1\}| + \int_{B_{1}^{+}} u^{p} \,\mathrm{d}x \leq C$$
(2.3.6)

for some p > 2. Choosing dimensional constants q > 1 and $\theta \in (0, 1/3)$ such that $p/q = (1 - \theta)/\theta$, we obtain

$$\begin{split} \int_{B_{1/2}^+} |\nabla u|^{3-3\theta} \, \mathrm{d}x &= \int_{\mathbb{R}^+} \mathrm{d}t \int_{\{u=t\} \cap B_{1/2}^+ \cap \{|\nabla u| \neq 0\}} \mathrm{d}\mathcal{H}^{n-1} h(t)^{p\theta-q(1-\theta)} |\nabla u|^{-\theta+2(1-\theta)} \\ &\leq \left(\int_{\mathbb{R}^+} \mathrm{d}t \int_{\{u=t\} \cap B_1^+ \cap \{|\nabla u| \neq 0\}} \mathrm{d}\mathcal{H}^{n-1} h(t)^p |\nabla u|^{-1} \right)^{\theta} \\ &\quad \cdot \left(\int_{\mathbb{R}^+} h(t)^{-q} \, \mathrm{d}t \int_{\{u=t\} \cap B_{1/2}^+} \mathrm{d}\mathcal{H}^{n-1} |\nabla u|^2 \right)^{1-\theta}. \end{split}$$

By Step 1 and (2.3.6), it follows that

$$\int_{B_{1/2}^+} |\nabla u|^{3-3\theta} \,\mathrm{d}x \le C,$$

which was the claim.

Step 3: Conclusion.

Combining Step 2 (rescaled) and Lemma 2.3.1, we deduce that our class of stable solutions satisfies

$$\|\nabla u\|_{L^{2+\gamma}(B_{1/4}^+)} \le C \|u\|_{L^1(B_1^+)} \quad \text{for } \varepsilon \le \varepsilon_0,$$
(2.3.7)

where $\gamma > 0$ is dimensional and $\varepsilon_0 > 0$ and C are universal.

To conclude, we apply a simple covering argument. Let $\delta \in (0, 1)$ be sufficiently small such that

$$\delta\left(\|DA\|_{L^{\infty}(B_{1}^{+})} + \|b\|_{L^{\infty}(B_{1}^{+})}\right) \leq \varepsilon_{0}.$$
(2.3.8)

First, we cover the lower boundary $\partial^0 B_{1/2}^+$ by a finite number of balls $B_{\delta/4}(y_i)$ with $y_i \in \partial^0 B_1^+$, taking $\delta > 0$ smaller if necessary so that $B_{\delta}(y_i) \subset B_1$. Next, we cover $\overline{B_{1/2}^+} \setminus (\cup_i B_{\delta/4}(y_i))$ by balls $B_{\delta/2}(z_i)$ with a smaller radius $\delta > 0$ such that $B_{\delta}(z_i) \subset B_1^+$. Thus we obtain a covering of $B_{1/2}^+$ by half-balls $\{B_{\delta/4}^+(y_i)\}_i$ (centered at the boundary) and interior balls $\{B_{\delta/2}(z_i)\}_i$, satisfying $B_{\delta}^+(y_i) \subset B_1^+$ and $B_{\delta}(z_i) \subset B_1^+$, respectively. Notice that, by (2.3.8), the radii δ and δ as well as the number of balls depend only on n, ε_0 , $\|DA\|_{L^{\infty}(B_1^+)}$, and $\|b\|_{L^{\infty}(B_1^+)}$.

Thanks to (2.3.8), the function $u(y_i + \delta \cdot)$ vanishing on $\partial^0 B_1^+$ is a stable solution of a semilinear equation in B_1^+ , with coefficients $A^{y_i,\delta} = A(y_i + \delta \cdot)$ and $b^{y_i,\delta} = \delta b(y_i + \delta \cdot)$ such that $\|DA^{y_i,\delta}\|_{L^{\infty}} + \|b^{y_i,\delta}\|_{L^{\infty}} \leq \varepsilon_0$. From (2.3.7) now we deduce

$$\|\nabla u\|_{L^{2+\gamma}(B^+_{\delta/4}(y_i))} \le C_{\delta} \|u\|_{L^1(B^+_{\delta}(y_i))}, \qquad (2.3.9)$$

where C_{δ} depends only on n, c_0 , C_0 , and δ . For the interior balls $B_{\delta/4}(z_i)$, we need the following interior energy estimates from Theorem 1.1.1 above (rescaled):

$$\|\nabla u\|_{L^{2+\gamma}(B_{\widetilde{\delta}/2}(z_i))} \le C_{\widetilde{\delta}} \|u\|_{L^1(B_{\widetilde{\delta}}(z_i))},\tag{2.3.10}$$

where C_{δ} depends only on n, c_0, C_0 , and δ .

By (2.3.9) and (2.3.10), we finally obtain

$$\begin{aligned} \|\nabla u\|_{L^{2+\gamma}(B^+_{1/2})} &\leq \sum_{i} \|\nabla u\|_{L^{2+\gamma}(B^+_{\delta/4}(y_j))} + \sum_{i} \|\nabla u\|_{L^{2+\gamma}(B^-_{\delta/2}(z_i))} \\ &\leq C_{\delta} \sum_{i} \|u\|_{L^{1}(B^+_{\delta}(y_i))} + C_{\widetilde{\delta}} \sum_{i} \|u\|_{L^{1}(B^-_{\delta}(z_i))} \\ &\leq C \|u\|_{L^{1}(B^+_{1})}, \end{aligned}$$

where the last constant depends only on n, c_0 , C_0 , $||DA||_{L^{\infty}(B_1^+)}$, and $||b||_{L^{\infty}(B_1^+)}$. This concludes the proof of the theorem.

Remark 2.3.2. It is also possible to deduce a higher integrability of the gradient from Lemma 2.3.1 directly by applying Gehring's lemma [59]. However, by that method, the integrability exponent in Theorem 2.1.1 would no longer be dimensional (i.e., depending only on n), but would additionally depend on the ellipticity constants.⁴ Thus, the techniques

⁴Indeed, combining Lemma 2.3.1 with the analogous interior estimates in [52, Proposition 1.3], by Poincaré's inequality and a scaling and covering argument, it is not hard to show that the (say) even reflection of ∇u with respect to $\{x_n = 0\}$ satisfies $\left(R^{-n}\int_{B_R(x)}|\nabla u|^2\right)^{1/2} \leq C_1R^{-n}\int_{B_{2R}(x)}|\nabla u|$ for any ball $B_{2R}(x) \subset B_1$, where $C_1 = C_1(n, c_0, C_0)$ is a universal constant. Applying Gehring's lemma (for instance, by Theorem 6.38 in [62]) we now obtain an estimate $\|\nabla u\|_{L^p(B_{1/2}^+)} \leq C \|\nabla u\|_{L^2(B_1^+)}$ for some $p = p(n, C_1) > 2$ and $C = C(n, C_1)$.

in [24] give a more precise control of the integrability exponent than Gehring's lemma. For instance, following the proof above, it is easy to see that one can take any $\gamma(n) < \frac{4}{3n-2}$.

We conclude this section by stating a corollary of the higher integrability and Hessian estimates that will be useful in the next chapter. It consists of two simple estimates on annuli that can be proven by a standard covering argument, combining Theorem 2.1.1 (respectively Theorem 2.1.4 and Lemma 2.3.1) with the analogous interior estimates Theorem 1.1.1 (respectively in Proposition 1.1.3 and Remark 1.3.4) obtained in Chapter 1.

Corollary 2.3.3. Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a nonnegative stable solution of -Lu = f(u) in B_1^+ , with u = 0 on $\partial^0 B_1^+$. Assume that $f \in C^1(\mathbb{R})$ is nonnegative and nondecreasing. Assume that L satisfies conditions (2.1.3), (2.1.4), and (2.1.5) in $\Omega = B_1^+$, and

$$\|DA\|_{L^{\infty}(B_{1}^{+})} + \|b\|_{L^{\infty}(B_{1}^{+})} \le \varepsilon$$

for some $\varepsilon > 0$. Let $0 < \rho_1 < \rho_2 < \rho_3 < \rho_4 \le 1$. Then

$$\|\nabla u\|_{L^{2+\gamma}(A^+_{\rho_2,\rho_3})} \le C_{\varepsilon,\rho_i} \|u\|_{L^1(A^+_{\rho_1,\rho_4})}$$

and

$$||D^2u||_{L^1(A^+_{\rho_2,\rho_3})} \le C_{\varepsilon,\rho_i} ||u||_{L^1(A^+_{\rho_1,\rho_4})}$$

where C_{ε,ρ_i} is a constant depending only on $n, c_0, C_0, \varepsilon, \rho_1, \rho_2, \rho_3$, and ρ_4 .

Chapter 3

Boundary regularity in $C^{1,1}$ domains

This chapter establishes the boundary Hölder continuity of stable solutions to semilinear elliptic problems in the optimal range of dimensions $n \leq 9$, for $C^{1,1}$ domains. We consider equations -Lu = f(u) in a bounded $C^{1,1}$ domain $\Omega \subset \mathbb{R}^n$, with u = 0 on $\partial\Omega$, where L is a linear elliptic operator with variable coefficients and $f \in C^1$ is nonnegative, nondecreasing, and convex. The stability of u amounts to the nonnegativity of the principal eigenvalue of the linearized equation -L - f'(u). Our result is new even for the Laplacian, for which [24] proved the Hölder continuity in C^3 domains.

3.1 Introduction

Let $\Omega \subset \mathbb{R}^n$ be a bounded domain and $f \colon \mathbb{R} \to \mathbb{R}$ a C^1 function. In this chapter, we consider stable solutions $u \colon \overline{\Omega} \to \mathbb{R}$ to the semilinear problem

$$\begin{cases} -Lu = f(u) & \text{in } \Omega\\ u = 0 & \text{on } \partial\Omega, \end{cases}$$
(3.1.1)

where L is a second order linear elliptic differential operator of the form

$$Lu = a_{ij}(x)u_{ij} + b_i(x)u_i, \quad a_{ij}(x) = a_{ji}(x).$$
(3.1.2)

We assume that the coefficient matrix $A(x) = (a_{ij}(x))$ is uniformly elliptic in Ω , i.e., there are positive constants c_0 , C_0 such that

$$c_0|p|^2 \le a_{ij}(x)p_ip_j \le C_0|p|^2 \quad \text{for all } p \in \mathbb{R}^n.$$
(3.1.3)

This last condition is denoted by $c_0 \leq A(x) \leq C_0$. In addition, we will always assume that

$$a_{ij} \in C^{0,1}(\overline{\Omega}), \quad b_i \in L^{\infty}(\Omega) \cap C^0(\Omega).$$
 (3.1.4)

For some auxiliary results, we will further need that

$$b_i \in C^0(\overline{\Omega}). \tag{3.1.5}$$

A strong solution $u \in C^0(\overline{\Omega}) \cap W^{2,n}_{\text{loc}}(\Omega)$ of (3.1.1) is *stable* if the principal eigenvalue of the linearized equation at u is nonnegative. Equivalently, the solution u is stable if there is a function $\varphi \in W^{2,n}_{\text{loc}}(\Omega)$ satisfying

$$\begin{cases} J_u \varphi \le 0 & \text{a.e. in } \Omega, \\ \varphi > 0 & \text{in } \Omega, \end{cases}$$
(3.1.6)

where $J_u := L + f'(u)$ denotes the Jacobi operator (the linearization) at u. For variational problems, the stability condition amounts to the nonnegativity of the second variation. In particular, the class of stable solutions contains (local and global) minimizers. Instead, here we are interested in non-variational equations as above. For fundamental properties of the principal eigenvalue of linear non self-adjoint operators such as J_u , we refer to the classic work of Berestycki, Nirenberg, and Varadhan [9].

The aim of this chapter is to investigate the regularity up to the boundary of stable solutions to (3.1.1). Here, the question reduces to showing that solutions are bounded, since the linear theory then allows to prove further smoothness properties. Our present work extends the boundary regularity results of Cabré, Figalli, Ros-Oton, and Serra [24] and of Cabré [19] for the Laplacian to the above operators with variable coefficients. When $n \ge 10$, examples of singular stable solutions have been known for a long time. In the breakthrough article [24], the authors solved the long-standing conjecture: if $n \le 9$, then all stable solutions are bounded (when $L = \Delta$). Their proof was quite delicate and relied on a contradiction-compactness argument which did not allow to quantify the constants in the estimates. An alternative quantitative proof has been recently found in [19], although it only applies to the Laplacian in flat domains. Generalizing and combining ideas form these two works, we will give, for the first time (even for the Laplacian), a quantitative proof valid in curved domains.

For $L = \Delta$, a key assumption needed in [24] was for the domain to be of class C^3 . As we explain next, our analysis will allow us to weaken this condition to a $C^{1,1}$ regularity assumption. For this, starting from a curved boundary, we flatten it out (locally) by a change of variables. This transformation does not alter the semilinear nature of the equation but modifies the coefficients, which now involve first and second order derivatives of the flattening map. Namely, the new coefficients a_{ij} include first derivatives of this map, while the b_i contain second derivatives. Now, the crucial point is to obtain universal a priori estimates independent of the nonlinearity (in the spirit of [24]) and having a specific dependence on the coefficients. Our bounds will depend only on the ellipticity constants in (3.1.3) and on the norms $\|\nabla a_{ij}\|_{L^{\infty}}$ and $\|b_i\|_{L^{\infty}}$ of the coefficients, corresponding to the flattening map of a $C^{1,1}$ domain. As mentioned above, thanks to a device of [19] for flat domains, our estimates in the new coordinates will all be quantitative. Combined with the interior bounds that we established in [52], they will yield a global estimate.

We believe that our ideas can also be applied to study the boundary regularity of stable solutions for other equations. Our method provides a robust, direct way of proving quantitative estimates up to the boundary. In particular, when L is the *p*-Laplacian, we could extend the optimal interior bounds of Cabré, Miraglio, and Sanchón in [25] up to the boundary. By contrast, the previous work [24] relies on an intricate blow-up and Liouville theorem argument. The authors of [24] need this in order to apply a result of theirs only available on a flat boundary, which they could only prove by contradiction-compactness. This critical step does not allow them to quantify the constants in their inequalities.

Variational problems have a long history of regularity results for stable solutions, starting with the pioneering work of Crandall and Rabinowitz [44] in the seventies. For exponential and power nonlinearities, they showed that stable solutions are bounded in smooth domains when $n \leq 9$ (see also Joseph and Lundgren [64] for an exhaustive analysis of the radial case). Their result is optimal, since the logarithm $u(x) = \log(1/|x|^2) \in W_0^{1,2}(\Omega)$ solves (3.1.1) (in the weak sense) with $\Omega = B_1$, $L = \Delta$, and $f(u) = 2(n-2)e^u$, and is stable for $n \geq 10$. This last fact follows immediately from Hardy's inequality. Surprisingly, [44] appears to be the only variational paper where variable coefficients have been considered. Namely, the a priori estimates in [44] apply to self-adjoint operators in divergence form, with merely bounded coefficients. However, the methods used cannot be extended to treat more general nonlinearities.

The motivation for considering exponential nonlinearities in [44] came from problems in combustion theory, namely, from the so called explosion or Gelfand problem [60] (recalled in Subsection 3.1.2 below). In the nineties, Brezis [11] asked whether the optimal dimension could be the same for more general nonlinearities. He was interested in a natural class of nonlinearities for which the Gelfand problem admits stable solutions, namely: nonnegative, nondecreasing, convex, and superlinear ones. This question motivated a series of works trying to establish global a priori estimates for stable solutions to (3.1.1) in the model case, i.e., when $L = \Delta$. First, Nedev [78] proved their boundedness for $n \leq 3$. Then, Cabré and Capella [21] reached the optimal dimension $n \leq 9$ in the radial case. Later, Cabré [16] and Villegas [95] showed the boundedness when $n \leq 4$. Afterwards, Cabré and Ros-Oton [27] proved the boundedness for $n \leq 7$ when Ω is a domain of double revolution. Finally, Cabré, Figalli, Ros-Oton, and Serra [24] solved the conjecture in C^3 domains.

Concerning non-variational problems, there is only one paper, to the best of our knowledge, studying the regularity of stable solutions in our setting. In [40], Cowan and Ghoussoub consider operators of the form (3.1.2), assuming that $a_{ij} = \delta_{ij}$ and $b_i \in C^{\infty}(\overline{\Omega})$, where $\Omega \subset \mathbb{R}^n$ is a smooth bounded domain. They showed that stable solutions are bounded when $n \leq 9$ for the exponential nonlinearity. In particular, adding an advection term does not modify the optimal dimension in this case. This is in accordance with our present work, where our mild smoothness assumption on the coefficients guarantees the invariance of the optimal dimension for general nonlinearities. It is worth noting that the interest of the authors in [40] was in singular nonlinearities appearing in the modeling of MEMS devices, namely, nonlinearities f = f(u) defined for $u \in [0, 1)$ which blow up at u = 1.

Finally, we would also like to mention the recent work of Costa, de Souza, and Montenegro [38] for a more general non-variational setting. In that paper, the authors consider the Gelfand problem for systems of equations including operators of the form (3.1.2). While they are mostly concerned with the existence of stable solutions to this problem, they also address the question of regularity, but only for the Laplacian. More precisely, they study stable solutions $u: \mathbb{R}^n \to \mathbb{R}^m$ (with $m \geq 2$) of $-\Delta u = F(u)$, where $F: \mathbb{R}^m \to \mathbb{R}^m$ satisfies natural assumptions analogous to ours. In the radial case, they are able to show that they are bounded for $n \leq 9$, which is the optimal dimension for scalar equations. For convex $C^{1,1}$ domains, they show their boundedness for $n \leq 3$ by adapting the interior estimates of Cabré [16]. We believe that our techniques can also be used to reach the optimal dimension for systems of equations in any $C^{1,1}$ domain.

3.1.1 Main results

Our main result provides two types of a priori estimates for strong stable solutions on domains of class $C^{1,1}$. The first one is an energy estimate valid in all dimensions. It has been announced in the previous chapter (Theorem 2.1.5 above), where we have proved it in flat domains (see Theorem 2.1.5 above or Theorem 3.1.4 below). Here, we will complete the proof, which involves a covering and approximation procedure. The second estimate is a bound of the Hölder norm in the optimal range of dimensions $n \leq 9$. As usual, here and throughout the chapter, when we write $C = C(\ldots)$ for a constant C we mean that C depends only on the quantities appearing inside the parentheses.

Theorem 3.1.1. Let $\Omega \subset \mathbb{R}^n$ be a bounded domain of class $C^{1,1}$ and let L satisfy conditions (3.1.3) and (3.1.4) in Ω . Assume that $f \in C^1(\mathbb{R})$ is nonnegative, nondecreasing, and convex.

Let $u \in C^0(\overline{\Omega}) \cap W^{2,n}_{\text{loc}}(\Omega)$ be a nonnegative stable solution of -Lu = f(u) in Ω , with u = 0 on $\partial\Omega$.

Then

$$\|\nabla u\|_{L^{2+\gamma}(\Omega)} \le C \|u\|_{L^{1}(\Omega)},\tag{3.1.7}$$

where $\gamma = \gamma(n) > 0$ and $C = C(\Omega, n, c_0, C_0, \|\nabla a_{ij}\|_{L^{\infty}(\Omega)}, \|b_i\|_{L^{\infty}(\Omega)})$. In addition,

$$\|u\|_{C^{\alpha}(\overline{\Omega})} \le C \|u\|_{L^{1}(\Omega)} \quad \text{if } n \le 9,$$

$$(3.1.8)$$

where $\alpha = \alpha(n, c_0, C_0) > 0$ and $C = C(\Omega, n, c_0, C_0, \|\nabla a_{ij}\|_{L^{\infty}(\Omega)}, \|b_i\|_{L^{\infty}(\Omega)}).$

The proof of Theorem 3.1.1 relies on analogous boundary estimates in half-balls, given next, as well as in the interior bounds from Chapter 1. As before, below, we always write $\mathbb{R}^n_+ = \{x \in \mathbb{R}^n : x_n > 0\}$ and, for each $\rho > 0$, we let

$$B_{\rho}^{+} := \mathbb{R}^{n}_{+} \cap B_{\rho}.$$

Moreover, for any open set $\Omega \subset \mathbb{R}^n_+$, we denote its lower and upper boundaries by

$$\partial^0 \Omega = \{ x_n = 0 \} \cap \partial \Omega, \quad \partial^+ \Omega = \mathbb{R}^n_+ \cap \partial \Omega.$$

The Hölder estimate (3.1.8) will be a consequence of the following:

Theorem 3.1.2. Let L satisfy conditions (3.1.3), (3.1.4), and (3.1.5) in $\Omega = B_1^+$. Assume that $f \in C^1(\mathbb{R})$ is nonnegative, nondecreasing, and convex.

Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a nonnegative stable solution to -Lu = f(u) in B_1^+ , with u = 0 on $\partial^0 B_1^+$.

Then

$$\|u\|_{C^{\alpha}(\overline{B^{+}_{1/2}})} \le C \|u\|_{L^{1}(B^{+}_{1})} \quad if \ n \le 9,$$
(3.1.9)

where $\alpha = \alpha(n, c_0, C_0) > 0$ and $C = C(n, c_0, C_0, \|\nabla a_{ij}\|_{L^{\infty}(B_1^+)}, \|b_i\|_{L^{\infty}(B_1^+)}).$

Remark 3.1.3. In contrast to Theorem 3.1.1 above, here we require additional hypotheses on the solution and the coefficients. Namely, we need third weak derivatives of u to remove the nonlinearity from the stability condition, making our bounds independent of f. We further need the continuity of b_i up to the boundary, assumption (3.1.5), to control certain surface integrals over $\partial^0 B_1^+$ appearing in the proof.

To prove the estimate in $C^{1,1}$ domains from the one in half-balls, we carry out an approximation and flattening procedure. We consider an exhaustion by smooth domains Ω_k and, in each of them, we construct a stable solution u_k to a semilinear equation with more regular coefficients. The smoothness of the data will guarantee that u_k is in $W^{3,p}(\Omega_k)$ and hence, flattening the boundary $\partial \Omega_k$, we may apply Theorem 3.1.2. Thanks to the $C^{1,1}$ regularity assumption, the constants in the bounds for u_k in half-balls will be independent of k. By convexity of f, the functions u_k converge to the original solution and taking limits we deduce the theorem.

As mentioned above, the energy estimate (3.1.7) in $C^{1,1}$ domains uses the analogue result in half-balls. In the following result from Chapter 2 (see Theorem 2.1.1), we obtained such a bound via Hessian estimates for stable solutions in the spirit of Sternberg and Zumbrun [91]. To prove Theorem 3.1.2, we will need both the energy and Hessian estimates.

Theorem 3.1.4 (Theorem 2.1.4). Let L satisfy conditions (3.1.3), (3.1.4), and (3.1.5) in $\Omega = B_1^+$. Assume that $f \in C^1(\mathbb{R})$ is nonnegative and nondecreasing. Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a nonnegative stable solution to -Lu = f(u) in

 B_1^+ , with u = 0 on $\partial^0 B_1^+$.

Then

$$\||\nabla u| D^2 u\|_{L^1(B_{1/2}^+)} \le C \|\nabla u\|_{L^2(B_1^+)}^2, \tag{3.1.10}$$

and

$$\|\nabla u\|_{L^{2+\gamma}(B^+_{1/2})} \le C \|u\|_{L^1(B^+_1)},\tag{3.1.11}$$

where $\gamma = \gamma(n) > 0$ and $C = C(n, c_0, C_0, \|\nabla a_{ij}\|_{L^{\infty}(B^+)}, \|b_i\|_{L^{\infty}(B^+)}).$

Application: Regularity of the extremal solution in $C^{1,1}$ 3.1.2domains.

Let $f: [0, +\infty) \to \mathbb{R}$ satisfy f(0) > 0 and be nondecreasing, convex, and superlinear at $+\infty$, meaning that

$$\lim_{u \to +\infty} \frac{f(u)}{u} = +\infty.$$

Given a constant $\lambda > 0$, we consider the problem

$$\begin{cases} -Lu = \lambda f(u) & \text{in } \Omega \\ u > 0 & \text{in } \Omega \\ u = 0 & \text{on } \partial \Omega, \end{cases}$$
(3.1.12)

where $\Omega \subset \mathbb{R}^n$ is a $C^{1,1}$ bounded domain, and L satisfies conditions (3.1.3) and (3.1.4) in Ω.

The boundary value problem (3.1.12) is the *Gelfand problem* mentioned above. It was first presented by Barenblatt in a volume edited by Gelfand [60]. Originally, (3.1.12) was introduced to study ignition and explosion phenomena in the theory of thermal combustion. In that framework, u can be understood as the temperature of a combustible mixture, while λ measures the relative strength of the reaction f(u) with respect to the diffusion-advection processes modeled by L. When λ is large, solutions are not expected to exist, which is interpreted as the occurrence of an explosion.

Stable solutions play a prominent role in the Gelfand problem, as evidenced by the next proposition below. For an account of the history and references for (3.1.12), we refer the reader to the monograph of Dupaigne [48]. Here, instead, we only recall a basic, well-known result concerning the existence of solutions to (3.1.12).

Note that by the identification $a_{ij} \in C^{0,1}(\overline{\Omega}) = W^{1,\infty}(\Omega)$ we can always write our operator L in divergence form

$$Lu = \operatorname{div} \left(A(x)\nabla u \right) + \widehat{b}(x) \cdot \nabla u, \qquad (3.1.13)$$

where $\hat{b}(x) = (\hat{b}_i(x))$ is the vector field given by

$$\hat{b}_i(x) = b_i(x) - \partial_k a_{ki}(x). \tag{3.1.14}$$

In particular, since we always assume that $\partial_k a_{ij} \in L^{\infty}(\Omega)$ and $b_i \in L^{\infty}(\Omega)$, we also have $\hat{b}_i \in L^{\infty}(\Omega)$. The following result for non-variational problems has appeared in a slightly different form in [8,38]. For the classical variational version, see, for instance, [11,48].

Proposition 3.1.5. There exists a constant $\lambda^* \in (0, +\infty)$ such that:

- (i) For each $\lambda \in (0, \lambda^*)$ there is a unique strong stable solution $u_{\lambda} \in C^0(\overline{\Omega}) \cap W^{2,n}_{\text{loc}}(\Omega)$ of (3.1.12). Moreover, we have $u_{\lambda} < u_{\lambda'}$ in Ω for $\lambda < \lambda'$.
- (ii) For $\lambda > \lambda^*$ there is no strong solution.

Assume moreover that

$$\operatorname{div} \widehat{b} \in L^{\infty}(\Omega), \tag{3.1.15}$$

so that the adjoint operator

$$L^{T}\zeta = \operatorname{div}\left(A(x)\nabla\zeta\right) - \widehat{b}(x)\cdot\nabla\zeta - \operatorname{div}\widehat{b}(x)\zeta$$

is well defined for $\zeta \in W^{2,n}_{\text{loc}}(\Omega)$ and has bounded coefficients. Then:

(iii) For $\lambda = \lambda^*$ there exists a unique L^1 -weak solution u^* , in the following sense: $u^* \in L^1(\Omega), f(u^*) \operatorname{dist}(\cdot, \partial \Omega) \in L^1(\Omega)$, and

$$-\int_{\Omega} u^{\star} L^{T} \zeta \, \mathrm{d}x = \lambda^{\star} \int_{\Omega} f(u^{\star}) \zeta \, \mathrm{d}x \quad \text{for all } \zeta \in W^{2,n}(\Omega) \text{ with } L^{T} \zeta \in L^{\infty}(\Omega) \text{ and } \zeta|_{\partial\Omega} = 0$$

The solution u^* is called the extremal solution of (3.1.12) and satisfies $u_{\lambda} \uparrow u^*$ as $\lambda \uparrow \lambda^*$.

Remark 3.1.6. The uniqueness of u^* is due to Martel [73]. Although he proved it in the model case $L = \Delta$, the same ideas extend to the operators considered in this chapter.

Remark 3.1.7. The additional regularity (3.1.15) of the drift \hat{b}_i is needed in (iii) to guarantee that $u^* \in L^1(\Omega)$ and $f(u^*) \operatorname{dist}(\cdot, \partial \Omega) \in L^1(\Omega)$. For this, testing the equation with the principal eigenfunction ϕ of L^T , by superlinearity of f, it is easy to show that

$$\int_{\Omega} f(u_{\lambda}) \phi \, \mathrm{d}x \le C \quad \text{ for } \lambda \in (0, \lambda^{\star}),$$

where C does not depend on λ . By regularity $\phi \in W^{2,p}(\Omega)$ for all $p < \infty$, hence $\phi \in C^1(\overline{\Omega})$, and by maximum principle $\phi \geq c \operatorname{dist}(\cdot, \partial \Omega)$, whence $f(u^*)\operatorname{dist}(\cdot, \partial \Omega) \in L^1(\Omega)$. Now, testing the equation with the unique solution to $-L^T \vartheta = 1$ in $\Omega, \vartheta = 0$ on $\partial\Omega$, we also have

$$\int_{\Omega} u_{\lambda} \, \mathrm{d}x = \lambda \int_{\Omega} f(u_{\lambda}) \vartheta \, \mathrm{d}x \quad \text{ for } \lambda \in (0, \lambda^{\star}).$$

Again, by regularity $\vartheta \leq C\phi$, and using the inequality above, we conclude that $u^* \in L^1(\Omega)$.

Since, a priori, the extremal solution u^* is only in $L^1(\Omega)$, it is natural to investigate further regularity properties. In this direction, Brezis and Vázquez [14, Problem 1] asked whether the extremal solution for the model operator $L = \Delta$ was always in $W_0^{1,2}(\Omega)$, this being the natural energy space of the variational problem. Similarly, as explained above, Brezis [11, Open problem 1] asked if the extremal solution u^* was always bounded for $n \leq 9$. Both of these questions were answered positively by Cabré, Figalli, Ros-Oton, and Serra [24] for the Laplacian in C^3 domains. For this, they applied their a priori estimates to the classical stable solutions $\{u_\lambda\}_{\lambda<\lambda^*}$ and, using that they are bounded in $L^1(\Omega)$, they passed to the limit as $\lambda \uparrow \lambda^*$. By this same procedure, our main theorem, Theorem 3.1.1, extends their result to operators with coefficients in $C^{1,1}$ domains: **Corollary 3.1.8.** Let $\Omega \subset \mathbb{R}^n$ be a bounded domain of class $C^{1,1}$ and let L satisfy conditions (3.1.3), (3.1.4), and (3.1.15) in Ω . Assume that $f \in C^1(\mathbb{R})$ is positive, nondecreasing, convex, and superlinear at $+\infty$.

Then the extremal solution u^* to (3.1.12) is in $W_0^{1,2+\gamma}(\Omega)$ for some $\gamma = \gamma(n) > 0$. Moreover, if $n \leq 9$, then u^* is bounded (and hence is a strong solution in $W^{2,p}(\Omega)$ for all $p < \infty$).

3.1.3 Structure of the proof

By approximation, it will suffice to prove Theorem 3.1.1 in smooth domains. Flattening the boundary, we further reduce the problem to half-balls and, hence, the core of the proof is to show the Hölder estimate in Theorem 3.1.2. Moreover, by a scaling and covering argument, we may assume that the operator L is close to the Laplacian, i.e., the coefficients satisfy A(0) = I and $\|DA\|_{L^{\infty}} + \|b\|_{L^{\infty}} \leq \varepsilon$, with ε small.

As in the previous chapters, to obtain a priori estimates, we will use the stability of the solution via a more convenient integral inequality. Following the notation above, we denote the norm induced by the positive definite matrix $A(x) = (a_{ij}(x))$ by

$$|p|_{A(x)} := (a_{ij}(x)p_ip_j)^{1/2}$$
 for $p \in \mathbb{R}^n$.

In Chapter 1, we showed that if u is stable, then

$$\int_{\Omega} f'(u)\xi^2 \,\mathrm{d}x \le \int_{\Omega} \left| \nabla \xi - \frac{1}{2}\xi A^{-1}(x)\widehat{b}(x) \right|_{A(x)}^2 \,\mathrm{d}x \quad \text{for all } \xi \in C_c^{\infty}(\Omega), \tag{3.1.16}$$

where the vector field $\hat{b}(x) = (\hat{b}_i(x))$ is given by (3.1.14) above (and introduced in (2.1.7)).

Using the integral stability inequality (3.1.16) and thanks to the energy and Hessian estimates in Theorem 3.1.4, we will be able to prove two key auxiliary results: Propositions 3.1.9 and 3.1.10 below, which we comment on next. Combined, they will yield the Hölder estimate (3.1.9) in Theorem 3.1.2.

The first proposition provides a weighted L^2 estimate for the radial derivative

$$u_r = \frac{x}{|x|} \cdot \nabla u \quad (r = |x|)$$

in a half-ball by the L^2 norm of the full gradient in a half-annulus, under a smallness condition on the coefficients. This bound requires $n \leq 9$ and will be essential in the proof of the Hölder regularity of stable solutions. As above, here and throughout this chapter, a constant depending only on n, c_0 , and C_0 will be called *universal*.

Proposition 3.1.9. Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a nonnegative stable solution of -Lu = f(u) in B_1^+ , with u = 0 on $\partial^0 B_1^+$. Assume that $f \in C^1(\mathbb{R})$ is nonnegative and nondecreasing. Assume that L satisfies conditions (3.1.3), (3.1.4), and (3.1.5) in $\Omega = B_1^+$, and

$$\|DA\|_{L^{\infty}(B_{1}^{+})} + \|b\|_{L^{\infty}(B_{1}^{+})} \le \varepsilon$$

for some $\varepsilon > 0$.

If $3 \le n \le 9$ and $\varepsilon \le \varepsilon_0$, then

$$\int_{B_{\rho}^{+}} r^{2-n} u_{r}^{2} \,\mathrm{d}x \leq C \int_{B_{2\rho}^{+} \setminus B_{\rho}^{+}} r^{2-n} |\nabla u|^{2} \,\mathrm{d}x + C\varepsilon \int_{B_{4\rho}^{+}} r^{3-n} |\nabla u|^{2} \,\mathrm{d}x$$

for all $\rho \leq 1/4$, where $\varepsilon_0 > 0$ and C are universal constants.

Although Proposition 3.1.9 requires $n \geq 3$, adding superfluous variables, we may be able to use it to prove the C^{α} estimate when $n \leq 2$ as well.

To prove Proposition 3.1.9, we use the integral stability inequality (3.1.16) with appropriate test functions. Letting $\xi = \mathbf{c}\eta$ in (3.1.16) with $\Omega = B_1^+$, where \mathbf{c} , η are smooth functions satisfying $\mathbf{c} = 0$ on $\partial^0 B_1^+$ and supp $\eta \subset B_1$, if we integrate by parts, then (3.1.16) becomes

$$\int_{B_1^+} \mathbf{c} J_u \mathbf{c} \, \eta^2 \, \mathrm{d}x \le \int_{B_1^+} \mathbf{c}^2 \left| \nabla \eta - \frac{1}{2} \eta A^{-1}(x) b(x) \right|_{A(x)}^2 \, \mathrm{d}x. \tag{3.1.17}$$

By approximation, we may choose

$$\mathbf{c}(x) = x \cdot \nabla u(x) = ru_r \quad \text{and} \quad \eta(x) = |x|_{A^{-1}(0)}^{\frac{2-n}{2}} \zeta(x),$$

where $\zeta \in C_c^{\infty}(B_1)$ is a cut-off. A test function of this type appeared for the first time in the work of Cabré and Capella [21] for the Laplacian in the radial case. A similar choice was used in [24] to establish the boundedness in C^3 domains. Our function is a linear transformation of the latter one.

This choice of test function will lead to the desired inequality, but also produces weighted Hessian errors that we must control. For this, we invoke Theorem 3.1.4, which is why we need the continuity of the coefficient $b_i \in C^0(\overline{B_1^+})$ and the assumptions $f \ge 0$ and $f' \ge 0$ on the nonlinearity. By contrast, for the Laplacian in C^3 domains, no such errors arise, which is why the previous works [19, 24] did not need any assumptions on f at this step.

In the second proposition, we control the L^1 norm of our stable solution by the L^1 norm of its radial derivative. This estimate is an extension of a device in [19] for the Laplacian in half-balls, and it is also the key step which makes our proofs quantitative. In addition to a smallness condition on the coefficients, this is the only place where we need the assumption A(0) = I and where we use the convexity of f crucially (aside from the approximation argument for $C^{1,1}$ domains). It is worth mentioning that such a tool is also available for interior estimates, where the proof is entirely different and only requires f to be nonnegative; see [19,52].

Proposition 3.1.10. Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a nonnegative stable solution of -Lu = f(u) in B_1^+ , with u = 0 on $\partial^0 B_1^+$. Assume that $f \in C^1(\mathbb{R})$ is nonnegative, nondecreasing, and convex. Assume that L satisfies conditions (3.1.3), (3.1.4), and (3.1.5) in $\Omega = B_1^+$, and that

$$A(0) = I$$
 and $||DA||_{L^{\infty}(B_1^+)} + ||b||_{L^{\infty}(B_1^+)} \le \varepsilon$

for some $\varepsilon > 0$.

If $\varepsilon \leq \varepsilon_0$, then

$$\|u\|_{L^1(B_1^+ \setminus B_{1/2}^+)} \le C \|u_r\|_{L^1(B_1^+ \setminus B_{1/2}^+)}$$

where $\varepsilon_0 > 0$ and C are universal constants.

The proof of the Hölder estimate (3.1.9) in Theorem 3.1.2 requires the previous results. Combining the energy estimate (3.1.11) with Proposition 3.1.10 on dyadic annuli and rescaling, it is easy to show that the weighted Dirichlet energy in a ball is controlled by the weighted L^2 norm of the radial derivative in a larger ball. Applying Proposition 3.1.9 and by hole-filling, we now deduce a decay of the former quantity, leading to a C^{α} bound.
3.1.4 Outline of the chapter

In Section 3.2, we prove Proposition 3.1.9, the weighted inequality for the radial derivative. Section 3.3 focuses on the proof of Proposition 3.1.10, controlling the solution by its radial derivative. In Section 3.4, we obtain the Hölder estimates in Theorem 3.1.2. Finally, in Section 3.5 we proof our main result, Theorem 3.1.1.

3.2 The boundary weighted L^2 estimate for radial derivatives

Here we obtain the weighted estimates in half-balls for the radial derivative leading to Proposition 3.1.9. We will test the stability inequality (3.1.17) with the functions

$$\mathbf{c} := x \cdot \nabla u \quad \text{and} \quad \eta = r^{\frac{2-n}{2}} \zeta,$$

where $\zeta \in C_c^{\infty}(B_1)$ is a cut-off. This is a valid test function, since $x \cdot \nabla u = 0$ on $\partial^0 B_1^+$. Thus, the proof is essentially the same as in the interior case; see Chapter 1 or [52].

First we test with a generic cut-off η , not necessarily a power function:

Lemma 3.2.1. Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a stable solution of -Lu = f(u)in B_1^+ , with u = 0 on $\partial^0 B_1^+$, for some function $f \in C^1(\mathbb{R})$. Assume that L satisfies conditions (3.1.3) and (3.1.4) in $\Omega = B_1^+$, and that

$$||DA||_{L^{\infty}(B_1^+)} + ||b||_{L^{\infty}(B_1^+)} \le \varepsilon$$

for some $\varepsilon > 0$. Then

$$\begin{split} &\int_{B_1^+} |\nabla u|_{A(x)}^2 \Big((n-2)\eta^2 + x \cdot \nabla(\eta^2) \Big) \,\mathrm{d}x \\ &+ \int_{B_1^+} \Big(-2(x \cdot \nabla u)A(x)\nabla u \cdot \nabla(\eta^2) - |x \cdot \nabla u|^2 |\nabla \eta|_{A(x)}^2 \Big) \,\mathrm{d}x \\ &\leq C\varepsilon \int_{B_1^+} |D^2 u| |\nabla u| |x|^2 \eta^2 \,\mathrm{d}x + C\varepsilon \int_{B_1^+} |\nabla u|^2 \Big(|x|^2 |\nabla(\eta^2)| + (|x| + \varepsilon |x|^2) \eta^2 \Big) \,\mathrm{d}x, \end{split}$$

for all $\eta \in C_c^{\infty}(B_1)$, where C is universal.

Proof. We will test the stability inequality (3.1.17) with $\mathbf{c} = x \cdot \nabla u$. First, we compute the left-hand side of (3.1.17), i.e., $\int_{B_1^+} (x \cdot \nabla u) J_u[x \cdot \nabla u] \eta^2 dx$. Computing, we have

$$L[x \cdot \nabla u] = x_k a_{ij}(x) u_{ijk} + 2a_{ij}(x) u_{ij} + x_k b_i(x) u_{ik} + b_i(x) u_i \quad \text{for a.e. } x \in B_1^+.$$
(3.2.1)

For the zero order term, integrating by parts and using the equation, we have

$$\int_{B_1^+} f'(u)(x \cdot \nabla u)^2 \eta^2 \, \mathrm{d}x = \int_{B_1^+} \nabla [f(u)] \cdot (x \cdot \nabla u) x \, \eta^2 \, \mathrm{d}x = \int_{B_1^+} Lu \, \mathrm{div} \left\{ (x \cdot \nabla u) x \, \eta^2 \right\} \, \mathrm{d}x$$
$$= -\int_{B_1^+} x \cdot \nabla [a_{ij}(x)u_{ij}] \, (x \cdot \nabla u) \, \eta^2 \, \mathrm{d}x + \int_{B_1^+} b_i(x)u_i \, \mathrm{div} \left\{ (x \cdot \nabla u) x \, \eta^2 \right\} \, \mathrm{d}x,$$
(3.2.2)

where in the last line we have integrated by parts again. Combining (3.2.1) and (3.2.2), it follows that

$$\int_{B_{1}^{+}} (x \cdot \nabla u) J_{u}[x \cdot \nabla u] \eta^{2} dx = \int_{B_{1}^{+}} (x \cdot \nabla u) L[x \cdot \nabla u] \eta^{2} dx + \int_{B_{1}^{+}} f'(u) (x \cdot \nabla u)^{2} \eta^{2} dx
= \int_{B_{1}^{+}} 2(x \cdot \nabla u) a_{ij}(x) u_{ij} \eta^{2} dx - \int_{B_{1}^{+}} x \cdot \nabla a_{ij}(x) u_{ij} (x \cdot \nabla u) \eta^{2} dx
+ \int_{B_{1}^{+}} x_{k} b_{i}(x) u_{ik}(x \cdot \nabla u) \eta^{2} dx + \int_{B_{1}^{+}} b_{i}(x) u_{i} \Big((x \cdot \nabla u) \eta^{2} + \operatorname{div} \{ (x \cdot \nabla u) x \eta^{2} \} \Big) dx.$$
(3.2.3)

Notice that the first integrand in the right-hand side of (3.2.3) can be written as

$$2(x \cdot \nabla u)a_{ij}(x)u_{ij} = \operatorname{div}\left(2(x \cdot \nabla u)A(x)\nabla u - |\nabla u|^2_{A(x)}x\right) + (n-2)|\nabla u|^2_{A(x)} - 2(x \cdot \nabla u)\partial_i a_{ij}(x)u_j + x \cdot \nabla a_{ij}(x)u_iu_j.$$
(3.2.4)

Hence, substituting (3.2.4) in (3.2.3) we deduce

$$\int_{B_1^+} (x \cdot \nabla u) J_u[x \cdot \nabla u] \eta^2 dx$$

$$= \int_{B_1^+} \operatorname{div} \left(2(x \cdot \nabla u) A(x) \nabla u - |\nabla u|^2_{A(x)} x \right) \eta^2 dx + \int_{B_1^+} (n-2) |\nabla u|^2_{A(x)} \eta^2 dx$$

$$+ \int_{B_1^+} \left(-2\partial_i a_{ij}(x) u_j \left(x \cdot \nabla u \right) + x \cdot \nabla a_{ij}(x) \left\{ u_i u_j - u_{ij} \left(x \cdot \nabla u \right) \right\} \right) \eta^2 dx$$

$$+ \int_{B_1^+} b_i(x) x_k u_{ik}(x \cdot \nabla u) \eta^2 dx + \int_{B_1^+} b_i(x) u_i \left((x \cdot \nabla u) \eta^2 + \operatorname{div} \left\{ (x \cdot \nabla u) x \eta^2 \right\} \right) dx.$$
(3.2.5)

Thus, integrating by parts the divergence term in (3.2.5) and applying the coefficient estimates $\|\nabla a_{ij}\|_{L^{\infty}} + \|b_i\|_{L^{\infty}} \leq \varepsilon$, we obtain the lower bound

$$\int_{B_{1}^{+}} (x \cdot \nabla u) J_{u}[x \cdot \nabla u] \eta^{2} dx$$

$$\geq -\int_{B_{1}^{+}} 2(x \cdot \nabla u) A(x) \nabla u \cdot \nabla(\eta^{2}) + \int_{B_{1}^{+}} |\nabla u|^{2}_{A(x)} \Big((n-2)\eta^{2} + x \cdot \nabla(\eta^{2}) \Big) dx \quad (3.2.6)$$

$$- C\varepsilon \int_{B_{1}^{+}} \Big(|D^{2}u| |\nabla u| |x|^{2} \eta^{2} + |\nabla u|^{2} |x| \eta^{2} + |\nabla u|^{2} |x|^{2} |\nabla(\eta^{2})| \Big) dx.$$

On the other hand, testing the integral stability inequality (3.1.17) with $\mathbf{c} = x \cdot \nabla u$, we deduce the upper bound

$$\begin{split} \int_{B_1^+} (x \cdot \nabla u) J_u[x \cdot \nabla u] \eta^2 \, \mathrm{d}x \\ &\leq \int_{B_1^+} |x \cdot \nabla u|^2 \left| \nabla \eta - \frac{1}{2} \eta A^{-1}(x) \widehat{b}(x) \right|_{A(x)}^2 \mathrm{d}x \\ &\leq \int_{B_1^+} |x \cdot \nabla u|^2 |\nabla \eta|_{A(x)}^2 \, \mathrm{d}x + C\varepsilon \int_{B_1} |\nabla u|^2 |x|^2 \left(|\nabla (\eta^2)| + \varepsilon \eta^2 \right) \mathrm{d}x. \end{split}$$
(3.2.7)

Combining (3.2.6) and (3.2.7) and rearranging terms yields the claim.

Recall our notation for the radial derivative

$$r = |x|, \qquad u_r = \frac{x}{|x|} \cdot \nabla u.$$

Given $\rho \in (0, 1/2]$, we consider a cut-off $\zeta \in C_c^{\infty}(B_{2\rho})$ with $0 \leq \zeta \leq 1$, $\zeta = 1$ in B_{ρ} , and $|\nabla \zeta| \leq C/\rho$ in supp $|\nabla \zeta| \subset \overline{B_{2\rho}} \setminus B_{\rho}$. For $a \geq 0$, by approximation, we may take the singular test function $\eta = r^{-a/2}\zeta$ in Lemma 3.2.1 (see Lemma 1.5.4 above), which yields:

Lemma 3.2.2. Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a stable solution of -Lu = f(u)in B_1^+ , with u = 0 on $\partial^0 B_1^+$, for some function $f \in C^1(\mathbb{R})$. Assume that L satisfies conditions (3.1.3) and (3.1.4) in $\Omega = B_1^+$, and that

$$\|DA\|_{L^{\infty}(B_{1}^{+})} + \|b\|_{L^{\infty}(B_{1}^{+})} \le \varepsilon$$

for some $\varepsilon > 0$.

If $0 \le a \le \min\{10, n\} - 2$, then

$$(n-2-a)\int_{B_{\rho}^{+}}r^{-a}|\nabla u|^{2} \,\mathrm{d}x + \frac{a(8-a)}{4}\int_{B_{\rho}^{+}}r^{-a}u_{r}^{2} \,\mathrm{d}x$$
$$\leq C\int_{B_{2\rho}^{+}\setminus B_{\rho}^{+}}r^{-a}|\nabla u|^{2} \,\mathrm{d}x + C\varepsilon\int_{B_{2\rho}^{+}}r^{2-a}|D^{2}u||\nabla u| \,\mathrm{d}x$$
$$+ C\varepsilon\int_{B_{2\rho}^{+}}(r^{1-a}+\varepsilon r^{2-a})|\nabla u|^{2} \,\mathrm{d}x.$$

for all $\rho \leq 1/2$, where C is a universal constant.

The proof of this lemma is the same as the one for interior estimates in Lemma 1.5.4, hence we omit it. Lemma 3.2.2 now allows us to prove Proposition 3.1.9:

Proof of Proposition 3.1.9. Since $3 \le n \le 9$, we have that $\min\{10, n\} - 2 = n - 2$ and we may choose the exponent a = n - 2 in Lemma 3.2.2, leading us to the inequality

$$\frac{(n-2)(10-n)}{4} \int_{B_{\rho}^{+}} r^{2-n} u_{r}^{2} dx$$

$$\leq C \int_{B_{2\rho}^{+} \setminus B_{\rho}^{+}} r^{2-n} |\nabla u|^{2} dx + C\varepsilon \int_{B_{2\rho}^{+}} r^{4-n} |D^{2}u| |\nabla u| dx$$

$$+ C\varepsilon \int_{B_{2\rho}^{+}} (r^{3-n} + \varepsilon r^{4-n}) |\nabla u|^{2} dx$$
(3.2.8)

for $\rho \leq 1/2$.

It remains to control the weighted Hessian error in (3.2.8). For this, combining the boundary "Hessian times the gradient" estimate (3.1.10) in Theorem 3.1.4 with the analogous interior estimates in Theorem 1.1.2 above, by a simple scaling and covering argument we have

$$\int_{B^+_{\delta/2} \setminus B^+_{\delta/4}} r^{4-n} |D^2 u| |\nabla u| \, \mathrm{d}x \le C \int_{B^+_{\delta} \setminus B^+_{\delta/8}} r^{3-n} |\nabla u|^2 \, \mathrm{d}x \quad \text{for all } \delta \in (0,1) \text{ and } \varepsilon \le \varepsilon_0,$$
(3.2.9)

where $\varepsilon_0 > 0$ and C are universal. Letting $\delta = 2^{3-k}\rho$ in (3.2.9) and summing in $k \in \mathbb{N}$, we obtain

$$\int_{B_{2\rho}^+} r^{4-n} |D^2 u| |\nabla u| \, \mathrm{d}x \le C \int_{B_{4\rho}^+} r^{3-n} |\nabla u|^2 \, \mathrm{d}x \quad \text{for all } \rho \le 1/4 \text{ and } \varepsilon \le \varepsilon_0.$$
(3.2.10)

Applying (3.2.10) in (3.2.8), using that (10 - n)(n - 2) > 0, we deduce the claim.

3.3 In half-annuli the radial derivative controls the function in L^1

Here we take advantage of the homogeneity of the equation to control the L^1 norm of a solution by the L^1 norm of its radial derivative. This is an extension of a device due to Cabré [19] which provided quantitative proofs of the regularity of stable solutions for the Laplacian in flat domains. Our proofs remain quantitative thanks to this idea.

Let $\tau \geq 1$ be a parameter close to 1. Given any function $v: B_1^+ \to \mathbb{R}$, we denote its L^{∞} rescaling by $v^{\tau} := v(\tau \cdot)$. Consider the elliptic operator $Lv = a_{ij}(x)v_{ij} + b_i(x)v_i$, with coefficients $a_{ij} \in C^{0,1}(\overline{B_1^+})$ and $b_i \in C^0(\overline{B_1^+})$. For each τ , we define L^{τ} to be the operator given by the rescaling

$$L^{\tau}v := \tau^{-2}a_{ij}^{\tau}(x)v_{ij} + \tau^{-1}b_i^{\tau}(x)v_i.$$
(3.3.1)

Our principal motivation for considering L^{τ} is the invariance property $(Lv)^{\tau} = L^{\tau}v^{\tau}$. In particular, given $u \in W^{3,p}(B_1^+)$ a solution to -Lu = f(u) in B_1^+ , we have

$$-L^{\tau}u^{\tau} = f(u^{\tau}) \quad \text{in } B^{+}_{1/\tau}.$$
(3.3.2)

Notice that if $1 \leq \tau < 1 + \delta$, then this last equation is satisfied in $B^+_{1/(1+\delta)} \subset B^+_{1/\tau}$. To prove Proposition 3.1.10, we will take a derivative of (3.3.2) with respect to τ .

Before giving the proof of the proposition, it is convenient to recall the following simple corollary of the Hessian and higher integrability estimates proven in Chapter 2. Here and throughout this section, we use the notation for half-annuli from [19], namely, for $\rho_2 > \rho_1 > 0$, we let

$$A_{\rho_1,\rho_2}^+ := B_{\rho_2}^+ \setminus \overline{B_{\rho_1}^+} = \{ x \in \mathbb{R}^n \colon x_n > 0, \rho_1 < |x| < \rho_2 \}.$$

Corollary 3.3.1 (Corollary 2.3.3 in Chapter 2). Let $u \in W^{3,p}(B_1^+)$, for some p > n, be a nonnegative stable solution of -Lu = f(u) in B_1^+ , with u = 0 on $\partial^0 B_1^+$. Assume that $f \in C^1(\mathbb{R})$ is nonnegative and nondecreasing. Assume that L satisfies conditions (3.1.3), (3.1.4), and (3.1.5) in $\Omega = B_1^+$, and

$$\|DA\|_{L^{\infty}(B_{1}^{+})} + \|b\|_{L^{\infty}(B_{1}^{+})} \le \varepsilon$$

for some $\varepsilon > 0$. Let $0 < \rho_1 < \rho_2 < \rho_3 < \rho_4 \leq 1$.

Then

$$\|\nabla u\|_{L^{2+\gamma}(A^+_{\rho_2,\rho_3})} \le C_{\varepsilon,\rho_i} \|u\|_{L^1(A^+_{\rho_1,\rho_4})}$$

and

 $\|D^2 u\|_{L^1(A^+_{\rho_2,\rho_3})} \le C_{\varepsilon,\rho_i} \|u\|_{L^1(A^+_{\rho_1,\rho_4})},$

where C_{ε,ρ_i} is a constant depending only on $n, c_0, C_0, \varepsilon, \rho_1, \rho_2, \rho_3$, and ρ_4 .

Proof of Proposition 3.1.10. Considering the rescaled function $u(\frac{\cdot}{6})$, we may assume that we have a stable solution in B_6^+ .

Let $\zeta \in C_c^{\infty}(A_{4,5})$ be a nonnegative cut-off function with $\zeta = 1$ in $A_{4,1,4,9}$. We consider the function $\xi := x_n \zeta$, which satisfies

$$\xi \ge 0$$
 in $A_{4,5}^+$, $\xi = 0$ on $\partial^0 A_{4,5}^+$, $\xi = \xi_{\nu} = 0$ on $\partial^+ A_{4,5}^+$, and $\xi = x_n$ in $A_{4,1,4,9}^+$.

Multiplying (3.3.2) (rescaled) by ξ for each $1 \leq \tau \leq 1.1$ and integrating in $A_{4,5}^+$, we have

$$\int_{A_{4,5}^+} (L^\tau u^\tau) \xi \, \mathrm{d}x = -\int_{A_{4,5}^+} f(u^\tau) \, \mathrm{d}x \quad \text{for all } \tau \in [1, 1.1]. \tag{3.3.3}$$

Differentiating (3.3.3) with respect to τ and integrating, we also have

$$\int_{1}^{1.1} \frac{\mathrm{d}}{\mathrm{d}\tau} \left(\int_{A_{4,5}^+} (L^\tau u^\tau) \xi \,\mathrm{d}x \right) \mathrm{d}\tau = \int_{A_{4,5}^+} \left(f(u) - f(u^{1.1}) \right) \xi \,\mathrm{d}x.$$
(3.3.4)

Our claim will be a consequence of this last identity. For this, we first establish lower bounds for the left-hand side of (3.3.4) by using that $L^{\tau}u^{\tau} \leq 0$. Later, with the help of the stability inequality and the convexity of f, we obtain upper bounds of the right-hand side. Finally, we will control the remaining Hessian errors by applying Corollary 3.3.1.

Step 1. We prove that

$$\frac{\mathrm{d}}{\mathrm{d}\tau} \int_{A_{4,5}^+} (L^\tau u^\tau) \xi \,\mathrm{d}x \ge c \|u\|_{L^1(A_{4,7,4.8}^+)} - C\|u_r\|_{L^1(A_{4,5.5}^+)} - C\varepsilon \|D^2 u\|_{L^1(A_{4,5.5}^+)} - C\varepsilon \|\nabla u\|_{L^1(A_{4,5.5}^+)}$$

for all $\tau \in [1, 1.1]$, where c and C are universal constants.

By definition (3.3.1), we have

$$\int_{A_{4,5}^+} (L^\tau u^\tau) \xi \, \mathrm{d}x = \int_{A_{4,5}^+} \tau^{-2} a_{ij}^\tau(x) u_{ij}^\tau \xi \, \mathrm{d}x + \int_{A_{4,5}^+} \tau^{-1} b_i^\tau(x) u_i^\tau \xi \, \mathrm{d}x \tag{3.3.5}$$

for all $\tau \in [1, 1.1]$.

On the one hand, since $\frac{\mathrm{d}u^{\tau}}{\mathrm{d}\tau} = \tau^{-1}x \cdot \nabla u^{\tau}$, differentiating under the integral sign

$$\frac{\mathrm{d}}{\mathrm{d}\tau} \left\{ \int_{A_{4,5}^+} \tau^{-2} a_{ij}^\tau(x) u_{ij}^\tau \,\xi \mathrm{d}x \right\}
= \int_{A_{4,5}^+} \tau^{-3} a_{ij}^\tau(x) [x \cdot \nabla u^\tau]_{ij} \,\xi \mathrm{d}x - 2 \int_{A_{4,5}^+} \tau^{-3} a_{ij}^\tau(x) u_{ij}^\tau \,\xi \mathrm{d}x + \int_{A_{4,5}^+} \tau^{-3} x \cdot \nabla a_{ij}^\tau(x) u_{ij}^\tau \,\xi \mathrm{d}x
= \int_{A_{4,5}^+} \tau^{-1} L^\tau [x \cdot \nabla u^\tau] \,\xi \mathrm{d}x - 2 \int_{A_{4,5}^+} \tau^{-1} (L^\tau u^\tau) \,\xi \mathrm{d}x - \int_{A_{4,5}^+} \tau^{-2} b_i^\tau(x) [x \cdot \nabla u^\tau]_i \,\xi \mathrm{d}x
+ \int_{A_{4,5}^+} \tau^{-3} x \cdot \nabla a_{ij}^\tau(x) u_{ij}^\tau \,\xi \mathrm{d}x + 2 \int_{A_{4,5}^+} \tau^{-2} b_i^\tau(x) u_i^\tau \,\xi \mathrm{d}x.$$
(3.3.6)

On the other hand, since $\operatorname{supp} \xi^{1/\tau} \subset A^+_{4\tau,5\tau} \subset A^+_{4,5.5}$, by a change of variables

$$\int_{A_{4,5}^+} \tau^{-1} b_i^{\tau}(x) u_i^{\tau} \xi \, \mathrm{d}x = \int_{A_{4,5}^+} b_i^{\tau}(x) (u_i)^{\tau} \xi \, \mathrm{d}x = \int_{A_{4,5.5}^+} \tau^{-n} b_i(x) u_i \xi^{1/\tau} \, \mathrm{d}x$$

and taking a derivative

$$\frac{\mathrm{d}}{\mathrm{d}\tau} \left\{ \int_{A_{4,5}^+} \tau^{-1} b_i^{\tau}(x) u_i^{\tau} \xi \,\mathrm{d}x \right\} = \frac{\mathrm{d}}{\mathrm{d}\tau} \left\{ \int_{A_{4,5.5}^+} \tau^{-n} b_i(x) u_i \xi^{1/\tau} \,\mathrm{d}x \right\}$$

$$= -n \int_{A_{4,5.5}^+} \tau^{-(n+1)} b_i(x) u_i \xi^{1/\tau} \,\mathrm{d}x - \int_{A_{4,5.5}^+} \tau^{-(n+2)} b_i(x) u_i x \cdot \nabla \xi^{1/\tau} \,\mathrm{d}x \qquad (3.3.7)$$

$$= -n \int_{A_{4,5}^+} \tau^{-2} b_i^{\tau}(x) u_i^{\tau} \xi \,\mathrm{d}x - \int_{A_{4,5}^+} \tau^{-3} b_i^{\tau}(x) u_i^{\tau} x \cdot \nabla \xi \,\mathrm{d}x,$$

where in the last line we have undone the change of variables. Thus, combining (3.3.6) and (3.3.7), by (3.3.5) we deduce

$$\begin{split} \frac{\mathrm{d}}{\mathrm{d}\tau} \left\{ \int_{A_{4,5}^+} (L^\tau u^\tau) \xi \, \mathrm{d}x \right\} \\ &= \int_{A_{4,5}^+} \tau^{-1} L^\tau [x \cdot \nabla u^\tau] \xi dx - 2 \int_{A_{4,5}^+} \tau^{-1} (L^\tau u^\tau) \xi dx \\ &+ \int_{A_{4,5}^+} \tau^{-3} x \cdot \nabla a_{ij}^\tau (x) u_{ij}^\tau \xi dx - \int_{A_{4,5}^+} \tau^{-2} b_i^\tau (x) [x \cdot \nabla u^\tau]_i \xi dx \\ &+ (2-n) \int_{A_{4,5}^+} \tau^{-2} b_i^\tau (x) u_i^\tau \xi dx - \int_{A_{4,5}^+} \tau^{-3} b_i^\tau (x) u_i^\tau x \cdot \nabla \xi \, \mathrm{d}x, \end{split}$$

and hence, by the bounds $1 \le \tau \le 1.1$ and $\|\nabla a_{ij}\|_{L^{\infty}} + \|b_i\|_{L^{\infty}} \le \varepsilon$, we have

$$\frac{\mathrm{d}}{\mathrm{d}\tau} \left\{ \int_{A_{4,5}^+} (L^\tau u^\tau) \xi \,\mathrm{d}x \right\} \ge \int_{A_{4,5}^+} \tau^{-1} L^\tau [x \cdot \nabla u^\tau] \xi dx - 2 \int_{A_{4,5}^+} \tau^{-1} (L^\tau u^\tau) \xi dx - C\varepsilon \|D^2 u\|_{L^1(A_{4,5,5}^+)} - C\varepsilon \|\nabla u\|_{L^1(A_{4,5,5}^+)}.$$
(3.3.8)

Next, we bound the two terms in the right-hand side of (3.3.8) by below.

For the first term, we write $L^{\tau}v = \operatorname{div}(\tau^{-2}A^{\tau}(x)v) + \tau^{-1}\widehat{b}^{\tau}(x) \cdot \nabla v$ in divergence form as in (3.1.13) and integrate by parts. Recalling that ξ and ξ_{ν} vanish on $\partial^{+}A^{+}_{4,5}$, and ξ and $x \cdot \nabla u^{\tau}$ vanish on $\partial^{0}A^{+}_{4,5}$, integrating by parts twice we have

$$\int_{A_{4,5}^+} \tau^{-1} L^{\tau} [x \cdot \nabla u^{\tau}] \xi \, \mathrm{d}x
= \int_{A_{4,5}^+} \tau^{-3} (x \cdot \nabla u^{\tau}) \mathrm{div} \left(A^{\tau}(x) \nabla \xi \right) \, \mathrm{d}x + \int_{A_{4,5}^+} \tau^{-2} \widehat{b}^{\tau}(x) \cdot \nabla [x \cdot \nabla u^{\tau}] \xi \, \mathrm{d}x \qquad (3.3.9)
\ge -C \|u_r\|_{L^1(A_{4,5.5}^+)} - C\varepsilon \|D^2 u\|_{L^1(A_{4,5.5}^+)} - C\varepsilon \|\nabla u\|_{L^1(A_{4,5.5}^+)},$$

where in the last line we have used that $1 \le \tau \le 1.1$, as well as the uniform ellipticity and the bounds $\|\nabla a_{ij}\|_{L^{\infty}} + \|b_i\|_{L^{\infty}} \le \varepsilon$.

The lower bounds for the second term in (3.3.8) are the most delicate. Given $\rho_1 \in (4.1, 4.2)$ and $\rho_2 \in (4.8, 4.9)$, we consider the solution φ of the mixed boundary value problem

$$\begin{cases} -\Delta \varphi = 1 & \text{in } A^+_{\rho_1,\rho_2} \\ \varphi = 0 & \text{on } \partial^0 A^+_{\rho_1,\rho_2} \\ \varphi_{\nu} = 0 & \text{on } \partial^+ A^+_{\rho_1,\rho_2} \end{cases}$$

Notice that $\varphi \geq 0$ in A_{ρ_1,ρ_2}^+ by the maximum principle. Moreover, we have the a priori bounds $|\varphi| + |\nabla \varphi| \leq C$ in A_{ρ_1,ρ_2}^+ , where C = C(n) is a dimensional constant, and hence

$$\xi \ge c\varphi \quad \text{in } A^+_{\rho_1,\rho_2},\tag{3.3.10}$$

for some small dimensional c = c(n) > 0; for the details see [19, Appendix B].

Using (3.3.10) and by the nonnegativity of $-L^{\tau}u^{\tau} = f(u^{\tau}) \ge 0$, we have

$$-\int_{A_{4,5}^+} \tau^{-1}(L^\tau u^\tau) \xi \, \mathrm{d}x \ge c \int_{A_{\rho_1,\rho_2}^+} \tau^{-1}(-L^\tau u^\tau) \,\varphi \, \mathrm{d}x.$$
(3.3.11)

Since A(0) = I, we have $|A^{\tau}(x) - I| \leq \varepsilon \tau |x|$ and writing

$$L^{\tau}u^{\tau} = \tau^{-2}\Delta u^{\tau} + \tau^{-2} \operatorname{tr} \left((A^{\tau}(x) - I)D^{2}u^{\tau} \right) + \tau^{-1}b^{\tau}(x) \cdot \nabla u^{\tau},$$

by the bounds for φ , τ , and the coefficients, the right-hand side of (3.3.11) can be further bounded by below as

$$\int_{A_{\rho_{1},\rho_{2}}^{+}} \tau^{-1}(-L^{\tau}u^{\tau}) \varphi \, dx
\geq \tau^{-3} \int_{A_{\rho_{1},\rho_{2}}^{+}} (-\Delta u^{\tau}) \varphi \, dx - C\varepsilon \int_{A_{\rho_{1},\rho_{2}}^{+}} \tau^{-1} \left(|D^{2}u^{\tau}| |x| + |\nabla u^{\tau}| \right) dx \qquad (3.3.12)
\geq c \int_{A_{\rho_{1},\rho_{2}}^{+}} (-\Delta u^{\tau}) \varphi \, dx - C\varepsilon ||D^{2}u||_{L^{1}(A_{4,5.5}^{+})} - C\varepsilon ||\nabla u||_{L^{1}(A_{4,5.5}^{+})}.$$

Following [19], we integrate by parts the Laplacian in (3.3.12) as

$$\int_{A_{\rho_{1},\rho_{2}}^{+}} (-\Delta u^{\tau}) \varphi \, \mathrm{d}x = \int_{A_{\rho_{1},\rho_{2}}^{+}} u^{\tau} \, \mathrm{d}x - \int_{\partial^{+} A_{\rho_{1},\rho_{2}}^{+}} (u^{\tau})_{\nu} \, \mathrm{d}\mathcal{H}^{n-1}$$

$$\geq c \|u\|_{L^{1}(A_{\tau\rho_{1},\tau\rho_{2}}^{+})} - \int_{\partial^{+} B_{\rho_{1}}^{+}} |(u^{\tau})_{r}| \, \mathrm{d}\mathcal{H}^{n-1} - \int_{\partial^{+} B_{\rho_{2}}^{+}} |(u^{\tau})_{r}| \, \mathrm{d}\mathcal{H}^{n-1}$$

$$\geq c \|u\|_{L^{1}(A_{4.7,4.8}^{+})} - C \int_{\partial^{+} B_{\tau\rho_{1}}^{+}} |u_{r}| \, \mathrm{d}\mathcal{H}^{n-1} - C \int_{\partial^{+} B_{\tau\rho_{2}}^{+}} |u_{r}| \, \mathrm{d}\mathcal{H}^{n-1},$$
(3.3.13)

where in the last line we have used that $\tau \rho_1 \leq 1.1 \cdot 4.2 \leq 4.7$ and $\tau \rho_2 \geq 4.8$. Now, combining (3.3.11), (3.3.12), and (3.3.13), we deduce

$$-2\int_{A_{4,5}^{+}} \tau^{-1}(L^{\tau}u^{\tau}) \xi \, \mathrm{d}x \ge c \|u\|_{L^{1}(A_{4,7,4.8}^{+})} - C\int_{\partial^{+}B_{\tau\rho_{1}}^{+}} |u_{r}| \, \mathrm{d}\mathcal{H}^{n-1} - C\int_{\partial^{+}B_{\tau\rho_{2}}^{+}} |u_{r}| \, \mathrm{d}\mathcal{H}^{n-1} - C\varepsilon \|D^{2}u\|_{L^{1}(A_{4,5.5}^{+})} - C\varepsilon \|\nabla u\|_{L^{1}(A_{4,5.5}^{+})}.$$

$$(3.3.14)$$

Integrating (3.3.14) in $\rho_1 \in (4.1, 4.2)$ and $\rho_2 \in (4.8, 4.9)$, using that $\tau \rho_1 \ge 4.1 \ge 4$ and $\tau \rho_2 \le 1.1 \cdot 4.9 \le 5.5$, we finally obtain

$$-2\int_{A_{4,5}^+} \tau^{-1}(L^{\tau}u^{\tau})\xi \,\mathrm{d}x \ge c \|u\|_{L^1(A_{4,7,4.8}^+)} - C\|u_r\|_{L^1(A_{4,5.5}^+)} - C\varepsilon \|D^2u\|_{L^1(A_{4,5.5}^+)} - C\varepsilon \|\nabla u\|_{L^1(A_{4,5.5}^+)}.$$
(3.3.15)

Applying (3.3.9) and (3.3.15) in (3.3.8) now yields the claim.

Step 2. We prove that for every $\delta \in (0,1)$ and $\varepsilon_0 > 0$, we have

$$\int_{A_{4,5}^+} \left(f(u) - f(u^{1.1}) \right) \xi \, \mathrm{d}x$$

$$\leq C_{\varepsilon_0} \left(\delta \|u\|_{L^1(A_{3,6}^+)} + \delta \|D^2 u\|_{L^1(A_{3,8,5.8}^+)} + \delta^{-1-2\frac{2+\gamma}{\gamma}} \|u_r\|_{L^1(A_{3,6}^+)} \right)$$

for all $\varepsilon \leq \varepsilon_0$, where $\gamma = \gamma(n) > 0$ and C_{ε_0} depends only on n, c_0, C_0 , and ε_0 .

Let $\phi \in C_c^{\infty}(A_{3.9,5.1})$ be a nonnegative test function with $\phi = 1$ in $A_{4,5}$. Since $\xi = 0$ on $\partial A_{4,5}^+$ and $u - u^{1.1} = 0$ on $\partial^0 A_{3.9,5.1}^+$, the functions ξ and $(u - u^{1.1})\phi$ are valid test functions in the integral stability inequality (3.1.16) with $\Omega = B_1^+$.

Since f is nondecreasing, we have $f' \ge 0$. By convexity $f(u) - f(u^{1,1}) \le f'(u)(u - u^{1,1})$, hence, multiplying by ξ , integrating, and using the stability inequality (3.1.16) twice, we obtain

$$\begin{split} &\int_{A_{4,5}^+} \left(f(u) - f(u^{1.1}) \right) \xi \, \mathrm{d}x \le \int_{A_{4,5}^+} f'(u)(u - u^{1.1}) \xi \, \mathrm{d}x \\ &\le \left(\int_{A_{4,5}^+} f'(u) \xi^2 \, \mathrm{d}x \right)^{1/2} \left(\int_{A_{3.9,5.1}^+} f'(u) \left((u - u^{1.1}) \phi \right)^2 \, \mathrm{d}x \right)^{1/2} \\ &\le \left(\int_{A_{4,5}^+} |\nabla \xi - \frac{1}{2} \xi A^{-1}(x) \widehat{b}(x)|_{A(x)}^2 \, \mathrm{d}x \right)^{1/2} \\ &\quad \cdot \left(\int_{A_{3.9,5.1}^+} |\nabla \{ (u - u^{1.1}) \phi \} - \frac{1}{2} (u - u^{1.1}) \phi A^{-1}(x) \widehat{b}(x) \big|_{A(x)}^2 \, \mathrm{d}x \right)^{1/2}. \end{split}$$
(3.3.16)

Using that $\|\xi\|_{C^1} + \|\phi\|_{C^1} \leq C$ and the coefficient bounds, from (3.3.16) it follows that

$$\int_{A_{4,5}^+} (f(u) - f(u^{1.1})) \xi \, dx
\leq C(1+\varepsilon) \|\nabla(u-u^{1.1})\|_{L^2(A_{3,9,5,1}^+)} + C(1+\varepsilon)^2 \|u-u^{1.1}\|_{L^2(A_{3,9,5,1}^+)}
\leq C_{\varepsilon_0} \|\nabla(u-u^{1.1})\|_{L^2(A_{3,9,5,1}^+)},$$
(3.3.17)

where in the last line we have applied the Poincaré inequality for functions vanishing

on $\partial^0 A^+_{3.9,5.1}$. It remains to control the norm $\|\nabla(u - u^{1.1})\|_{L^2(A^+_{3.9,5.1})}$ in (3.3.17). First we interpolate between L^1 and $L^{2+\gamma}$. Letting $q = \frac{2(1+\gamma)}{2+\gamma}$, we have

$$\|\nabla(u-u^{1,1})\|_{L^2(A_{3,9,5,1}^+)} \le C \|\nabla u\|_{L^{2+\gamma}(A_{3,9,5,1,1,1})}^{1/q} \|\nabla(u-u^{1,1})\|_{L^1(A_{3,9,5,1}^+)}^{1/q'}.$$
 (3.3.18)

From (3.3.18), by Corollary 3.3.1 we deduce

$$\begin{aligned} \|\nabla(u-u^{1,1})\|_{L^{2}(A^{+}_{3,9,5,1})} &\leq C_{\varepsilon_{0}} \|u\|_{L^{1}(A^{+}_{3,6})}^{1/q} \|\nabla(u-u^{1,1})\|_{L^{1}(A^{+}_{3,9,5,1})}^{1/q'} \\ &\leq \delta \|u\|_{L^{1}(A^{+}_{3,6})} + C_{\varepsilon_{0}} \delta^{-q'/q} \|\nabla(u-u^{1,1})\|_{L^{1}(A^{+}_{3,9,5,1})}. \end{aligned}$$
(3.3.19)

By Corollary C.2 (with $\tilde{\delta} = \delta^{1+q'/q}$), we also have

$$\begin{aligned} \|\nabla(u-u^{1.1})\|_{L^{1}(A^{+}_{3.9,5.1})} \\ &\leq C\delta^{1+\frac{q'}{q}} \|D^{2}(u-u^{1.1})\|_{L^{1}(A^{+}_{3.8,5.2})} + C\delta^{-1-\frac{q'}{q}} \|u-u^{1.1}\|_{L^{1}(A^{+}_{3.8,5.2})}. \end{aligned}$$
(3.3.20)

Hence, applying (3.3.20) in (3.3.19), we obtain the following estimate for the Dirichlet energy

$$\begin{aligned} \|\nabla(u-u^{1.1})\|_{L^{2}(A^{+}_{3.9,5.1})} \\ &\leq \delta \|u\|_{L^{1}(A^{+}_{3,6})} + C_{\varepsilon_{0}}\delta \|D^{2}u\|_{L^{1}(A^{+}_{3.8,5.8})} + C_{\varepsilon_{0}}\delta^{-1-2\frac{g'}{q}} \|u-u^{1.1}\|_{L^{1}(A^{+}_{3.8,5.2})}. \end{aligned}$$
(3.3.21)

Finally, since $\frac{\mathrm{d}}{\mathrm{d}\tau}u^{\tau}(x) = ru_r(\tau x)$, we have $u(x) - u^{1.1}(x) = -r \int_1^{1.1} u_r(\tau x) \,\mathrm{d}\tau$ and hence

$$\|u - u^{1.1}\|_{L^1(A_{3,8,5,2}^+)} \le C \|u_r\|_{L^1(A_{3,8,5,2\cdot 1,1}^+)} \le C \|u_r\|_{L^1(A_{3,6}^+)}.$$
(3.3.22)

Using (3.3.22) in (3.3.21), and by (3.3.17) we deduce the claim.

Step 3. Conclusion.

Combining Steps 1 and 2 in (3.3.4), for $\delta \in (0, 1)$ and (say) $\varepsilon \leq 1$, we have

$$\|u\|_{L^{1}(A^{+}_{4.7,4.8})} \leq C\delta \|u\|_{L^{1}(A^{+}_{3,6})} + C(\varepsilon + \delta) \|D^{2}u\|_{L^{1}(A^{+}_{3.8,5.8})} + C\varepsilon \|\nabla u\|_{L^{1}(A^{+}_{4,5.5})} + C\delta^{-1-2\frac{2+\gamma}{\gamma}} \|u_{r}\|_{L^{1}(A^{+}_{3,6})}.$$

$$(3.3.23)$$

Thanks to Corollary 3.3.1 we can control the Hessian and gradient errors in (3.3.23) by the L^1 norm of the function. Namely, we have $\|D^2u\|_{L^1(A^+_{3.8,5.8})} \leq C\|u\|_{L^1(A^+_{3,6})}$ and by Hölder's inequality $\|\nabla u\|_{L^1(A^+_{4,5.5})} \leq C\|\nabla u\|_{L^{2+\gamma}(A^+_{4,5.5})} \leq C\|u\|_{L^1(A^+_{3,6})}$. It follows that

$$\|u\|_{L^{1}(A^{+}_{4,7,4.8})} \le C\delta \|u\|_{L^{1}(A^{+}_{3,6})} + C\delta^{-1-2\frac{2+\gamma}{\gamma}} \|u_{r}\|_{L^{1}(A^{+}_{3,6})} \quad \text{for } \varepsilon \le \delta.$$
(3.3.24)

Now, proceeding as in [19], if we write the function in terms of the radial derivative as $u(s\sigma) = u(t\sigma) - \int_s^t u_r(r\sigma) \, dr$ for $s \in (3,6), t \in (4.7,4.8)$, and $\sigma \in \mathbb{S}^{n-1}$, then integrating in $\int_3^6 s^{n-1} \, ds \int_{\partial^+ B_1^+} d\sigma$ it is not hard to show that

$$\|u\|_{L^{1}(A_{3,6}^{+})} \leq C\left(\|u\|_{L^{1}(A_{4,7,4.8}^{+})} + \|u_{r}\|_{L^{1}(A_{3,6}^{+})}\right).$$
(3.3.25)

Combining (3.3.25) and (3.3.24), we deduce

$$\|u\|_{L^{1}(A_{3,6}^{+})} \leq C\delta \|u\|_{L^{1}(A_{3,6}^{+})} + C\delta^{-1-2\frac{2+\gamma}{\gamma}} \|u_{r}\|_{L^{1}(A_{3,6}^{+})} \quad \text{for } \varepsilon\delta,$$

and choosing $\delta > 0$ universal small in this last inequality, we can absorb the L^1 norm of u into the left-hand side, concluding the proof.

3.4 Boundary C^{α} estimate

We prove the Hölder estimate in half-balls. At this point, the proof amounts to combining Propositions 3.1.9 and 3.1.10 with Theorem 3.1.4, to deduce a decay of the weighted Dirichlet energy and applying a scaling and covering argument.

Proof of the Hölder estimate (3.1.9) in Theorem 3.1.2. We may assume that $3 \leq n \leq 9$. Indeed, when n = 2, we recover the estimate by applying Theorem 3.1.2 to the function $\tilde{u}(x_1, x_2, x_3) := u(x_2, x_3)$, a stable solution to the elliptic equation $c_0 \tilde{u}_{x_1x_1} + L\tilde{u} = f(\tilde{u})$ in $B_1^+ \subset \mathbb{R}^3$ where L acts only in the (x_2, x_3) variables. Similarly, when n = 1, one considers the function $\tilde{u}(x_1, x_2, x_3) := u(x_3)$.

Throughout the proof, C denotes a generic universal constant unless stated otherwise. The proof is divided in three steps. In Step 1, we prove the decay of the weighted Dirichlet energy under the assumption A(0) = I. Later, in Step 2, we remove this assumption and prove a C^{α} estimate in universally small balls. Finally, in Step 3, we deduce the theorem by a scaling an covering argument. **Step 1:** Under the assumption that

$$A(0) = I \quad and \quad \|DA\|_{L^{\infty}(B_{1}^{+})} + \|b\|_{L^{\infty}(B_{1}^{+})} \le \varepsilon,$$

we prove that if $\varepsilon \leq \varepsilon_0$, then

$$\int_{B_{\rho}^{+}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C \|u\|_{L^1(B_1^{+})}^2 \rho^{2\alpha} \quad \text{for all } \rho \le 1/16, \tag{3.4.1}$$

where $\varepsilon_0 > 0$, $\alpha > 0$, and C are universal constants.

First we write the weighted Dirichlet integral as an infinite sum on dyadic annuli, applying Corollary 3.3.1 on each annulus. We treat the case $\rho = 1/2$ and recover the result for general ρ by rescaling. This is the same approach used for the interior estimates in Chapter 1.

Let $r_j := 2^{-j}$ for $j \ge 0$. Then

$$\int_{B_{1/2}^+} r^{2-n} |\nabla u|^2 \,\mathrm{d}x = \sum_{j=0}^\infty \int_{A_{r_{j+2},r_{j+1}}^+} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C \sum_{j=0}^\infty r_j^{2-n} \int_{A_{r_{j+2},r_{j+1}}^+} |\nabla u|^2 \,\mathrm{d}x, \quad (3.4.2)$$

and hence, we must control each of the summands in (3.4.2).

Combining Corollary 3.3.1 and Proposition 3.1.10 applied to the functions $u(r_j \cdot)$ and by Hölder inequality, we have

$$r_{j}^{2-n} \int_{A_{r_{j+2},r_{j+1}}^{+}} |\nabla u|^{2} \,\mathrm{d}x \le C r_{j}^{2-n} \int_{A_{r_{j+3},r_{j}}^{+}} u_{r}^{2} \,\mathrm{d}x \quad \text{for } \varepsilon \le \varepsilon_{0}.$$
(3.4.3)

Therefore, using (3.4.3) in (3.4.2), we obtain

$$\int_{B_{1/2}^+} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C \sum_{j=0}^\infty r_j^{2-n} \int_{A_{r_{j+3},r_j}^+} u_r^2 \,\mathrm{d}x \le C \int_{B_1^+} r^{2-n} u_r^2 \,\mathrm{d}x \quad \text{for } \varepsilon \le \varepsilon_0.$$
(3.4.4)

Applying (3.4.4) to the functions $u(2\rho \cdot)$, we deduce

$$\int_{B_{\rho}^+} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C \int_{B_{2\rho}^+} r^{2-n} u_r^2 \,\mathrm{d}x \quad \text{ for all } \rho \le 1/2 \text{ and } \varepsilon \le \varepsilon_0,$$

and, by Proposition 3.1.9 (with 2ρ in place of ρ), it follows that

$$\int_{B_{\rho}^{+}} r^{2-n} |\nabla u|^{2} \,\mathrm{d}x \leq C \int_{B_{4\rho}^{+} \setminus B_{2\rho}^{+}} r^{2-n} |\nabla u|^{2} \,\mathrm{d}x + C\varepsilon \int_{B_{8\rho}^{+}} r^{3-n} |\nabla u|^{2} \,\mathrm{d}x$$
for all $\rho \leq 1/8$ and $\varepsilon \leq \varepsilon_{0}$.
$$(3.4.5)$$

Now, splitting the last integral into $B_{8\rho}^+ = (B_{8\rho}^+ \setminus B_{\rho}^+) \cup B_{\rho}^+$, since $r^{3-n} \leq r^{2-n}$ in B_1^+ , and using that the bounds $\rho \leq 1/8$ and $\varepsilon \leq \varepsilon_0$ are universal, from (3.4.5) we deduce

$$\int_{B_{\rho}^{+}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C \int_{B_{8\rho}^{+} \setminus B_{\rho}^{+}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x + C\varepsilon \int_{B_{\rho}^{+}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x$$
for all $\rho \le 1/8$ and $\varepsilon \le \varepsilon_0$.

Taking $\varepsilon_0 > 0$ universal smaller if necessary, we can absorb the last integral into the left-hand side, which yields

$$\int_{B_{\rho}^{+}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C \int_{B_{8\rho}^{+} \setminus B_{\rho}^{+}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \quad \text{for all } \rho \le 1/8 \text{ and } \varepsilon \le \varepsilon_0.$$
(3.4.6)

Hole-filling (3.4.6), we also have

$$\int_{B_{\rho}^{+}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le \theta \int_{B_{8\rho}^{+}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \quad \text{for all } \rho \le 1/8 \text{ and } \varepsilon \le \varepsilon_0, \tag{3.4.7}$$

where $\theta = \frac{C}{1+C} \in (0,1)$ is universal. Iterating (3.4.7), for $8^{-(k+1)} < \rho \le 8^{-k}$ we deduce

$$\int_{B_{\rho}^{+}} r^{2-n} |\nabla u|^{2} \, \mathrm{d}x \le \theta^{k} \int_{B_{8^{k}\rho}^{+}} r^{2-n} |\nabla u|^{2} \, \mathrm{d}x \le \frac{1}{\theta} \rho^{2\alpha} \int_{B_{1}^{+}} r^{2-n} |\nabla u|^{2} \, \mathrm{d}x,$$

with $\alpha = -\frac{1}{2}\log_8 \theta > 0$, and hence

$$\int_{B_{\rho}^{+}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C\rho^{2\alpha} \int_{B_1^{+}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \quad \text{for all } \rho \le 1/8 \text{ and } \varepsilon \le \varepsilon_0.$$
(3.4.8)

Finally, we can estimate the integral in the right-hand side of (3.4.8) by splitting $B_1^+ = (B_1^+ \setminus B_{1/8}^+) \cup B_{1/8}^+$ and applying (3.4.6) with $\rho = 1/8$ to bound the integral in the annulus, which results in

$$\int_{B_{\rho}^{+}} r^{2-n} |\nabla u|^2 \,\mathrm{d}x \le C\rho^{2\alpha} \|\nabla u\|_{L^2(B_1^{+})} \quad \text{for all } \rho \le 1/8 \text{ and } \varepsilon \le \varepsilon_0.$$
(3.4.9)

Applying the energy estimate (3.1.11) in Theorem 3.1.4 (rescaled) to (3.4.9) now yields the claim.

Step 2: Assuming

$$\|DA\|_{L^{\infty}(B_{1}^{+})} + \|b\|_{L^{\infty}(B_{1}^{+})} \le \varepsilon,$$

we prove that if $\varepsilon \leq \varepsilon_0$, then

$$\|u\|_{C^{\alpha}(\overline{B^{+}_{\rho_{0}}})} \leq C \|u\|_{L^{1}(B^{+}_{1})},$$

where $\varepsilon_0 > 0$, $\alpha > 0$, $\rho_0 > 0$, and C are universal.

Notice that for each $y \in \partial^0 B_1^+$ and each half-ball $B_d(y) \subset B_1^+$, there is a rotation matrix $R = R(y) \in SO(n)$ such that the function $u^{y,d}(x) := u\left(y + \frac{d}{\sqrt{C_0}}A^{1/2}(y)Rx\right)$ satisfies u = 0 on $\partial^0 B_1^+$, and is a stable solution to a semilinear equation in B_1^+ with coefficients

$$\begin{aligned} A^{y,d}(x) &:= R^T A^{-1/2}(y) A \Big(y + \frac{d}{\sqrt{C_0}} A^{1/2}(y) Rx \Big) A^{-1/2}(y) R, \\ b^{y,d}(x) &:= \frac{d}{\sqrt{C_0}} R^T A^{-1/2}(y) b \Big(y + \frac{d}{\sqrt{C_0}} A^{1/2}(y) Rx \Big). \end{aligned}$$

The new matrix is uniformly elliptic $\frac{c_0}{C_0} \leq A^{y,d} \leq \frac{C_0}{c_0}$ with $A^{y,d}(0) = I$, and the coefficients can be bounded by $\|DA^{y,d}\|_{L^{\infty}(B_1^+)} + \|b^{y,d}\|_{L^{\infty}(B_1^+)} \leq Cd\left(\|DA\|_{L^{\infty}(B_1^+)} + \|b\|_{L^{\infty}(B_1^+)}\right)$. Choosing d > 0 universal sufficiently small so that $Cd \leq 1$, we further have

$$\|DA^{y,d}\|_{L^{\infty}(B_{1}^{+})} + \|b^{y,d}\|_{L^{\infty}(B_{1}^{+})} \le \varepsilon \quad \text{ for all } y \in \partial^{0}B_{1-d}^{+}$$

Since $\varepsilon \leq \varepsilon_0$ (with $\varepsilon_0 > 0$ as in Step 1), by (3.4.1) it follows that

$$\int_{B_{\rho}^{+}} r^{2-n} |\nabla u^{y,d}|^2 \, \mathrm{d}x \le C \|u^{y,d}\|_{L^1(B_1^{+})}^2 \rho^{2\alpha} \quad \text{for } y \in \partial^0 B_{1-d}^{+} \text{ and } \rho \le 1/8,$$

and using that $r^{2-n} \ge \rho^{2-n}$ in B^+_{ρ} , we also have

$$\int_{B_{\rho}^{+}} |\nabla u^{y,d}|^2 \,\mathrm{d}x \le C \|u^{y,d}\|_{L^1(B_1^{+})}^2 \rho^{2\alpha+n-2} \quad \text{for } y \in \partial^0 B_{1-d}^+ \text{ and } \rho \le 1/8.$$
(3.4.10)

Let us express (3.4.10) in terms of the original function u. By the change of variables $z = y + \frac{d}{\sqrt{C_0}} A^{1/2}(y) Rx$ and by uniform ellipticity, using that $B^+_{\sqrt{c_0}\rho} \subset A^{1/2}(y) R(B^+_{\rho})$, the left-hand side of (3.4.10) can be bounded from below by

$$\int_{B_{\rho}^{+}} |\nabla u^{y,d}|^{2} \,\mathrm{d}x \ge c \,d^{2-n} \int_{B_{d\sqrt{\frac{c_{0}}{C_{0}}\rho}}(y)} |\nabla u|^{2} \,\mathrm{d}z,$$
(3.4.11)

where c > 0 is a universal constant. Similarly, we also have $\|u^{y,d}\|_{L^{1}(B_{1}^{+})} \leq Cd^{-n}\|u\|_{L^{1}(B_{1}^{+})}$ and hence, from (3.4.10) and (3.4.11) we deduce

$$\int_{B^+_{d\sqrt{\frac{c_0}{C_0}\rho}}(y)} |\nabla u|^2 \,\mathrm{d}z \le C d^{-2} \|u\|_{L^1(B^+_1)}^2 \rho^{n-2+2\alpha} \quad \text{for } y \in \partial^0 B^+_{1-d} \text{ and } \rho \le 1/8.$$
(3.4.12)

We let $\rho_0 := \frac{d}{16} \sqrt{\frac{c_0}{C_0}}$. Making *d* smaller if necessary, we may assume that $B_{2\rho_0}^+ \subset B_{1-d}^+$. Dividing ρ by $d\sqrt{\frac{c_0}{C_0}}$ in (3.4.12), using that *d* is universal, and by Cauchy-Schwarz we have

$$\int_{B_{\rho}^{+}(y)} |\nabla u| \, \mathrm{d}z \le C ||u||_{L^{1}(B_{1}^{+})} \rho^{n-1+\alpha} \quad \text{for } y \in \partial^{0} B_{2\rho_{0}}^{+} \text{ and } \rho \le 2\rho_{0}.$$
(3.4.13)

With (3.4.13) on hand, we are finally ready to prove the boundary Hölder estimate.

Let $x = (x', x_n) \in B^+_{\rho_0} \subset \mathbb{R}^n \times \mathbb{R}_+$. Since u = 0 on $\partial^0 B^+_1$, by the Poincaré inequality

$$\|u\|_{L^{1}(B^{+}_{2x_{n}}(x',0))} \leq Cx_{n} \|\nabla u\|_{L^{1}(B^{+}_{2x_{n}}(x',0))}.$$
(3.4.14)

Applying (3.4.13) with $\rho = 2x_n$ and y = (x', 0), from (3.4.14) we deduce

$$\|u\|_{L^{1}(B^{+}_{2x_{n}}(x',0))} \leq C \|u\|_{L^{1}(B^{+}_{1})}(x_{n})^{n+\alpha}.$$
(3.4.15)

By the interior Hölder estimates from Theorem 1.1.1 (see Chapter 1) applied in the ball $B_{x_n}(x) \subset B_1^+$, we have

$$\|u\|_{L^{\infty}(B_{x_n/2}(x))} + (x_n)^{\alpha} [u]_{C^{\alpha}(\overline{B}_{x_n/2}(x))} \le C \|u\|_{L^1(B_{x_n}(x))} (x_n)^{-n},$$
(3.4.16)

where $\alpha > 0$ and C are universal constants (since we are assuming a universal bound $\varepsilon \leq \varepsilon_0$ on the coefficients). Since $B_{x_n}(x) \subset B^+_{2x_n}(x', 0)$, combining (3.4.16) and (3.4.15)

$$\|u\|_{L^{\infty}(B_{x_n/2}(x))} + (x_n)^{\alpha} [u]_{C^{\alpha}(\overline{B}_{x_n/2}(x))} \le C \|u\|_{L^1(B_1^+)} (x_n)^{\alpha} \quad \text{for } x \in B_{\rho_0}^+.$$
(3.4.17)

In particular, from (3.4.17) it follows that $|u(x)| \leq C ||u||_{L^1(B_1^+)}(x_n)^{\alpha}$ in $B_{\rho_0}^+$, and we have controlled the L^{∞} norm of u in $B_{\rho_0}^+$. To bound the Hölder norm in $B_{\rho_0}^+$, consider

 $x, y \in B_{\rho_0}^+$ such that $x \neq y$. Without loss of generality we may assume $y_n \leq x_n$. On the one hand, if $|x - y| \leq x_n/2$, then from (3.4.17) we deduce

$$\frac{|u(x) - u(y)|}{|x - y|^{\alpha}} \le [u]_{C^{\alpha}(\overline{B}_{x_n/2}(x))} \le C ||u||_{L^1(B_1^+)}.$$

On the other hand, if $|x - y| > x_n/2$, then by the L^{∞} estimates in (3.4.17) it follows that

$$|u(x) - u(y)| \le u(x) + u(y) \le C ||u||_{L^{1}(B_{1}^{+})} ((x_{n})^{\alpha} + (y_{n})^{\alpha}) \le C ||u||_{L^{1}(B_{1}^{+})} (x_{n})^{\alpha} \le C ||u||_{L^{1}(B_{1}^{+})} ||x - y||^{\alpha}.$$

Combined, the two inequalities above yield a bound for $[u]_{C^{\alpha}(B^{+}_{oo})}$, which was the claim.

Step 3: Conclusion.

Arguing as in the proof of the higher integrability estimate in Chapter 2, by a scaling and covering argument and using the interior estimates from Chapter 1, is is not hard to deduce the theorem in its final form from Step 2.

3.5 Approximation and proof in $C^{1,1}$ domains

Here we give the complete proof of our main result, Theorem 3.1.1, which establishes a priori estimates in $C^{1,1}$ domains. By an approximation argument (carried out in the proof at the end of this section), it will suffice to prove this result in smooth domains.

First, we comment on the invariance of our class of solutions under general transformations flattening the boundary. This will allow us to reduce the problem in smooth domains to proving estimates in half-balls, which have already been obtained in Theorem 3.1.2 above.

Given a smooth bounded domain $\Omega \subset \mathbb{R}^n$, we can always write it as the superlevel set of a smooth function $\Phi \in C^{\infty}(\mathbb{R}^n)$, namely,

$$\Omega = \{ x \in \mathbb{R}^n \colon \Phi(x) > 0 \} = \{ \Phi > 0 \}.$$

Moreover, Φ can be chosen so that $\nabla \Phi \neq 0$ on $\partial \Omega$; see Appendix E.

Let $x_0 \in \partial \Omega$. Upon rotating the coordinate axes, we may assume that $\nabla \Phi(x_0) = \partial_n \Phi(x_0) e_n$, with $\partial_n \Phi(x_0) = |\nabla \Phi(x_0)| > 0$. Then, writing $x = (x', x_n) \in \mathbb{R}^{n-1} \times \mathbb{R}$, the map $\Psi(x) := ((x - x_0)', \frac{\Phi(x)}{\partial_n \Phi(x_0)})$ is a local diffeomorphism around x_0 which flattens out the boundary $\partial \Omega$. More precisely, by a quantitative version of the Inverse Function Theorem, we have the following:

Lemma 3.5.1. There are numbers $0 < R_1 < R_2$ and $\rho > 0$ depending only on $\|\nabla \Phi\|_{C^{0,1}(\mathbb{R}^n)}$ and $\||\nabla \Phi|^{-1}\|_{L^{\infty}(\partial\Omega)}$ such that

$$\Psi(B_{R_1}(x_0) \cap \Omega) \subset B^+_{\rho/2} \subset B^+_{\rho} \subset \Psi(B_{R_2}(x_0) \cap \Omega).$$
(3.5.1)

Proof. By a translation, we may assume that $x_0 = 0 \in \partial \Omega$. Since the map Ψ satisfies $\Psi(0) = 0$ and $D\Psi(0) = I$, choosing $R_2 > 0$ small such that

$$|D\Psi(x) - D\Psi(z)| \le 1/2$$
 for all $x, z \in B_{R_2}$, (3.5.2)

by Lemma 1.3 in [67, Chapter XIV] we deduce that for all $y \in B_{R_2/2}$ there is a unique $x \in B_{R_2}$ such that $\Psi(x) = y$. Thus, we obtain the second inclusion

$$B^+_{R_2/2} \subset \Psi(B_{R_2} \cap \Omega).$$

Using that $[D\Psi]_{C^{0,1}(\mathbb{R}^n)} \leq [\nabla\Phi]_{C^{0,1}(\mathbb{R}^n)} |||\nabla\Phi|^{-1}||_{L^{\infty}(\partial\Omega)}$, it is easy to check that condition (3.5.2) is fulfilled if

$$R_2 \le (4[\nabla\Phi]_{C^{0,1}(\mathbb{R}^n)} \| |\nabla\Phi|^{-1} \|_{L^{\infty}(\partial\Omega)})^{-1}.$$
(3.5.3)

To show the first inclusion in (3.5.1), we proceed as above but considering the inverse map Ψ^{-1} instead. If $R_1 > 0$ is sufficiently small such that

$$|D\Psi^{-1}(\widetilde{x}) - D\Psi^{-1}(\widetilde{z})| \le 1/2 \quad \text{for all } \widetilde{x}, \widetilde{z} \in B_{2R_1}, \tag{3.5.4}$$

then, again by Lemma 1.3 in [67, Chapter XIV], we deduce

$$\Psi(B_{R_1} \cap \Omega) \subset B_{2R_1}^+. \tag{3.5.5}$$

For (3.5.4) to hold in this case, it suffices to take $R_1 > 0$ sufficiently small such that

$$[D\Psi^{-1}]_{C^{0,1}(B_{2R_1})}4R_1 \le 1/2.$$
(3.5.6)

It remains to estimate $[D\Psi^{-1}]_{C^{0,1}(B_{2R_1})}$. Let $\tilde{x} = \Psi(x)$, $\tilde{y} = \Psi(y) \in B_{2R_1}$. If $R_1 \leq R_2$, with R_2 as in (3.5.3), then $\frac{1}{\partial_n \Phi(x)} \leq 2 ||\nabla \Phi|^{-1}||_{L^{\infty}(\partial\Omega)}$ and hence

$$\begin{split} |D\Psi^{-1}(\widetilde{x}) - D\Psi^{-1}(\widetilde{y})|^2 \\ &\leq 2 \frac{|\nabla'\Phi(x) - \nabla'\Phi(y)|^2}{|\partial_n\Phi(x)|^2} + \frac{(2|\nabla'\Phi(y)|^2 + |\partial_n\Phi(0)|^2) |\partial_n\Phi(x) - \partial_n\Phi(y)|^2}{|\partial_n\Phi(x)|^2 |\partial_n\Phi(y)|^2} \\ &\leq 8(1 + 6\|\nabla\Phi\|^2_{L^{\infty}(\mathbb{R}^n)} \||\nabla\Phi|^{-1}\|^2_{L^{\infty}(\partial\Omega)}) \||\nabla\Phi|^{-1}\|^2_{L^{\infty}(\partial\Omega)} [\nabla\Phi]^2_{C^{0,1}(\mathbb{R}^n)} |x - y|^2. \end{split}$$

Moreover, under the same assumption on R_1 , we have

$$|x - y| \le [D\Psi^{-1}]_{L^{\infty}(B_{2R_1})} |\widetilde{x} - \widetilde{y}| \le (1 + 8 \|\nabla\Phi\|_{L^{\infty}}^2 \||\nabla\Phi|^{-1}\|_{L^{\infty}(\partial\Omega)}^2)^{1/2} |\widetilde{x} - \widetilde{y}|.$$

Combining the last two bounds, it follows that if $R_1 \leq R_2$ is such that

$$R_{1} \leq \left(16\sqrt{2}\left(1+8\|\nabla\Phi\|_{L^{\infty}(\mathbb{R}^{n})}^{2}\||\nabla\Phi|^{-1}\|_{L^{\infty}(\partial\Omega)}^{2}\right)\||\nabla\Phi|^{-1}\|_{L^{\infty}(\partial\Omega)}[\nabla\Phi]_{C^{0,1}(\mathbb{R}^{n})}\right)^{-1}, \quad (3.5.7)$$

then (3.5.6) holds, and hence (3.5.5) also.

Finally, choosing R_1 , $R_2 > 0$ satisfying (3.5.3), (3.5.7), and $8R_1 \leq R_2$, we may take $\rho = 4R_1$, which concludes the proof of (3.5.1).

Consider now the operator $Lu(x) = a_{ij}(x)u_{ij} + b_i(x)u_i$ introduced in (3.1.2) acting on functions $u \in C^2(B_{R_2}(x_0) \cap \overline{\Omega})$, with R_2 as in Lemma 3.5.1. In the new coordinates $\widetilde{x} = \Psi(x)$, the function $\widetilde{u} = u \circ \Psi^{-1}$ satisfies

$$\widetilde{L}\widetilde{u} := (Lu)(\Psi^{-1}(\widetilde{x})) = \widetilde{a}_{ij}(\widetilde{x})\widetilde{u}_{ij} + \widetilde{b}_i(\widetilde{x})\widetilde{u}_i,$$

where the new coefficients are given by

$$\widetilde{a}_{ij} \circ \Psi(x) = a_{kl}(x)\partial_k \Psi_i(x)\partial_l \Psi_j(x)$$

and

$$b_i \circ \Psi(x) = b_k(x)\partial_k\Psi_i(x) + a_{jk}(x)\partial_{jk}^2\Psi_i(x)$$

If $0 < c_0 \leq A(x) \leq C_0$, then, taking $R_2 = R_2(\|\nabla \Phi\|_{C^{0,1}(\mathbb{R}^n)}, \||\nabla \Phi|^{-1}\|_{L^{\infty}(\partial\Omega)}) > 0$ smaller if necessary (uniformly in the norms of $\nabla \Phi$), the new matrix $\widetilde{A}(\widetilde{x}) = (\widetilde{a}_{ij}(\widetilde{x}))$ is uniformly elliptic with (say)

$$0 < \frac{1}{2}c_0 \le \widetilde{A}(\widetilde{x}) \le \frac{3}{2}C_0.$$

If u is a stable solution of -Lu = f(u) in $B_{R_2}(x_0) \cap \Omega$, then \tilde{u} is a stable solution of $-\tilde{L}\tilde{u} = f(\tilde{u})$ in $\Psi(B_{R_2}(x_0) \cap \Omega) \subset \mathbb{R}^n_+$. We have already shown that \tilde{u} solves this equation. To show that it is stable, notice that if $\varphi > 0$ is the function in the definition of stability (3.1.6), then $\tilde{\varphi} = \varphi \circ \Psi^{-1}$ satisfies the same condition with respect to \tilde{L} .

Finally, notice that the norms of the coefficients $||DA||_{L^{\infty}}$ and $||b||_{L^{\infty}}$ involve the norms $||DA||_{L^{\infty}}$, $||b||_{L^{\infty}}$, $||D\Phi||_{L^{\infty}}$, and $||D^2\Phi||_{L^{\infty}}$. This dependence will be crucial to extend our results to $C^{1,1}$ domains, which are given by a $C^{1,1} = W^{2,\infty}$ function Φ .

Now, thanks to all the preliminaries, we can finally upgrade our estimates in half-balls to $C^{1,1}$ domains by an approximation argument:

Proof of Theorem 3.1.1. We proceed in two steps. First, we prove the theorem in smooth domains for solutions in $u \in W^{3,p}(\Omega)$, that is, with weak derivatives integrable up to the boundary. Then, we approximate our $C^{1,1}$ domain from the interior by smooth domains and apply the first step on a suitable sequence of stable solutions.

As commented above (see also Appendix E), recall that given a bounded domain $\Omega \subset \mathbb{R}^n$ of class $C^{1,1}$, there is a function $\Phi \in C^{1,1}(\mathbb{R}^n)$ such that $\Omega = \{\Phi > 0\}$ and $\nabla \Phi \neq 0$ on $\partial \Omega$. If Ω is smooth, then the function Φ can be chosen to be C^{∞} . The main purpose of the function Φ is to quantify the dependence of our bounds on the domain. In the proof below, diam(Ω) denotes the diameter of Ω , that is, diam(Ω) = $\sup_{x, u \in \Omega} |x - y|$.

Step 1: Under the additional assumptions that Ω is smooth, L satisfies (3.1.5), and $u \in W^{3,p}(\Omega)$ for some p > n, we prove that

$$\|\nabla u\|_{L^{2+\gamma}(\Omega)} \le C \|u\|_{L^1(\Omega)},$$

where $\gamma > 0$ is dimensional and C is a constant depending only on n, c_0 , C_0 , $||DA||_{L^{\infty}(\Omega)}$, $||b||_{L^{\infty}(\Omega)}$, $||\nabla\Phi||_{C^{0,1}(\mathbb{R}^n)}$, $|||\nabla\Phi|^{-1}||_{L^{\infty}(\partial\Omega)}$, and diam(Ω). In addition,

$$||u||_{C^{\alpha}(\overline{\Omega})} \le C||u||_{L^{1}(\Omega)} \quad \text{if } n \le 9,$$

where $\alpha > 0$ is universal and C is a constant depending only on n, c_0 , C_0 , $||DA||_{L^{\infty}(\Omega)}$, $||b||_{L^{\infty}(\Omega)}$, $||\nabla \Phi||_{C^{0,1}(\mathbb{R}^n)}$, and $|||\nabla \Phi|^{-1}||_{L^{\infty}(\partial\Omega)}$.

Let $0 < R_1 < R_2$ be the functions of $\|\nabla \Phi\|_{C^{0,1}(\mathbb{R}^n)}$ and $\||\nabla \Phi|^{-1}\|_{L^{\infty}(\partial\Omega)}$ constructed in Lemma 3.5.1 above.

Let $\delta := R_1/3 > 0$. Since Ω is bounded, it is contained in a ball of radius diam $(\Omega) < \infty$. Hence, we can cover $\overline{\Omega}$ by N balls $\{B_i\}_i$ of radius δ , where $N \leq C \operatorname{diam}(\Omega)^n \delta^{-n}$ for some

¹Indeed, since $\widetilde{A}(\widetilde{x})p \cdot p = A(x)D\Psi(x)p \cdot D\Psi(x)p$, by ellipticity $c_0|D\Psi(x)p|^2 \leq \widetilde{A}(\widetilde{x})p \cdot p \leq C_0|D\Psi(x)p|^2$. Moreover, since $|D\Psi(x)p|^2 = |p'|^2 + |\frac{\nabla'\Phi(x)}{\partial_n\Phi(x_0)} \cdot p' + \frac{\partial_n\Phi(x)}{\partial_n\Phi(x_0)}p_n|^2$, choosing $R_2 > 0$ smaller (as before, with the stated dependence) such that $\frac{9}{10} \leq \frac{\partial_n\Phi(x)}{\partial_n\Phi(x_0)} \leq \frac{|\nabla\Phi(x)|}{\partial\Phi(x_0)} \leq \frac{11}{10}$ and $\frac{|\nabla'\Phi(x)|}{\partial\Phi(x_0)} \leq \frac{1}{100}$, it is easy to check that $\frac{1}{2}|p|^2 \leq |D\Psi(x)p|^2 \leq \frac{3}{2}|p|^2$ as claimed.

dimensional C^2 We write $2B_i$ to denote the ball centered at the same point but with twice the radius. We label the balls B_i in a way such that the first N' < N are close to the boundary, in the sense that $2B_i \cap \partial \Omega \neq \emptyset$, and the remaining N - N' are interior, i.e., they satisfy the inclusion $2B_i \subset \Omega$.

For each $i \leq N'$, by definition, there is a boundary point $x_i \in 2B_i \cap \partial\Omega$ and hence $B_i \subset B_{3\delta}(x_i) = B_{R_1}(x_i)$. In particular, flattening the boundary as explained above and applying the energy estimate (3.1.11) in Theorem 3.1.2 rescaled, we deduce

$$\|\nabla u\|_{L^{2+\gamma}(B_i\cap\Omega)} \le \|\nabla u\|_{L^{2+\gamma}(B_{R_1}(x_i)\cap\Omega)} \le C\|u\|_{L^1(B_{R_2}(x_i)\cap\Omega)} \quad \text{for all } i \le N', \qquad (3.5.8)$$

where $C = C(n, c_0, C_0, ||DA||_{L^{\infty}(\Omega)}, ||b||_{L^{\infty}(\Omega)}, ||\nabla \Phi||_{C^{0,1}(\mathbb{R}^n)}, |||\nabla \Phi|^{-1}||_{L^{\infty}(\partial\Omega)})$. Therefore, by (3.5.8) and interior estimates (Theorem 1.1.1 above), we deduce

$$\begin{aligned} \|\nabla u\|_{L^{2+\gamma}(\Omega)} &\leq \sum_{i \leq N'} \|\nabla u\|_{L^{2+\gamma}(B_i \cap \Omega)} + \sum_{i > N'} \|\nabla u\|_{L^{2+\gamma}(B_i)} \\ &\leq C \sum_{i \leq N'} \|u\|_{L^1(B_{R_2}(x_i) \cap \Omega)} + C \sum_{i > N'} \|u\|_{L^1(2B_i)} \\ &\leq C \|u\|_{L^1(\Omega)}, \end{aligned}$$

where C depends only on n, c_0 , C_0 , $\|DA\|_{L^{\infty}(\Omega)}$, $\|b\|_{L^{\infty}(\Omega)}$, $\|\nabla\Phi\|_{C^{0,1}(\mathbb{R}^n)}$, $\||\nabla\Phi|^{-1}\|_{L^{\infty}(\partial\Omega)}$, and diam(Ω). The first claim follows.

Assume now that $n \leq 9$.

We prove the L^{∞} estimate first. Let $x \in \Omega$. If $\operatorname{dist}(x, \partial \Omega) < R_1$, then $x \in B_{R_1}(x_0) \cap \Omega$ for some $x_0 \in \partial \Omega$. Hence, flattening the boundary, by Theorem 3.1.2 we deduce

$$|u(x)| \le ||u||_{L^{\infty}(B_{R_1}(x_0)\cap\Omega)} \le C ||u||_{L^1(B_{R_2}\cap\Omega)} \le C ||u||_{L^1(\Omega)},$$

where $C = C(n, c_0, C_0, ||DA||_{L^{\infty}(\Omega)}, ||b||_{L^{\infty}(\Omega)}, ||\nabla\Phi||_{C^{0,1}(\mathbb{R}^n)}, |||\nabla\Phi|^{-1}||_{L^{\infty}(\partial\Omega)})$. Otherwise, if dist $(x, \partial\Omega) \ge R_1$, then by interior estimates (rescaled) we have

$$|u(x)| \le ||u||_{L^{\infty}(B_{R_1/2}(x))} \le C ||u||_{L^1(B_{R_1}(x))} \le C ||u||_{L^1(\Omega)},$$

where again $C = C(n, c_0, C_0, \|DA\|_{L^{\infty}(\Omega)}, \|b\|_{L^{\infty}(\Omega)}, \|\nabla\Phi\|_{C^{0,1}(\mathbb{R}^n)}, \||\nabla\Phi|^{-1}\|_{L^{\infty}(\partial\Omega)})$. The two inequalities yield the desired L^{∞} bound $\|u\|_{L^{\infty}(\Omega)} \leq C \|u\|_{L^{1}(\Omega)}$.

Next, we estimate the C^{α} seminorm. Let $x, y \in \Omega$ with $x \neq y$ and let $\delta := 2R_1/3 > 0$. We distinguish three cases:

• If
$$|x - y| \ge \delta/2$$
, then

$$\frac{|u(x) - u(y)|}{|x - y|^{\alpha}} \le 2^{\alpha} \widetilde{\delta}^{-\alpha} \left(|u(x)| + |u(y)| \right) \le 2^{\alpha + 1} \widetilde{\delta}^{-\alpha} ||u||_{L^{\infty}(\Omega)}$$

and we may apply the L^{∞} estimate $||u||_{L^{\infty}(\Omega)} \leq C ||u||_{L^{1}(\Omega)}$.

²To see this, given $k \in \mathbb{N}$ and R > 0, consider the set $A_{k,R} = \{Rl/k : l \in \mathbb{Z} \text{ with } -k \leq l \leq k\}^n \subset \mathbb{R}^n$. Notice that $A_{k,R}$ is a discrete set with $N = (2k+1)^n$ elements. For each $x \in B_R$, there is a $y \in A_{k,R}$ such that $|x - y| \leq \sqrt{n}\frac{R}{2k}$. Hence, if $\sqrt{n}\frac{R}{2\delta} < k \leq \sqrt{n}\frac{R}{2\delta} + 1$, then $B_R \subset \bigcup_{y \in A_{k,R}} B_{\delta}(y)$ and the number of balls can be estimated by $N \leq (\sqrt{n}\frac{R}{\delta} + 3)^n \leq (2\sqrt{n})^n \left(\frac{R}{\delta}\right)^n$ by taking δ smaller in terms of R and n.

• If $|x-y| < \tilde{\delta}/2$ and (say) dist $(x, \partial \Omega) < \tilde{\delta}$, then $x, y \in B_{\tilde{\delta}/2}(x) \subset B_{\frac{3}{2}\tilde{\delta}}(x_0) \subset B_{R_1}(x_0)$ for some $x_0 \in \partial \Omega$. Hence, flattening the boundary, by Theorem 3.1.2 we deduce

$$\frac{|u(x) - u(y)|}{|x - y|^{\alpha}} \le [u]_{C^{\alpha}(\overline{B_{\delta/2}(x) \cap \Omega})} \le [u]_{C^{\alpha}(\overline{B_{R_1}(x_0) \cap \Omega})} \le C ||u||_{L^1(B_{R_2}(x_0) \cap \Omega)}$$
$$\le C ||u||_{L^1(\Omega)},$$

where C has the same dependence as above.

• If $|x-y| < \tilde{\delta}/2$ and $\min\{\operatorname{dist}(x,\partial\Omega), \operatorname{dist}(y,\partial\Omega)\} \ge \tilde{\delta}$, then $x, y \in B_{\tilde{\delta}/2}(x) \subset B_{\tilde{\delta}}(x) \subset \Omega$ and by interior estimates (rescaled) we deduce

$$\frac{|u(x) - u(y)|}{|x - y|^{\alpha}} \le [u]_{C^{\alpha}(\overline{B}_{\tilde{\delta}/2}(x))} \le C ||u||_{L^{1}(B_{\bar{\delta}}(x))} \le C ||u||_{L^{1}(\Omega)},$$

where C has the same dependence as above.

The three inequalities yield the bound

$$[u]_{C^{\alpha}(\overline{\Omega})} \le C \|u\|_{L^{1}(\Omega)},$$

where $C = C(n, c_0, C_0, \|DA\|_{L^{\infty}(\Omega)}, \|b\|_{L^{\infty}(\Omega)}, \|\nabla\Phi\|_{C^{0,1}(\mathbb{R}^n)}, \||\nabla\Phi|^{-1}\|_{L^{\infty}(\partial\Omega)})$. This concludes the proof of Step 1.

Step 2: Conclusion: Approximation argument.

Let $\Omega_k = \{\Phi_k > 0\}$ be an exhaustion of $\Omega = \{\Phi > 0\}$ by C^{∞} sets, with norms satisfying

$$\|\nabla \Phi_k\|_{C^{0,1}(\mathbb{R}^n)} + \||\nabla \Phi_k|^{-1}\|_{L^{\infty}(\partial \Omega_k)} \le C$$
(3.5.9)

for some constant C depending only on Φ and Ω ; see Appendix E.

For each k, let $b_i^k := b_i * \eta_k$, where $(\eta_k)_k$ is a regularizing sequence such that $b_i^k \in C^{\infty}(\overline{\Omega_k})$. In particular, $b_i^k \to b_i$ locally uniformly in Ω . We define the operator

$$L_k := a_{ij}(x)\partial_{ij} + b_i^k(x)\partial_i,$$

where $a_{ij} \in C^{0,1}(\overline{\Omega})$ is the same coefficient as in the statement of the theorem. By elliptic regularity, all bounded strong solutions of the problem $-L_k u_k = f(u_k)$ in Ω_k , $u_k = 0$ on $\partial \Omega_k$, belong to $W^{3,p}(\Omega_k)$.

We will distinguish the two cases f(0) > 0 and f(0) = 0.

Case f(0) > 0. Let $\varepsilon_k \in (0, 1)$ with $\varepsilon_k \downarrow 0$. For each k, we will construct a stable solution $u_k \in W^{3,p}(\Omega_k)$ to the problem

$$\begin{cases} -L_k u_k = (1 - \varepsilon_k) f(u_k) & \text{in } \Omega_k \\ u_k > 0 & \text{in } \Omega_k \\ u_k = 0 & \text{on } \partial \Omega_k \end{cases}$$
(3.5.10)

by monotone iteration starting at 0. This is the so called *minimal solution* of (3.5.10), that is, the smallest positive supersolution of (3.5.10), and it is well-known to be stable.³

³To prove that it is stable one argues by contradiction, considering ϕ , the principal eigenfunction of $L + (1 - \varepsilon_k) f'(u_k)$ in Ω_k , and showing that $u_k - \delta \phi$ would be a positive supersolution for $\delta > 0$ sufficiently small; see [8].

For the monotone iteration to converge, we need a barrier function. We claim that $u \in C^0(\overline{\Omega}) \cap W^{2,n}_{\text{loc}}(\Omega)$ is a barrier for (3.5.10), in fact, we have u > 0 on $\overline{\Omega_k} \subset \Omega$ and

$$-L_k u \ge (1 - \varepsilon_k) f(u)$$
 in Ω_k

Indeed, using the equation satisfied by u and by monotonicity of f, we have

$$-L_k u - (1 - \varepsilon_k) f(u) = \varepsilon_k f(u) + (L - L_k) u \ge \varepsilon_k f(0) - \|b - b^k\|_{L^{\infty}(\Omega_k)} \|\nabla u\|_{L^{\infty}(\Omega_k)},$$

and the right-hand side is nonnegative by choosing the regularizing sequence η_k in terms of ε_k , f(0), and $\|\nabla u_k\|_{L^{\infty}(\Omega_k)}$ so that $\|b - b^k\|_{L^{\infty}(\Omega_k)}$ is sufficiently small. Here, recall that by L^p estimates we have $u \in C^0(\overline{\Omega}) \cap W^{2,p}_{\text{loc}}(\Omega)$ for all $p < \infty$, and hence $u \in C^{1,\alpha}(\overline{\Omega_k})$ for all $\alpha \in (0,1)$ and $k \in \mathbb{N}$.

Next, we carry out the monotone iteration. Let $k \in \mathbb{N}$ and $u_k^{(0)} = 0$. For each $l \in \mathbb{N}$, we consider the unique solution $u_k^{(l)}$ to the problem $-L_k u_k^{(l)} = (1 - \varepsilon_k) f(u_k^{(l-1)})$ in Ω_k , $u_k^{(l)} = 0$ on $\partial \Omega_k$. Since u is a barrier, by maximum principle, it is not hard to show that $0 \le u_k^{(l-1)} \le u_k^{(l)} \le u$ in Ω_k , and by global regularity the monotone limit $u_k := \lim_{l \to \infty} u_k^{(l)}$ converges uniformly in $C^2(\overline{\Omega_k})$ norm and solves (3.5.10). By construction, u_k is below any supersolution of (3.5.10) and hence it is the minimal solution.

Since $\overline{\Omega_k} \subset \Omega_{k+1} \subset \Omega$, by maximum principle we have

$$0 \le u_k \le u_{k+1} \le u \quad \text{in } \Omega_k.$$

Let $u^*(x) := \lim_{k\to\infty} u_k(x)$ for $x \in \Omega$. Using that $u \in C^0(\overline{\Omega})$, by L^p estimates we have $u_k \to u^*$ weakly in $W^{2,p}_{\text{loc}}(\Omega)$ for all $p < \infty$. Since $u^* \leq u$ in Ω , we can extend u^* up to the boundary to a function $u^* \in C^0(\overline{\Omega}) \cap W^{2,p}_{\text{loc}}(\Omega)$. By weak convergence, it follows that u^* is a strong solution to $-Lu^* = f(u^*)$ in Ω , $u^* = 0$. Moreover, u^* is a stable solution. To see this, taking a positive function $\varphi \in W^{2,n}_{\text{loc}}(\Omega)$ as in the stability inequality (3.1.6) for u, using that $u^* \leq u$ and by the convexity of f, we have

$$J_{u^{\star}}\varphi = (L + f'(u^{\star}))\varphi \le (L + f'(u))\varphi = J_{u}\varphi \le 0 \quad \text{in } \Omega.$$

Finally, by the uniqueness of stable solutions for convex nonlinearities (see Appendix F), it follows that $u = u^*$.

Applying Step 1 to the minimal solutions u_k , by the bounds $||b^k||_{L^{\infty}(\Omega_k)} \leq ||b||_{L^{\infty}(\Omega)}$ and (3.5.9), and by the monotonicity of the sequence, for $k \geq l+1$ we obtain

$$\|\nabla u_k\|_{L^{2+\gamma}(\Omega_l)} \le \|\nabla u_k\|_{L^{2+\gamma}(\Omega_k)} \le C \|u_k\|_{L^1(\Omega_k)} \le C \|u\|_{L^1(\Omega)}.$$

Now, since $\nabla u_k \to \nabla u$ uniformly on compacts, letting $k \to \infty$ and $l \to \infty$ in this last estimate and by monotone convergence, we deduce the higher integrability in $C^{1,1}$ domains.

Assuming moreover that $n \leq 9$, by Step 1 applied to u_k , for $k \geq l+1$ we have

$$\|u_k\|_{C^{\alpha}(\overline{\Omega_l})} \le \|u_k\|_{C^{\alpha}(\overline{\Omega_k})} \le C\|u_k\|_{L^1(\Omega_k)} \le C\|u^{\star}\|_{L^1(\Omega)} = C\|u\|_{L^1(\Omega)}.$$

Hence, letting $k \to \infty$ and $l \to \infty$, we deduce the Hölder estimate in $C^{1,1}$ domains.

Case f(0) = 0. Without loss of generality, we may assume that u > 0 in Ω . Since 0 is a stable solution, by Proposition F.1 we deduce that $f(u) = \mu_1[L,\Omega]u$,⁴ and hence u is a principal eigenfunction of L.

⁴Here we are following the notation in Appendix F, namely, $\mu_1[L,\Omega]$ denotes the principal eigenvalue of L in Ω with the sign convention $L\varphi = -\mu\varphi$.

Since $a_{ij} \in C(\overline{\Omega_k})$ and $b_i^k \in L^{\infty}(\Omega_k)$, by standard existence theory, there is a principal eigenvalue $\mu_k := \mu_1[L_k, \Omega_k]$ and eigenfunction $\varphi_k \in W^{2,p}(\Omega_k)$ for all $p < \infty$, satisfying $\varphi_k > 0$ in Ω_k , $\|\varphi_k\|_{L^1(\Omega_k)} = \|u\|_{L^1(\Omega)}$, $-L_k\varphi_k = \mu_k\varphi_k$ in Ω_k , and $\varphi_k = 0$ on $\partial\Omega_k$. Moreover, recalling that $a_{ij} \in C^{0,1}(\overline{\Omega_k})$ and $b_i^k \in C^{\infty}(\overline{\Omega_k})$, we further have $\varphi_k \in W^{3,p}(\Omega_k)$ for $p < \infty$. In particular, since φ_k are stable, by Step 1 we have the bounds

$$\|\nabla \varphi_k\|_{L^{2+\gamma}(\Omega_k)} \le C \|u\|_{L^1(\Omega)}, \tag{3.5.11}$$

and

$$\|\varphi_k\|_{C^{\alpha}(\overline{\Omega_k})} \le C \|u\|_{L^1(\Omega)} \quad \text{if } n \le 9.$$
(3.5.12)

To deduce the final estimates it suffices to extract a subsequence converging to u. For this, we essentially follow the proof of Theorem 2.1 in [9]. Namely, by Harnack inequality, for $k \geq l+1$ we have $\|\varphi_k\|_{L^{\infty}(\Omega_l)} \leq C_l \inf_{\Omega_l} \varphi_k \leq C_l \|\varphi_k\|_{L^1(\Omega_l)} \leq C_l \|u\|_{L^1(\Omega)}$, hence, by interior estimates (up to a subsequence) $\varphi_k \to \varphi$ weakly in $W_{\text{loc}}^{2,p}(\Omega)$ for some positive $\varphi \in$ $W_{\text{loc}}^{2,p}(\Omega)$. Moreover, since $\varphi_k \in W_0^{1,2+\gamma}(\Omega_k)$, the extension $\varphi_k \chi_{\Omega_k}$ is bounded in $W_0^{1,2+\gamma}(\Omega)$ and by compactness $\varphi_k \chi_{\Omega_k} \to \varphi$ weakly in $W_0^{1,2+\gamma}(\Omega)$, hence strongly in $L^{2+\gamma}(\Omega)$. In particular, by strong convergence $\|\varphi\|_{L^1(\Omega)} = \lim_k \|\varphi_k\|_{L^1(\Omega_k)} = \|u\|_{L^1(\Omega)}$ and by weak lower semicontinuity, from (3.5.11), we deduce

$$\|\nabla\varphi\|_{L^{2+\gamma}(\Omega)} \le C \|u\|_{L^1(\Omega)}.$$
 (3.5.13)

By (3.5.12), using that $\varphi_k \to \varphi$ converges locally uniformly in Ω , it is also clear that

$$\|\varphi\|_{C^{\alpha}(\overline{\Omega})} \le C \|u\|_{L^{1}(\Omega)} \quad \text{if } n \le 9.$$

$$(3.5.14)$$

Passing to the limit in the equation, we see that $\varphi \in W_0^{1,2+\gamma}(\Omega) \cap W_{\text{loc}}^{2,p}(\Omega)$ solves $-L\varphi = \mu^*\varphi$ where $\mu^* = \lim_k \mu_1[L_k, \Omega_k]$. In fact, $\mu^* = \mu_1[L, \Omega]$ by the characterization of the principal eigenfunction and the maximum principle.

It follows that both u and φ are positive principal eigenfunctions of L in Ω , with $\|u\|_{L^1(\Omega)} = \|\varphi\|_{L^1(\Omega)}$, hence $u = \varphi$ and (3.5.13), (3.5.14) are already the claimed estimates.

Appendix A

Stability is not equivalent to the integral inequality

Let $u \in C^2(\overline{\Omega})$ be a solution to -Lu = f(u) in Ω with u = 0 on $\partial\Omega$. Recall that u is a stable solution if

$$J_u \varphi = L \varphi + f'(u) \varphi \le 0 \quad \text{in } \Omega, \tag{A.1}$$

for some function $\varphi \in C^2(\overline{\Omega})$ with $\varphi > 0$ in Ω and $\varphi = 0$ on $\partial\Omega$. This is the stability condition (1.1.3) presented in the Introduction and is equivalent to the nonnegativity of the first Dirichlet eigenvalue of J_u (with the sign convention $J_u\varphi = -\mu\varphi$). There, we also showed that stable solutions satisfy the integral inequality (1.1.8), which reads

$$\int_{\Omega} f'(u)\xi^2 \,\mathrm{d}x \le \int_{\Omega} |\nabla\xi - \frac{1}{2}A^{-1}(x)\widehat{b}(x)\xi|^2_{A(x)} \quad \text{for all } \xi \in C^{\infty}_c(\Omega). \tag{A.2}$$

Our goal in this appendix is to show that the integral inequality (A.2) does not imply the stability condition (A.1) in general. The main reason is that the problem is not variational, due to the drift in L. We also give conditions under which the equivalence holds. Namely, writing the operator in divergence form $Lu = \operatorname{div}(A(x)\nabla u) + \hat{b}(x) \cdot \nabla u$, we show that if $A^{-1}(x)\hat{b}(x)$ is the gradient of a scalar function, then the problem is variational and the two conditions are equivalent.

First we write the integrals in (A.2) as the quadratic form associated to a linear selfadjoint operator. Integrating by parts, we have

$$\int_{\Omega} \left(|\nabla \xi - \frac{1}{2} \xi A^{-1}(x) \widehat{b}(x)|_{A(x)}^2 - f'(u) \xi^2 \right) \mathrm{d}x = -\int_{\Omega} \xi \widetilde{J}_u \xi \,\mathrm{d}x, \tag{A.3}$$

where \widetilde{J}_u is the operator

$$\widetilde{J}_{u}\xi := \operatorname{div}(A(x)\nabla\xi) - \left\{ \frac{1}{2}\operatorname{div}(\widehat{b}(x)) + \frac{1}{4}|\widehat{b}(x)|_{A^{-1}(x)}^{2} \right\} \xi + f'(u)\xi.$$
(A.4)

Hence, by the variational characterization of eigenvalues, (A.2) amounts to the nonnegativity of the principal eigenvalue of \tilde{J}_u .

We can now state our example of a solution satisfying (A.2) but not (A.1):

Example A.1. Consider the operator $Lv = \Delta v + \hat{b}(x) \cdot \nabla v$ with vector field

$$\widehat{b}(x) = \frac{-x_2 e_1 + x_1 e_2}{\sqrt{x_1^2 + x_2^2}}.$$

For each constant c > 0, we let $f(u) = (\lambda_1 + c)u + 1$, where λ_1 denotes the least Dirichlet eigenvalue of the Laplacian in the unit ball B_1 .

If c > 0 is sufficiently small, then the unique solution u to the boundary value problem

$$\begin{cases} -Lu = f(u) & \text{in } B_1 \\ u = 0 & \text{on } \partial B_1 \end{cases}$$

satisfies the integral stability condition (A.2) but is not a stable solution, i.e., the stability condition (A.1) does not hold.¹

Proof. The problem for u is equivalent to solving

$$\begin{cases} -\Delta u - \hat{b}(x) \cdot \nabla u - (\lambda_1 + c)u = 1 & \text{in } B_1 \\ u = 0 & \text{on } \partial B_1. \end{cases}$$
(A.5)

Notice that the drift $\hat{b} \in L^{\infty}(B_1)$ has a weak derivative $D\hat{b} \in L^p(B_1)$ for $1 \leq p < 2$, and satisfies the identities $|\hat{b}(x)| = 1$ and $\operatorname{div} \hat{b}(x) = 0$ for a.e. $x \in B_1$. Moreover, since \hat{b} is tangent to spheres, the derivative $\hat{b}(x) \cdot \nabla$ vanishes on radial functions. In particular, the principal eigenfunction of the Laplacian is also an eigenfunction of the adjoint operator $L^T = \Delta - \operatorname{div}(\hat{b}(x) \cdot) = \Delta - \hat{b}(x) \cdot \nabla$, with eigenvalue λ_1 . Since the point spectrum of L^T is discrete, for c > 0 small, we deduce that $\lambda_1 + c$ is not an eigenvalue of the adjoint operator. The Fredholm alternative now gives that (A.5) has a unique solution.

Let φ_1 and ξ_1 be positive principal eigenfunctions of J_u and \tilde{J}_u , respectively. Since φ_1 and ξ_1 are positive in B_1 , they must be radial. It follows that

$$J_u\varphi_1 = \Delta\varphi_1 + (\lambda_1 + c)\varphi_1 = -\mu_1\varphi_1 \quad \text{and} \quad \widetilde{J}_u\xi_1 = \Delta\xi_1 + (\lambda_1 + c - 1/4)\xi_1 = -\widetilde{\mu_1}\xi_1,$$

where μ_1 and $\tilde{\mu_1}$ are the least eigenvalues of each operator. By uniqueness, the functions are multiples of the principal eigenfunction of the Laplacian. Therefore, we have $\mu_1 = -c < 0$ and $\tilde{\mu_1} = 1/4 - c > 0$ for c sufficiently small. This means that u is not stable but (A.2) holds, which was the claim.

Next we investigate the relation between the failure of the equivalence and the form of the drift \hat{b} . Let $\varphi_1 \in C^2(\overline{\Omega})$ be the unique positive principal eigenfunction of J_u with $\int \varphi_1^2 dx = 1$. In particular, the function satisfies $\varphi_1 > 0$ in Ω , $\varphi_1 = 0$ on $\partial\Omega$, and $J_u \varphi_1 = -\mu_1 \varphi_1$, where $\mu_1 \in \mathbb{R}$ is the least eigenvalue of J_u . Consider a test function $\xi \in C_c^{\infty}(\Omega)$. Multiplying $J_u \varphi_1$ by ξ^2 / φ_1 and integrating by parts in Ω , we have

$$-\mu_{1} = \int_{\Omega} (J_{u}\varphi_{1}) \frac{\xi^{2}}{\varphi_{1}} dx = \int_{\Omega} \left(-A(x)\nabla\varphi_{1} \cdot \nabla\left(\frac{\xi^{2}}{\varphi_{1}}\right) + \widehat{b}(x) \cdot \frac{\xi^{2}}{\varphi_{1}}\nabla\varphi_{1} + f'(u)\xi^{2} \right) dx$$
$$= \int_{\Omega} \left(|\xi\nabla\log\varphi_{1}|^{2}_{A(x)} - 2A(x)\xi\nabla\log\varphi_{1} \cdot \nabla\xi + \xi\widehat{b}(x) \cdot \xi\nabla\log\varphi_{1} \right) dx$$
$$+ \int_{\Omega} f'(u)\xi^{2} dx.$$

¹The function u can be given explicitly in terms of Bessel functions of the first kind \mathcal{J}_{α} as

$$u(x) = \frac{1}{(\lambda_1 + c)\mathcal{J}_{\frac{n-2}{2}}(\sqrt{\lambda_1 + c})} |x|^{\frac{2-n}{2}} \mathcal{J}_{\frac{n-2}{2}}\left(\sqrt{\lambda_1 + c} |x|\right) - \frac{1}{\lambda_1 + c}.$$

Using that

$$\begin{aligned} |\xi \nabla \log \varphi_1|^2_{A(x)} &- 2A(x)\xi \nabla \log \varphi_1 \cdot \left(\nabla \xi - \frac{1}{2}\xi A^{-1}(x)\widehat{b}(x)\right) \\ &= \left|\nabla \xi - \frac{1}{2}\xi A^{-1}(x)\widehat{b}(x) - \xi \nabla \log \varphi_1\right|^2_{A(x)} - \left|\nabla \xi - \frac{1}{2}\xi A^{-1}(x)\widehat{b}(x)\right|^2_{A(x)}, \end{aligned}$$

in the integral above, by (A.3) we obtain the identity

$$-\mu_1 = \int_{\Omega} \xi \widetilde{J}_u \xi \,\mathrm{d}x + \int_{\Omega} \left| \nabla \xi - \frac{1}{2} \xi A^{-1}(x) \widehat{b}(x) - \xi \nabla \log \varphi_1 \right|_{A(x)}^2 \mathrm{d}x. \tag{A.6}$$

Now, assuming the integral stability inequality (A.2), we can minimize (A.3) among smooth functions ξ with $\xi = 0$ on $\partial\Omega$ and $\int_{\Omega} \xi^2 dx = 1$. The unique positive minimizer ξ_1 satisfies $\widetilde{J}_u \xi_1 = -\widetilde{\mu}_1 \xi_1$, where $\widetilde{\mu}_1 \ge 0$ is the least eigenvalue of \widetilde{J}_u . Letting $\xi = \xi_1$ in (A.6) yields

$$-\mu_1 = -\widetilde{\mu}_1 + \int_{\Omega} \left| \nabla \left(\log \xi_1 - \log \varphi_1 \right) - \frac{1}{2} A^{-1}(x) \widehat{b}(x) \right|_{A(x)}^2 \xi_1^2 \, \mathrm{d}x, \tag{A.7}$$

and from (A.7) we see that we always have $\mu_1 \leq \tilde{\mu_1}$, with equality if and only if

$$\nabla \log\left(\frac{\xi_1}{\varphi_1}\right) = \frac{1}{2}A^{-1}(x)\widehat{b}(x).$$
(A.8)

This can only happen when the drift \hat{b} is of a special form. Notice that the vector field from Example A.1 is the curl of $\sqrt{x_1^2 + x_2^2} e_3$ and so, by the Helmholtz decomposition, cannot be written as the gradient of a function.

Conversely, assume that $\hat{b}(x) = A(x)\nabla w(x)$ for some function $w \in C^2(\overline{\Omega})$. In this case, the problem can be cast in variational form and conditions (A.1) and (A.2) are equivalent. Indeed, the solutions of -Lu = f(u) in Ω are critical points of the functional $\mathcal{E}(u) = \int_{\Omega} e^{w(x)} \left(\frac{1}{2}|\nabla u|^2_{A(x)} - F(u)\right) dx$, where $F(u) = \int_0^u f(t) dt$. The integral stability inequality (A.2) amounts to the nonnegativity of the second variation

$$\frac{\mathrm{d}^2}{\mathrm{d}^2 t}\Big|_{t=0} \mathcal{E}\big(u+t\varphi\big) = \int_{\Omega} e^{w(x)} \Big(|\nabla\varphi|^2_{A(x)} - f'(u)\varphi^2\Big) \,\mathrm{d}x = -\int_{\Omega} e^{w(x)}\varphi J_u\varphi \,\mathrm{d}x$$

since, letting $\varphi = e^{-w/2}\xi$ in this expression, we have

$$\frac{\mathrm{d}^2}{\mathrm{d}^2 t}\Big|_{t=0} \mathcal{E}\left(u+t e^{-w/2}\xi\right) = \int_{\Omega} \left(|\nabla\xi-\frac{1}{2}\xi\nabla w(x)|^2_{A(x)} - f'(u)\xi^2\right)\mathrm{d}x = -\int_{\Omega}\xi\widetilde{J}_u\xi\,\mathrm{d}x.$$

In particular, since $-\int_{\Omega} \xi \widetilde{J}_u \xi \, dx \ge \widetilde{\mu}_1 \|\xi\|_{L^2(\Omega)}^2$ and taking $\varphi = \varphi_1$ to be the principal eigenfunction of J_u above, we have

$$\mu_1 \int_{\Omega} e^{w(x)} \varphi_1^2 \, \mathrm{d}x = -\int_{\Omega} (e^{w/2} \varphi_1) \widetilde{J}_u(e^{w/2} \varphi_1) \, \mathrm{d}x \ge \widetilde{\mu_1} \int_{\Omega} e^{w(x)} \varphi_1^2 \, \mathrm{d}x$$

and we obtain the reverse inequality $\mu_1 \geq \tilde{\mu_1}$.

Appendix B

A trace inequality

First we prove a simple lemma to control the L^p norm in the ball by the L^p norms of the trace and the gradient:

Lemma B.1. For $p \ge 1$ and $u \in W^{1,p}(B_1)$, we have

$$\|u\|_{L^{p}(B_{1})}^{p} \leq 2^{p-1} \left(\|u\|_{L^{p}(\partial B_{1})}^{p} + \|\nabla u\|_{L^{p}(B_{1})}^{p} \right).$$

Proof. By approximation, we may assume that $u \in C^{\infty}(\overline{B}_1)$. For $r \in (0,1)$ and $\sigma \in \partial B_1$, we have $u(r\sigma) = u(\sigma) - \int_r^1 \sigma \cdot \nabla u(t\sigma) dt$ and hence

$$r^{n-1}|u(r\sigma)|^{p} \leq 2^{p-1}r^{n-1}|u(\sigma)|^{p} + 2^{p-1}r^{n-1}\int_{r}^{1}|\nabla u(t\sigma)|^{p} dt$$

$$\leq 2^{p-1}|u(\sigma)|^{p} + 2^{p-1}\int_{0}^{1}t^{n-1}|\nabla u(t\sigma)|^{p} dt.$$
(B.1)

Integrating (B.1) in $\int_0^1 dr \int_{\partial B_1} d\mathcal{H}^{n-1}(\sigma)$ now yields the claim.

We prove a Sobolev trace inequality with best exponent:

Proposition B.2. For $1 , let <math>p^* := \frac{n-1}{n-p}p$. Then

$$\|u\|_{L^{p^{\star}}(\partial B_{1})}^{p} \leq C\left(\|u\|_{L^{p}(\partial B_{1})}^{p} + \|\nabla u\|_{L^{p}(B_{1})}^{p}\right)$$

for all $u \in W^{1,p}(B_1)$, where C is a constant depending only on n and p.

Proof. By approximation, we may assume that $u \in C^{\infty}(\overline{B}_1)$. Recall the standard Sobolev inequality

$$\|u\|_{L^{p_{S}}(B_{1})}^{p} \leq C(\|u\|_{L^{p}(B_{1})}^{p} + \|\nabla u\|_{L^{p}(B_{1})}^{p}),$$
(B.2)

where $p_S := \frac{n}{n-p}p$ is the Sobolev exponent and C depends only on n and p. By the divergence theorem we have

$$\int_{\partial B_1} |u|^{p^*} \, \mathrm{d}\mathcal{H}^{n-1} = \int_{B_1} \operatorname{div}(x|u|^{p^*}) \, \mathrm{d}x = n \int_{B_1} |u|^{p^*} \, \mathrm{d}x + p^* \int_{B_1} |u|^{p^*-2} u(x \cdot \nabla u) \, \mathrm{d}x,$$

whence

$$\int_{\partial B_1} |u|^{p^*} \,\mathrm{d}\mathcal{H}^{n-1} \le n \int_{B_1} |u|^{p^*} \,\mathrm{d}x + p^* \int_{B_1} |u|^{p^*-1} |\nabla u| \,\mathrm{d}x. \tag{B.3}$$

The last term in (B.3) can be bounded by the Hölder inequality as

$$\int_{B_1} |u|^{p^*-1} |\nabla u| \, \mathrm{d}x \le \left(\int_{B_1} |u|^{(p^*-1)\frac{p}{p-1}} \, \mathrm{d}x \right)^{\frac{p-1}{p}} \|\nabla u\|_{L^p(B_1)},$$

and noticing that $(p^{\star} - 1)\frac{p}{p-1} = p_S$ we deduce

$$\|u\|_{L^{p^{\star}}(\partial B_{1})}^{p^{\star}} \leq n\|u\|_{L^{p^{\star}}(B_{1})}^{p^{\star}} + p^{\star}\|\nabla u\|_{L^{p}(B_{1})}\|u\|_{L^{p_{S}}(B_{1})}^{p^{\star}-1}.$$
(B.4)

Since $p^* < p_S$, by Hölder we have $||u||_{L^{p^*}(B_1)} \leq C||u||_{L^{p_S}(B_1)}$, and applying the Sobolev inequality (B.2) in (B.4), we obtain the trace Sobolev inequality

$$\|u\|_{L^{p^{\star}}(\partial B_{1})}^{p} \leq C(\|u\|_{L^{p}(B_{1})}^{p} + \|\nabla u\|_{L^{p}(B_{1})}^{p}),$$
(B.5)

where C depends only on n and p. Applying Lemma B.1 in (B.5) now yields the claim. \Box

Appendix C

Two interpolation inequalities

We recall two interpolation inequalities in cubes by Cabré [19] (with elementary proofs in that paper). In the first one, the L^p norm of the gradient is bounded by a weighted L^1 norm of the Hessian and the L^p norm of the function:

Proposition C.1 ([19]). Let $Q = (0, 1)^n \subset \mathbb{R}^n$, $p \ge 1$, and $u \in C^2(\overline{Q})$.

Then, for every $\delta \in (0, 1)$,

$$\|\nabla u\|_{L^{p}(Q)}^{p} \leq C_{p}\left(\delta \| |\nabla u|^{p-1} D^{2} u\|_{L^{1}(Q)} + \delta^{-p} \|u\|_{L^{p}(Q)}^{p}\right),$$

where C_p is a constant depending only on n and p.

Proof. We first prove the claim for n = 1. In this case, we have $|u'(x_0)| = \min_{x \in [0,1]} |u'(x)|$ for some $x_0 \in [0,1]$. For $0 < y < \frac{1}{3} < \frac{2}{3} < z < 1$, since (u(y) - u(z))/(y - z) is equal to u' at some point, we have $|u'(x_0)| \le 3(|u(y)| + |u(z)|)$ and integrating in y and z, we deduce

$$|u'(x_0)| \le 6 ||u||_{L^1(0,1)}.$$

Integrating $\frac{d}{dx}(|u'|)^p$ from x_0 to $x \in (0, 1)$, we obtain

$$|u'(x)|^p \le p \int_0^1 |u'|^{p-1} |u''| \, \mathrm{d}x + |u'(x_0)|^p.$$

Now, applying the previous inequality, integrating in $x \in (0, 1)$, and by Hölder, we deduce

$$||u'||_{L^{p}(0,1)}^{p} \leq p|||u'|^{p-1}u''||_{L^{1}(0,1)} + 6^{p}||u||_{L^{p}(0,1)}^{p}.$$

Rescaling and covering, it is then easy to conclude the claim with a parameter $\delta > 0$.

For n > 1, we apply the last inequality to $x_n \mapsto u(x', x_n)$ for each $x' \in (0, 1)^{n-1}$ and then integrate in x'. This leads to the claimed inequality with $||u_{x_n}||_{L^p(Q)}^p$ on the left-hand side. Applying this to the remaining variables and summing yields the claim. \Box

By a simple covering argument, we also have the analogous estimate in half-annuli. We state it for p = 1, since this is the form in which we will use it:

Corollary C.2. Let $u \in C^2(\overline{B_1^+})$ and $0 < \rho_1 < \rho_2 < \rho_3 < \rho_4 \leq 1$. Then, for every $\widetilde{\delta} \in (0, 1)$, we have

$$\|\nabla u\|_{L^{1}(B^{+}_{\rho_{3}}\setminus B^{+}_{\rho_{2}})} \leq C_{\rho_{i}}\left(\widetilde{\delta}\|D^{2}u\|_{L^{1}(B^{+}_{\rho_{4}}\setminus B^{+}_{\rho_{1}})} + \widetilde{\delta}^{-1}\|u\|_{L^{1}(B^{+}_{\rho_{4}}\setminus B^{+}_{\rho_{1}})}\right),$$

where C_{ρ_i} is a constant depending only on n, ρ_1 , ρ_2 , ρ_3 , and ρ_4 .

Proof. We cover $\overline{B_{\rho_3}^+} \setminus B_{\rho_2}^+$ by disjoint cubes Q_j of side-length $\varepsilon > 0$, with ε sufficiently small such that $Q_j \subset B_{\rho_4}^+ \setminus \overline{B_{\rho_1}^+}$. Since ε and the number of cubes depend only on n and ρ_i , applying Proposition C.1 in each cube and summing in j yields the claim. \Box

The second inequality controls the L^p norm of the function by the L^p norm of the gradient and the L^1 norm of the function:

Proposition C.3 ([19]). Let $Q = (0, 1)^n \subset \mathbb{R}^n$, $p \ge 1$, and $u \in C^2(\overline{Q})$.

Then, for every $\delta \in (0, 1)$,

$$||u||_{L^{p}(Q)}^{p} \leq C\left(\widetilde{\delta}^{p} ||\nabla u||_{L^{p}(Q)}^{p} + \widetilde{\delta}^{-n(p-1)} ||u||_{L^{1}(Q)}^{p}\right),$$

where C is a constant depending only on n.

Proof. Let $u_Q = \frac{1}{|Q|} \int_Q u \, dx$. By Poincaré's inequality $||u - u_Q||_{L^p(Q)} \leq C_p ||\nabla u||_{L^p(Q)}$ and hence

$$||u||_{L^{p}(Q)} \leq ||u - u_{Q}||_{L^{p}(Q)} + ||u_{Q}||_{L^{p}(Q)} \leq C_{p}(||\nabla u||_{L^{p}(Q)} + ||u||_{L^{1}(Q)}).$$

A scaling and covering argument then leads to the result.

Appendix D

Absorbing errors in larger balls

We recall a celebrated device of Simon [89] which allows to absorb errors in large balls when controlling quantities in smaller balls:

Lemma D.1 ([89]). Let $\beta \geq 0$ and $C_0 > 0$. Let \mathcal{B} be the class of all open balls B contained in the unit ball B_1 of \mathbb{R}^n and let $\sigma: \mathcal{B} \to [0, +\infty)$ satisfy the following subadditivity property:

$$\sigma(B) \leq \sum_{j=1}^{N} \sigma(B^{j}) \quad \text{whenever } N \in \mathbb{Z}^{+}, \{B^{j}\}_{j=1}^{N} \subset \mathcal{B}, \text{ and } B \subset \bigcup_{j=1}^{N} B^{j}.$$

It follows that there exists a constant $\delta > 0$, which depends only on n and β , such that if

$$\rho^{\beta}\sigma\left(B_{\rho/2}(y)\right) \leq \delta\rho^{\beta}\sigma\left(B_{\rho}(y)\right) + C_{0} \quad \text{whenever } B_{\rho}(y) \subset B_{1},$$

then

$$\sigma(B_{1/2}) \le CC_0$$

for some constant C which depends only on n and β .

Proof. The idea is to obtain an inequality for the quantity

$$S := \sup\{\rho^{\beta}\sigma(B_{\rho/2}(y)) \colon B_{\rho}(y) \subset B_1\}.$$

Given $\varepsilon \in (0, 1/2)$, there are $N = N(\varepsilon)$ points $y_i \in \overline{B_{1/2}}$ such that $B_{1/2} \subset \bigcup_{i=1}^N B_{\varepsilon/4}(y_i)$. Notice that $B_{\varepsilon}(y_i) \subset B_1$.

Let $\rho > 0$ and $y \in \mathbb{R}^n$ be such that $B_{\rho}(y) \subset B_1$, and consider $\varepsilon \in (0, 1/2)$ as above. Since $B_{\rho/2}(y) \subset \bigcup_i B_{\varepsilon \rho/4}(y+y_i)$ and $B_{\varepsilon \rho}(y+y_i) \subset B_{\rho}(y) \subset B_1$, by assumption we have

$$\rho^{\beta}\sigma(B_{\rho/2}(y)) \leq \rho^{\beta} \sum_{i=1}^{N} \sigma(B_{\varepsilon\rho/4}(y+y_{i}))$$
$$\leq \delta\varepsilon^{-\beta} \sum_{i=1}^{N} \rho^{\beta}\varepsilon^{\beta}\sigma(B_{\varepsilon\rho/2}(y+y_{i})) + 2^{\beta}\varepsilon^{-\beta}NC_{0}$$
$$\leq \delta\varepsilon^{-\beta}NS + 2^{\beta}\varepsilon^{-\beta}NC_{0}$$

and hence, taking the supremum in ρ and y we deduce

$$S(1 - \delta \varepsilon^{-\beta} N) \le 2^{\beta} \varepsilon^{-\beta} N C_0.$$

Taking $\delta > 0$ such that $\delta \varepsilon^{-\beta} N = 1/2$, it follows that $S \leq 2^{\beta+1} \varepsilon^{-\beta} N C_0$, and hence the claim.

Appendix E

Approximating $C^{1,1}$ domains by smooth ones from the interior

In this appendix, we show that bounded domains of class $C^{1,1}$ can be approximated by smooth sets from the interior, satisfying certain uniform bounds. This is a well-known result in the literature, and is valid more generally for domains of class $C^{k,\alpha}$ with $k \ge 1$ and $\alpha \in [0, 1]$. We have included an elementary proof for the sake of completeness. Our proof follows the approach suggested by Gilbarg and Trudinger in [63, Problem 6.9].

Recall the definition of $C^{1,1}$ domains:

Definition E.1. A bounded domain $\Omega \subset \mathbb{R}^n$ is of class $C^{1,1}$ if at each point $x_0 \in \partial \Omega$ there is a ball $B = B_{\rho}(x_0)$ and a one-to-one mapping Ψ of B onto $U \subset \mathbb{R}^n$ such that:

- (i) $\Psi(B \cap \Omega) \subset \mathbb{R}^n_+$
- (ii) $\Psi(B \cap \partial \Omega) \subset \partial \mathbb{R}^n_+$
- (iii) $\Psi \in C^{1,1}(\overline{B})$ and $\Psi^{-1} \in C^{1,1}(\overline{U})$

Equivalently, Ω is of class $C^{1,1}$ if each point of $\partial\Omega$ has a neighborhood in which $\partial\Omega$ is the graph of a $C^{1,1}$ function of n-1 of the coordinates.

Every such domain can be written as the positive set of a $C^{1,1}$ function. Namely, we have:

Lemma E.2. Let $\Omega \subset \mathbb{R}^n$ be a bounded domain of class $C^{1,1}$. Then there is a function $\Phi \in C^{1,1}(\mathbb{R}^n)$ such that $\Omega = \{\Phi > 0\}, \ \Phi = 0$ on $\partial\Omega$, and $\nabla\Phi(x) \neq 0$ for all $x \in \partial\Omega$.

Proof. By compactness, $\partial\Omega$ may be covered by finitely many balls $\{B_j = B_{\rho_j}(x_j)\}_{j=1}^N$, with $x_j \in \partial\Omega$ and $\rho_j > 0$, such that there are flattening maps $\Psi_j \in C^{1,1}(\overline{B_j})$ as in Definition E.1.

Let $\rho > 0$ be sufficiently small so that the set $B_0 := \{x \in \Omega : \operatorname{dist}(x, \partial \Omega) > \rho\}$ satisfies $\overline{\Omega} \subset \bigcup_{j=0}^N B_j$, and consider a partition of unity $\{\eta_j\}_{j=0}^N$ subordinated to the covering $\{B_j\}_{j=0}^N$. The function

$$\Phi = \eta_0 + \sum_{j=1}^N \eta_j \Psi_j^n$$

now satisfies the desired properties.

Remark E.3. Notice that, by construction, Φ is compactly supported and takes negative values in a bounded neighborhood of $\partial\Omega$ outside $\overline{\Omega}$.

Regularizing Φ and taking appropriate superlevel sets, we obtain the following approximation result:

Lemma E.4. Let $\Omega = \{\Phi > 0\} \subset \mathbb{R}^n$ be a bounded domain of class $C^{1,1}$, with $\Phi \in C^{1,1}(\mathbb{R}^n)$ as in Lemma E.2 above. Then, there is an exhaustion of Ω by smooth sets $\Omega_k = \{\Phi_k > 0\}$,¹ where the functions $\Phi_k \in C^{\infty}(\mathbb{R}^n)$ satisfy

$$\|\nabla\Phi_k\|_{C^1(\mathbb{R}^n)} + \||\nabla\Phi_k|^{-1}\|_{L^\infty(\partial\Omega_k)} \le C,$$

for some constant C depending only on Φ and Ω . Moreover, we have that $\partial \Omega_k \to \partial \Omega$ in the sense of the Hausdorff distance.²

Proof. Since $\Omega = \{\Phi > 0\}$ and $\nabla \Phi \neq 0$ on $\partial \Omega$, by continuity it follows that Φ is comparable to the distance function $d(x) = \operatorname{dist}(x, \partial \Omega)$ in $\overline{\Omega}$, say

$$L^{-1}d(x) \le \Phi(x) \le Ld(x) \quad \text{for } x \in \overline{\Omega},$$
 (E.1)

for some $L \ge 1$. From now on, for x outside Ω , we let $d(x) = -\text{dist}(x, \partial \Omega)$.

Consider a mollifying sequence $(\eta^{\varepsilon})_{\varepsilon>0}$ with $\operatorname{supp} \eta^{\varepsilon} \subset B_{\varepsilon}$. We will choose the functions

$$\Phi_k := \Phi * \eta^{\varepsilon_k} - 2L\varepsilon_k \tag{E.2}$$

for some appropriate sequence $\varepsilon_k \downarrow 0$. Since $\|\nabla \Phi_k\|_{C^1} = \|\nabla \Phi_k\|_{L^{\infty}} + \|D^2 \Phi_k\|_{L^{\infty}}$, recalling that $\|D^2 \Phi_k\|_{L^{\infty}(\mathbb{R}^n)} = [\nabla \Phi_k]_{C^{0,1}(\mathbb{R}^n)}$ (because Φ_k is C^{∞}) and $\Phi \in C^{1,1}$, the stated uniform bounds for $\|\nabla \Phi_k\|_{C^1}$ hold by the standard properties of convolutions.

Given $\delta > 0$, from (E.1) we see that

$$\Phi(x) > \delta/L \quad \text{for } x \in \{x \in \mathbb{R}^n \colon d(x) > \delta\} = \{d > \delta\},\$$

and taking the convolution with η^{ε} , we deduce

$$\Phi * \eta^{\varepsilon}(x) > \delta/L \quad \text{for } x \in \{d > \delta + \varepsilon\}.$$
(E.3)

Similarly, for $\delta > 0$ we have

$$\Phi(x) \le L\widetilde{\delta} \quad \text{for } x \in \{d \le \widetilde{\delta}\},\$$

and regularizing we obtain

$$\Phi * \eta^{\varepsilon}(x) \le L\widetilde{\delta} \quad \text{for } x \in \{d \le \widetilde{\delta} - \varepsilon\}.$$
(E.4)

Letting $\tilde{\delta} = \delta/L^2$, since $L\tilde{\delta} = \delta/L$, by (E.3) and (E.4) we have

$$\{d>\delta+\varepsilon\}\subset\{\Phi*\eta^\varepsilon>\delta/L\}\subset\{d>\delta/L^2-\varepsilon\},$$

and the choice $\delta = 2L^2 \varepsilon$ now yields the inclusions

$$\{d > \varepsilon(2L^2 + 1)\} \subset \{\Phi * \eta^\varepsilon > 2L\varepsilon\} \subset \{d > \varepsilon\}.$$
 (E.5)

Next, we construct the sequence ε_k in (E.2) inductively. Given $\varepsilon_1 > 0$ small, we define $\varepsilon_{k+1} := \varepsilon_k / \{2(2L^2 + 1)\}$. Hence, by (E.5), the sets $\Omega_k := \{\Phi_k > 0\} = \{\Phi * \eta^{\varepsilon_k} > 2L\varepsilon_k\}$ satisfy

$$\overline{\Omega_k} \subset \{d \ge \varepsilon_k\} \subset \{d > \varepsilon_{k+1}(2L^2 + 1)\} \subset \Omega_{k+1} \subset \Omega.$$
(E.6)

¹By an exhaustion we mean that $\overline{\Omega_k} \subset \Omega_{k+1} \subset \Omega$ and $\Omega = \bigcup_k \Omega_k$.

²By this we mean that $\max\{\sup_{x\in\partial\Omega} \operatorname{dist}(x,\partial\Omega_k), \sup_{x\in\partial\Omega_k} \operatorname{dist}(x,\partial\Omega)\} \to 0 \text{ as } k \to \infty.$

They clearly exhaust Ω , since $\varepsilon_k \downarrow 0$ and thus $\Omega = \bigcup_k \{d \ge \varepsilon_{k-1}\} \subset \bigcup_k \Omega_k$. Furthermore, the inclusions (E.6) show that $\partial \Omega_k$ is at a Hausdorff distance of at most $\varepsilon_k(2L^2 + 1)$ from $\partial \Omega_k$ and hence $\partial \Omega_k \to \partial \Omega$ with respect to this distance.

It remains to prove the lower bound for $|\nabla \Phi_k|$ on $\partial \Omega_k$, which will also show that the boundary $\partial \Omega_k$ is smooth. Let $x \in \mathbb{R}^n$ with $d(x) = |x - x_0|$ for some $x_0 \in \partial \Omega$. Since

$$\nabla\Phi(x)\cdot\frac{\nabla\Phi(x_0)}{|\nabla\Phi(x_0)|} \ge |\nabla\Phi(x_0)| - [\nabla\Phi]_{C^{0,1}(\mathbb{R}^n)}d(x),$$

we have the lower bound

$$\nabla \Phi(x) \cdot \frac{\nabla \Phi(x_0)}{|\nabla \Phi(x_0)|} \ge \frac{1}{2} \| |\nabla \Phi|^{-1} \|_{L^{\infty}(\partial \Omega)}^{-1} \quad \text{for } x \in \{ -\rho < d < \rho \},$$

where $\rho > 0$ depends only on $\||\nabla \Phi|^{-1}\|_{L^{\infty}(\partial\Omega)}$ and $[\nabla \Phi]_{C^{0,1}(\mathbb{R}^n)}$. Taking the convolution with η^{ε_k} in this last inequality, we obtain

$$\nabla \Phi_k(x) \cdot \frac{\nabla \Phi(x_0)}{|\nabla \Phi(x_0)|} \ge \frac{1}{2} \||\nabla \Phi|^{-1}\|_{L^{\infty}(\partial\Omega)}^{-1} \quad \text{for } x \in \{-(\rho - \varepsilon_k) < d < \rho - \varepsilon_k\}.$$
(E.7)

By (E.6), the boundary $\partial\Omega_k$ is at a distance of at most $\varepsilon_k(2L^2 + 1)$ from $\partial\Omega$. Hence, choosing $\varepsilon_1 > 0$ sufficiently small such that $\rho - \varepsilon_1 > \varepsilon_1(2L^2 + 1)$, from (E.7) and the definition of $\varepsilon_k = \varepsilon_1/\{2(2L^2 + 1)\}^{k-1}$ we deduce

$$|\nabla \Phi_k(x)| \ge \nabla \Phi_k(x) \cdot \frac{\nabla \Phi(x_0)}{|\nabla \Phi(x_0)|} \ge \frac{1}{2} |||\nabla \Phi|^{-1}||_{L^{\infty}(\partial \Omega)}^{-1} \quad \text{on } \partial \Omega_k$$

and therefore $\||\nabla \Phi_k|^{-1}\|_{L^{\infty}(\partial\Omega_k)} \leq 2\||\nabla \Phi|^{-1}\|_{L^{\infty}(\partial\Omega)}$, which concludes the proof. \Box

Appendix F

On the uniqueness of stable solutions

Here, we prove the uniqueness of stable solutions to nonvariational equations involving convex nonlinearities. For this, we employ some fundamental results of Berestycki, Nirenberg, and Varadhan [9] on the principal eigenfunction. Compare the following statement with Proposition 1.3.1 in Dupaigne's book [48]:

Proposition F.1. Given $\Omega \subset \mathbb{R}^n$ a bounded domain, let $u_1, u_2 \in C^0(\overline{\Omega}) \cap W^{2,n}_{\text{loc}}(\Omega)$ be two stable solutions of the equation -Lu = f(u) in Ω , with u = 0 on $\partial\Omega$.

Assume that $f \in C^1(\mathbb{R})$ is convex.

Then either $u_1 = u_2$ or $f(u) = \mu_1[L, \Omega]u$ on the ranges of u_1 and u_2 .

Remark F.2. Here $\mu_1[L,\Omega]$ denotes the principal (or smallest) eigenvalue of L in Ω , with the sign convention $-L\varphi = \mu_1\varphi$. It is characterized by (see [9])

 $\mu_1[L,\Omega] = \sup\left\{\mu\colon \text{ there is a function } \varphi > 0 \in W^{2,n}_{\text{loc}}(\Omega) \text{ satisfying } L\varphi + \mu\varphi \le 0 \text{ in } \Omega\right\}.$

Moreover, L can be any uniformly elliptic second order operator. In particular, we allow zero order terms.

Remark F.3. In the proof of Theorem 3.1.1 above, we only need a weaker version of Proposition F.1. Namely, we could assume additionally that $u_1 \leq u_2$, which admits a shorter proof. However, the present statement might be more useful in applications.

Proof. Assume $u_1 \neq u_2$ and consider the difference $w := u_2 - u_1$. Let

$$\Omega^+ := \{ x \in \Omega \colon u_2(x) > u_1(x) \} = \{ w > 0 \},\$$

and assume $\Omega^+ \neq 0$ (otherwise we exchange the roles of u_1 and u_2). By convexity

$$-Lw = f(u_2) - f(u_1) \le f'(u_2)w,$$

whence

$$J_{u_2}w = Lw + f'(u_2)w \ge 0 \quad \text{in } \Omega.$$
(F.1)

By the monotonicity of the principal eigenvalue with respect to the domain, since u_2 is stable, we have

$$\mu_1[J_{u_2}, \Omega^+] \ge \mu_1[J_{u_2}, \Omega] \ge 0.$$
(F.2)

Since w > 0 in Ω^+ , by (F.1) and (F.2) it follows that

$$\begin{cases} J_{u_2}w + \mu_1[J_{u_2}, \Omega^+]w \ge 0 & \text{in } \Omega^+\\ w = 0 & \text{on } \partial \Omega^+. \end{cases}$$

Applying [9, Corollary 2.2], we deduce that w is a positive principal eigenfunction of J_{u_2} in Ω^+ , that is,

$$J_{u_2}w + \mu_1[J_{u_2}, \Omega^+]w = 0 \quad \text{in } \Omega^+.$$
 (F.3)

Using the equation $-Lw = f(u_2) - f(u_1)$ in Ω^+ , from (F.3) we see that

$$f(u_2) - f(u_1) - f'(u_2)(u_2 - u_1) = \mu_1[J_{u_2}, \Omega^+](u_2 - u_1) \ge 0$$
 in Ω^+ . (F.4)

By convexity we also have $f(u_2) - f(u_1) - f'(u_2)(u_2 - u_1) \leq 0$ and hence, by (F.4), it follows that $\mu_1[J_{u_2}, \Omega^+] = 0$ and f is affine in the union of the intervals $[u_1(x), u_2(x)]$ when $x \in \Omega^+$.

For instance, if f is of the form f(u) = au + b in the ranges above, since -Lw = awin Ω^+ with w > 0, we must have $a = \mu_1[L, \Omega^+]$. Moreover, to have nontrivial solutions of $-Lu = \mu_1[L, \Omega^+]u + b$, the Fredholm alternative forces b = 0. Therefore, we see that $f(u) = \mu_1[L, \Omega^+]u$ in the ranges of u_1 and u_2 in Ω^+ .

If $\Omega^- := \{w < 0\} \neq \emptyset$, then, arguing as above with -w in place of w, we deduce

$$(J_{u_1} + \mu_1[J_{u_1}, \Omega^-])w = 0$$
 in Ω^- ,

with $\mu_1[J_{u_1}, \Omega^-] = 0$ and $f(u) = \mu_1[L, \Omega^-]u$ in the ranges of u_2 and u_1 in Ω^- .

The regularity of f and the continuity of the solutions forces $\mu_1[L, \Omega^+] = \mu_1[L, \Omega^-]$. However, by the stability of u_2

$$0 \le \mu_1[J_{u_2}, \Omega] = \mu_1[L + \mu_1[L, \Omega^+], \Omega] = \mu_1[L, \Omega] - \mu_1[L, \Omega^+],$$

but

$$\mu_1[L,\Omega] - \mu_1[L,\Omega^+] < 0$$

by the strict monotonicity of μ_1 , since $\Omega^+ \subsetneq \Omega$ with $|\Omega \setminus \Omega^+| = |\Omega^-| > 0$ by assumption.¹ This contradiction forces either Ω^+ or Ω^- to be empty, hence $f(u) = \mu_1[L, \Omega]$ in the ranges of u_1 and u_2 , as we claimed.

¹To show that $\mu_1[L,\Omega] < \mu_1[L,\Omega^+]$, first notice that we already have $\mu_1[L,\Omega] \le \mu_1[L,\Omega^+]$ by definition of μ_1 . Suppose $\mu_1[L,\Omega] = \mu_1[L,\Omega^+]$ and consider φ_1 a positive principal eigenfunction of L in Ω . Hence $L\varphi_1 + \mu_1[L,\Omega^+]\varphi_1 \le 0$ in Ω^+ and $\varphi_1 > 0$ in Ω^+ . By Corollary 2.1 in [9] we must have $\varphi_1 = 0$ on $\partial\Omega^+ \cap \Omega \neq \emptyset$, contradicting the positivity in Ω .
Part II

A nonlocal Weierstrass extremal field theory

Chapter 4

A Weierstrass extremal field theory for the fractional Laplacian

In this chapter, we extend part of the Weierstrass extremal field theory in the Calculus of Variations to a nonlocal framework. Our model case is the energy functional for the fractional Laplacian (the Gagliardo-Sobolev seminorm), for which such a theory was still unknown until our work [23].

We build a null-Lagrangian and a calibration for nonlinear equations involving the fractional Laplacian in the presence of a field of extremals. Thus, our construction assumes the existence of a family of solutions to the Euler-Lagrange equation whose graphs produce a foliation. Then, the minimality of each leaf in the foliation follows from the existence of the calibration. As an application, we show that monotone solutions to fractional semilinear equations are minimizers.

4.1 Introduction

The Weierstrass extremal field theory, a classical tool from the Calculus of Variations, provides a sufficient condition for the minimality of critical points. Namely, if an extremal of an elliptic functional can be embedded in a family of critical points whose graphs produce a foliation (in particular, the graphs do not intersect each other), then the given extremal is a minimizer. The proof of this result is based on the construction of a *calibration*, that is, an auxiliary functional satisfying certain properties (see Definition 4.1.1). This theory has found important applications in the context of minimal surfaces, among others.

The purpose of this chapter is to extend the classical Weierstrass field theory to the setting of functionals associated to nonlocal equations, starting here with the simplest one. Our main result is the construction of a calibration for the fractional functional

$$\mathcal{E}_{s,F}(w) := \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|w(x) - w(y)|^2}{|x - y|^{n+2s}} \,\mathrm{d}x \,\mathrm{d}y - \int_{\Omega} F(w(x)) \,\mathrm{d}x,$$

where $s \in (0, 1)$, $c_{n,s}$ is a positive normalizing constant, $F \in C^1(\mathbb{R})$,

$$Q(\Omega) := (\mathbb{R}^n \times \mathbb{R}^n) \setminus (\Omega^c \times \Omega^c) = (\Omega \times \Omega) \cup (\Omega \times \Omega^c) \cup (\Omega^c \times \Omega), \tag{4.1.1}$$

and $\Omega \subset \mathbb{R}^n$ is a given bounded domain. Here and throughout the paper, $\Omega^c = \mathbb{R}^n \setminus \Omega$.

The Euler-Lagrange equation for the functional $\mathcal{E}_{s,F}$ is the semilinear equation

$$(-\Delta)^s u = F'(u) \quad \text{in } \Omega,$$

where

$$(-\Delta)^s u(x) = c_{n,s} \operatorname{P.V.} \int_{\mathbb{R}^n} \frac{u(x) - u(y)}{|x - y|^{n+2s}} \,\mathrm{d}y$$

is the fractional Laplacian and P.V. stands for the principal value.

Our construction does not use the Caffarelli-Silvestre extension problem for the fractional Laplacian. This is relevant, since in the next chapter, it allows us to treat more general nonlocal functionals of the form¹

$$\mathcal{E}_{\mathrm{N}}(w) := \frac{1}{2} \iint_{Q(\Omega)} G_{\mathrm{N}}(x, y, w(x), w(y)) \,\mathrm{d}x \,\mathrm{d}y,$$

where the Lagrangian $G_{\rm N}(x, y, a, b)$ is required to satisfy the natural ellipticity condition

$$\partial_{ab}^2 G_N(x, y, a, b) + \partial_{ab}^2 G_N(y, x, b, a) \le 0.$$
 (4.1.2)

As in the classical local theory, our calibration is built in the presence of a field of extremals, namely, a one-parameter family of critical points of $\mathcal{E}_{s,F}$ (or of \mathcal{E}_{N}) whose graphs form a foliation (see Definition 4.1.2). In particular, the graphs do not intersect each other. For the construction, it suffices to have subsolutions, respectively supersolutions, on each respective side of a given extremal —something very useful for some applications.

As a first application of our calibration, we establish that monotone solutions to translation invariant nonlocal equations are minimizers. This is related to a celebrated conjecture of De Giorgi for the Allen-Cahn equation. More precisely, if u is a solution satisfying $\partial_{x_n} u > 0$ in \mathbb{R}^n , then it is a minimizer² among functions w such that

$$\lim_{\tau \to -\infty} u(x', \tau) \le w(x', x_n) \le \lim_{\tau \to +\infty} u(x', \tau)$$

for $x = (x', x_n) \in \mathbb{R}^{n-1} \times \mathbb{R}$. This result was only known for those nonlocal functionals for which an existence and regularity theory of minimizers is available. We elaborate on this further in Subsection 4.1.4.

As a second application, in Chapter 5 below, we establish that minimizers of nonlocal elliptic functionals are viscosity solutions. Although this was previously known for problems where a weak comparison principle is available (see [5, 66, 87]), we can prove it in more general scenarios by using the calibration technique. This has been motivated by the theory of nonlocal minimal surfaces, where the calibration argument of Cabré [18] greatly simplified the original proof (that minimizers are viscosity solutions) from [32].

4.1.1 The notion of calibration

A fundamental problem in the Calculus of Variations consists of finding conditions for a function to be a minimizer of a given functional. More precisely, given a functional $\mathcal{E}: \mathcal{A} \to \mathbb{R}$ defined on some set of admissible functions \mathcal{A} , and given $u \in \mathcal{A}$, one wishes to know whether u minimizes \mathcal{E} among competitors in \mathcal{A} having the same Dirichlet condition as u.

¹The subindices L and N will be used throughout the work to denote local and nonlocal objects, respectively.

 $^{^{2}}$ Monotone solutions are easily seen to be strictly stable solutions and, as a result, to be minimizers with respect to small compactly supported perturbations. Our result gives a more precise neighborhood in which the solution is minimizing.

In classical local problems, the Dirichlet condition refers to the value of u on the boundary of the domain Ω , while in nonlocal problems one prescribes the value in all the exterior of Ω , namely, in $\Omega^c = \mathbb{R}^n \setminus \Omega$.

One effective strategy to establish the minimality of a given function $u \in \mathcal{A}$ consists of constructing a calibration:

Definition 4.1.1. A functional $C: \mathcal{A} \to \mathbb{R}$ is a *calibration* for the functional \mathcal{E} and the admissible function $u \in \mathcal{A}$ if the following conditions hold:

 $(\mathcal{C}1) \ \mathcal{C}(u) = \mathcal{E}(u).$

(C2) $\mathcal{C}(w) \leq \mathcal{E}(w)$ for all $w \in \mathcal{A}$ with the same Dirichlet condition as u.

(C3) $\mathcal{C}(w) = \mathcal{C}(\widetilde{w})$ for all $w, \widetilde{w} \in \mathcal{A}$ with the same Dirichlet condition as u.

Functionals satisfying (C3) are known as *null-Lagrangians* (see, for instance, [54, Chapter 8] and [61, Section 1.4]). It is, however, convenient to relax this last condition to the less stringent

 $(\mathcal{C}3')$ $\mathcal{C}(u) \leq \mathcal{C}(w)$ for all $w \in \mathcal{A}$ with the same Dirichlet condition as u.

In this work we still refer to functionals satisfying (C1), (C2), and (C3') as calibrations.³

Recall the meaning of the Dirichlet condition for local and nonlocal problems given right before Definition 4.1.1.

Once a calibration is available, the minimality of u follows immediately both in the local and nonlocal cases. Indeed, if C is a calibration for \mathcal{E} and $u \in \mathcal{A}$, then, for every $w \in \mathcal{A}$ with the same Dirichlet condition as u, applying (C1), (C3'), and (C2) (in this order) we obtain

$$\mathcal{E}(u) = \mathcal{C}(u) \le \mathcal{C}(w) \le \mathcal{E}(w).$$

Therefore, u is a minimizer.

4.1.2 The classical theory of calibrations

Calibrations arose in the development of the classical theory of the Calculus of Variations. Historically, a fundamental question was to determine necessary and sufficient conditions for a function to be a minimizer. A satisfactory answer has been obtained for functionals —that we often call "energies", following PDE terminology— of the form

$$\mathcal{E}_{\mathrm{L}}(w) := \int_{\Omega} G_{\mathrm{L}}(x, w(x), \nabla w(x)) \,\mathrm{d}x.$$
(4.1.3)

In this framework, the function $G_{\rm L}(x,\lambda,q)$ is called the Lagrangian of $\mathcal{E}_{\rm L}$.

A necessary condition for minimality is the vanishing of the first variation of \mathcal{E}_{L} . That is, every minimizer is a critical point of \mathcal{E}_{L} (an *extremal*) and must satisfy the associated Euler-Lagrange equation. If the Lagrangian $G_{L}(x, \lambda, q)$ is convex in the variables (λ, q) , then the functional \mathcal{E}_{L} is convex and, in this case, every extremal is a minimizer. Although many models from Physics exhibit such a convexity property, it is well known that very relevant nonconvex energies appear in applications. This is the case of the Allen-Cahn energy, among many others. For such energy functionals, the Dirichlet problem may admit

³In the literature, functionals satisfying (C1), (C2), and (C3') are sometimes called *subcalibrations*.

several extremals, not all of them being minimizers. Still, if $G_{\rm L}$ is not convex in (λ, q) , one often has that the Lagrangian $G_{\rm L}(x, \lambda, q)$ is convex with respect to the gradient variable q, which amounts to the *ellipticity* of the problem.

For nonconvex elliptic problems, one is interested in having sufficient conditions for an extremal to be a minimizer. Two of such conditions, due to Jacobi and Weierstrass, are well known. First, if a solution is strictly stable,⁴ then it is a *local minimizer* in a certain topology, that is, a minimizer in a small neighborhood. This is Jacobi's condition. The real difficulty is proving minimality in a larger, more interesting class of competitors (or perhaps even *absolute minimality*). In this direction, the Weierstrass sufficient condition yields minimality among functions taking values in a precise region. To further elaborate on this, we need to introduce the notion of *field*. Essentially, this is a collection of ordered functions $u^t : \overline{\Omega} \to \mathbb{R}$, with t in some interval $I \subset \mathbb{R}$, enjoying some regularity for the joint function $(x,t) \mapsto u^t(x)$. The key point is that the graphs of these functions produce a foliation of a certain region \mathcal{G} in $\mathbb{R}^n \times \mathbb{R}$, which allows to carry out a subtle convexity argument to bound the nonconvex functional by below with a calibration.

While fields are a classical concept in local problems, we can extend their definition to include both the local and nonlocal settings, as follows.

Definition 4.1.2. Given a domain $D \subset \mathbb{R}^n$ (not necessarily bounded) and an interval $I \subset \mathbb{R}$ (not necessarily bounded, nor open), we say that a family $\{u^t\}_{t \in I}$ of functions $u^t \colon \overline{D} \to \mathbb{R}$ is a *field in* D if

- the function $(x,t) \mapsto u^t(x)$ is continuous in $\overline{D} \times I$;
- for each $x \in \overline{D}$, the function $t \mapsto u^t(x)$ is C^1 and increasing in I.

We say that $\{u^t\}_{t\in I}$ is a C^2 field in D if, additionally, the function $(x,t) \mapsto u^t(x)$ is C^2 in $\overline{D} \times I$.

Given a functional \mathcal{E} acting on functions defined in \overline{D} , and given a subdomain $\Omega \subset D$, we say that $\{u^t\}_{t\in I}$ is a *field of extremals*⁵ in Ω (roughly speaking, since we should refer to \mathcal{E} , D, and Ω) when it is a field in D and each of the functions u^t is a critical point of \mathcal{E} in Ω .

In the local setting, we will take $D = \Omega$. For nonlocal Lagrangians we will set $D = \mathbb{R}^n$ and $\Omega \subset \mathbb{R}^n$.

Given a field in D as above, the region

$$\mathcal{G} = \{(x,\lambda) \in \overline{D} \times \mathbb{R} \colon \lambda = u^t(x) \text{ for some } t \in I\} \subset \mathbb{R}^n \times \mathbb{R}$$

is foliated by the graphs of the functions u^t , which do not intersect each other (since $u^t(x)$ is increasing in t). In particular, we can uniquely define a *leaf-parameter function*

$$t: \mathcal{G} \to I, \quad (x,\lambda) \mapsto t(x,\lambda) \quad \text{determined by} \quad u^{t(x,\lambda)}(x) = \lambda.$$
 (4.1.4)

The function t is continuous in \mathcal{G} by the assumptions in Definition 4.1.2. We will often refer to the functions u^t (or their graphs) as the "*leaves*" of the field.

⁴A solution u is said to be *strictly stable* if the principal eigenvalue of the linearized equation at u is positive.

⁵The term *extremal field* is also often used in the literature, but we find it ambiguous.

Having defined what a field is, we can now state the classical theorem of Weierstrass, which was first proven for scalar ODEs:

Weierstrass sufficient condition. For the functional \mathcal{E}_{L} in (4.1.3), assuming ellipticity (i.e., that the Lagrangian $G_{L}(x, \lambda, q)$ is convex in the gradient variable q), if a critical point is embedded in a field of extremals,⁶ then the critical point is a minimizer among functions taking values in the foliated region \mathcal{G} and having the same boundary values as the given critical point. (4.1.5)

The proof of (4.1.5) is based on the construction of a calibration. For this, given an interval $I \subset \mathbb{R}$, a bounded domain $\Omega \subset \mathbb{R}^n$, and a C^2 field of extremals $\{u^t\}_{t \in I}$ in Ω , one considers the set of admissible functions $\mathcal{A}_{\mathrm{L}} = \{w \in C^1(\overline{\Omega}) : \operatorname{graph} w \subset \mathcal{G}\}$. Then, the functional $\mathcal{C}_{\mathrm{L}} : \mathcal{A}_{\mathrm{L}} \to \mathbb{R}$, defined through the Legendre transform of the Lagrangian G_{L} as

$$\mathcal{C}_{\mathrm{L}}(w) := \int_{\Omega} \left\{ \partial_{q} G_{\mathrm{L}}(x, u^{t}(x), \nabla u^{t}(x)) \cdot \left(\nabla w(x) - \nabla u^{t}(x) \right) \right\} \Big|_{t=t(x, w(x))} \, \mathrm{d}x + \int_{\Omega} G_{\mathrm{L}}(x, u^{t}(x), \nabla u^{t}(x)) \Big|_{t=t(x, w(x))} \, \mathrm{d}x,$$

$$(4.1.6)$$

is a calibration for \mathcal{E}_{L} and each critical point $u^{t_0}, t_0 \in I$. While condition (C1) in Definition 4.1.1 follows directly from (4.1.6), and condition (C2) amounts to ellipticity (i.e., the convexity of the Lagrangian in the gradient variable $q = \nabla w(x)$), it is a remarkable fact that the null-Lagrangian property (C3) holds. Its proof will be recalled in Section 4.3.

As an illustrative example, in the presence of a field of extremals, the functional⁷

$$\mathcal{E}_{1,F}(w) = \frac{1}{2} \int_{\Omega} |\nabla w(x)|^2 \,\mathrm{d}x - \int_{\Omega} F(w(x)) \,\mathrm{d}x$$

(which is typically nonconvex), admits a calibration. It is given by

$$\mathcal{C}_{1,F}(w) = \int_{\Omega} \left\{ \nabla u^{t}(x) \cdot (\nabla w(x) - \nabla u^{t}(x)) + \frac{1}{2} |\nabla u^{t}(x)|^{2} \right\} \Big|_{t=t(x,w(x))} \mathrm{d}x - \int_{\Omega} F(w(x)) \,\mathrm{d}x.$$
(4.1.7)

Notice that, although the functional $\mathcal{E}_{1,F}$ is not convex, its Lagrangian is elliptic.

The Weierstrass sufficient condition (4.1.5) naturally leads to the question of when it is possible to embed a solution of the Euler-Lagrange equation into a field of extremals in a large portion of space. An important case corresponds to functionals which are invariant with respect to some translations. A first example are those Lagrangians which do not depend on a direction of space and, at the same time, the extremal is monotone in that same direction. Such a solution can be translated along the invariant direction to produce a field of extremals. This applies to layer solutions of the Allen-Cahn equation; see Subsection 4.1.4 below. A second example are those Lagrangians $G_L(x, \lambda, q)$ which do not depend on the function variable λ . In this case, a field can be obtained by translating the solution in the vertical direction. This can be used, for instance, to show that minimal graphs are minimizing minimal surfaces; see Section 4.2.

⁶In fact, it suffices that the leaves of the field above and below the graph of the given critical point are, respectively, super and subsolutions to the Euler-Lagrange equation. This, which is well known, will be easily seen within the proofs of our main results.

⁷The subindex 1 in the definition of the energy functional refers to the fractional parameter s in $\mathcal{E}_{s,F}$, which are the nonlocal analogues of $\mathcal{E}_{1,F}$ treated later. As s tends to 1, one recovers $\mathcal{E}_{1,F}$ from $\mathcal{E}_{s,F}$ after a suitable normalization; see, for instance, [47].

Fields of extremals can also be built in the presence of a concrete explicit solution. Here, using the precise solution and PDE at hand, one may be able to construct a field in a more or less explicit way. This approach has been applied in the theory of minimal surfaces to establish the minimality of Simons and Lawson cones, as explained also in Section 4.2.

4.1.3 Nonlocal calibrations

While the theory of calibrations for local equations is well understood, there are only two papers, to the best of our knowledge, dealing with nonlocal ones. In [18] the first author found an explicit calibration for the fractional perimeter, as explained in Section 4.2. Pagliari [80] investigated the abstract structure of calibrations for the fractional total variation.⁸

We now present our main result, which builds a calibration for the functional

$$\mathcal{E}_{s,F}(w) = \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|w(x) - w(y)|^2}{|x - y|^{n+2s}} \,\mathrm{d}x \,\mathrm{d}y - \int_{\Omega} F(w(x)) \,\mathrm{d}x \tag{4.1.8}$$

in the presence of a field of extremals. Recall (4.1.1) for the meaning of $Q(\Omega)$, Definition 4.1.2 for the notion of field, and (4.1.4) for the leaf-parameter function t. The calibration properties (C1), (C2), (C3), and (C3') have been introduced in Definition 4.1.1. As we will explain in Section 4.4, the regularity assumptions on the field can be significantly weakened.

Theorem 4.1.3. Let $I \subset \mathbb{R}$ be an interval, $\Omega \subset \mathbb{R}^n$ a bounded domain, and $s \in (0, 1)$. Let $\{u^t\}_{t \in I}$ be a C^2 field in \mathbb{R}^n in the sense of Definition 4.1.2 satisfying

$$|u^t(x)| + |\partial_t u^t(x)| \le C$$
 for all $x \in \mathbb{R}^n$ and $t \in I$,

for some constant C. Consider the admissible functions

$$\mathcal{A}_s = \{ w \in C^0(\mathbb{R}^n) \cap L^\infty(\mathbb{R}^n) : \operatorname{graph} w \subset \mathcal{G} \},\$$

where

$$\mathcal{G} = \{ (x, \lambda) \in \mathbb{R}^n \times \mathbb{R} \colon \lambda = u^t(x) \text{ for some } t \in I \}.$$

Given $t_0 \in I$ and $F \in C^1(\mathbb{R})$, let $\mathcal{C}_{s,F}$ be the functional

$$\mathcal{C}_{s,F}(w) := c_{n,s} \operatorname{P.V.} \iint_{Q(\Omega)} \int_{u^{t_0}(x)}^{w(x)} \frac{u^t(x) - u^t(y)}{|x - y|^{n+2s}} \Big|_{t = t(x,\lambda)} d\lambda \, dx \, dy - \int_{\Omega} F(w(x)) \, dx \\
+ \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|u^{t_0}(x) - u^{t_0}(y)|^2}{|x - y|^{n+2s}} \, dx \, dy$$
(4.1.9)

defined for $w \in \mathcal{A}_s$, where $c_{n,s}$ is the positive constant in (4.1.8).

Taking $C = C_{s,F}$ and $\mathcal{E} = \mathcal{E}_{s,F}$ as in (4.1.8), we have the following:

(a) $C_{s,F}$ satisfies (C1) and (C2) with $u = u^{t_0}$.

⁸This functional involves the fractional perimeter of each sublevel set of a given function. The author succeeded in constructing a calibration to prove that halfspaces are minimizers, but other fields of extremals are not mentioned in this work.

(b) Assume in addition that the family $\{u^t\}_{t\in I}$ satisfies

$$\begin{aligned} (-\Delta)^s u^t - F'(u^t) &\geq 0 \quad in \ \Omega \quad for \ t \geq t_0, \\ (-\Delta)^s u^t - F'(u^t) &\leq 0 \quad in \ \Omega \quad for \ t \leq t_0. \end{aligned}$$

Then, $C_{s,F}$ satisfies (C3') with $u = u^{t_0}$. In particular, u^{t_0} minimizes $\mathcal{E}_{s,F}$ among functions w in \mathcal{A}_s such that $w \equiv u^{t_0}$ in Ω^c .

(c) Assume in addition that $\{u^t\}_{t\in I}$ is a field of extremals in Ω , that is, a field in \mathbb{R}^n satisfying

$$(-\Delta)^s u^t - F'(u^t) = 0$$
 in Ω for all $t \in I$.

Then, the functional $C_{s,F}$ satisfies (C3) with $u = u^{t_0}$. Therefore, $C_{s,F}$ is a calibration for $\mathcal{E}_{s,F}$ and u^{t_0} . As a consequence, for every $t \in I$, the extremal u^t minimizes $\mathcal{E}_{s,F}$ among functions w in \mathcal{A}_s such that $w \equiv u^t$ in Ω^c .

The meaning of the principal value P.V. in the definition of the functional $C_{s,F}$ will be made precise in Section 4.4; see Remark 4.4.3.

Even when the nonlocal energy functional is as simple as $\mathcal{E}_{s,F}$ (the energy functional associated to the fractional Laplacian) the form of a calibration, if any could exist, was not known prior to our work [23].

Our first attempts at constructing a calibration for $\mathcal{E}_{s,F}$ consisted on trying to "nonlocalize" the expression (4.1.7) for the local calibration, mainly by substituting gradients by fractional ones or double integrals of differences. This strategy seems to lead to functionals that are not calibrations. We comment on these attempts with more detail in Appendix H.

A second failed approach consisted of trying to find a satisfactory calibration using the extension problem for the fractional Laplacian. Indeed, applying the local theory in the extended space gives a calibration in terms of a certain field of extremals "upstairs", but it was not clear at all how to write it in terms of the given field "downstairs" (the reason being that the functional is too involved). Thus, the extension has not been useful to us; see Appendix G.

We were puzzled for a long time until we revisited the work of the first author [18], which found a calibration for the fractional perimeter. It was written in terms of the Euler-Lagrange and Neumann operators associated to the fractional perimeter. We then realized that such a structure was also present, but hidden, in the classical local calibration $C_{\rm L}$ in (4.1.6). More precisely, for every $t_0 \in I$, in Theorem 4.3.1 we will see that

$$\mathcal{C}_{\mathrm{L}}(w) = \int_{\Omega} \int_{u^{t_0}(x)}^{w(x)} \mathcal{L}_{\mathrm{L}}(u^{t})(x) \Big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x + \int_{\partial\Omega} \int_{u^{t_0}(x)}^{w(x)} \mathcal{N}_{\mathrm{L}}(u^{t})(x) \Big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}\mathcal{H}^{n-1}(x) + \mathcal{E}_{\mathrm{L}}(u^{t_0}),$$

$$(4.1.10)$$

where \mathcal{L}_{L} and \mathcal{N}_{L} are, respectively, the Euler-Lagrange and Neumann operators associated to the functional \mathcal{E}_{L} in (4.1.3). To the best of our knowledge, this is the first time that the local calibration has been written in this way. From this expression, the null-Lagrangian property follows easily.⁹ Instead, traditionally one exhibited the null-Lagrangian property

⁹Indeed, if $\{u^t\}_{t\in I}$ is a field of extremals, then $\mathcal{L}_{L}(u^t) \equiv 0$ and the calibration depends only on the value of w on $\partial\Omega$.

of $C_{\rm L}$ by either expressing the functional as the flux of a divergence-free vector field in $\Omega \times \mathbb{R}$ or by certain straightforward although opaque analytic computations; see Section 4.3. Neither of these approaches reveals the exact role played by the Euler-Lagrange and Neumann operators in the calibration.

Once we found (4.1.10) for the local case, a simple extension of this expression easily led us to the nonlocal calibration of Theorem 4.1.3. The key point is that each of the terms in (4.1.10) has a clear nonlocal counterpart. In fact, the same procedure works for general nonlocal functionals \mathcal{E}_{N} of the form

$$\mathcal{E}_{\mathrm{N}}(w) = \frac{1}{2} \iint_{Q(\Omega)} G_{\mathrm{N}}(x, y, w(x), w(y)) \,\mathrm{d}x \,\mathrm{d}y$$

with Lagrangian $G_N(x, y, a, b)$ satisfying the natural ellipticity condition (4.1.2).

4.1.4 An application to monotone solutions

Our main motivation to find a calibration came from the study of *monotone solutions* to the fractional Allen-Cahn equation

$$(-\Delta)^s u = u - u^3$$
 in \mathbb{R}^n

(see [30, 31], for instance, and [42] for more general integro-differential operators). Note that when the operator is the classical Laplacian, these solutions are related to a famous conjecture of De Giorgi; see [26] for instance.

The following is an application of our main theorem to monotone solutions of translation invariant equations.

Corollary 4.1.4. Given $s \in (0,1)$ and $F \in C^3(\mathbb{R})$, let $u : \mathbb{R}^n \to \mathbb{R}$ be a bounded solution of

$$(-\Delta)^s u = F'(u) \quad in \mathbb{R}^n. \tag{4.1.11}$$

Assume that u is increasing in x_n .

Then, for each bounded domain $\Omega \subset \mathbb{R}^n$, u is a minimizer of $\mathcal{E}_{s,F}$ among continuous functions $w : \mathbb{R}^n \to \mathbb{R}$ satisfying

$$\lim_{\tau \to -\infty} u(x',\tau) \le w(x',x_n) \le \lim_{\tau \to +\infty} u(x',\tau) \quad \text{for all } (x',x_n) \in \mathbb{R}^{n-1} \times \mathbb{R}$$
(4.1.12)

and such that $w \equiv u$ in Ω^c .

Let us mention that we only assume $F \in C^3$ for simplicity, to ensure that u is of class C^2 independently of s. We could weaken the regularity assumptions, but this is not the purpose of the thesis.

To prove Corollary 4.1.4, we define the one-parameter family of functions $u^t(x) := u(x', x_n + t)$, where $x = (x', x_n) \in \mathbb{R}^{n-1} \times \mathbb{R}$. Thanks to the monotonicity assumption, $u_{x_n} > 0$, the family $\{u^t\}_{t \in \mathbb{R}}$ is a field in \mathbb{R}^n in the sense of Definition 4.1.2. Moreover, it is a field of extremals on account of the translation invariance of the equation (4.1.11). Thus, Theorem 4.1.3 gives that, on each bounded domain $\Omega \subset \mathbb{R}^n$, u is a minimizer of the energy $\mathcal{E}_{s,F}$ among the admissible functions in the statement.

Let us point out that this minimality result was already known and can be proven without using calibrations. Nevertheless, this alternative proof (described in the next paragraph) requires an existence and regularity theorem for minimizers. Such a result is not available for many other nonlocal equations. In the next chapter, we construct calibrations in a general nonlocal setting, and thus the proof above will allow us to show minimality of monotone solutions for the fist time.

Now, we briefly discuss the proof of Corollary 4.1.4 which does not use calibrations; for the details, see [26]. One considers a minimizer within the region given by (4.1.12) and with the same exterior datum as the monotone solution. Its existence and regularity can be proved in the case of equation (4.1.11). Now, the monotone solution can be translated, starting from infinity, until it touches the minimizer from one side, something that must happen in Ω , not in the exterior. The strong comparison principle then yields that the translated solution and the minimizer must coincide. Moreover, by the exterior condition they must be equal to the original solution. In particular, this proves that the monotone solution is a minimizer. Furthermore, this proof gives uniqueness of solution with the exterior data of u.

4.1.5 Outline of the chapter

For the proofs of our main theorems, the reader may skip sections 4.2 and 4.3.

In Section 4.2 we briefly comment on the classical perimeter functional and review the work of the first author on the calibration for the nonlocal perimeter [18]. Section 4.3 is devoted to recalling some known facts from the classical theory of calibrations and proving the new expression (4.1.10); see Theorem 4.3.1. In Section 4.4 we prove Theorem 4.1.3 under weaker assumptions on the field.

4.2 The classical and nonlocal perimeters

In this section we recall different notions for the perimeter of a set. First we introduce the classical perimeter functional and its calibration. We will mention several results concerning fields of extremals in this setting. Later we revisit the work of the first author [18] on the construction of a calibration for the nonlocal perimeter. Here we will focus on identifying the key feature that leads to the calibration properties in this nonlocal framework. This will suggest a candidate structure to search for in local functionals, which will lead to (4.1.10) and then allow us to treat the fractional Laplacian case.

As mentioned in the Introduction, some relevant applications of calibrations concern the theory of minimal surfaces. In broad terms, a minimal surface $\Sigma \subset \mathbb{R}^n$ is a critical point of the (n-1)-dimensional area functional. Given a domain $\Omega \subset \mathbb{R}^n$, the classical perimeter of a (regular) set $F \subset \mathbb{R}^n$ inside Ω is defined by

$$\mathcal{P}_{\mathcal{L}}(F) := \mathcal{H}^{n-1}(\Omega \cap \partial F), \tag{4.2.1}$$

where \mathcal{H}^{n-1} is the (n-1)-dimensional Hausdorff measure. Here we interpret the boundary of F as the surface $\Sigma = \partial F$. The critical points E of \mathcal{P}_{L} (known as *minimal sets* in the literature) satisfy $H_{\mathrm{L}}[\partial E] = 0$ in Ω , where $H_{\mathrm{L}}[\Sigma]$ denotes the mean curvature of Σ . The variation of \mathcal{P}_{L} is taken with respect to perturbations preserving the boundary datum $E \cap \partial \Omega$.

We are interested in showing that certain minimal sets E minimize \mathcal{P}_{L} among sets Fwith the same boundary condition $F \cap \partial \Omega = E \cap \partial \Omega$. Assume that for a family of minimal sets $\{E^t\}_{t \in \mathbb{R}}$, the surfaces ∂E^t form a foliation of Ω . For $x \in \Omega$, we let t(x) be the unique $t \in \mathbb{R}$ such that $x \in \partial E^t$. Denote the outward unit normal vector to ∂F by $\nu_{\partial F}$. Then, the perimeter functional admits as a calibration

$$\mathcal{C}_{\mathcal{P}_{\mathrm{L}}}(F) := \int_{\Omega \cap \partial F} X \cdot \nu_{\partial F} \,\mathrm{d}\mathcal{H}^{n-1}, \qquad (4.2.2)$$

where

$$X(x) := \nu_{\partial E^t} \Big|_{t=t(x)}(x)$$

is the vector field given by the normal vectors to the surfaces ∂E^t . Notice that Definition 4.1.1 can be easily modified to involve subsets of \mathbb{R}^n instead of functions. Then, properties ($\mathcal{C}1$) and ($\mathcal{C}2$) are easy to check directly, while the null-Lagrangian property ($\mathcal{C}3$) follows from the divergence theorem and the fact that divX = 0. As a consequence, each E^t minimizes \mathcal{P}_{L} and, therefore, each ∂E^t is a minimizing minimal surface.

This discussion leads to the question of when it is possible to embed a minimal surface in a field of extremals. The simplest situation is when the minimal surface is a graph. If $u: \Omega' \subset \mathbb{R}^{n-1} \to \mathbb{R}$ is a minimal graph, then the graphs of the translations $u^t = u + t$ give a field of extremals in $\Omega = \Omega' \times \mathbb{R}$. By the calibration $\mathcal{C}_{\mathcal{P}_L}$ in (4.2.2), every minimal graph is a minimizing minimal surface.¹⁰ We point out that the functional $\mathcal{C}_{\mathcal{P}_L}$ can also be obtained by integrating a closed differential form; see Chapter 1 in [37].

Another interesting situation is when the minimal surface is not a graph but has an explicit expression. Here a field of extremals can still be obtained in some cases. For instance, this is done for the Simons cone and for the more general Lawson cones in Bombieri, De Giorgi, and Giusti [10] and in Davini [45]. The strategy here consists of using the symmetries of the cone to reduce the minimal surface equation to an ODE in the plane. It is then shown that the solutions of this ODE do not intersect each other, and thus give a foliation. We remark that, although the cone is an explicit extremal, the field of extremals itself is not explicit. An alternative is to build fields made of sub and supersolutions, which are easier to obtain and suffice to show the minimality (see footnote 6). Explicit examples of such fields have been found for Lawson cones, simplifying the proof of the minimality; see De Philippis and Paolini [46] for the Simons cone and Liu [71] for Lawson cones. We mention that the case of minimal surfaces of codimension greater than 1 can also be treated, where the appropriate notion of calibration involves the use of differential forms; see [77].

The perimeter functional \mathcal{P}_{L} in (4.2.1) has a nonlocal analogue. Given a nonnegative symmetric kernel K = K(z), with $z \in \mathbb{R}^n$, the K-nonlocal perimeter of a set $F \subset \mathbb{R}^n$ inside Ω is defined by

$$\mathcal{P}_{\mathrm{N}}(F) := \frac{1}{2} \iint_{Q(\Omega)} \left| \mathbb{1}_{F}(x) - \mathbb{1}_{F}(y) \right| K(x-y) \,\mathrm{d}x \,\mathrm{d}y,$$

where $Q(\Omega)$ was defined in (4.1.1). It is well known that the Euler-Lagrange operator associated to \mathcal{P}_N is the nonlocal mean curvature H_K , which is defined for F at boundary points $x \in \partial F$ by

$$H_K[F](x) := \int_{\mathbb{R}^n} \left(\mathbb{1}_{F^c}(y) - \mathbb{1}_F(y) \right) K(x-y) \, \mathrm{d}y,$$

¹⁰Notice that restricting the area functional to the class of graphs yields a convex functional. In particular, every minimal graph minimizes area in this smaller class, but it is not a priori clear if they are minimizers with respect to all surfaces. The calibration is used to prove this stronger fact.

as introduced in [32]. In particular, if a (sufficiently regular) set E minimizes \mathcal{P}_{N} with respect to sets F with the same exterior values $F \setminus \Omega = E \setminus \Omega$, then $H_{K}[E](x) = 0$ for $x \in \partial E \cap \Omega$.

In [18], the first author showed that, given a measurable function $\phi \colon \mathbb{R}^n \to \mathbb{R}$, the functional¹¹

$$\mathcal{C}_{\mathcal{P}_{N}}(F) := \frac{1}{2} \iint_{Q(\Omega)} \operatorname{sign} \left(\phi(x) - \phi(y) \right) \left(\mathbb{1}_{F}(x) - \mathbb{1}_{F}(y) \right) K(x-y) \, \mathrm{d}x \, \mathrm{d}y \tag{4.2.3}$$

is a calibration for the nonlocal perimeter \mathcal{P}_N and each superlevel set

$$E^t := \{ x \in \mathbb{R}^n : \phi(x) > t \},\$$

assuming that these sets have zero nonlocal mean curvature. As a consequence, each E^t is a minimizer of \mathcal{P}_N with respect to sets that coincide with E^t outside Ω .

Properties (C1) and (C2) are easy to check directly from expression (4.2.3). However, showing the null-Lagrangian property (C3) requires an alternative expression for $C_{\mathcal{P}_{N}}$. For this, [18] wrote (4.2.3) in terms of the sets E^{t} as follows. Assume for simplicity that ϕ is smooth and $\nabla \phi(x) \neq 0$ for all x. Then the level sets are smooth surfaces

$$\partial E^t = \{ x \in \mathbb{R}^n \colon \phi(x) = t \},\$$

which have zero Lebesgue measure in \mathbb{R}^n , and it can be readily checked that

$$\operatorname{sign}(\phi(x) - \phi(y)) = \left(\mathbb{1}_{(E^t)^c}(y) - \mathbb{1}_{E^t}(y)\right)\Big|_{t=\phi(x)} \quad \text{for a.e. } (x, y) \in \mathbb{R}^n \times \mathbb{R}^n.$$
(4.2.4)

By skew-symmetry of sign $(\phi(x) - \phi(y))$, using (4.2.4) and splitting the integration domain into $(\Omega \times \mathbb{R}^n) \cup (\Omega^c \times \Omega)$, we arrive at the alternative expression

$$\mathcal{C}_{\mathcal{P}_{N}}(F) = \int_{\Omega \cap F} H_{K}[E^{t}](x) \big|_{t=\phi(x)} dx + \int_{F \setminus \Omega} \left\{ \int_{\Omega} \left(\mathbb{1}_{(E^{t})^{c}}(y) - \mathbb{1}_{E^{t}}(y) \right) K(x-y) dy \right\} \Big|_{t=\phi(x)} dx,$$

$$(4.2.5)$$

see [18] for details. Thus, if for all t we have $H_K[E^t] = 0$ in Ω , then the quantity $\mathcal{C}_{\mathcal{P}_N}(F)$ depends only on the exterior condition $F \setminus \Omega$, which makes it to be a null-Lagrangian.¹²

Passing from (4.2.3) to (4.2.5) is the crucial step in [18]. To our knowledge, the structure of the alternative expression (4.2.5) for $C_{\mathcal{P}_{N}}$ is the only way to prove the null-Lagrangian property. The two terms in (4.2.5) bring out the true dependence of $C_{\mathcal{P}_{N}}$ on the data:¹³ the first one involves the Euler-Lagrange equation of \mathcal{P}_{N} at each superlevel set E^{t} , while the second one depends only on the set F outside Ω . The existence of such a structure for the nonlocal perimeter suggested that it could also be present, although hidden, in other calibrations, even in the local case, as we will see in next section.

¹³The structure in (4.2.5) also appears in the local framework, where the calibration $C_{\mathcal{P}_{L}}$ given by (4.2.2) can be written as

$$\mathcal{C}_{\mathcal{P}_{\mathrm{L}}}(F) = \int_{\Omega \cap F} H_{\mathrm{L}}[\partial E^{t}](x)|_{t=t(x)} \,\mathrm{d}x - \int_{\partial \Omega \cap F} X \cdot \nu_{\partial \Omega} \,\mathrm{d}\mathcal{H}^{n-1}$$

Moreover, it is not difficult to see that the calibration for the *fractional perimeter*, i.e., the K-nonlocal perimeter with $K(z) = |z|^{-n-2s}$ for $s \in (0, 1)$, recovers $\mathcal{C}_{\mathcal{P}_{L}}$ in the limit when $s \to 1$.

 $^{^{11}}$ Let us point out that the idea of using the sign function comes from the Legendre transform of the absolute value that appears in the fractional perimeter functional.

¹²As mentioned in the Introduction, to show minimality one does not actually need the full null-Lagrangian property (C3) but rather the weaker condition (C3'). For instance, to prove that the set E^0 minimizes \mathcal{P}_N , from (4.2.5) it can be shown that it suffices for the E^t "above" and "below" E^0 to be super and subsolutions, respectively. For more details see [18] and compare with Theorem 4.1.3 and Proposition 4.3.3 in the following sections.

4.3 The theory of calibrations for local equations, and a novelty

The purpose of this section is twofold: first, to review the classical theory of fields of extremals and calibrations for "local" functionals and, second, to give a new proof of the calibration properties in this setting. Inspired by the structure of the calibration (4.2.5) for the nonlocal perimeter, we will find an alternative expression for the classical calibration (4.3.8) (a new expression to the best of our knowledge), which involves only the Euler-Lagrange and Neumann operators acting on the field.

Consider an energy functional of the form

$$\mathcal{E}_{\mathrm{L}}(w) := \int_{\Omega} G_{\mathrm{L}}(x, w(x), \nabla w(x)) \,\mathrm{d}x, \qquad (4.3.1)$$

where the Lagrangian $G_{\rm L}(x,\lambda,q)$ is of class C^2 in all arguments.

A function that plays an important role when studying minimality to nonconvex energy functionals of the form (4.3.1) is the so-called *Weierstrass excess function*. It is defined for $x \in \Omega, \lambda \in \mathbb{R}$, and $q, \tilde{q} \in \mathbb{R}^n$ by

$$E(x,\lambda,q,\widetilde{q}) := G_{L}(x,\lambda,\widetilde{q}) - G_{L}(x,\lambda,q) - \partial_{q}G_{L}(x,\lambda,q) \cdot (\widetilde{q}-q).$$
(4.3.2)

It is well known (see [61]) that if $u \in C^1(\overline{\Omega})$ is a minimizer of \mathcal{E}_{L} with respect to small $C_c^0(\Omega)$ perturbations,¹⁴ then it must satisfy the Weierstrass necessary condition

$$E(x, u(x), \nabla u(x), \xi) \ge 0 \quad \text{for all } x \in \Omega, \xi \in \mathbb{R}^n.$$
(4.3.3)

Note that condition (4.3.3) on the excess function is automatically satisfied by every function u whenever $G_{L}(x, \lambda, q)$ is convex with respect to the variable q, i.e., when the problem is elliptic. The Dirichlet energy and more generally the Lagrangian associated to the p-Laplacian are important elliptic examples where (4.3.3) is thus automatically satisfied.

Given an interval $I \subset \mathbb{R}$ and $\{u^t\}_{t \in I}$ a C^2 field in Ω (in the sense of Definition 4.1.2), we let

$$\mathcal{G}_{\mathrm{L}} := \left\{ (x, \lambda) \in \overline{\Omega} \times \mathbb{R} : \lambda = u^{t}(x) \text{ for some } t \in I \right\}$$

and consider the set of admissible functions

$$\mathcal{A}_{\mathcal{L}} := \left\{ w \in C^1(\overline{\Omega}) : \operatorname{graph} w \subset \mathcal{G}_{\mathcal{L}} \right\}.$$

In the classical theory, one employs the Legendre transform of G_L to define the functional $\mathcal{C}_L : \mathcal{A}_L \to \mathbb{R}$ by

$$\mathcal{C}_{\mathrm{L}}(w) := \int_{\Omega} \left\{ \partial_{q} G_{\mathrm{L}}(x, u^{t}(x), \nabla u^{t}(x)) \cdot \left(\nabla w(x) - \nabla u^{t}(x) \right) \right\} \Big|_{t=t(x, w(x))} \mathrm{d}x + \int_{\Omega} G_{\mathrm{L}}(x, u^{t}(x), \nabla u^{t}(x)) \Big|_{t=t(x, w(x))} \mathrm{d}x.$$

$$(4.3.4)$$

Under the assumption that $\{u^t\}_{t\in I}$ is a field of extremals and every leaf u^t satisfies the Weierstrass necessary condition (4.3.3), it is well known that $C_{\rm L}$ is a calibration for the functional $\mathcal{E}_{\rm L}$ and each $u^{t,15}$ see [1,3,61]. As an illustrative example, the *p*-Dirichlet energy

$$\mathcal{E}_{p\text{-Dir}}(w) = \frac{1}{p} \int_{\Omega} |\nabla w(x)|^p \,\mathrm{d}x$$

¹⁴This type of local minimizers are often referred to as *strong minimizers* in the literature.

¹⁵The positivity of the excess function E for every leaf u^t is only required to show property (C2). Properties (C1) and (C3) follow directly from the existence of the field of extremals.

admits the calibration

$$\mathcal{C}_{p\text{-Dir}}(w) = \int_{\Omega} \left\{ \left| \nabla u^{t}(x) \right|^{p-2} \nabla u^{t}(x) \cdot \left(\nabla w(x) - \nabla u^{t}(x) \right) + \frac{1}{p} \left| \nabla u^{t}(x) \right|^{p} \right\} \Big|_{t=t(x,w(x))} \mathrm{d}x.$$

We will give a new proof that the functional $C_{\rm L}$ is a calibration. As mentioned before, the key point in our approach is to rewrite $C_{\rm L}$ in an alternative form involving only those operators which are of interest to the theory of PDE: the Euler-Lagrange and Neumann operators. These arise when computing the first variation of $\mathcal{E}_{\rm L}$ at $u \in C^2(\overline{\Omega})$ in a direction of $\eta \in C^{\infty}(\overline{\Omega})$, that is,

$$\frac{\mathrm{d}}{\mathrm{d}\varepsilon}\mathcal{E}_{\mathrm{L}}(w+\varepsilon\eta)\Big|_{\varepsilon=0} = \int_{\Omega}\mathcal{L}_{\mathrm{L}}(w)(x)\,\eta(x)\,\mathrm{d}x + \int_{\partial\Omega}\mathcal{N}_{\mathrm{L}}(w)(x)\,\eta(x)\,\mathrm{d}\mathcal{H}^{n-1}(x).$$
(4.3.5)

Here in (4.3.5), $\mathcal{L}_{\rm L}$ denotes the Euler-Lagrange operator

$$\mathcal{L}_{\mathcal{L}}(w)(x) := -\operatorname{div}\left(\partial_{q}G_{\mathcal{L}}(x, w(x), \nabla w(x))\right) + \partial_{\lambda}G_{\mathcal{L}}(x, w(x), \nabla w(x))$$
(4.3.6)

and \mathcal{N}_{L} denotes the Neumann operator

$$\mathcal{N}_{\mathrm{L}}(w)(x) := \partial_q G_{\mathrm{L}}(x, w(x), \nabla w(x)) \cdot \nu_{\partial\Omega}(x), \qquad (4.3.7)$$

where $\nu_{\partial\Omega}$ is the outward unit normal vector to $\partial\Omega$.

The following identity is our new result.

Theorem 4.3.1. Given an interval $I \subset \mathbb{R}$ and a bounded domain $\Omega \subset \mathbb{R}^n$, let $\{u^t\}_{t \in I}$ be a C^2 field in Ω in the sense of Definition 4.1.2. Let $G_L = G_L(x, \lambda, q)$ be a C^2 function. Then, for any $t_0 \in I$, the functional \mathcal{C}_L defined in (4.3.4) can be written as

$$\mathcal{C}_{\mathrm{L}}(w) = \int_{\Omega} \int_{u^{t_0}(x)}^{w(x)} \mathcal{L}_{\mathrm{L}}(u^t)(x) \Big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x + \int_{\partial\Omega} \int_{u^{t_0}(x)}^{w(x)} \mathcal{N}_{\mathrm{L}}(u^t)(x) \Big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}\mathcal{H}^{n-1}(x) + \mathcal{E}_{\mathrm{L}}(u^{t_0}).$$

$$(4.3.8)$$

The proof of the theorem follows a typical strategy for showing the null-Lagrangian property, as seen for instance in [1]. However, a new trick will allow us to identify in the expression the operators \mathcal{L}_{L} and \mathcal{N}_{L} acting on the leaves. This is the first time we have seen the calibration written this way.

Proof of Theorem 4.3.1. In order to prove the result it will be enough to show that

$$\mathcal{C}_{\mathrm{L}}(w) - \mathcal{C}_{\mathrm{L}}(\widetilde{w}) = \int_{\Omega} \int_{\widetilde{w}(x)}^{w(x)} \mathcal{L}_{\mathrm{L}}(u^{t})(x) \Big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x + \int_{\partial\Omega} \int_{\widetilde{w}(x)}^{w(x)} \mathcal{N}_{\mathrm{L}}(u^{t})(x) \Big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}\mathcal{H}^{n-1}(x).$$

$$(4.3.9)$$

for any given $w, \tilde{w} \in \mathcal{A}_{L}$. That is, we only need to take $\tilde{w} = u^{t_0}$ and use the easy equality $\mathcal{C}_{L}(u^{t_0}) = \mathcal{E}_{L}(u^{t_0})$.

First, let us briefly describe the proof of identity (4.3.9). We consider $C_{\rm L}$ acting on the convex combination $w_{\theta} := (1 - \theta)\widetilde{w} + \theta w$ and express the left-hand side of (4.3.9) as $\int_0^1 \frac{\mathrm{d}}{\mathrm{d}\theta} C_{\rm L}(w_{\theta}) \,\mathrm{d}\theta$. While in the literature the functions w and \widetilde{w} are assumed to have the same boundary conditions, here we do not impose such a restriction. Then, we compute the derivative in θ using the expression of $C_{\rm L}(w_{\theta})$ as an integral in x and integrating by parts. Finally, after applying Fubini's theorem to interchange the order of integration, the key point is to make the change of variables $\theta \mapsto w_{\theta}(x)$ for each x. This yields the final expression.

Next, let us proceed with the proof. We let $\zeta := w - \widetilde{w}$ and hence $w_{\theta} = \widetilde{w} + \theta \zeta$. Since

$$C_{\rm L}(w) - C_{\rm L}(\widetilde{w}) = \int_0^1 \frac{\mathrm{d}}{\mathrm{d}\theta} C_{\rm L}(w_\theta) \,\mathrm{d}\theta \qquad (4.3.10)$$

and

$$\frac{\mathrm{d}}{\mathrm{d}\theta} \mathcal{C}_{\mathrm{L}}(w_{\theta}) = \int_{\Omega} \frac{\mathrm{d}}{\mathrm{d}\theta} \Big\{ G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big|_{t=t(x, w_{\theta}(x))} \Big\} \mathrm{d}x + \int_{\Omega} \frac{\mathrm{d}}{\mathrm{d}\theta} \left(\Big\{ \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \cdot (\nabla w_{\theta} - \nabla u^{t}) \Big\} \Big|_{t=t(x, w_{\theta}(x))} \right) \mathrm{d}x,$$
(4.3.11)

we must compute each of the integrands in (4.3.11).

By the chain rule, the first integrand can be written as

$$\frac{\mathrm{d}}{\mathrm{d}\theta} \Big\{ G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big|_{t=t(x, w_{\theta}(x))} \Big\}
= \partial_{\lambda} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big|_{t=t(x, w_{\theta}(x))} \frac{\mathrm{d}}{\mathrm{d}\theta} \big(u^{t(x, w_{\theta}(x))} \big)
+ \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big|_{t=t(x, w_{\theta}(x))} \cdot \frac{\mathrm{d}}{\mathrm{d}\theta} \big(\nabla u^{t} \big|_{t=t(x, w_{\theta}(x))} \big)$$

and using that $\frac{\mathrm{d}}{\mathrm{d}\theta} \left(u^{t(x,w_{\theta}(x))}(x) \right) = \frac{\mathrm{d}}{\mathrm{d}\theta} w_{\theta}(x) = \zeta(x)$, we deduce

$$\frac{\mathrm{d}}{\mathrm{d}\theta} \Big\{ G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big|_{t=t(x, w_{\theta}(x))} \Big\}
= \partial_{\lambda} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big|_{t=t(x, w_{\theta}(x))} \zeta + \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big|_{t=t(x, w_{\theta}(x))} \cdot \frac{\mathrm{d}}{\mathrm{d}\theta} \big(\nabla u^{t} \big|_{t=t(x, w_{\theta}(x))} \big).$$
(4.3.12)

Similarly, the second integrand in (4.3.11) is

$$\frac{\mathrm{d}}{\mathrm{d}\theta} \left(\left\{ \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \cdot (\nabla w_{\theta} - \nabla u^{t}) \right\} \Big|_{t=t(x,w_{\theta}(x))} \right) \\
= \left\{ \partial_{t} \left[\partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \right] \cdot (\nabla w_{\theta} - \nabla u^{t}) \right\} \Big|_{t=t(x,w_{\theta}(x))} \frac{\mathrm{d}}{\mathrm{d}\theta} \left[t(x, w_{\theta}(x)) \right] \\
+ \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \Big|_{t=t(x,w_{\theta}(x))} \cdot \frac{\mathrm{d}}{\mathrm{d}\theta} \nabla w_{\theta} \\
- \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \Big|_{t=t(x,w_{\theta}(x))} \cdot \frac{\mathrm{d}}{\mathrm{d}\theta} \left(\nabla u^{t} \Big|_{t=t(x,w_{\theta}(x))} \right) \right) \\
= \left\{ \partial_{t} \left[\partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \right] \cdot \left(\nabla w_{\theta} - \nabla u^{t} \right) \right\} \Big|_{t=t(x,w_{\theta}(x))} \partial_{\lambda} t(x, w_{\theta}(x)) \zeta \\
+ \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \Big|_{t=t(x,w_{\theta}(x))} \cdot \nabla \zeta \\
- \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \Big|_{t=t(x,w_{\theta}(x))} \cdot \frac{\mathrm{d}}{\mathrm{d}\theta} \left(\nabla u^{t} \Big|_{t=t(x,w_{\theta}(x))} \right).$$
(4.3.13)

Adding (4.3.12) and (4.3.13), substituting in (4.3.11), and rearranging terms, we see that

$$\frac{\mathrm{d}}{\mathrm{d}\theta} \mathcal{C}_{\mathrm{L}}(w_{\theta}) = \int_{\Omega} \partial_{\lambda} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big|_{t=t(x, w_{\theta}(x))} \zeta \,\mathrm{d}x + \int_{\Omega} \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big|_{t=t(x, w_{\theta}(x))} \cdot \nabla \zeta \,\mathrm{d}x
+ \int_{\Omega} \Big\{ \partial_{t} \big[\partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big] \cdot \big(\nabla w_{\theta} - \nabla u^{t} \big) \Big\} \Big|_{t=t(x, w_{\theta}(x))} \partial_{\lambda} t(x, w_{\theta}(x)) \zeta \,\mathrm{d}x.$$

$$(4.3.14)$$

The second term in (4.3.14) can be integrated by parts as

$$\int_{\Omega} \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big|_{t=t(x, w_{\theta}(x))} \cdot \nabla \zeta \, \mathrm{d}x$$

$$= \int_{\partial\Omega} \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \big|_{t=t(x, w_{\theta}(x))} \cdot \nu_{\partial\Omega} \zeta \, \mathrm{d}\mathcal{H}^{n-1}$$

$$- \int_{\Omega} \mathrm{div} (\partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t})) \big|_{t=t(x, w_{\theta}(x))} \zeta \, \mathrm{d}x$$

$$- \int_{\Omega} \partial_{t} [\partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t})] \big|_{t=t(x, w_{\theta}(x))} \cdot \nabla [t(x, w_{\theta}(x))] \zeta \, \mathrm{d}x.$$
(4.3.15)

We now claim that

$$\nabla \left[t(x, w_{\theta}(x)) \right] = \left(\nabla w_{\theta} - \nabla u^{t} \right) \Big|_{t=t(x, w_{\theta}(x))} \partial_{\lambda} t(x, w_{\theta}(x)), \qquad (4.3.16)$$

which leads to the identity we wish to prove. Indeed, if (4.3.16) holds, then, substituting (4.3.15) in (4.3.14), we have

$$\frac{\mathrm{d}}{\mathrm{d}\theta} \mathcal{C}_{\mathrm{L}}(w_{\theta}) = \int_{\Omega} \left\{ \partial_{\lambda} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) - \mathrm{div} \left(\partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \right) \right\} \Big|_{t=t(x, w_{\theta}(x))} \zeta \,\mathrm{d}x
+ \int_{\partial\Omega} \partial_{q} G_{\mathrm{L}}(x, u^{t}, \nabla u^{t}) \Big|_{t=t(x, w_{\theta}(x))} \cdot \nu_{\partial\Omega} \zeta \,\mathrm{d}\mathcal{H}^{n-1}
= \int_{\Omega} \mathcal{L}_{\mathrm{L}}(u^{t}) \Big|_{t=t(x, w_{\theta}(x))} \zeta \,\mathrm{d}x + \int_{\partial\Omega} \mathcal{N}_{\mathrm{L}}(u^{t}) \Big|_{t=t(x, w_{\theta}(x))} \zeta \,\mathrm{d}\mathcal{H}^{n-1}.$$
(4.3.17)

Thus, using (4.3.17) in (4.3.10), by Fubini's theorem we deduce

$$\begin{aligned} \mathcal{C}_{\mathrm{L}}(w) &- \mathcal{C}_{\mathrm{L}}(\widetilde{w}) \\ &= \int_{\Omega} \int_{0}^{1} \mathcal{L}_{\mathrm{L}}(u^{t}) \big|_{t=t(x,w_{\theta}(x))} \zeta(x) \,\mathrm{d}\theta \,\mathrm{d}x + \int_{\partial\Omega} \int_{0}^{1} \mathcal{N}_{\mathrm{L}}(u^{t}) \big|_{t=t(x,w_{\theta}(x))} \zeta(x) \,\mathrm{d}\theta \,\mathrm{d}\mathcal{H}^{n-1} \\ &= \int_{\Omega} \int_{\widetilde{w}(x)}^{w(x)} \mathcal{L}_{\mathrm{L}}(u^{t}) \big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x + \int_{\partial\Omega} \int_{\widetilde{w}(x)}^{w(x)} \mathcal{N}_{\mathrm{L}}(u^{t}) \big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}\mathcal{H}^{n-1}, \end{aligned}$$

where in the last line we have applied the change of variables $\lambda = \widetilde{w}(x) + \theta \zeta(x)$ for each x.

To show the claim (4.3.16), we use the definition of the leaf-parameter function $t(x, \lambda)$. Differentiating $u^t(x)|_{t=t(x,\lambda)} = \lambda$ with respect to λ , we have

$$\partial_t u^t \big|_{t=t(x,\lambda)} \partial_\lambda t(x,\lambda) = 1.$$
(4.3.18)



Figure 4.1

Moreover, taking the gradient of the identity $u^{t(x,w_{\theta}(x))}(x) = w_{\theta}(x)$, we obtain

$$\nabla u^{t}(x)\big|_{t=t(x,w_{\theta}(x))} + \partial_{t}u^{t}(x)\big|_{t=t(x,w_{\theta}(x))}\nabla \big[t(x,w_{\theta}(x))\big] = \nabla w_{\theta}(x).$$
(4.3.19)

Multiplying (4.3.19) by $\partial_{\lambda} t(x, w_{\theta}(x))$, applying (4.3.18) with $\lambda = w_{\theta}(x)$, and rearranging terms leads to (4.3.16) and concludes the proof.

Remark 4.3.2. The expression (4.3.8) can be deduced in a more geometric way using the divergence theorem in \mathbb{R}^{n+1} . As we see next, this gives an alternative proof of Theorem 4.3.1. Consider the vector field $X: \overline{\Omega} \times \mathbb{R} \to \mathbb{R}^{n+1} = \mathbb{R}^n \times \mathbb{R}$ given by

$$X(x,\lambda) = (X^x(x,\lambda), X^\lambda(x,\lambda)),$$

where

$$X^{x}(x,\lambda) := -\partial_{q}G_{\mathrm{L}}(x,u^{t}(x),\nabla u^{t}(x))\big|_{t=t(x,\lambda)},$$

$$X^{\lambda}(x,\lambda) := \Big\{-\partial_{q}G_{\mathrm{L}}(x,u^{t}(x),\nabla u^{t}(x)) \cdot \nabla u^{t}(x) + G_{\mathrm{L}}(x,u^{t}(x),\nabla u^{t}(x))\Big\}\Big|_{t=t(x,\lambda)}$$

Then, an easy computation from [1] shows

$$\operatorname{div} X(x,\lambda) = \mathcal{L}_{\mathrm{L}}(u^{t})(x) \big|_{t=t(x,\lambda)}$$

where div is the divergence in \mathbb{R}^{n+1} , i.e., $\operatorname{div} X(x,\lambda) = \operatorname{div}_x X^x(x,\lambda) + \partial_\lambda X^\lambda(x,\lambda)$. From the definition of X, it can also be checked that \mathcal{C}_{L} can be written in the compact form

$$\mathcal{C}_{\mathrm{L}}(w) = \int_{\Gamma_w} X \cdot \nu_{\Gamma_w} \,\mathrm{d}\mathcal{H}^n,$$

where $\Gamma_w \subset \mathbb{R}^{n+1}$ is the graph of w and ν_{Γ_w} is the unit vector normal to Γ_w pointing "upwards". In coordinates, ν_{Γ_w} reads $\nu_{\Gamma_w}(x, w(x)) = (1 + |\nabla w(x)|^2)^{-1/2} (-\nabla w(x), 1)$.

Consider now the regions between the graphs of w and \tilde{w} , distinguishing the parts above and below each function

$$R^+ = \{ (x, \lambda) \in \Omega \times \mathbb{R} \colon \widetilde{w}(x) < \lambda < w(x) \},\$$

$$R^{-} = \{ (x, \lambda) \in \Omega \times \mathbb{R} \colon w(x) < \lambda < \widetilde{w}(x) \},\$$

as well as their lateral boundaries on $\partial\Omega \times \mathbb{R}$, that is, $S^+ = \overline{R^+} \cap (\partial\Omega \times \mathbb{R})$ and $S^- = \overline{R^-} \cap (\partial\Omega \times \mathbb{R})$; see Figure 4.1. Applying the divergence theorem to the field X separately in each of the regions R^+ and R^- , we see that

$$\mathcal{C}_{\mathrm{L}}(w) = \mathcal{C}_{\mathrm{L}}(\widetilde{w}) + \int_{R^{+}} \operatorname{div} X \, \mathrm{d}\mathcal{H}^{n+1} - \int_{R^{-}} \operatorname{div} X \, \mathrm{d}\mathcal{H}^{n+1} - \int_{S^{+}} X \cdot \nu_{\partial\Omega} \, \mathrm{d}\mathcal{H}^{n} + \int_{S^{-}} X \cdot \nu_{\partial\Omega} \, \mathrm{d}\mathcal{H}^{n},$$

where we have extended the outer normal $\nu_{\partial\Omega}$ parallel to the surface $\partial\Omega \times \mathbb{R}$; see Figure 4.1. It is also immediate to check that

$$X \cdot \nu_{\partial \Omega} = -\mathcal{N}_{\mathrm{L}}(u^t) \big|_{t=t(x,\lambda))}$$

on $S^+ \cup S^-$, with \mathcal{N}_L as in (4.3.7). Thus, we obtain the passage from (4.3.4) to (4.3.8) as an application of the divergence theorem.

Next we prove the key null-Lagrangian property (C3) for the calibration, which follows readily from the new identity (4.3.8) for C_L :

Proposition 4.3.3. Under the same hypotheses as in Theorem 4.3.1, assume that, for some $t_0 \in I$, the leaves of the field $\{u^t\}_{t \in I}$ satisfy

$$\mathcal{L}_{\mathcal{L}}(u^{t}) \geq 0 \quad in \ \Omega \quad for \ t \geq t_{0},
\mathcal{L}_{\mathcal{L}}(u^{t}) \leq 0 \quad in \ \Omega \quad for \ t \leq t_{0},$$
(4.3.20)

where \mathcal{L}_{L} is the Euler-Lagrange operator introduced in (4.3.6).

Then, for all w in \mathcal{A}_{L} such that $w \equiv u^{t_{0}}$ on $\partial\Omega$, the functional \mathcal{C}_{L} defined in (4.3.4) satisfies

$$\mathcal{C}_{\mathrm{L}}(u^{t_0}) \leq \mathcal{C}_{\mathrm{L}}(w)$$

Assume, moreover, that the leaves $\{u^t\}_{t\in I}$ satisfy the Euler-Lagrange equation in Ω , that is,

$$\mathcal{L}_{\mathcal{L}}(u^t) = 0 \quad in \ \Omega \quad for \ all \ t \in I.$$

$$(4.3.21)$$

Then, for all w as above, we have

$$\mathcal{C}_{\mathrm{L}}(w) = \mathcal{C}_{\mathrm{L}}(u^{t_0}).$$

Proof. Notice that, by (4.3.8), we have $C_{L}(u^{t_0}) = \mathcal{E}_{L}(u^{t_0})$. Hence, assuming (4.3.20), since the boundary integral in (4.3.8) vanishes ($w \equiv u^{t_0}$ on $\partial \Omega$), it suffices to show that

$$\int_{u^{t_0}(x)}^{w(x)} \mathcal{L}_{\mathcal{L}}(u^t)(x) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \ge 0.$$

$$(4.3.22)$$

However, this is clear by (4.3.20) and the fact that u^t are increasing with respect to t. If we additionally have (4.3.21), then the integral in (4.3.22) is zero and the claim follows.

The remaining calibration properties (C1) and (C2) can be directly obtained from the original definition (4.3.4) of $C_{\rm L}$. First, we prove property (C1):

Proposition 4.3.4. Assume the same hypotheses of Theorem 4.3.1. Then, for all $t \in I$, the functional C_L defined in (4.3.4) satisfies

$$\mathcal{C}_{\mathrm{L}}(u^t) = \mathcal{E}_{\mathrm{L}}(u^t).$$

Proof. Given $t_0 \in I$, from the definition of the leaf-parameter function it follows that $t(x, u^{t_0}(x)) = t_0$. In particular, $\nabla u^{t_0}(x) = \nabla u^t(x) \big|_{t=t(x, u^{t_0}(x))}$, and substituting in the definition of \mathcal{C}_{L} in (4.3.4) we see that

$$\mathcal{C}_{\mathrm{L}}(u^{t_0}) = \int_{\Omega} G_{\mathrm{L}}(x, u^{t_0}(x), \nabla u^{t_0}(x)) \,\mathrm{d}x = \mathcal{E}_{\mathrm{L}}(u^{t_0}).$$

Since t_0 was arbitrary, the claim follows.

Finally, we show property (C2):

Proposition 4.3.5. Under the same hypotheses as in Theorem 4.3.1, the energy \mathcal{E}_{L} can be decomposed in terms of \mathcal{C}_{L} and the excess function E as

$$\mathcal{E}_{\mathrm{L}}(w) = \mathcal{C}_{\mathrm{L}}(w) + \int_{\Omega} \mathrm{E}(x, u^{t}(x), \nabla u^{t}(x), \nabla w(x)) \big|_{t=t(x, w(x))} \,\mathrm{d}x,^{16}$$

for all w in \mathcal{A}_{L} .

As a consequence, if each leaf of the field $\{u^t\}_{t\in I}$ satisfies the Weierstrass necessary condition (4.3.3), then

$$\mathcal{C}_{\mathrm{L}}(w) \leq \mathcal{E}_{\mathrm{L}}(w),$$

for all w in \mathcal{A}_{L} .

Proof. For each $x \in \Omega$, take $\lambda = w(x) = u^t(x)|_{t=t(x,w(x))}, q = \nabla u^t(x)|_{t=t(x,w(x))}$, and $\tilde{q} = \nabla w(x)$. Substituting in (4.3.2) and integrating in Ω , comparing this expression with the definition of \mathcal{C}_{L} in (4.3.4), the identity follows.

Remark 4.3.6. If the energy functional includes reaction terms on a portion of the boundary $\Gamma_{\mathcal{N}} \subset \partial \Omega$, i.e.,

$$\widetilde{\mathcal{E}}_{\mathrm{L}}(w) = \int_{\Omega} G_{\mathrm{L}}(x, w(x), \nabla w(x)) \,\mathrm{d}x - \int_{\Gamma_{\mathcal{N}}} F(w(x)) \,\mathrm{d}\mathcal{H}^{n-1}(x),$$

then we can still apply the methods from the local theory of calibrations to show that

$$\widetilde{\mathcal{C}_{\mathrm{L}}}(w) = \int_{\Omega} \partial_q G_{\mathrm{L}}(x, u^t(x), \nabla u^t(x)) \cdot \left(\nabla w(x) - \nabla u^t(x)\right) \big|_{t=t(x, w(x))} \,\mathrm{d}x \\ + \int_{\Omega} G_{\mathrm{L}}(x, u^t(x), \nabla u^t(x)) \big|_{t=t(x, w(x))} - \int_{\Gamma_{\mathcal{N}}} F(w(x)) \,\mathrm{d}\mathcal{H}^{n-1}(x) \,\mathrm{d}x$$

is a calibration. Hence, one can establish the minimality of the leaves among competitors with the same boundary data only on $\partial \Omega \setminus \Gamma_N$. Note that in this scenario, extremals satisfy the equation

$$\begin{cases} \mathcal{L}_{\mathrm{L}}(u) = 0 & \text{in } \Omega, \\ \mathcal{N}_{\mathrm{L}}(u) = F'(u) & \text{in } \Gamma_{\mathcal{N}}. \end{cases}$$

In particular, this allows to treat the extension problem for the fractional Laplacian as explained in Appendix G. In this setting, one considers the Dirichlet energy in a domain of the extended space \mathbb{R}^{n+1}_+ with an additional potential energy on the part of its boundary lying on $\mathbb{R}^n = \partial \mathbb{R}^{n+1}_+$.

¹⁶This identity is known as the Weierstrass representation formula

4.4 The calibration for the fractional Laplacian

In this section we construct a calibration for the functional

$$\mathcal{E}_{s,F}(w) = \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|w(x) - w(y)|^2}{|x - y|^{n+2s}} \,\mathrm{d}x \,\mathrm{d}y - \int_{\Omega} F(w(x)) \,\mathrm{d}x$$

where $s \in (0, 1)$, $c_{n,s}$ is a positive normalizing constant, and $F \in C^1(\mathbb{R})$. It involves the Gagliardo-Sobolev seminorm and a nonlinear potential term.

If u is a critical point of $\mathcal{E}_{s,F}$ with respect to functions with the same exterior data as u, then u satisfies the nonlocal semilinear equation

$$(-\Delta)^s u - F'(u) = 0 \quad \text{in } \Omega. \tag{4.4.1}$$

In this setting, recall the standard subspace of locally integrable functions given by

$$L_s^1(\mathbb{R}^n) := \Big\{ u \in L_{\text{loc}}^1(\mathbb{R}^n) \colon \|u\|_{L_s^1(\mathbb{R}^n)} = \int_{\mathbb{R}^n} \frac{|u(y)|}{1 + |y|^{n+2s}} \, \mathrm{d}y < +\infty \Big\}.$$

If $u \in L^1_s(\mathbb{R}^n)$ is C^2 in a neighborhood of $x \in \mathbb{R}^n$, then the fractional Laplacian $(-\Delta)^s u(x)$ is well defined. More generally, we only need u to be $C^{2s+\alpha}$ in a neighborhood of x for some small $\alpha > 0$ such that $2s + \alpha$ is not an integer. Here C^{β} denotes the space $C^{k,\gamma}$ of functions with Hölder continuous k-th order derivatives, where $k = \lfloor \beta \rfloor$, $\gamma = \beta - k$. In particular, for $(-\Delta)^s u$ to be well defined in a domain Ω , we just need that the function $u \in L^1_s(\mathbb{R}^n)$ is smooth in a neighborhood of Ω . The function could be extremely wild outside the domain, as long as it satisfies the growth assumption defining $L^1_s(\mathbb{R}^n)$.

As explained in the Introduction, to build the calibration we will assume the existence of a field in \mathbb{R}^n . In particular, we are given a family of functions $u^t \colon \mathbb{R}^n \to \mathbb{R}$, with $t \in I$ for some interval $I \subset \mathbb{R}$, satisfying certain regularity assumptions. For clarity reasons, in the statement of Theorem 4.1.3 we have assumed that the function $(x,t) \mapsto u^t(x)$ belongs to $C^2(\mathbb{R}^n \times I) \cap L^{\infty}(\mathbb{R}^n \times I)$ and $(x,t) \mapsto \partial_t u^t(x)$ is in $L^{\infty}(\mathbb{R}^n \times I)$. We have also assumed that $t \mapsto u^t(x)$ is increasing in I for all $x \in \mathbb{R}^n$. Thus, the graphs of u^t produce a foliation of a certain region in $\mathbb{R}^n \times \mathbb{R}$. Nevertheless, these conditions can be weakened, as presented next, to yield a more satisfactory theory for the fractional Laplacian.

The following is a weaker definition of field than the one in the statement of Theorem 4.1.3, but which suffices to establish the result. On the one hand, we allow the functions u^t in the field to be "wild" outside a neighborhood of Ω , as long as their fractional Laplacian is under control. On the other hand, the leaves u^t can touch each other (but not cross) outside Ω . That is, we need u^t to be increasing in t in $\overline{\Omega}$, but only nondecreasing outside. In particular, the graphs of u^t will only produce a foliation in a certain region of $\overline{\Omega} \times \mathbb{R}$.

Definition 4.4.1. Given an interval $I \subset \mathbb{R}$, a bounded domain $\Omega \subset \mathbb{R}^n$, and $s \in (0, 1)$, we say that a family $\{u^t\}_{t \in I}$ of functions $u^t \colon \mathbb{R}^n \to \mathbb{R}$ is a *field* (to be precise, we should say a *field associated to the s-fractional Laplacian*, not to be inconsistent with Definition 4.1.2) when the following conditions hold:

- (i) The function $(x,t) \mapsto u^t(x)$ is continuous in $\overline{\Omega} \times I$.
- (ii) The function $t \mapsto u^t(x)$ is
 - increasing in I for all $x \in \overline{\Omega}$

- nondecreasing in I for a.e. $x \in \Omega^c$
- C^1 in I for a.e. $x \in \mathbb{R}^n$

(iii) For each compact interval $J \subset I$, we have

$$\sup_{t \in J} \left\{ \|\partial_t u^t\|_{L^{\infty}(\mathbb{R}^n)} + \|u^t\|_{L^1_s(\mathbb{R}^n)} + \|u^t\|_{C^{2s+\alpha}(N)} \right\} < \infty,$$

for some bounded domain $N \subset \mathbb{R}^n$, with $\overline{\Omega} \subset N$, and some $\alpha > 0$.

Essentially, one needs a reasonable regularity of the joint function, as well as some further regularity separately in each of the variables, locally uniformly in the parameter t.

By properties (i) and (ii), the leaf-parameter function $t = t(x, \lambda)$ from (4.1.4) is welldefined and continuous in the region

$$\mathcal{G} := \{ (x, \lambda) \in \overline{\Omega} \times \mathbb{R} \colon \lambda = u^t(x) \text{ for some } t \in I \}.$$
(4.4.2)

The potential F will play no role in the construction of the calibration for $\mathcal{E}_{s,F}$ and, hence, we can focus on the first term. Let

$$\mathcal{E}_s(u) := \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|u(x) - u(y)|^2}{|x - y|^{n+2s}} \,\mathrm{d}x \,\mathrm{d}y$$

and consider the energy space $\dot{H}^s(\Omega) := \{ u \in L^1_{\text{loc}}(\mathbb{R}^n) : \mathcal{E}_s(u) < \infty \}.$

Let $\{u^t\}_{t\in I}$ be a field in the sense of Definition 4.4.1, and let $t_0 \in I$. We consider the functional

$$\mathcal{C}_{s}(w) := \int_{\Omega} \int_{u^{t_{0}}(x)}^{w(x)} (-\Delta)^{s} u^{t}(x) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x + \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|u^{t_{0}}(x) - u^{t_{0}}(y)|^{2}}{|x - y|^{n+2s}} \, \mathrm{d}x \, \mathrm{d}y \quad (4.4.3)$$

acting on continuous functions $w \in C^0(\overline{\Omega})$ with the same exterior datum as u^{t_0} , and such that their graph is contained in $\mathcal{G} \subset \overline{\Omega} \times \mathbb{R}$ when restricted to $\overline{\Omega}$. Here \mathcal{G} has been introduced in (4.4.2). We denote this set of admissible functions by \mathcal{A}_{s,t_0} , that is,

$$\mathcal{A}_{s,t_0} := \left\{ w \in C^0(\overline{\Omega}) \colon w = u^{t_0} \text{ on } \partial\Omega, \quad w = u^{t_0} \text{ a.e. in } \Omega^c, \operatorname{graph}(w_{|\overline{\Omega}}) \subset \mathcal{G} \right\}.$$
(4.4.4)

Remark 4.4.2. The functional C_s is well defined in the set \mathcal{A}_{s,t_0} . Let us check this. For $x \in \Omega$ and λ between $u^{t_0}(x)$ and w(x), we have that

$$t(x,\lambda) \in [t_{\min}, t_{\max}] \subset I,$$

where

$$t_{\min} = \min_{x \in \overline{\Omega}} t(x, w(x))$$
 and $t_{\max} = \max_{x \in \overline{\Omega}} t(x, w(x)).^{17}$ (4.4.5)

Then, on the one hand, since the fractional Laplacians $(-\Delta)^s u^t(x)$ are uniformly bounded in $x \in \Omega$ and $t \in [t_{\min}, t_{\max}]$ by (iii), the iterated integral in the first term in (4.4.3) is finite. On the other hand, taking into account the identity

$$\frac{c_{n,s}}{2} \iint_{Q(\Omega)} \frac{|u^t(x) - u^t(y)|^2}{|x - y|^{n+2s}} \, \mathrm{d}x \, \mathrm{d}y = \int_{\Omega} u^t(x) (-\Delta)^s u^t(x) \, \mathrm{d}x,$$

the second integral in (4.4.3) is finite thanks to the uniform boundedness in $x \in \Omega$ and $t \in [t_{\min}, t_{\max}]$ of each $u^t(x)$ and of the fractional Laplacians $(-\Delta)^s u^t(x)$.

¹⁷Note here that $t(x, u^{t_0}(x)) \equiv t_0$ and that $t_{min} \leq t_0 \leq t_{max}$ since $w \equiv u^{t_0}$ on $\partial \Omega$.

Remark 4.4.3. The functional C_s coincides, in the set A_{s,t_0} , with the functional $C_{s,F}$ appearing in Theorem 4.1.3 when F = 0. Indeed, we can write

$$\begin{split} \int_{\Omega} \int_{u^{t_0}(x)}^{w(x)} (-\Delta)^s u^t(x) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x \\ &= \int_{\Omega} \mathrm{d}x \int_{u^{t_0}(x)}^{w(x)} \mathrm{d}\lambda \lim_{\varepsilon \downarrow 0} \int_{\mathbb{R}^n \setminus \{|x-y| > \varepsilon\}} \mathrm{d}y \, c_{n,s} \frac{u^t(x) - u^t(y)}{|x-y|^{n+2s}} \Big|_{t=t(x,\lambda)} \\ &= \lim_{\varepsilon \downarrow 0} \iint_{(\Omega \times \mathbb{R}^n) \setminus \{|x-y| > \varepsilon\}} \mathrm{d}x \, \mathrm{d}y \int_{u^{t_0}(x)}^{w(x)} \mathrm{d}\lambda \, c_{n,s} \frac{u^t(x) - u^t(y)}{|x-y|^{n+2s}} \Big|_{t=t(x,\lambda)} \\ &= c_{n,s} \lim_{\varepsilon \downarrow 0} \iint_{Q(\Omega) \setminus \{|x-y| > \varepsilon\}} \mathrm{d}x \, \mathrm{d}y \int_{u^{t_0}(x)}^{w(x)} \mathrm{d}\lambda \, \frac{u^t(x) - u^t(y)}{|x-y|^{n+2s}} \Big|_{t=t(x,\lambda)} \end{split}$$

by the regularity of the field and where the last equality follows from the fact that $u^{t_0}(x) = w(x)$ for almost every $x \in \Omega^c$. This last expression gives meaning to the principal value in the definition of $\mathcal{C}_{s,F}$ in Theorem 4.1.3 in the Introduction.

Notice that the expression of C_s in (4.4.3) only involves the Euler-Lagrange equation of the field $\{u^t\}_{t\in I}$ and the energy of the leaf u^{t_0} . This will give both (C1) and the null-Lagrangian property (C3). To prove (C2), we next show, in Lemma 4.4.4, that the functional (4.4.3) can be recast in a useful alternative form. We include Figure 4.2 for the convenience of the reader, to better identify the terms involved in the new expression (of Lemma 4.4.4).

To simplify the statements and proofs below, for $\varepsilon > 0$ we use the truncated kernel $K_{\varepsilon}(z) = c_{n,s}|z|^{-n-2s} \mathbb{1}_{B_{\varepsilon}^{\varepsilon}}(z)$, and for $u \in L_s^1(\mathbb{R}^n)$ we let

$$(-\Delta)^s_{\varepsilon}u(x) = \int_{\mathbb{R}^n} (u(x) - u(y)) K_{\varepsilon}(x - y) \, \mathrm{d}y.$$

In particular, when ε goes to zero we recover the fractional Laplacian $(-\Delta)^s u^t(x) = \lim_{\varepsilon \downarrow 0} (-\Delta)^s_{\varepsilon} u^t(x)$. We also write

$$\mathcal{C}_{s}^{\varepsilon}(w) := \int_{\Omega} \int_{u^{t_{0}}(x)}^{w(x)} (-\Delta)_{\varepsilon}^{s} u^{t}(x) \big|_{t=t(x,\lambda)} d\lambda dx + \frac{1}{4} \iint_{Q(\Omega)} |u^{t_{0}}(x) - u^{t_{0}}(y)|^{2} K_{\varepsilon}(x-y) dx dy.$$

$$(4.4.6)$$

Lemma 4.4.4. Given an interval $I \subset \mathbb{R}$, a bounded domain $\Omega \subset \mathbb{R}^n$, and $s \in (0,1)$, let $\{u^t\}_{t\in I}$ be a field in the sense of Definition 4.4.1. Consider the set of admissible functions \mathcal{A}_{s,t_0} defined in (4.4.4).

Then, for each $\varepsilon > 0$ and $w \in \mathcal{A}_{s,t_0}$, with $t_0 \in I$, the functional $\mathcal{C}_s^{\varepsilon}$ defined in (4.4.6) satisfies

$$\begin{split} \mathcal{C}_{s}^{\varepsilon}(w) &= -\frac{c_{n,s}}{2} \iint_{Q(\Omega) \setminus \{|x-y| < \varepsilon\}} \mathrm{d}x \, \mathrm{d}y \int_{t(x,w(x))}^{t(y,w(y))} \frac{u^{t}(x) - u^{t}(y)}{|x-y|^{n+2s}} \partial_{t} u^{t}(y) \, \mathrm{d}t \\ &+ \frac{c_{n,s}}{4} \iint_{Q(\Omega) \setminus \{|x-y| < \varepsilon\}} \frac{|w(x) - u^{t(x,w(x))}(y)|^{2}}{|x-y|^{n+2s}} \, \mathrm{d}x \, \mathrm{d}y, \end{split}$$

where we have extended the leaf-parameter function $x \mapsto t(x, w(x))$ continuously outside Ω by letting $t(x, w(x)) = t_0$ for $x \in \Omega^c$.



Figure 4.2: The function w (in red) and the leaves u^t .

Proof. Throughout the proof, C_{ε} denotes a generic positive constant depending only on n, s, $|\Omega|$, and ε . It is easy to check that

$$(1+|y|^{n+2s})$$
 $K_{\varepsilon}(x-y) \le C_{\varepsilon}$ for all $x \in \Omega$ and $y \in \mathbb{R}^n$. (4.4.7)

By (4.4.7) we have

$$K_{\varepsilon}(x-y) |u^{t}(x) - u^{t}(y)| \le C_{\varepsilon} \left(||u^{t}||_{C^{0}(\overline{\Omega})} + |u^{t}(y)| \right) \left(1 + |y|^{n+2s} \right)^{-1}$$

for all $x \in \Omega$, a.e. $y \in \mathbb{R}^n$ and all $t \in I$. Hence, integrating in $y \in \mathbb{R}^n$,

$$\int_{\mathbb{R}^n} K_{\varepsilon}(x-y) \left| u^t(x) - u^t(y) \right| \mathrm{d}y \le C_{\varepsilon} \left(\|u^t\|_{C^0(\overline{\Omega})} + \|u^t\|_{L^1_s(\mathbb{R}^n)} \right) \quad \text{for all } x \in \Omega, \ t \in I.$$
(4.4.8)

Consider now t_{\min} and t_{\max} given in (4.4.5). By properties (i) and (iii) in Definition 4.4.1, for $x \in \Omega$ we have

$$\begin{split} \left| \int_{u^{t_0}(x)}^{w(x)} \mathrm{d}\lambda \int_{\mathbb{R}^n} \mathrm{d}y \, K_{\varepsilon}(x-y) \left| u^t(x) - u^t(y) \right| \right|_{t=t(x,\lambda)} \\ & \leq C_{\varepsilon} \sup_{t \in [t_{\min}, t_{\max}]} \left(\|u^t\|_{C^0(\overline{\Omega})} + \|u^t\|_{L^1_s(\mathbb{R}^n)} \right) \|w - u^{t_0}\|_{C^0(\overline{\Omega})} < \infty, \end{split}$$

and we can apply Fubini's theorem to get

$$\int_{\Omega} \mathrm{d}x \int_{u^{t_0}(x)}^{w(x)} \mathrm{d}\lambda (-\Delta)_{\varepsilon}^{s} u^{t}(x) \big|_{t=t(x,\lambda)}$$

$$= \int_{\Omega} \mathrm{d}x \int_{\mathbb{R}^{n}} \mathrm{d}y K_{\varepsilon}(x-y) \int_{u^{t_0}(x)}^{w(x)} (u^{t}(x) - u^{t}(y)) \big|_{t=t(x,\lambda)} \mathrm{d}\lambda.$$
(4.4.9)

Applying the change of variables $\lambda = u^t(x)$ for a.e. $x \in \Omega$ in (4.4.9), the integral becomes

$$\int_{\Omega} \mathrm{d}x \int_{\mathbb{R}^n} \mathrm{d}y \ K_{\varepsilon}(x-y) \int_{u^{t_0}(x)}^{w(x)} (u^t(x) - u^t(y)) \big|_{t=t(x,\lambda)} \mathrm{d}\lambda$$

$$= \int_{\Omega} \mathrm{d}x \int_{\mathbb{R}^n} \mathrm{d}y \ K_{\varepsilon}(x-y) \int_{t_0}^{t(x,w(x))} (u^t(x) - u^t(y)) \partial_t u^t(x) \,\mathrm{d}t.$$
(4.4.10)

Thanks to the extension of the leaf-parameter function by $t(x, w(x)) = t_0$ for $x \in \Omega^c$, using that $Q(\Omega) = (\Omega \times \mathbb{R}^n) \cup (\Omega^c \times \Omega)$, we can rewrite the right-hand side of (4.4.10) as

$$\int_{\Omega} \mathrm{d}x \int_{\mathbb{R}^n} \mathrm{d}y \ K_{\varepsilon}(x-y) \int_{t_0}^{t(x,w(x))} (u^t(x) - u^t(y)) \,\partial_t u^t(x) \,\mathrm{d}t = \iint_{Q(\Omega)} \mathrm{d}x \,\mathrm{d}y \ K_{\varepsilon}(x-y) \int_{t_0}^{t(x,w(x))} (u^t(x) - u^t(y)) \,\partial_t u^t(x) \,\mathrm{d}t.$$

$$(4.4.11)$$

The idea now is to use the symmetry of the domain $Q(\Omega)$ to symmetrize the right-hand side of the previous identity. This will allow us to integrate an exact differential of t, and this will lead to the identity claimed in the lemma.

Symmetrizing (4.4.11), we have

$$\iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \int_{t_0}^{t(x,w(x))} (u^t(x) - u^t(y)) \partial_t u^t(x) \, dt$$

= $\frac{1}{2} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \int_{t_0}^{t(x,w(x))} (u^t(x) - u^t(y)) \partial_t u^t(x) \, dt$ (4.4.12)
 $- \frac{1}{2} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \int_{t_0}^{t(y,w(y))} (u^t(x) - u^t(y)) \partial_t u^t(y) \, dt.$

Splitting the integral $\int_{t_0}^{t(y,w(y))} dt$ into the sum $\int_{t_0}^{t(x,w(x))} dt + \int_{t(x,w(x))}^{t(y,w(y))} dt$ and rearranging terms, the right-hand side of (4.4.12) becomes

$$\frac{1}{2} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \int_{t_0}^{t(x,w(x))} (u^t(x) - u^t(y)) \partial_t u^t(x) \, dt
- \frac{1}{2} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \int_{t_0}^{t(y,w(y))} (u^t(x) - u^t(y)) \partial_t u^t(y) \, dt
= -\frac{1}{2} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \int_{t(x,w(x))}^{t(y,w(y))} (u^t(x) - u^t(y)) \partial_t u^t(y) \, dt
+ \frac{1}{2} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \int_{t_0}^{t(x,w(x))} (u^t(x) - u^t(y)) (\partial_t u^t(x) - \partial_t u^t(y)) \, dt.$$
(4.4.13)

Let us show that the integrals in the right-hand side of (4.4.13) are well defined. For

the first integral, taking absolute values and using Fubini's theorem, we have

$$\frac{1}{2} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \left| \int_{t(x,w(x))}^{t(y,w(y))} |u^{t}(x) - u^{t}(y)| |\partial_{t}u^{t}(y)| \, dt \right| \\
\leq \frac{1}{2} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \int_{t_{\min}}^{t_{\max}} |u^{t}(x) - u^{t}(y)| ||\partial_{t}u^{t}||_{L^{\infty}(\mathbb{R}^{n})} \, dt \\
= \frac{1}{2} \int_{t_{\min}}^{t_{\max}} dt \, ||\partial_{t}u^{t}||_{L^{\infty}(\mathbb{R}^{n})} \iint_{Q(\Omega)} K_{\varepsilon}(x-y) |u^{t}(x) - u^{t}(y)| \, dx \, dy \\
\leq \int_{t_{\min}}^{t_{\max}} dt \, ||\partial_{t}u^{t}||_{L^{\infty}(\mathbb{R}^{n})} \int_{\Omega} dx \int_{\mathbb{R}^{n}} dy K_{\varepsilon}(x-y) |u^{t}(x) - u^{t}(y)|,$$
(4.4.14)

where in the last line we have used that $Q(\Omega) = (\Omega \times \mathbb{R}^n) \cup (\mathbb{R}^n \times \Omega)$ (not a disjoint union) and the symmetry of K_{ε} . Applying the bound (4.4.8) in (4.4.14) and by property (iii) in Definition 4.4.1, we deduce the finiteness of (4.4.14). It follows that the first integral is well defined.

The second integral in the right-hand side of (4.4.13) can be integrated explicitly as

$$\frac{1}{2} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \int_{t_0}^{t(x,w(x))} (u^t(x) - u^t(y)) (\partial_t u^t(x) - \partial_t u^t(y)) \, dt
= \frac{1}{4} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \int_{t_0}^{t(x,w(x))} \frac{d}{dt} |u^t(x) - u^t(y)|^2 \, dt
= \frac{1}{4} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) |w(x) - u^{t(x,w(x))}(y)|^2
- \frac{1}{4} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) |u^{t_0}(x) - u^{t_0}(y)|^2.$$
(4.4.15)

Concatenating the equalities (4.4.9), (4.4.10), (4.4.11), (4.4.12), and (4.4.13), and using (4.4.15), we conclude

$$\begin{split} \int_{\Omega} \mathrm{d}x \int_{u^{t_0}(x)}^{w(x)} \mathrm{d}\lambda(-\Delta)_{\varepsilon}^{s} u^{t}(x) \big|_{t=t(x,\lambda)} \\ &= -\frac{1}{2} \iint_{Q(\Omega)} \mathrm{d}x \, \mathrm{d}y K_{\varepsilon}(x-y) \int_{t(x,w(x))}^{t(y,w(y))} (u^{t}(x) - u^{t}(y)) \partial_{t} u^{t}(y) \, \mathrm{d}t \\ &\quad + \frac{1}{4} \iint_{Q(\Omega)} \mathrm{d}x \, \mathrm{d}y K_{\varepsilon}(x-y) |w(x) - u^{t(x,w(x))}(y)|^{2} \\ &\quad - \frac{1}{4} \iint_{Q(\Omega)} \mathrm{d}x \, \mathrm{d}y K_{\varepsilon}(x-y) |u^{t_0}(x) - u^{t_0}(y)|^{2}, \end{split}$$

which is the claim of the lemma.

Having Lemma 4.4.4 at hand, we can now prove the calibration property (C2).

Lemma 4.4.5. Given an interval $I \subset \mathbb{R}$, a bounded domain $\Omega \subset \mathbb{R}^n$, and $s \in (0,1)$, let $\{u^t\}_{t\in I}$ be a field in the sense of Definition 4.4.1. Consider the set of admissible functions \mathcal{A}_{s,t_0} defined in (4.4.4).

Then, for all $w \in \mathcal{A}_{s,t_0}$, with $t_0 \in I$, we have

$$\mathcal{C}_s(w) \le \mathcal{E}_s(w).$$

Proof. Let $\varepsilon > 0$. By Lemma 4.4.4, we have

$$C_{s}^{\varepsilon}(w) = -\frac{1}{2} \iint_{Q(\Omega)} dx \, dy K_{\varepsilon}(x-y) \int_{t(x,w(x))}^{t(y,w(y))} (u^{t}(x) - u^{t}(y)) \partial_{t} u^{t}(y) \, dt + \frac{1}{4} \iint_{Q(\Omega)} |w(x) - u^{t(x,w(x))}(y)|^{2} K_{\varepsilon}(x-y) \, dx \, dy,$$
(4.4.16)

where $t(x, w(x)) = t_0$ for $x \in \Omega^c$. We claim that for a.e. $(x, y) \in Q(\Omega)$ we have

$$-\frac{1}{2} \int_{t(x,w(x))}^{t(y,w(y))} (u^t(x) - u^t(y))\partial_t u^t(y) \, \mathrm{d}t \le -\frac{1}{2} \int_{t(x,w(x))}^{t(y,w(y))} (w(x) - u^t(y))\partial_t u^t(y) \, \mathrm{d}t. \quad (4.4.17)$$

Indeed, first of all note that $\partial_t u^t(y) \ge 0$ for a.e. $y \in \mathbb{R}^n$ by property (ii) in Definition 4.4.1. Moreover, the quantities $u^t(x)$, $u^t(y)$, w(x), and $\partial_t u^t(y)$ are finite for a.e. $(x, y) \in Q(\Omega)$ and all $t \in I$. For those $(x, y) \in Q(\Omega)$, if $t(x, u(x)) \le t(y, u(y))$ then by property (ii) we have $w(x) = u^{t(x,w(x))}(x) \le u^t(x)$ for $t \in [t(x, u(x)), t(y, u(y))]$, and the claim follows in this case. When $t(x, u(x)) \ge t(y, u(y))$ the argument is similar.

The right-hand side of (4.4.17) can be integrated explicitly as

$$-\frac{1}{2} \int_{t(x,w(x))}^{t(y,w(y))} (w(x) - u^{t}(y))\partial_{t}u^{t}(y) dt = \frac{1}{4} \int_{t(x,w(x))}^{t(y,w(y))} \frac{d}{dt} |w(x) - u^{t}(y)|^{2} dt$$
$$= \frac{1}{4} |w(x) - w(y)|^{2} - \frac{1}{4} |w(x) - u^{t(x,w(x))}(y)|^{2}.$$
(4.4.18)

Now, using (4.4.17) and (4.4.18) in (4.4.16), it follows that

$$\mathcal{C}_s^{\varepsilon}(w) \leq \mathcal{E}_{K_{\varepsilon}}(w).$$

Finally, by property (iii) in Definition 4.4.1, $(-\Delta)^s_{\varepsilon} u^t$ converge to $(-\Delta)^s u^t$ in $L^1(\Omega)$, locally uniformly in t, as $\varepsilon \downarrow 0$. This is enough to pass to the limit in the inequality above and conclude the proof.

We can finally give the proof of Theorem 4.1.3. We will show the identity

$$\mathcal{C}_{s,F}(w) = \int_{\Omega} \int_{u^{t_0}(x)}^{w(x)} \left((-\Delta)^s u^t(x) - F'(u^t(x)) \right) \Big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x + \mathcal{E}_{s,F}(u^{t_0}) \tag{4.4.19}$$

for $w \in \mathcal{A}_{s,t_0}$, and that this functional is a calibration for $\mathcal{E}_{s,F}$ and each u^t when the family $\{u^t\}_{t\in I}$ is a field of extremals, that is, when each u^t solves the semilinear equation (4.4.1). In particular, each of the u^t will be a minimizer. More generally, we show that u^{t_0} minimizes $\mathcal{E}_{s,F}$ if the u^t above u^{t_0} are supersolutions of (4.4.1) and the u^t below are subsolutions.

Proof of Theorem 4.1.3. First of all, note that for $x \in \Omega$ we have

$$F(w(x)) - F(u^{t_0}(x)) = \int_{u^{t_0}(x)}^{w(x)} F'(\lambda) \, \mathrm{d}\lambda = \int_{u^{t_0}(x)}^{w(x)} F'(u^{t(x,\lambda)}) \, \mathrm{d}\lambda$$

and, thanks to Remark 4.4.3, the functional $\mathcal{C}_{s,F}$ given by (4.1.9) can be written simply as

$$\mathcal{C}_{s,F}(w) = \mathcal{C}_s(w) - \int_{\Omega} F(w(x)) \,\mathrm{d}x$$

where C_s has been introduced in (4.4.3). This proves (4.4.19).

(a) From (4.4.19) it is clear that $\mathcal{C}_{s,F}(u^{t_0}) = \mathcal{E}_{s,F}(u^{t_0})$, which is condition (C1). To obtain (C2) we apply Lemma 4.4.5, which gives

$$\mathcal{C}_{s,F}(w) = \mathcal{C}_s(w) - \int_{\Omega} F(w(x)) \, \mathrm{d}x \le \mathcal{E}_s(u) - \int_{\Omega} F(w(x)) \, \mathrm{d}x = \mathcal{E}_{s,F}(w).$$

(b) To show (C3'), that is, $C_{s,F}(w) \geq C_{s,F}(u^{t_0}) = \mathcal{E}_{s,F}(u^{t_0})$, by (4.4.19) it suffices to show that

$$\int_{u^{t_0}(x)}^{w(x)} \left((-\Delta)^s u^t(x) - F'(u^t(x)) \right) \Big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \ge 0$$

for $x \in \Omega$. But this is clear from the monotonicity of the field u^t and the hypotheses in (b).

On the other hand, we have already seen in subsection 4.1.1 how properties (C1), (C2), and (C3') yield the minimality of $u = u^{t_0}$.

(c) By (4.4.19), using that each u^t satisfies the Euler-Lagrange equation (4.4.1), we have that

$$\mathcal{C}_{s,F}(w) - \mathcal{E}_{s,F}(u^{t_0}) = \int_{\Omega} \int_{u^{t_0}(x)}^{w(x)} \left((-\Delta)^s u^t(x) - F'(u^t(x)) \right) \Big|_{t=t(x,\lambda)} \mathrm{d}\lambda \,\mathrm{d}x = 0$$

Hence $\mathcal{C}_{s,F}(w) = \mathcal{E}_{s,F}(u^{t_0}) = \mathcal{C}_{s,F}(u^{t_0})$ for all $w \in \mathcal{A}_{s,t_0}$. In particular, the functional $\mathcal{C}_{s,F}$ satisfies all three properties $(\mathcal{C}1), (\mathcal{C}2)$, and $(\mathcal{C}3)$, and thus it is a calibration. Choosing $t_0 = t$ for each $t \in I$, we deduce the minimality of u^t .

Proof of Corollary 4.1.4. First of all, by a simple argument from [1], it suffices to prove the corollary for the class of functions $w \in C^0$ satisfying the strict inequality

$$\lim_{\tau \to -\infty} u(x',\tau) < w(x',x_n) < \lim_{\tau \to +\infty} u(x',\tau) \quad \text{ for all } (x',x_n) \in \mathbb{R}^{n-1} \times \mathbb{R}.$$
(4.4.20)

Indeed, if w satisfies the non-strict inequality (4.1.12), then for $\theta \in (0, 1)$ we consider $w_{\theta} := (1 - \theta)u + \theta w$, which satisfies (4.4.20) by the strict monotonicity of u. Hence, applying the corollary in the strict case, we have $\mathcal{E}_{s,F}(u) \leq \mathcal{E}_{s,F}(w_{\theta})$. Letting $\theta \to 1^-$ yields the result for w. Hence, we may assume (4.4.20).

Since u is bounded and $F \in C^3$, by regularity theory for the fractional Laplacian, we have that u is at least C^2 and $\nabla u \in L^{\infty}(\mathbb{R}^n)$; see [84]. For each $t \in \mathbb{R}$, we consider the family of translations $u^t(x) = u(x', x_n + t)$. By the monotonicity and the regularity properties of u, the family $\{u^t\}_{t\in\mathbb{R}}$ is a field in the sense of Definition 4.4.1. Moreover, by the translation invariance of the equation, it is a field of extremals. Hence, we can apply Theorem 4.1.3 to conclude that u is a minimizer in the set of admissible functions \mathcal{A}_s with w = u in Ω^c .

Chapter 5

Null-Lagrangians and calibrations for general nonlocal functionals and an application to the viscosity theory

In this chapter we build a null-Lagrangian and a calibration for general nonlocal elliptic functionals in the presence of a field of extremals. Thus, our construction assumes the existence of a family of solutions to the Euler-Lagrange equation whose graphs produce a foliation. Then, as a consequence of the calibration, we show the minimality of each leaf in the foliation. Our model case is the energy functional for the fractional Laplacian, for which such a null-Lagrangian was recently discovered by us.

As a first application of our calibration, we show that monotone solutions to translation invariant nonlocal equations are minimizers. Our second application is perhaps surprising, since here "minimality" is assumed instead of being concluded. We will see that the foliation framework is large enough to provide a proof showing that minimizers of nonlocal elliptic functionals are viscosity solutions.

5.1 Introduction

Null-Lagrangians and calibrations have played a prominent role in the Calculus of Variations, since they provide sufficient conditions for the minimality of critical points. Important examples are those calibrations constructed in the presence of a *field of extremals*, i.e., a foliation by critical points. These notions have their origin in the classical extremal field theory of Weierstrass and are a powerful tool to prove minimality of solutions to PDEs. Especially, they have found many relevant applications in the context of minimal surfaces.

In Chapter 4 we initiated the study of calibrations for nonlocal problems. There, we treated the simplest nonlocal model: the energy functional for the fractional Laplacian (the Gagliardo-Sobolev seminorm). In the present chapter, we extend the theory to a wide class of nonlocal functionals. Our main result is the construction of a calibration for the energy functional¹

$$\mathcal{E}_{\mathrm{N}}(w) := \frac{1}{2} \iint_{Q(\Omega)} G_{\mathrm{N}}(x, y, w(x), w(y)) \,\mathrm{d}x \,\mathrm{d}y,$$

¹Consistent with the notation in [23], the subindices N and L are used throughout the text to denote nonlocal and local objects, respectively.

where, given a bounded domain $\Omega \subset \mathbb{R}^n$, we have written

$$Q(\Omega) := (\mathbb{R}^n \times \mathbb{R}^n) \setminus (\Omega^c \times \Omega^c) = (\Omega \times \Omega) \cup (\Omega \times \Omega^c) \cup (\Omega^c \times \Omega).$$
(5.1.1)

Throughout the chapter $\Omega^c = \mathbb{R}^n \setminus \Omega$, as above. The Lagrangian $G_N(x, y, a, b)$ is required to satisfy the natural ellipticity condition

$$\partial_{ab}^2 G_{\rm N}(x, y, a, b) + \partial_{ab}^2 G_{\rm N}(y, x, b, a) \le 0,$$

on which we elaborate below (see the comments before Theorem 5.1.3 and also Section 5.2).

As in the local theory, as well as in our fractional Laplacian theory developed in the previous chapter, our calibration for \mathcal{E}_{N} is built in the presence of a field of extremals. As mentioned above, this is a one-parameter family of critical points of \mathcal{E}_{N} whose graphs form a foliation (see Definition 5.1.2). For the construction, it suffices to have subsolutions and supersolutions on each respective side of a given extremal, a fact that is sometimes very useful.

A first application of our calibration concerns the minimality of monotone solutions to translation invariant nonlocal equations. More precisely, we prove that if u is a solution (with an appropriate regularity and growth at infinity, which will depend on the Lagrangian G_N) satisfying $\partial_{x_n} u > 0$ in \mathbb{R}^n , then it is a minimizer among functions w satisfying

$$\lim_{\tau \to -\infty} u(x', \tau) \le w(x', x_n) \le \lim_{\tau \to +\infty} u(x', \tau)$$

for $x = (x', x_n) \in \mathbb{R}^{n-1} \times \mathbb{R}$. This result, which is related to a celebrated conjecture of De Giorgi for the Allen-Cahn equation, was only known for those nonlocal functionals for which an existence and regularity theory of minimizers is available. We explain this further in Subsection 5.1.4.

As a second application, we show that minimizers of nonlocal elliptic functionals are viscosity solutions. This type of result was previously known for problems where a weak comparison principle is available; see [5, 66, 87]. However, we can prove it in more general scenarios by using the calibration technique; see Subsection 5.1.5. This strategy was previously used by Cabré [18] in the context of nonlocal minimal surfaces.

5.1.1 Examples

Our theory covers several important elliptic functionals \mathcal{E}_{N} given by a Lagrangian G_{N} as above:

• The case

$$G_{\rm N}(x, y, a, b) = \frac{|a - b|^p}{2p|x - y|^{n + ps}}$$

with $p \in [1, \infty)$ and $s \in (0, 1)$, corresponds to the fractional *p*-Dirichlet Lagrangian, which gives rise to the fractional *p*-Laplace equation. More generally, considering

$$G_{\rm N}(x, y, a, b) = \frac{|a - b|^p}{2p|x - y|^{n + ps}} - \frac{1}{2|\Omega|} \mathbb{1}_{\Omega \times \Omega}(x, y)(F(a, x) + F(b, y)),$$

we can add a reaction term in the Euler-Lagrange equation. For instance, if we take p = 2 we recover (up to a multiplicative constant) the Lagrangian associated to the fractional

semilinear equation $(-\Delta)^s u = \partial_u F(u, x)$ in Ω , treated in our previous work [23]. Recall the expression for the fractional Laplacian:

$$(-\Delta)^{s} u(x) = c_{n,s} \text{ P.V.} \int_{\mathbb{R}^{n}} \frac{u(x) - u(y)}{|x - y|^{n+2s}} \, \mathrm{d}y,$$

where $c_{n,s}$ is a positive normalizing constant and P.V. stands for the principal value.

• The Lagrangian

$$G_{\mathrm{N}}(x, y, a, b) = \frac{G\left(\frac{a-b}{|x-y|}\right)}{|x-y|^{n+s-1}},$$

where $s \in (0,1)$, $G''(\tau) = (1 + \tau^2)^{-(n+s+1)/2}$, and G(0) = G'(0) = 0, recovers the fractional perimeter for subgraphs; see [41].

• The general structure

$$G_{\rm N}(x, y, a, b) = G(x - y, a - b)$$

appears in the leading terms of the previous examples and gives rise to translation invariant equations. However, it is also of interest to treat functionals where the interactions occur only inside Ω , that is, when G_N is of the form

$$G_{\mathcal{N}}(x, y, a, b) = \mathbb{1}_{\Omega \times \Omega}(x, y) \ G(x - y, a - b).$$

These Lagrangians appear, for instance, in the macroelastic energy from Peridynamics; see [88]. In this case, G might be compactly supported in the (x - y)-variable.

• The case

$$G_{\mathcal{N}}(x, y, a, b) = -\mathbb{1}_{\Omega \times \Omega}(x, y)K(x - y)ab + \frac{1}{2|\Omega|}\mathbb{1}_{\Omega \times \Omega}(x, y)(F(a) + F(b))$$

corresponds to convolution-type operators. Functionals of this type appear in numerous problems, but most notably in the framework of constrained minimization (not treated in our setting); see, for instance, [4, 7, 35, 70] where the first term is the interaction energy and the second one is the entropy. Let us point out that, when the kernel K is even, the ellipticity condition boils down to the nonnegativity of K.

5.1.2 Calibrations and fields of extremals

As mentioned in the previous chapter, a fundamental problem in the Calculus of Variations is to find conditions for a function to be a minimizer of a given energy functional. More precisely, given a functional $\mathcal{E} \colon \mathcal{A} \to \mathbb{R}$ defined on some set of admissible functions \mathcal{A} , and given $u \in \mathcal{A}$, one wishes to know whether u minimizes \mathcal{E} among competitors in \mathcal{A} having the same Dirichlet condition as u. For nonlocal problems, given a bounded domain Ω , the Dirichlet condition refers to the value of the function in all the exterior of Ω , namely, in $\Omega^c = \mathbb{R}^n \setminus \Omega$.

One useful method to show the minimality of a given function $u \in \mathcal{A}$ consists of constructing a calibration. This is an auxiliary functional touching the energy \mathcal{E} by below at u and satisfying a null-Lagrangian property.

Definition 5.1.1. A functional $C: \mathcal{A} \to \mathbb{R}$ is a *calibration* for the functional \mathcal{E} and the admissible function $u \in \mathcal{A}$ if the following conditions hold:

- $(\mathcal{C}1) \ \mathcal{C}(u) = \mathcal{E}(u).$
- (C2) $\mathcal{C}(w) \leq \mathcal{E}(w)$ for all $w \in \mathcal{A}$ with the same Dirichlet condition as u.
- (C3) $\mathcal{C}(w) = \mathcal{C}(\widetilde{w})$ for all $w, \widetilde{w} \in \mathcal{A}$ with the same Dirichlet condition as u.

Functionals satisfying (C3) are known as *null-Lagrangians*. As in our previous work [23], it is convenient to relax this last condition to the less stringent

 $(\mathcal{C}3')$ $\mathcal{C}(u) \leq \mathcal{C}(w)$ for all $w \in \mathcal{A}$ with the same Dirichlet condition as u.

We still refer to functionals satisfying (C1), (C2), and (C3') as calibrations.

Once a calibration is available, the minimality of u among admissible functions with the same Dirichlet condition follows immediately. For this, simply apply (C1), (C3'), and (C2), in this order.

Historically, motivated by classical problems in Mechanics and Geometry, significant efforts have been put into rigorously understanding minimizers of general functionals of the form

$$\mathcal{E}_{\mathrm{L}}(w) := \int_{\Omega} G_{\mathrm{L}}(x, w(x), \nabla w(x)) \,\mathrm{d}x.$$
(5.1.2)

It is well-known that every minimizer is a critical point of \mathcal{E}_{L} (an *extremal*) and must satisfy the associated Euler-Lagrange equation. Conversely, if the Lagrangian $G_{L}(x, \lambda, q)$ is convex in the variables (λ, q) , then the functional \mathcal{E}_{L} is convex and every critical point is a minimizer. This convexity assumption is too restrictive for many relevant applications, such as the Allen-Cahn energy. For these functionals, the Dirichlet problem may admit several extremals, not all of them being minimizers. Nevertheless, one often has that the Lagrangian $G_{L}(x, \lambda, q)$ is convex with respect to the gradient variable q, which amounts to the *ellipticity* of the problem.

A systematic theory of calibrations has been developed for functionals \mathcal{E}_{L} of the form (5.1.2), namely, the *extremal field theory* going back to works of Weierstrass. The key idea is to assume the existence of a family of critical points $u^{t} : \overline{\Omega} \to \mathbb{R}$, with t in some interval $I \subset \mathbb{R}$, whose graphs do not intersect each other. Thus, the graphs of these functions produce a foliation of a certain region \mathcal{G} in $\mathbb{R}^{n} \times \mathbb{R}$, which allows to carry out a subtle convexity argument to bound the nonconvex functional by below with a calibration.

Next, we recall our definition of field for nonlocal problems, as introduced in [23]:

Definition 5.1.2. Given an interval $I \subset \mathbb{R}$ (not necessarily bounded, nor open), we say that a family $\{u^t\}_{t\in I}$ of functions $u^t \colon \mathbb{R}^n \to \mathbb{R}$ is a *field in* \mathbb{R}^n if

- the function $(x,t) \mapsto u^t(x)$ is continuous in $\mathbb{R}^n \times I$;
- for each $x \in \mathbb{R}^n$, the function $t \mapsto u^t(x)$ is C^1 and increasing in I.

Given a functional \mathcal{E} acting on functions defined in \mathbb{R}^n , and given a bounded domain $\Omega \subset \mathbb{R}^n$, we say that $\{u^t\}_{t \in I}$ is a *field of extremals* in Ω (for \mathcal{E}) when it is a field in \mathbb{R}^n and each of the functions u^t is a critical point of \mathcal{E} in Ω .

Given a field in \mathbb{R}^n as above, the region

$$\mathcal{G} = \{ (x, \lambda) \in \mathbb{R}^n \times \mathbb{R} \colon \lambda = u^t(x) \text{ for some } t \in I \} \subset \mathbb{R}^n \times \mathbb{R}$$

is foliated by the graphs of the functions u^t , which do not intersect each other (since $u^t(x)$ is increasing in t). In particular, we can uniquely define a *leaf-parameter function*

$$t: \mathcal{G} \to I, \quad (x,\lambda) \mapsto t(x,\lambda) \quad \text{determined by} \quad u^{t(x,\lambda)}(x) = \lambda.$$
 (5.1.3)

The function t is continuous in \mathcal{G} by the assumptions in Definition 5.1.2. We will often refer to the functions u^t (or their graphs) as the "*leaves*" of the field.

Next, let us recall the fundamental result of the classical extremal field theory. Namely, given an elliptic Lagrangian² $G_{\rm L}$ and $\{u^t\}_{t\in I}$ a smooth field of extremals in Ω , the functional

$$\mathcal{C}_{\mathrm{L}}(w) := \int_{\Omega} \left\{ \partial_{q} G_{\mathrm{L}}(x, u^{t}(x), \nabla u^{t}(x)) \cdot \left(\nabla w(x) - \nabla u^{t}(x) \right) \right\} \Big|_{t=t(x, w(x))} \mathrm{d}x + \int_{\Omega} G_{\mathrm{L}}(x, u^{t}(x), \nabla u^{t}(x)) \Big|_{t=t(x, w(x))} \mathrm{d}x,$$
(5.1.4)

is a calibration for the functional \mathcal{E}_{L} and each critical point $u^{t_0}, t_0 \in I$. In particular, each leaf u^{t_0} minimizes \mathcal{E}_{L} among competitors w satisfying $w = u^{t_0}$ on $\partial\Omega$ and whose graphs lie in the region \mathcal{G} . Moreover, in [23, Theorem 3.1] we found the following alternative expression for the calibration \mathcal{C}_{L} . For each $t_0 \in I$, we have that \mathcal{C}_{L} in (5.1.4) can be written as

$$\mathcal{C}_{\mathrm{L}}(w) = \int_{\Omega} \int_{u^{t_0}(x)}^{w(x)} \mathcal{L}_{\mathrm{L}}(u^{t})(x) \Big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x + \int_{\partial\Omega} \int_{u^{t_0}(x)}^{w(x)} \mathcal{N}_{\mathrm{L}}(u^{t})(x) \Big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}\mathcal{H}^{n-1}(x) + \mathcal{E}_{\mathrm{L}}(u^{t_0}),$$
(5.1.5)

where \mathcal{L}_{L} and \mathcal{N}_{L} are, respectively, the Euler-Lagrange and Neumann operators associated to the functional \mathcal{E}_{L} in (5.1.2). As in the fractional Laplacian framework treated in [23], our new nonlocal calibration given in Theorem 5.1.3 below will be based on this identity.

While the theory of calibrations for local equations is well understood, there are very few papers prior to [23] dealing with nonlocal ones, which we mention next. In [18], Cabré gave a calibration for the fractional perimeter. Independently, Pagliari [80] investigated the abstract structure of calibrations for the fractional total variation. This last functional involves the fractional perimeter of each sublevel set of a given function. The author succeeded in constructing a calibration to prove that the characteristic functions of halfspaces are minimizers, but other fields of extremals are not mentioned in that work. Our present work provides, as a particular case, a calibration for the fractional total variation in the presence of a general field of extremals. Moreover, we can relate our construction with the calibration for the fractional perimeter in [18] applied to each superlevel set; see Appendix J.

In Chapter 4 we constructed a calibration for the energy associated to semilinear equations involving the fractional Laplacian, that is, for energies of the form

$$\mathcal{E}_{s,F}(w) = \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|w(x) - w(y)|^2}{|x - y|^{n + 2s}} \,\mathrm{d}x \,\mathrm{d}y - \int_{\Omega} F(w(x)) \,\mathrm{d}x.$$

$$G_{\mathcal{L}}(x, u^{t}(x), q) \geq G_{\mathcal{L}}(x, u^{t}(x), \nabla u^{t}(x)) + \partial_{q}G_{\mathcal{L}}(x, u^{t}(x), \nabla u^{t}(x)) \cdot (q - \nabla u^{t}(x))$$

for all $x \in \Omega$, $q \in \mathbb{R}^n$, and $t \in I$; see [23] for more details.

²Recall that here ellipticity means that $G_{\rm L}(x,\lambda,q)$ is convex with respect to the gradient variable q. However, for (5.1.4) to be a calibration, a weaker condition than convexity in q suffices. One needs to assume that each u^t satisfies the Weierstrass sufficient condition, namely

Given $\{u^t\}_{t\in I}$ a field of extremals in Ω , we showed that

$$\begin{aligned} \mathcal{C}_{s,F}(w) &= c_{n,s} \operatorname{P.V.} \iint_{Q(\Omega)} \int_{u^{t_0}(x)}^{w(x)} \frac{u^t(x) - u^t(y)}{|x - y|^{n + 2s}} \bigg|_{t = t(x,\lambda)} \mathrm{d}\lambda \, \mathrm{d}x \, \mathrm{d}y - \int_{\Omega} F(w(x)) \, \mathrm{d}x \\ &+ \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|u^{t_0}(x) - u^{t_0}(y)|^2}{|x - y|^{n + 2s}} \, \mathrm{d}x \, \mathrm{d}y \end{aligned}$$

is a calibration for $\mathcal{E}_{s,F}$ and u^{t_0} , $t_0 \in I$. We recall that the expression of $\mathcal{C}_{s,F}$ was obtained by replacing the operators \mathcal{L}_{L} and \mathcal{N}_{L} appearing in (5.1.5) by their nonlocal counterparts.

5.1.3 Main result

Next, we present our main result, which builds a calibration for the functional

$$\mathcal{E}_{\mathrm{N}}(w) = \frac{1}{2} \iint_{Q(\Omega)} G_{\mathrm{N}}(x, y, w(x), w(y)) \,\mathrm{d}x \,\mathrm{d}y \tag{5.1.6}$$

when the Lagrangian $G_{\rm N}(x, y, a, b)$ satisfying the condition³

$$\partial_{ab}^2 G_N(x, y, a, b) + \partial_{ab}^2 G_N(y, x, b, a) \le 0.$$
 (5.1.7)

We will see that (5.1.7) guarantees the ellipticity of the problem (or a strong comparison principle, see Appendix I). It will also ensure that the calibrating functional defined in Theorem 5.1.3 below satisfies property (C2), thus mirroring the effect of ellipticity in the local case.

As in the classical theory, in this general nonlocal framework every extremal is a minimizer whenever the functional \mathcal{E}_N is convex. A sufficient condition to guarantee the convexity of \mathcal{E}_N is that the Lagrangian $(a, b) \mapsto G_N(x, y, a, b) + G_N(y, x, b, a)$ be convex. Contrary to the local case, this hypothesis does not guarantee the ellipticity assumption (5.1.7).⁴ This seems to be due to the great generality of (5.1.6). Nevertheless, in most examples we have in mind, the Lagrangian has a leading term of the form G(x, y, a - b) for which ellipticity does follow from convexity in the (a - b)-variable. For instance, consider the linear equation $(-\Delta)^s u = \lambda u$, with $\lambda \in \mathbb{R}$. This equation admits an energy functional

$$\frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|u(x) - u(y)|^2}{|x - y|^{n + 2s}} \, \mathrm{d}x \, \mathrm{d}y - \frac{\lambda}{2} \int_{\Omega} u(x)^2 \, \mathrm{d}x$$

which is elliptic in the sense of (5.1.7) but not always convex (when λ is large enough). Notice that the equation satisfies the strong comparison principle, while the availability of the weak comparison principle depends on λ .⁵

³Compare it with (5.2.15) in Section 5.2, where the Lagrangian additionally satisfies the pairwise symmetry condition (5.2.1) (a condition that can always be assumed without loss of generality).

⁴For pairwise symmetric Lagrangians (see (5.2.1) below) ellipticity reduces to $\partial_{ab}^2 G_N \leq 0$ while convexity amounts to the conditions $\partial_{aa}^2 G_N \geq 0$ and $\partial_{aa}^2 G_N \partial_{bb}^2 G_N \geq (\partial_{ab}^2 G_N)^2$. The reader can check the nonequivalence with the simple quadratic examples $G_N = \pm K(x-y)(a\pm b)^2$, with K > 0.

⁵We say that an operator \mathcal{L} satisfies the *strong comparison principle* if, given two functions u and v satisfying $\mathcal{L}u \leq \mathcal{L}v$ in Ω , $u \leq v$ in \mathbb{R}^n , and touching somewhere in Ω , then $u \equiv v$ in \mathbb{R}^n . By contrast, \mathcal{L} satisfies the *weak comparison principle* if, given two functions u and v satisfying $\mathcal{L}u \leq \mathcal{L}v$ in Ω and $u \leq v$ in Ω^c , then $u \leq v$ in Ω .

For the functional \mathcal{E}_N and its associated Euler-Lagrange operator

$$\mathcal{L}_{\mathcal{N}}(w)(x) := \frac{1}{2} \int_{\mathbb{R}^n} \left(\partial_a G_{\mathcal{N}}(x, y, w(x), w(y)) + \partial_b G_{\mathcal{N}}(y, x, w(y), w(x)) \right) dy$$
(5.1.8)

to be well defined for $x \in \Omega$, one needs to make growth and regularity assumptions on the Lagrangian G_N . These determine the class of admissible functions; see [15] for some examples of natural assumptions. In this respect, our main result (Theorem 5.1.3 below), which gives a calibration for general nonlocal Lagrangians satisfying the ellipticity hypothesis (5.1.7), is a formal result since it does not specify the precise class of admissible functions. In other words, the great generality of the functions does not allow for specifying the growth and regularity assumptions on G_N and on the admissible functions w. Thus, the theorem cannot take into account any integrability issues.⁶ However, we could give completely rigorous results for some specific families of Lagrangians, adapting the admissible class of functions to the concrete problem. Indeed, within the proof of the next theorem, there are only a few points that must be justified, namely, the interchange of certain integrals and the convergence of some expressions. Hence, in the following statement we use the term "sufficiently regular for G_N " in the sense that those functions make all integrals to be well defined.

Recall (5.1.1) for the meaning of $Q(\Omega)$, Definition 5.1.2 for the notion of field, and (5.1.3) for the leaf-parameter function t. The calibration properties (C1), (C2), (C3), and (C3') have been introduced in Definition 5.1.1.

Theorem 5.1.3. Let $I \subset \mathbb{R}$ be an interval and let $\Omega \subset \mathbb{R}^n$ be a bounded domain. Given a function $G_N = G_N(x, y, a, b)$ satisfying the ellipticity condition (5.1.7), let $\{u^t\}_{t \in I}$ be a field in \mathbb{R}^n (in the sense of Definition 5.1.2) which is sufficiently regular for G_N .

Given $t_0 \in I$, let \mathcal{E}_N be defined by (5.1.6) and \mathcal{C}_N be the functional

$$\mathcal{C}_{\mathrm{N}}(w)$$

$$:= \frac{1}{2} \iint_{Q(\Omega)} \int_{u^{t_0}(x)}^{w(x)} \Big\{ \partial_a G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) + \partial_b G_{\mathcal{N}}(y, x, u^t(y), u^t(x)) \Big\} \Big|_{t=t(x,\lambda)} \mathrm{d}\lambda \,\mathrm{d}x \,\mathrm{d}y \\ + \mathcal{E}_{\mathcal{N}}(u^{t_0}) \Big\}$$

defined in a set \mathcal{A}_N of sufficiently regular admissible functions $w \colon \mathbb{R}^n \to \mathbb{R}$ (for G_N) satisfying graph $w \subset \mathcal{G}$, where

$$\mathcal{G} = \{ (x, \lambda) \in \mathbb{R}^n \times \mathbb{R} \colon \lambda = u^t(x) \quad \text{for some } t \in I \}.$$

Taking $C = C_N$ and $\mathcal{E} = \mathcal{E}_N$ in Definition 5.1.1, we have the following:

(a) $C_{\rm N}$ satisfies (C1) and (C2) with $u = u^{t_0}$.

(b) Assume in addition that the family $\{u^t\}_{t\in I}$ satisfies

$$\mathcal{L}_{N}(u^{t}) \geq 0 \quad in \ \Omega \quad for \ t \geq t_{0}, \\ \mathcal{L}_{N}(u^{t}) \leq 0 \quad in \ \Omega \quad for \ t \leq t_{0},$$

where \mathcal{L}_{N} is the Euler-Lagrange operator associated to \mathcal{E}_{N} given by (5.1.8). Then, \mathcal{C}_{N} satisfies (C3') with $u = u^{t_{0}}$. In particular, $u^{t_{0}}$ minimizes \mathcal{E}_{N} among functions w in \mathcal{A}_{N} such that $w \equiv u^{t_{0}}$ in Ω^{c} .

 $^{^{6}}$ This is in contrast with Theorem 1.3 in [23], where we gave a fully rigorous result for the fractional Laplacian.

(c) Assume in addition that $\{u^t\}_{t\in I}$ is a field of extremals in Ω , that is, a field in \mathbb{R}^n satisfying

$$\mathcal{L}_{N}(u^{t}) = 0$$
 in Ω for all $t \in I$.

Then, the functional C_N satisfies (C3) with $u = u^{t_0}$. Therefore, C_N is a calibration for \mathcal{E}_N and u^{t_0} . As a consequence, for every $t \in I$, the extremal u^t minimizes \mathcal{E}_N among functions w in \mathcal{A}_N such that $w \equiv u^t$ in Ω^c .

As mentioned above, the class of functionals \mathcal{E}_{N} of the form (5.1.6) satisfying the ellipticity condition (5.1.7) includes the Gagliardo-Sobolev seminorm (for which we constructed a calibration in Chapter 4 above) as well as the fractional total variation (see Appendix J) and the examples in Subsection 5.1.1. Our calibration in Theorem 5.1.3 is a generalization of the one in the previous chapter. To guess the expression of \mathcal{C}_{N} above, we extrapolated our new identity (5.1.5) in the local theory. The key point is that each of the terms in (5.1.5) has a clear nonlocal counterpart; see (5.2.7) below.

An interesting feature of the calibrations considered in this chapter is their stability under the addition of functionals. Due to their special structure, calibrations given in terms of fields can be added together to obtain new ones. In particular, the local theory can be combined with the nonlocal one developed in this work to produce calibrations for energies involving both local and nonlocal interaction terms. We explain this further in Section 5.3.

5.1.4 An application to monotone solutions

Our interest in fields of extremals came from the study of *monotone solutions* to the fractional Allen-Cahn equation

$$(-\Delta)^s u = u - u^3 \quad \text{in } \mathbb{R}^n \tag{5.1.9}$$

(see [30, 31], as well as [42] for more general integro-differential operators). When the operator is the classical Laplacian, these solutions are related to a celebrated conjecture of De Giorgi; see [26].

In Corollary 4.1.4 above, we proved that monotone solutions of (5.1.9) are minimizers among competitors taking values in a precise region of space (the region specified in the next corollary). Thanks to Theorem 5.1.3 of the current chapter, the same proof allows to establish the minimality of monotone solutions to more general nonlocal translation invariant equations. More precisely, given a Lagrangian of the form $G_N = G_N(x - y, a, b)$, with associated energy functional \mathcal{E}_N defined by (5.1.6), the Euler-Lagrange operator \mathcal{L}_N given by (5.1.8) is translation invariant, that is, for all x and z in \mathbb{R}^n the identity

$$\mathcal{L}_{\mathrm{N}}(w)(x+z) = \mathcal{L}_{\mathrm{N}}(w(\cdot+z))(x)$$

holds. We then have the following:

Corollary 5.1.4. Let $G_N = G_N(x - y, a, b)$ be a function satisfying the ellipticity condition (5.1.7). Let u be a sufficiently regular solution for G_N (see the comments before Theorem 5.1.3) of $\mathcal{L}_N(u) = 0$ in \mathbb{R}^n . Assume that u is increasing in the x_n -variable, i.e.,

$$\partial_{x_n} u > 0 \quad in \ \mathbb{R}^n. \tag{5.1.10}$$
Then, for each bounded domain $\Omega \subset \mathbb{R}^n$, u is a minimizer of \mathcal{E}_N among sufficiently regular admissible functions w satisfying

 $\lim_{\tau \to -\infty} u(x', \tau) \le w(x', x_n) \le \lim_{\tau \to +\infty} u(x', \tau) \quad \text{for all} \ x = (x', x_n) \in \Omega$

and such that $w \equiv u$ in Ω^c .

This minimality result was already known for the fractional Laplacian $\mathcal{L}_{\mathrm{N}} = (-\Delta)^{s}$. For such an operator, it can be proven with an alternative argument (described in the Introduction of [23]) which does not use any calibration. However, such an alternative proof requires an existence and regularity theorem for minimizers, as explained in Appendix I. Such a result is not available for many general Lagrangians of the form $G_{\mathrm{N}}(x-y, a, b)$. For these functionals, Corollary 5.1.4 allows to establish the minimality of monotone solutions for the first time.

Notice that, given a monotone solution, the translation invariance of the equation is all what is needed in order to produce a field of extremals (by sliding the solution in the x_n -variable). Therefore, Corollary 5.1.4 also holds for translation invariant equations involving both local and nonlocal terms; see Section 5.3.

5.1.5 An application to the viscosity theory

Here we are interested in conditions to ensure that minimizers, or more generally weak solutions, are viscosity solutions. These are different notions of solutions that have been formulated both for differential and for nonlocal equations. Within the Calculus of Variations, it is natural to work with *weak solutions* belonging to the energy space. On the contrary, when dealing with fully nonlinear equations, it is more suitable to work with *viscosity solutions*. Here, the equation is transferred to act on smooth functions touching the extremal from one side.

In the local framework, it has been shown in the literature that minimizers of many relevant functionals are viscosity solutions. For the *p*-Laplace equation (here every weak solution is a minimizer), Juutinen, Lindqvist, and Manfredi [65] obtained the result by using a weak comparison principle. This allows to compare the minimizer with a function touching it by below and which is later slid upwards, forcing the equation to have the correct sign. For functionals of the form (5.1.2), assuming convexity (a stronger condition than ellipticity), Barron and Jensen [6] found a simpler variational argument. We comment on their strategy at the end of the present subsection as well as in Remark 5.4.9. Showing that non-minimizing weak solutions are viscosity solutions has also been treated in the literature. For instance, this has been done by Medina and Ochoa [74] for semilinear equations driven by the *p*-Laplacian. Their proof again uses a comparison principle.

Concerning nonlocal problems, the first results in this direction appeared in the geometric setting. Caffarelli, Roquejoffre, and Savin [32] showed that minimizers to the nonlocal perimeter are viscosity solutions of the homogeneous nonlocal mean curvature equation. Their proof is quite involved and uses a comparison principle. Later, Cabré [18] was able to show the same result via a simpler calibration argument (here we will give the analogue of this result in the functional setting). The case of nonlocal minimal graphs has also been treated by Cozzi and Lombardini [41]. In the functional setting, as far as we know, the first nonlocal result appeared in the work of Servadei and Valdinoci [87] for linear equations involving the fractional Laplacian. There, the authors employ a regularization by convolution that is not available for other operators. For equations driven by the fractional *p*-Laplacian, we mention the paper by Korvenpää, Kuusi, and Lindgren [66] where they treat the homogeneous problem, and the work by Barrios and Medina [5] for the semilinear one. In both cases, a comparison principle is needed.

Next, we state the main result of this subsection. We will show that every minimizer of our elliptic nonlocal functionals is a viscosity solution. In contrast with most of the previous works, the novelty of our result is that we do not need a weak comparison principle, allowing us to treat a bigger class of Lagrangians. This is achieved by a calibration argument. In a way, the information given by the weak comparison principle is already contained in the properties (C1)-(C3) satisfied by the calibration. Recall, as explained at the beginning of Subsection 5.1.3, that the weak comparison principle does not follow from ellipticity. However, the ellipticity of the Lagrangian (condition (5.1.7) above) suffices for the calibration argument in our proof.

Our theorem applies to general nonlocal elliptic functionals of the form (5.1.6). Since we do not make any growth and regularity assumptions on the Lagrangian G_N , as in the main theorem above, our result is only formal. Nevertheless, again, we could give a fully rigorous statement for specific families of Lagrangians. In fact, this is what we do in Section 5.4 for fractional semilinear equations.

Theorem 5.1.5. Let $G_N = G_N(x, y, a, b)$ be a function satisfying the ellipticity condition (5.1.7) and let $\Omega \subset \mathbb{R}^n$ be a bounded domain. Let u be a sufficiently regular minimizer of the functional \mathcal{E}_N given by (5.1.6).

Then, the function u is a viscosity solution of the associated Euler-Lagrange equation $\mathcal{L}_{N}(u) = 0$ in Ω .

Later in Section 5.4, we will give a more precise statement of this result, showing that minimizers by above (below) are viscosity supersolutions (subsolutions). Furthermore, while our theorem only applies to minimizers, we will explain how it can be used to prove that certain non-minimizing weak solutions are viscosity solutions. Here, the idea is to "freeze" the lower order terms; see Remark 5.4.10.

The proof of Theorem 5.1.5 is based on the following energy comparison result for ordered functions embedded in a *weak field* (that is, a "degenerate field" where the leaves are still ordered, but may touch each other; see Figure 5.1 and Definition 5.4.1). Thus, here we will need to extend the above theory of nonlocal calibrations to the more general setting of weak fields.

Theorem 5.1.6. Let $G_N = G_N(x, y, a, b)$ be a function satisfying the ellipticity condition (5.1.7). Given a bounded domain $\Omega \subset \mathbb{R}^n$, let u and v belong to $C(\overline{\Omega})$ and satisfy u = v a.e. in Ω^c and $u \leq v$ in Ω .

Assume that there exists a weak field $\{\varphi^t\}_{t\in[0,T]}$ for u and v (in the sense of Definition 5.4.1) which is sufficiently regular for G_N (see the comments at the beginning of Subsection 5.4.3).

Then, if $\mathcal{E}_{N}(u) < \infty$, we have

$$\mathcal{E}_{\mathrm{N}}(v) \leq \mathcal{E}_{\mathrm{N}}(u) + \int_{\Omega} \int_{u(x)}^{v(x)} \mathcal{L}_{\mathrm{N}}\left(\varphi^{t}(x)\right) \Big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x.$$

Let us point out that, in Section 5.4, we will prove analogous results to Theorems 5.1.5 and 5.1.6 in the fractional semilinear setting, giving in this case fully rigorous statements under precise regularity assumptions (see Theorems 5.4.6 and 5.4.3 respectively). Furthermore, the same proof will allow us to prove Theorems 5.1.5 and 5.1.6 in the more general



Figure 5.1: Example of a weak field for two functions u and v.

setting of "mixed" functionals involving both local and nonlocal terms (see Theorems 5.4.8 and 5.4.7 below).

The energy inequality in Theorem 5.1.6 is established as in our previous calibration arguments. Once available, we can prove Theorem 5.1.5. Indeed, assume that u is a minimizer and that a smooth function φ touches u by below at some contact point. Now, we slide φ upwards and take the maxima with u to obtain a weak field. The energy comparison with $v = \varphi^T$ will show that φ must be a supersolution at the contact point, since otherwise u would not be a minimizer. Applying the same procedure to smooth functions touching u by above, we will conclude that u is a viscosity solution.

Finally, let us mention that a simple variational proof of our result can be given without using the calibration argument. For this, in addition to the ellipticity condition (5.1.7), one needs to assume that the function $(a, b) \mapsto G_N(x, y, a, b) + G_N(y, x, b, a)$ is convex; see footnote 4. The proof is a nonlocal counterpart of the one by Barron and Jensen [6]. We explain this further in Remark 5.4.9.

5.1.6 Outline of the chapter

Section 5.2 contains the proofs of Theorem 5.1.3 and Corollary 5.1.4. In Section 5.3 we explain how to combine the local and nonlocal theory to obtain calibrations for mixed energy functionals. In Section 5.4 we apply the calibration formalism to the theory of viscosity solutions. First, we prove the fully rigorous results for the fractional Laplacian (Theorems 5.4.3 and 5.4.6) with all details in regularity and integrability issues. Then, we show Theorems 5.1.5 and 5.1.6 (contained, respectively, in the more general Theorems 5.4.8 and 5.4.7).

5.2 The calibration for general nonlocal functionals

Having obtained a calibration for the semilinear problem involving the fractional Laplacian in the previous chapter, we are now interested in extending this construction to a general class of nonlocal functionals. In this way, we plan to obtain a similar picture to that of the general local theory treated in Section 4.3 above. We find a functional C_N that, at least at the formal level, is a calibration for the nonlocal energy functional \mathcal{E}_N . We say at the formal level since the appropriate regularity assumptions on the field of extremals will depend on the concrete given functional \mathcal{E}_N and its associated nonlocal problem.

Consider the nonlocal energy functional \mathcal{E}_{N} of the form (5.1.6). Since $Q(\Omega)$ is invariant with respect to the reflection $(x, y) \mapsto (y, x)$, we may assume without loss of generality that the Lagrangian G_{N} is *pairwise symmetric*,⁷ that is,

$$G_{\mathcal{N}}(y, x, b, a) = G_{\mathcal{N}}(x, y, a, b) \quad \text{for all } (x, y) \in Q(\Omega) \text{ and } (a, b) \in \mathbb{R}^2.$$
(5.2.1)

In particular, from the pairwise symmetry it follows that

$$\partial_b G_{\mathcal{N}}(x, y, \widetilde{a}, \widetilde{b}) = \partial_a G_{\mathcal{N}}(y, x, \widetilde{b}, \widetilde{a}) \quad \text{for all } (x, y) \in Q(\Omega) \text{ and } (\widetilde{a}, \widetilde{b}) \in \mathbb{R}^2.$$
 (5.2.2)

The first variation of \mathcal{E}_{N} at u in the direction of $\eta \in C_{c}^{\infty}(\mathbb{R}^{n})$ (notice that η is not necessarily supported in Ω) is given by

$$\begin{split} &\frac{\mathrm{d}}{\mathrm{d}\varepsilon} \mathcal{E}_{\mathrm{N}}(u+\varepsilon\eta)\Big|_{\varepsilon=0} \\ &= \frac{1}{2} \iint_{Q(\Omega)} \partial_{a} G_{\mathrm{N}}(x,y,u(x),u(y)) \,\eta(x) \,\mathrm{d}x \,\mathrm{d}y + \frac{1}{2} \iint_{Q(\Omega)} \partial_{b} G_{\mathrm{N}}(x,y,u(x),u(y)) \,\eta(y) \,\mathrm{d}x \,\mathrm{d}y \\ &= \frac{1}{2} \iint_{Q(\Omega)} \partial_{a} G_{\mathrm{N}}(x,y,u(x),u(y)) \,\eta(x) \,\mathrm{d}x \,\mathrm{d}y + \frac{1}{2} \iint_{Q(\Omega)} \partial_{b} G_{\mathrm{N}}(y,x,u(y),u(x)) \,\eta(x) \,\mathrm{d}x \,\mathrm{d}y \\ &= \iint_{Q(\Omega)} \partial_{a} G_{\mathrm{N}}(x,y,u(x),u(y)) \,\eta(x) \,\mathrm{d}x \,\mathrm{d}y, \end{split}$$

where we have used the symmetry of $Q(\Omega)$ and the identity (5.2.2).

Writing the domain $Q(\Omega)$ as the disjoint union $Q(\Omega) = (\Omega \times \mathbb{R}^n) \cup (\Omega^c \times \Omega)$, we can split the last integral to obtain

$$\frac{\mathrm{d}}{\mathrm{d}\varepsilon}\mathcal{E}_{\mathrm{N}}(u+\varepsilon\eta)\Big|_{\varepsilon=0} = \int_{\Omega}\mathcal{L}_{\mathrm{N}}(u)(x)\,\eta(x)\,\mathrm{d}x + \int_{\Omega^{c}}\mathcal{N}_{\mathrm{N}}(u)(x)\,\eta(x)\,\mathrm{d}x,\tag{5.2.3}$$

where we have introduced the nonlinear operators

$$\mathcal{L}_{\mathrm{N}}(u)(x) := \int_{\mathbb{R}^n} \partial_a G_{\mathrm{N}}(x, y, u(x), u(y)) \,\mathrm{d}y$$

and

$$\mathcal{N}_{\mathrm{N}}(u)(x) := \int_{\Omega} \partial_a G_{\mathrm{N}}(x, y, u(x), u(y)) \,\mathrm{d}y.$$

Consistent with the terminology in [23], we refer to \mathcal{L}_N as the Euler-Lagrange operator associated to \mathcal{E}_N , while \mathcal{N}_N is its associated nonlocal Neumann operator.

⁷Here we follow the terminology of [49].

Since we are interested in minimization problems with respect to functions with the same exterior data, we only consider variations η that are compactly supported in Ω . Thus, an extremal u of \mathcal{E}_{N} will satisfy the Euler-Lagrange equation

$$\mathcal{L}_{\mathcal{N}}(u) = 0 \quad \text{in } \Omega. \tag{5.2.4}$$

Given an interval $I \subset \mathbb{R}$, let $u^t \colon \mathbb{R}^n \to \mathbb{R}$ be a field in \mathbb{R}^n (in the sense of Definition 5.1.2), with $t \in I$, which covers the region

$$\mathcal{G} := \{ (x, \lambda) \in \mathbb{R}^n \times \mathbb{R} : \lambda = u^t(x) \text{ for some } t \in I \}.$$

Let us also consider the class of admissible functions

$$\mathcal{A}_{\mathrm{N}} := \{ w \colon \mathbb{R}^n \to \mathbb{R} : w \text{ is sufficiently regular for } G_{\mathrm{N}} \text{ and } \operatorname{graph} w \subset \mathcal{G} \}, \qquad (5.2.5)$$

where "sufficiently regular" refers to the following issue. Since we are not making any growth or structure assumption on $G_{\rm N}$, the class of functions w for which $\mathcal{E}_{\rm N}(w)$ makes sense must be chosen according to each nonlocal functional under investigation. This will be the functions considered in $\mathcal{A}_{\rm N}$, which may contain further regularity restrictions so that the operators $\mathcal{L}_{\rm N}$ and $\mathcal{N}_{\rm N}$, as well as all the integrals in the proofs are well defined.

Let $t_0 \in I$. Our goal is to construct a calibration for \mathcal{E}_N and u^{t_0} . We define the functional \mathcal{C}_N on \mathcal{A}_N by

$$\mathcal{C}_{\mathrm{N}}(w) := \iint_{Q(\Omega)} \int_{u^{t_0}(x)}^{w(x)} \partial_a G_{\mathrm{N}}(x, y, u^t(x), u^t(y)) \big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x \,\mathrm{d}y + \mathcal{E}_{\mathrm{N}}(u^{t_0}). \tag{5.2.6}$$

By the above considerations and splitting the domain into $Q(\Omega) = (\Omega \times \mathbb{R}^n) \cup (\Omega^c \times \Omega)$, we can rewrite (5.2.6) as

$$\mathcal{C}_{\mathrm{N}}(w) = \int_{\Omega} \int_{u^{t_0}(x)}^{w(x)} \mathcal{L}_{\mathrm{N}}(u^t)(x) \big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x + \int_{\Omega^c} \int_{u^{t_0}(x)}^{w(x)} \mathcal{N}_{\mathrm{N}}(u^t)(x) \big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x + \mathcal{E}_{\mathrm{N}}(u^{t_0}).$$
(5.2.7)

Notice that (5.2.7) is the "canonical" nonlocal analogue of Theorem 3.1 in [23], and thus of the classical local Weierstrass calibration $C_{\rm L}$.

Next we show that if the field $\{u^t\}_{t\in I}$ is made up of supersolutions above u^{t_0} and subsolutions below, then u^{t_0} minimizes \mathcal{C}_N among functions in \mathcal{A}_N with the same exterior data. Furthermore, if all the functions u^t satisfy the Euler-Lagrange equation (i.e., u^t is a field of extremals), then \mathcal{C}_N is a null-Lagrangian and its value depends only on the exterior datum. The following result (properties (\mathcal{C}_3) and (\mathcal{C}_3') of the calibration) follows readily from expression (5.2.7) for \mathcal{C}_N . Note that here we do not need to assume the ellipticity of \mathcal{G}_N .

Proposition 5.2.1. Given an interval $I \subset \mathbb{R}$, a bounded domain $\Omega \subset \mathbb{R}^n$, and a function $G_N = G_N(x, y, a, b)$, let $\{u^t\}_{t \in I}$ be a field in \mathbb{R}^n (in the sense of Definition 5.1.2) which is sufficiently regular for G_N . Assume that, for $t_0 \in I$, the leaves satisfy the inequalities

$$\mathcal{L}_{N}(u^{t}) \geq 0 \quad in \ \Omega \quad for \ t \geq t_{0}, \quad and$$

$$\mathcal{L}_{N}(u^{t}) \leq 0 \quad in \ \Omega \quad for \ t \leq t_{0}.$$
 (5.2.8)

Consider the set of admissible functions \mathcal{A}_{N} defined in (5.2.5).

Then, for all w in \mathcal{A}_N such that $w \equiv u^{t_0}$ in Ω^c , the functional \mathcal{C}_N defined in (5.2.6) satisfies

$$\mathcal{C}_{\mathrm{N}}(u^{t_0}) \leq \mathcal{C}_{\mathrm{N}}(w).$$

Assume in addition that the leaves satisfy the Euler-Lagrange equation (5.2.4), that is,

$$\mathcal{L}_{N}(u^{t}) = 0 \quad in \ \Omega \quad for \ all \ t \in I.$$
(5.2.9)

Then, for all w in \mathcal{A}_{N} such that $w \equiv u^{t_{0}}$ in Ω^{c} , we have

$$\mathcal{C}_{\mathcal{N}}(u^{t_0}) = \mathcal{C}_{\mathcal{N}}(w).$$

Proof. First, notice that $\mathcal{C}_{N}(u^{t_0}) = \mathcal{E}_{N}(u^{t_0})$. Since $w \equiv u^{t_0}$ in Ω^c , by (5.2.7) we have

$$\mathcal{C}_{\mathrm{N}}(w) - \mathcal{C}_{\mathrm{N}}(u^{t_0}) = \int_{\Omega} \int_{u^{t_0}(x)}^{w(x)} \mathcal{L}_{\mathrm{N}}(u^t) \big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x.$$

Assuming (5.2.8), it suffices to show that for all $x \in \Omega$ we have

$$\int_{u^{t_0}(x)}^{w(x)} \mathcal{L}_{\mathcal{N}}(u^t) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \ge 0.$$
(5.2.10)

If $w(x) \ge u^{t_0}(x)$, then, using that the functions $\{u^t\}_{t\in I}$ are increasing in t, we have $t(x, \lambda) \ge t_0$ for $\lambda \in [u^{t_0}(x), w(x)]$. Hence, by assumption (5.2.8), $\mathcal{L}_{N}(u^t)|_{t=t(x,\lambda)} \ge 0$ and (5.2.10) follows in this case. The case $w(x) \le u^{t_0}(x)$ is treated similarly.

If we further assume (5.2.9), then the integral in (5.2.10) vanishes and the claim follows. \Box

The functional C_N can be rewritten in the following alternative form that we will use to verify the remaining calibration properties (C1) and (C2).

Lemma 5.2.2. Given an interval $I \subset \mathbb{R}$, a bounded domain $\Omega \subset \mathbb{R}^n$, and a pairwise symmetric function $G_N = G_N(x, y, a, b)$ in the sense of (5.2.1), let $\{u^t\}_{t \in I}$ be a field in \mathbb{R}^n (in the sense of Definition 5.1.2) which is sufficiently regular for G_N . Consider the set of admissible functions \mathcal{A}_N defined in (5.2.5).

Then, for all w in \mathcal{A}_N , the functional \mathcal{C}_N defined in (5.2.6) satisfies

$$\mathcal{C}_{N}(w) = \frac{1}{2} \iint_{Q(\Omega)} G_{N}(x, y, w(x), u^{t}(y)) \big|_{t=t(x, w(x))} \, dx \, dy + \frac{1}{2} \iint_{Q(\Omega)} \int_{t(x, w(x))}^{t(y, w(y))} \partial_{b} G_{N}(x, y, u^{t}(x), u^{t}(y)) \partial_{t} u^{t}(y) \, dt \, dx \, dy.$$
(5.2.11)

Proof. We will rewrite the integral term in the definition (5.2.6) of C_N . Applying the change of variables $\lambda \mapsto t$ with $u^t(x) = \lambda$ for each x, we have

$$\begin{split} \iint_{Q(\Omega)} \int_{u^{t_0}(x)}^{w(x)} \partial_a G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x \, \mathrm{d}y \\ &= \iint_{Q(\Omega)} \int_{t_0}^{t(x,w(x))} \partial_a G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \partial_t u^t(x) \, \mathrm{d}t \, \mathrm{d}x \, \mathrm{d}y. \end{split}$$

Symmetrizing this expression in (x, y) and using (5.2.2), we deduce

$$\iint_{Q(\Omega)} \int_{u^{t_0}(x)}^{w(x)} \partial_a G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x \, \mathrm{d}y$$

$$= \frac{1}{2} \iint_{Q(\Omega)} \int_{t_0}^{t(x,w(x))} \partial_a G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \partial_t u^t(x) \, \mathrm{d}t \, \mathrm{d}x \, \mathrm{d}y$$

$$+ \frac{1}{2} \iint_{Q(\Omega)} \int_{t_0}^{t(y,w(y))} \partial_b G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \partial_t u^t(y) \, \mathrm{d}t \, \mathrm{d}x \, \mathrm{d}y.$$
(5.2.12)

Splitting the integral $\int_{t_0}^{t(y,w(y))} \cdot dt$ in (5.2.12) into $\int_{t_0}^{t(x,w(x))} \cdot dt + \int_{t(x,w(x))}^{t(y,w(y))} \cdot dt$, we obtain

$$\iint_{Q(\Omega)} \int_{u^{t_0}(x)}^{w(x)} \partial_a G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \big|_{t=t(x,\lambda)} d\lambda \, dx \, dy \\
= \frac{1}{2} \iint_{Q(\Omega)} \int_{t_0}^{t(x,w(x))} \frac{d}{dt} \big\{ G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \big\} \, dt \, dx \, dy \\
+ \frac{1}{2} \iint_{Q(\Omega)} \int_{t(x,w(x))}^{t(y,w(y))} \partial_b G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \partial_t u^t(y) \, dt \, dx \, dy.$$
(5.2.13)

Integrating the derivative with respect to t in (5.2.13) and recalling, by definition of the leaf-parameter function, that $w(x) = u^{t(x,w(x))}(x)$, we have

$$\begin{aligned}
\iint_{Q(\Omega)} \int_{u^{t_0}(x)}^{w(x)} \partial_a G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x \, \mathrm{d}y \\
&= \frac{1}{2} \iint_{Q(\Omega)} G_{\mathcal{N}}(x, y, w(x), u^{t(x,w(x))}(y)) \, \mathrm{d}x \, \mathrm{d}y - \mathcal{E}_{\mathcal{N}}(u^{t_0}) \\
&+ \frac{1}{2} \iint_{Q(\Omega)} \int_{t(x,w(x))}^{t(y,w(y))} \partial_b G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \partial_t u^t(y) \, \mathrm{d}t \, \mathrm{d}x \, \mathrm{d}y.
\end{aligned}$$
(5.2.14)

Adding $\mathcal{E}_{N}(u^{t_0})$ to both sides of (5.2.14) now yields the claim.

In the next proposition we prove the calibration property (C1). This follows directly from Lemma 5.2.2. Here, ellipticity of G_N is still not needed.

Proposition 5.2.3. Given an interval $I \subset \mathbb{R}$, a bounded domain $\Omega \subset \mathbb{R}^n$, and a pairwise symmetric function $G_N = G_N(x, y, a, b)$ in the sense of (5.2.1), let $\{u^t\}_{t \in I}$ be a field in \mathbb{R}^n (in the sense of Definition 5.1.2) which is sufficiently regular for G_N .

Then, for all $t \in I$, the functional C_N defined in (5.2.6) satisfies

$$\mathcal{C}_{\mathrm{N}}(u^t) = \mathcal{E}_{\mathrm{N}}(u^t).$$

Proof. Let $t_0 \in I$. Choosing $w = u^{t_0}$ in (5.2.11), since $t(x, w(x)) = t_0$ for all x, we have

$$\mathcal{C}_{N}(u^{t_{0}}) = \frac{1}{2} \iint_{Q(\Omega)} G_{N}(x, y, u^{t_{0}}(x), u^{t_{0}}(y)) \, \mathrm{d}x \, \mathrm{d}y = \mathcal{E}_{N}(u^{t_{0}}).$$

Since $t_0 \in I$ was arbitrary, this proves the proposition.

It remains to prove the last calibration property (C2). We will now need the natural ellipticity assumption on the Lagrangian G_N :

$$\partial_{ab}^2 G_N(x, y, a, b) \le 0.$$
 (5.2.15)

Notice that (5.2.15) is simply the ellipticity condition (5.1.7) from the Introduction written for a pairwise symmetric Lagrangian. Moreover, this condition is related to a strong comparison principle, as explained in Appendix I.

Proposition 5.2.4. Given an interval $I \subset \mathbb{R}$, a bounded domain $\Omega \subset \mathbb{R}^n$, and a pairwise symmetric function $G_N = G_N(x, y, a, b)$ in the sense of (5.2.1), let $\{u^t\}_{t \in I}$ be a field in \mathbb{R}^n (in the sense of Definition 5.1.2) which is sufficiently regular for G_N . Consider the set of admissible functions \mathcal{A}_N defined in (5.2.5). Assume that the ellipticity condition $\partial_{ab}^2 G_N \leq 0$ holds.

Then, for all w in \mathcal{A}_N , the functional \mathcal{C}_N defined in (5.2.6) satisfies

$$\mathcal{C}_{\mathrm{N}}(w) \leq \mathcal{E}_{\mathrm{N}}(w).$$

Proof. If we compute the difference $\mathcal{E}_{N}(w) - \mathcal{C}_{N}(w)$, using the alternative expression (5.2.11) for \mathcal{C}_{N} , we obtain

$$\mathcal{E}_{N}(w) - \mathcal{C}_{N}(w) = \frac{1}{2} \iint_{Q(\Omega)} \left\{ G_{N}(x, y, w(x), w(y)) - G_{N}(x, y, w(x), u^{t(x, w(x))}(y)) \right\} dx dy$$

$$- \frac{1}{2} \iint_{Q(\Omega)} \int_{t(x, w(x))}^{t(y, w(y))} \partial_{b} G_{N}(x, y, u^{t}(x), u^{t}(y)) \partial_{t} u^{t}(y) dt dx dy.$$
(5.2.16)

Recalling that $u^{t(y,w(y))}(y) = w(y)$, we can write the first integral on the right-hand side of (5.2.16) as

$$\frac{1}{2} \iint_{Q(\Omega)} \left\{ G_{N}(x, y, w(x), w(y)) - G_{N}(x, y, w(x), u^{t(x, w(x))}(y)) \right\} dx dy
= \frac{1}{2} \iint_{Q(\Omega)} \int_{t(x, w(y))}^{t(y, w(y))} \frac{d}{dt} \left\{ G_{N}(x, y, w(x), u^{t}(y)) \right\} dt dx dy
= \frac{1}{2} \iint_{Q(\Omega)} \int_{t(x, w(y))}^{t(y, w(y))} \partial_{b} G_{N}(x, y, w(x), u^{t}(y)) \partial_{t} u^{t}(y) dt dx dy.$$
(5.2.17)

Plugging (5.2.17) into (5.2.16), we see that

$$\mathcal{E}_{\mathrm{N}}(w) - \mathcal{C}_{\mathrm{N}}(w) = \iint_{Q(\Omega)} \int_{t(x,w(x))}^{t(y,w(y))} \left\{ \partial_{b} G_{\mathrm{N}}(x,y,w(x),u^{t}(y)) - \partial_{b} G_{\mathrm{N}}(x,y,u^{t}(x),u^{t}(y)) \right\} \partial_{t} u^{t}(y) \,\mathrm{d}t \,\mathrm{d}x \,\mathrm{d}y.$$

Thus, it suffices to show that

$$\int_{t(x,w(x))}^{t(y,w(y))} \left\{ \partial_b G_{\mathcal{N}}(x,y,w(x),u^t(y)) - \partial_b G_{\mathcal{N}}(x,y,u^t(x),u^t(y)) \right\} \partial_t u^t(y) \, \mathrm{d}t \ge 0 \qquad (5.2.18)$$

for all $(x, y) \in Q(\Omega)$.

Let $(x, y) \in Q(\Omega)$ and assume first that $t(x, w(x)) \leq t(y, w(y))$. By monotonicity of the leaves u^t in I, for $t \in [t(x, w(x)), t(y, w(y))]$ we have

$$w(x) = u^{t(x,w(x))}(x) \le u^t(x),$$

and by ellipticity

$$\partial_b G_{\mathcal{N}}(x, y, w(x), u^t(y)) \partial_t u^t(y) \ge \partial_b G_{\mathcal{N}}(x, y, u^t(x), u^t(y)) \partial_t u^t(y)$$

Whence, (5.2.18) follows. The case $t(x, w(x)) \ge t(y, w(y))$ is treated similarly.

Finally, combining Propositions 5.2.1, 5.2.3, and 5.2.4, we easily conclude Theorem 5.1.3.

Proof of Theorem 5.1.3. (a) Property (C1) follows from Proposition 5.2.3 and property (C2) follows from Proposition 5.2.4.

- (b) This follows from the first part of Proposition 5.2.1.
- (c) This follows from the second part of Proposition 5.2.1.

Proof of Corollary 5.1.4. For each $t \in \mathbb{R}$ we define $u^t(x) := u(x', x_n + t)$, where $x = (x', x_n) \in \mathbb{R}^{n-1} \times \mathbb{R}$. By the monotonicity (5.1.10) of u and by translation invariance of the equation $\mathcal{L}_{N}(u) = 0$, it follows that the family $\{u^t\}_{t \in \mathbb{R}}$ is a field of extremals in \mathbb{R}^n in the sense of Definition 5.1.2. Hence, Theorem 5.1.3 yields the minimality of each u^t among competitors w with $w \equiv u$ in Ω^c and satisfying the assumption

$$\lim_{\tau \to -\infty} u(x',\tau) < w(x',x_n) < \lim_{\tau \to +\infty} u(x',\tau) \quad \text{for all} \quad x = (x',x_n) \in \Omega.$$

Finally, we can relax the previous strict inequalities by considering the competitor $(1 - \varepsilon)w + \varepsilon u$ and letting $\varepsilon \to 0$. In this way we recover the condition in the statement of Corollary 5.1.4 where the inequalities are not strict.

5.3 The calibration for functionals involving both local and nonlocal terms

The results derived in Section 5.2 may be combined with the classical local ones to yield a theory that applies to functionals involving both local and nonlocal interactions. These functionals appear when dealing with symmetric Lévy processes, where the infinitesimal generators are given by the sum of a second order differential operator and an integrodifferential one. Recently, mixed functionals have attracted great attention from different points of view; see [72,93] and references therein.

The mixed energy⁸

$$\mathcal{E}_{\mathrm{T}}(w) := \mathcal{E}_{\mathrm{N}}(w) + \mathcal{E}_{\mathrm{L}}(w)$$

= $\frac{1}{2} \iint_{Q(\Omega)} G_{\mathrm{N}}(x, y, w(x), w(y)) \,\mathrm{d}x \,\mathrm{d}y + \int_{\Omega} G_{\mathrm{L}}(x, w(x), \nabla w(x)) \,\mathrm{d}x$ (5.3.1)

admits a calibrating functional

$$\mathcal{C}_{\mathrm{T}}(w) := \mathcal{C}_{\mathrm{N}}(w) + \mathcal{C}_{\mathrm{L}}(w),$$

⁸Here in the notation we use the subscript M, which stands for "Mixed".

where \mathcal{E}_{L} , G_{L} , \mathcal{C}_{L} , and \mathcal{E}_{N} , G_{N} , \mathcal{C}_{N} are defined as in the Introduction. By combining identities (5.1.5) and (5.2.7), the functional \mathcal{C}_{T} may be written equivalently as

$$\begin{aligned} \mathcal{C}_{\mathrm{T}}(w) &= \int_{\Omega} \int_{u^{t_0}(x)}^{w(x)} \mathcal{L}_{\mathrm{T}}(u^t) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x \\ &+ \int_{\Omega^c} \int_{u^{t_0}(x)}^{w(x)} \mathcal{N}_{\mathrm{N}}(u^t) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x + \int_{\partial\Omega} \int_{u^{t_0}(x)}^{v(x)} \mathcal{N}_{\mathrm{L}}(u^t) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}\mathcal{H}^{n-1}(x) \\ &+ \mathcal{E}_{\mathrm{T}}(u^{t_0}), \end{aligned}$$

where the Euler-Lagrange operator of the mixed problem is

$$\mathcal{L}_{\mathrm{T}}(w) := \mathcal{L}_{\mathrm{N}}(w) + \mathcal{L}_{\mathrm{L}}(w),$$

and \mathcal{L}_L , \mathcal{N}_L , \mathcal{L}_N , and \mathcal{N}_N are the operators introduced above.

Since $C_{\rm T}$ shares the same structure as $C_{\rm L}$ and $C_{\rm N}$, a straightforward adaptation of the proofs in the sections above shows that $C_{\rm T}$ satisfies all three calibration properties. We mention that property (C2) requires both the local and nonlocal ellipticity conditions, that is, one must assume that both

$$\partial_{qq}^2 G_{\mathrm{L}}(x,\lambda,q) \ge 0 \text{ and } \partial_{ab}^2 G_{\mathrm{N}}(x,y,a,b) \le 0$$

hold.

As an application of this theory, we can prove the analogue of Corollary 5.1.4 for mixed functionals. Namely, if \mathcal{L}_{T} is translation invariant, i.e., $\mathcal{L}_{T}(u(\cdot + y))(x) = \mathcal{L}_{T}(u)(x + y)$ for all x and y in \mathbb{R}^{n} , then monotone solutions are minimizers among functions lying between the limits of the solution in the direction of monotonicity. The proof is identical to the one of Corollary 5.1.4.

Remark 5.3.1. Mixed energies appear in the following relevant frameworks. However, these minimization problems include constraints. Thus, one cannot directly apply the calibration theory developed above, since constrained minimizers need not be minimizers of the original functional and no foliation of extremals is expected. As examples of such frameworks, we mention the theory of aggregation equations [35], certain problems from astrophysics [70], the Thomas-Fermi theory [7], the Choquard-Pekar model [68], as well as the problem of finding the best constant in the Sobolev inequality [94].

5.4 Application to the viscosity theory

For the application of calibrations to prove that minimizers are viscosity solutions, we need to consider more general fields, namely, those which are not necessarily increasing in a bounded domain $\Omega \subset \mathbb{R}^n$, but only nondecreasing. The situation is that different leaves will coincide in certain subsets of Ω ; see Figure 5.1 in the Introduction. Such a field will appear when sliding a touching test function and truncating it with the minimizer.

5.4.1 An energy comparison result for fractional functionals

Given $s \in (0, 1)$ and a bounded domain $\Omega \subset \mathbb{R}^n$, let $u, v \in C(\overline{\Omega}) \cap L^1_s(\mathbb{R}^n)$ be two functions such that u = v a.e. in Ω^c and $u \leq v$ in Ω . Assume we are given functions $\varphi^t \colon \mathbb{R}^n \to \mathbb{R}$, with $t \in [0, T]$ and nondecreasing in t, such that they "interpolate" between the two functions: $\varphi^0 = u, \varphi^T = v$. When the family $\{\varphi^t\}_{t \in [0,T]}$ satisfies appropriate regularity assumptions, we will be able to construct a calibration involving the field.

Consider the region

$$\mathcal{G} = \left\{ (x, \lambda) \in \Omega \times \mathbb{R} : u(x) < \lambda < v(x) \right\}$$

as well as the sections

 $\Omega_t := \{ x \in \Omega \colon \varphi^t(x) > u(x) \} \quad \text{for each } t \in (0, T),$

which will be increasing in t, and

$$I_x := \{ t \in (0,T) \colon \varphi^t(x) > u(x) \} \quad \text{for each } x \in \Omega.$$

Definition 5.4.1. Given $s \in (0, 1)$ and a bounded domain $\Omega \subset \mathbb{R}^n$, let $u, v \in C(\overline{\Omega}) \cap L^1_s(\mathbb{R}^n)$ be such that u = v a.e. in Ω^c and $u \leq v$ in Ω . A family $\{\varphi^t\}_{t \in [0,T]}$ of functions $\varphi^t \colon \mathbb{R}^n \to \mathbb{R}$ is said to be a *weak field for u and v* (see Figure 5.1) if the following conditions are satisfied:

- (i) $\varphi^0 = u$ and $\varphi^T = v$.
- (ii) $\varphi^t = u$ a.e. in Ω^c , for all $t \in [0, T]$.
- (iii) The function $(x, t) \mapsto \varphi^t(x)$ is continuous in $\overline{\Omega} \times [0, T]$.
- (iv) For each $x \in \Omega$, the function $t \mapsto \varphi^t(x)$ is $C^1(I_x)$ and increasing in $\overline{I_x}$. Moreover, there exists a constant $C_0 > 0$ such that

$$|\partial_t \varphi^t(x)| \leq C_0$$
 for all $x \in \Omega$ and $t \in I_x$.

Moreover, the weak field is *regular by below*, if the following regularity condition holds:

(v) The functions $\{\varphi^t\}_{t\in(0,T)}$ are uniformly $C^{1,1}$ by below in Ω_t , uniformly in t, in the following sense. There exist a constant $C_0 > 0$ and a bounded domain $N \subset \mathbb{R}^n$, with $\overline{\Omega} \subset N$, such that, for each $t \in (0,T)$ and $x \in \Omega_t$, there is a function $\psi \in C^2(N)$ touching φ^t by below in N at x, that is, $\psi(x) = \varphi^t(x)$ and $\psi \leq \varphi^t$ a.e. in N, satisfying

$$D^2\psi \ge -C_0$$
 in N .

Remark 5.4.2. The more technical assumption in Definition 5.4.1, condition (v), is needed for the calibration of the fractional Laplacian to be well defined. An important consequence of (v) is that the fractional Laplacian $(-\Delta)^s \varphi^t(x)$ is bounded by above uniformly in $t \in$ (0,T) and $x \in \Omega_t$.

If $\{\varphi^t\}_{t\in[0,T]}$ is a weak field for u and v, then for each $(x,\lambda) \in \mathcal{G}$ there exists a unique $t = t(x,\lambda)$ such that $\varphi^{t(x,\lambda)}(x) = \lambda$. The existence is a consequence of (iii), since $\varphi^0(x) < \lambda < \varphi^T(x)$, while the uniqueness follows from (iv).

Recall that, given $s \in (0, 1)$ and $F \in C^1(\mathbb{R})$, we have

$$\mathcal{E}_{s,F}(w) = \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|w(x) - w(y)|^2}{|x - y|^{n + 2s}} \,\mathrm{d}x \,\mathrm{d}y - \int_{\Omega} F(w(x)) \,\mathrm{d}x$$

We also write

$$\mathcal{E}_{s}(w) = \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|w(x) - w(y)|^{2}}{|x - y|^{n+2s}} \,\mathrm{d}x \,\mathrm{d}y$$

following the notation introduced in Chapter 4. The weak field allows to compare the energies of u and v via the following theorem:

Theorem 5.4.3. Given a bounded domain $\Omega \subset \mathbb{R}^n$ and $s \in (0, 1)$, let $u, v \in C(\overline{\Omega}) \cap L^1_s(\mathbb{R}^n)$ be such that u = v a.e. in Ω^c and $u \leq v$ in Ω . Let $\{\varphi^t\}_{t \in [0,T]}$ be a weak field for u and vwhich is regular by below in the sense of Definition 5.4.1.

Then, given $F \in C^1(\mathbb{R})$, if $\mathcal{E}_s(u) < \infty$ we have

$$\mathcal{E}_{s,F}(v) \le \mathcal{E}_{s,F}(u) + \int_{\Omega \cap \{v > u\}} \int_{u(x)}^{v(x)} \left((-\Delta)^s \varphi^t(x) - F'(\varphi^t(x)) \right) \Big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x$$

Proof. We proceed as in the proof of the calibration properties although giving fewer details. The idea is to consider the analogue of the fractional calibration $C_{s,F}$ for $\mathcal{E}_{s,F}$ and v introduced in (4.1.9) above, and to use property (C2), i.e., $C_{s,F}(u) \leq \mathcal{E}_{s,F}(u)$, which gives the energy comparison.

For $\varepsilon > 0$, we consider the kernel $K_{\varepsilon} = c_{n,s} |\cdot|^{-n-2s} \mathbb{1}_{\mathbb{R}^n \setminus B_{\varepsilon}}$ and the truncated fractional Laplacian $(-\Delta)^s_{\varepsilon} \varphi(x) = \int_{\mathbb{R}^n} (\varphi(x) - \varphi(y)) K_{\varepsilon}(x-y) \, \mathrm{d}y.$

Since u and v differ only in the region $\Omega_T \subset \Omega$, it suffices to consider all functionals defined in $Q(\Omega_T)$ instead of on the larger set

$$Q(\Omega) = Q(\Omega_T) \cup ((\Omega \setminus \Omega_T) \times \Omega_T^c) \cup (\Omega^c \times (\Omega \setminus \Omega_T)).$$

By a slight modification of the proof of Lemma 4.4.4 above, we have the identity

$$\int_{\Omega_T} \int_{v(x)}^{u(x)} (-\Delta)_{\varepsilon}^{s} \varphi^{t}(x) \big|_{t=t(x,\lambda)} dx + \frac{1}{4} \iint_{Q(\Omega_T)} |v(x) - v(y)|^2 K_{\varepsilon}(x-y) dx dy$$

$$= -\frac{1}{2} \iint_{Q(\Omega_T)} dx dy K_{\varepsilon}(x-y) \int_{t(x,u(x))}^{t(y,u(y))} (\varphi^{t}(x) - \varphi^{t}(y)) \partial_t \varphi^{t}(y) dt \qquad (5.4.1)$$

$$+ \frac{1}{4} \iint_{Q(\Omega_T)} |u(x) - \varphi^{t(x,u(x))}(y)|^2 K_{\varepsilon}(x-y) dx dy.$$

Moreover, since φ^t is nondecreasing in t, the first term in the right-hand side of (5.4.1) can be bounded by

$$-\frac{1}{2}\iint_{Q(\Omega_T)} \mathrm{d}x \,\mathrm{d}y K_{\varepsilon}(x-y) \int_{t(x,u(x))}^{t(y,u(y))} (\varphi^t(x) - \varphi^t(y)) \,\partial_t \varphi^t(y) \,\mathrm{d}t$$

$$\leq -\frac{1}{2}\iint_{Q(\Omega_T)} \mathrm{d}x \,\mathrm{d}y K_{\varepsilon}(x-y) \int_{t(x,u(x))}^{t(y,u(y))} (u(x) - \varphi^t(y)) \,\partial_t \varphi^t(y) \,\mathrm{d}t \qquad (5.4.2)$$

$$= \frac{1}{4}\iint_{Q(\Omega_T)} \mathrm{d}x \,\mathrm{d}y K_{\varepsilon}(x-y) \big(|u(x) - u(y)|^2 - |u(x) - \varphi^{t(x,u(x))}(y)|^2 \big),$$

where in the last line we have integrated $-(u(x) - \varphi^t(y))\partial_t \varphi^t(y) = \frac{\mathrm{d}}{\mathrm{d}t}|u(x) - \varphi^t(y)|^2$. Hence, writing $\mathcal{E}_{K_{\varepsilon}}(u) = \frac{1}{2} \iint_{Q(\Omega)} |u(x) - u(y)|^2 K_{\varepsilon}(x-y) \,\mathrm{d}x \,\mathrm{d}y$, combining (5.4.1) and (5.4.2) we deduce

$$\mathcal{E}_{K_{\varepsilon}}(u) - \mathcal{E}_{K_{\varepsilon}}(v) = \frac{1}{4} \iint_{Q(\Omega_{T})} \left(|u(x) - u(y)|^{2} - |v(x) - v(y)|^{2} \right) K_{\varepsilon}(x - y) \, \mathrm{d}x \, \mathrm{d}y$$

$$\geq \int_{\Omega_{T}} \int_{v(x)}^{u(x)} (-\Delta)_{\varepsilon}^{s} \varphi^{t}(x) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x.$$
(5.4.3)

Finally, thanks to property (v) and by Fatou's lemma, we can pass to the limit as $\varepsilon \downarrow 0$ inside the integrals in (5.4.3) to obtain

$$\mathcal{E}_s(v) \le \mathcal{E}_s(u) + \int_{Q(\Omega_T)} \int_{u(x)}^{v(x)} (-\Delta)^s \varphi^t(x) \big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x.$$

Since $F(u) - F(v) = \int_{v(x)}^{u(x)} F'(\lambda) d\lambda = \int_{v(x)}^{u(x)} F'(\varphi^t(x)) \Big|_{t=t(x,\lambda)} d\lambda$, adding the potential term now yields the result.

5.4.2 Minimizers of fractional functionals are viscosity solutions

We recall the definition of viscosity solution in the nonlocal setting. The following is taken from Definition 2.2 in [34]:

Definition 5.4.4. Given bounded domain $\Omega \subset \mathbb{R}^n$, $s \in (0, 1)$, and $F \in C^1(\mathbb{R})$, we say that $u \in C(\Omega) \cap L^1_s(\mathbb{R}^n)$ is a viscosity supersolution of the semilinear equation

$$(-\Delta)^s v = F'(v)$$
 in Ω ,

if whenever the following happens

- x_0 is any point in Ω
- N is a neighborhood of x_0 in Ω
- φ is some C^2 function in \overline{N}
- $\varphi(x_0) = u(x_0)$
- $\varphi(x) < u(x)$ for every $x \in N \setminus \{x_0\}$,

then, the function

$$\overline{\varphi}(x) := \begin{cases} \varphi(x) & \text{for } x \in N \\ u(x) & \text{for } x \in \mathbb{R}^n \setminus N, \end{cases}$$

satisfies $(-\Delta)^s \overline{\varphi}(x_0) \ge F'(\overline{\varphi}(x_0)).$

We also have the analogous definition of viscosity subsolution. We say that u is a viscosity solution if it is both a viscosity supersolution and subsolution.

Our main results in this section deal with minimizers of the energy functional $\mathcal{E}_{s,F}$. In fact, it is enough to consider one-sided minimizers, which are defined as follows:

Definition 5.4.5. Given a bounded domain $\Omega \subset \mathbb{R}^n$, $s \in (0, 1)$, and $F \in C^1(\mathbb{R})$, we say that a function $u \colon \mathbb{R}^n \to \mathbb{R}$ is a *one-sided minimizer by above* of the functional $\mathcal{E}_{s,F}$ if $\mathcal{E}_{s,F}(u) < \infty$ and for all functions v such that $v \geq u$ in Ω and v = u in Ω^c we have

$$\mathcal{E}_{s,F}(v) \ge \mathcal{E}_{s,F}(u).$$

We also have the analogous definition of one-sided minimizer by below.

We will now prove that one-sided minimizers by above are viscosity supersolutions. This is a consequence of Theorem 5.4.3.

Theorem 5.4.6. Given a bounded domain $\Omega \subset \mathbb{R}^n$, $s \in (0,1)$, and $F \in C^1(\mathbb{R})$, let $u \colon \mathbb{R}^n \to \mathbb{R}$ in $C(\Omega)$ be a one-sided minimizer by above of the functional $\mathcal{E}_{s,F}$.

Then u is a viscosity supersolution.

Proof. We proceed by contradiction. Suppose that u is not a viscosity supersolution. Then there exist $x_0 \in \Omega$, a neighborhood $N \subset \Omega$ of x_0 , and a function $\varphi \in C^2(\overline{N})$, with $\varphi(x_0) = u(x_0)$ and $\varphi(x) < u(x)$ for all $x \in \Omega$, such that the extension $\overline{\varphi}$ satisfies

$$(-\Delta)^s \overline{\varphi}(x_0) < F'(\overline{\varphi}(x_0)).$$

We will now construct a function above u which has less energy, thus violating the one-sided minimality by above. The idea of the proof is to raise the function φ to produce a local foliation whose leaves are strict subsolutions.

Recall the truncations introduced in the proof of Theorem 5.4.3 above. Namely, for $\varepsilon > 0$ we let $K_{\varepsilon} = c_{n,s} |\cdot|^{-n-2s} \mathbb{1}_{\mathbb{R}^n \setminus B_{\varepsilon}}$ and $(-\Delta)^s_{\varepsilon} \varphi(x) = \int_{\mathbb{R}^n} (\varphi(x) - \varphi(y)) K_{\varepsilon}(x-y) \, \mathrm{d}y$. Since $(-\Delta)^s \overline{\varphi}(x_0) - F'(\overline{\varphi}(x_0)) =: -4c_0 < 0$, by continuity of F' and of the fractional

Laplacian, there is a smaller neighborhood N of x_0 , with $N \subset N$, such that

$$(-\Delta)^s_{\varepsilon}\overline{\varphi}(x) - F'(\overline{\varphi}(x)) < -2c_0$$

for all $x \in \widetilde{N}$ and $\varepsilon \in (0, \varepsilon_0)$, for some small $\varepsilon_0 > 0$.

For $0 \le t \le T$, where $0 < T \le \min_{\partial N} (u - \varphi)$, we define the functions

$$\varphi^t(x) := \begin{cases} \max\{u(x), \varphi(x) + t\} & \text{for } x \in N\\ u(x) & \text{for } x \in \mathbb{R}^n \setminus N. \end{cases}$$

It is clear that the family $\{\varphi^t\}_{t\in[0,T]}$ is a weak field for u and φ^T which is regular by below, in the sense of Definition 5.4.1.

For 0 < t < T, $\varepsilon > 0$, and $x \in N$ such that $\varphi^t(x) > u(x)$, we have

$$(-\Delta)^s_{\varepsilon} \varphi^t(x) \le (-\Delta)^s_{\varepsilon} \overline{\varphi}(x) + T \int_{\mathbb{R}^n \setminus N} c_{n,s} |x-y|^{-n-2s} \, \mathrm{d}y.$$

From this inequality and the continuity of F', taking a sufficiently small T, we obtain

$$(-\Delta)^s_{\varepsilon}\varphi^t(x) - F'(\varphi^t(x)) < -c_0$$

for all $x \in \widetilde{N}$, $\varepsilon \in (0, \varepsilon_0)$, and $t \in (0, T)$ such that $\varphi^t(x) > u(x)$. Letting $v = \varphi^T$, by Theorem 5.4.3, we conclude that

$$\begin{aligned} \mathcal{E}_{s,F}(v) &\leq \mathcal{E}_{s,F}(u) + \int_{\Omega \cap \{v > u\}} \int_{u(x)}^{v(x)} \left\{ (-\Delta)^s \varphi^t(x) - F'(\varphi^t(x)) \right\} \Big|_{t=t(x,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x \\ &\leq \mathcal{E}_{s,F}(u) - c_0 \Big| \{ (x,\lambda) \in \Omega \times \mathbb{R} : u(x) < \lambda < v(x) \} \Big| \\ &< \mathcal{E}_{s,F}(u), \end{aligned}$$

which contradicts the minimality of u.

5.4.3More general nonlocal functionals

We now extend the previous approach to the more general setting of mixed functionals. First, we note that the notion of weak field needs to be adapted to arbitrary mixed functionals \mathcal{E}_{T} of the form (5.3.1). This class of energies is too big to allow for a general definition of weak field. However, it can be done for specific families of Lagrangians $G_{\rm L}$ and $G_{\rm N}$, as the ones given in the Introduction. Under precise growth and regularity assumptions

on the Lagrangians, it suffices to modify condition (v) in Definition 5.4.1 appropriately so that there are no integrability issues in our proofs below. Thus, in the next theorems, by a weak field which is "sufficiently regular for $G_{\rm L}$ and $G_{\rm N}$ " we mean a weak field satisfying the required additional regularity conditions.

Next, we prove the following energy comparison result in the presence of a sufficiently regular weak field, which contains Theorem 5.1.6 in the Introduction (and will be proven as in Theorem 5.4.3 above):

Theorem 5.4.7. Let $G_{\rm L} = G_{\rm L}(x,\lambda,q)$ be a function satisfying $\partial_{qq}^2 G_{\rm L}(x,\lambda,q) \ge 0$, and let $G_{\rm N} = G_{\rm N}(x,y,a,b)$ be a pairwise symmetric function satisfying $\partial_{ab}^2 G_{\rm N}(x,y,a,b) \le 0$.

Given a bounded domain $\Omega \subset \mathbb{R}^n$, let $u, v \in C(\overline{\Omega})$ such that u = v a.e. in Ω^c and $u \leq v$ in Ω . Assume that there exists $\{\varphi^t\}_{t \in [0,T]}$, a weak field for u and v (in the sense of Definition 5.4.1) which is sufficiently regular for G_L and G_N .

Then, if $\mathcal{E}_{\mathrm{T}}(u) = \mathcal{E}_{\mathrm{L}}(u) + \mathcal{E}_{\mathrm{N}}(u) < \infty$ (defined in Section 5.3), we have

$$\mathcal{E}_{\mathrm{T}}(v) \leq \mathcal{E}_{\mathrm{T}}(u) + \int_{\Omega} \int_{u(x)}^{v(x)} \mathcal{L}_{\mathrm{T}}(\varphi^{t}(x)) \big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x,$$

where $\mathcal{L}_{T} = \mathcal{L}_{L} + \mathcal{L}_{N}$ is the Euler-Lagrange operator associated to \mathcal{E}_{T} .

Proof. We consider the calibration functional constructed in Section 5.3, that is,

$$\mathcal{C}_{\mathrm{T}}(w) = \mathcal{E}_{\mathrm{T}}(v) + \int_{\Omega} \int_{v(x)}^{w(x)} \mathcal{L}_{\mathrm{T}}(\varphi^{t}) \big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x.$$

Following the strategy there, one can show, by using the ellipticity conditions, that $C_{\rm T}$ also satisfies property (C_2) in the new framework of weak fields. In particular,

$$\mathcal{E}_{\mathrm{T}}(u) \geq \mathcal{C}_{\mathrm{T}}(u) = \mathcal{E}_{\mathrm{T}}(v) + \int_{\Omega} \int_{v(x)}^{u(x)} \mathcal{L}_{\mathrm{T}}(\varphi^{t}) \big|_{t=t(x,\lambda)} \,\mathrm{d}\lambda \,\mathrm{d}x,$$

which yields the desired result.

With this result at hand, we can easily show that one sided minimizers by above are viscosity supersolutions of the Euler-Lagrange equation. Here it is clear how to adapt Definitions 5.4.4 and 5.4.5 to the case of mixed energy functionals. The following result includes Theorem 5.1.5 in the Introduction (and is proven as Theorem 5.4.6 above):

Theorem 5.4.8. Let $G_{\rm L} = G_{\rm L}(x,\lambda,q)$ be a function satisfying $\partial_{qq}^2 G_{\rm L}(x,\lambda,q) \ge 0$, and let $G_{\rm N} = G_{\rm N}(x,y,a,b)$ be a pairwise symmetric function satisfying $\partial_{ab}^2 G_{\rm N}(x,y,a,b) \le 0$. Let $\Omega \subset \mathbb{R}^n$ be a bounded domain and let u be a sufficiently regular one-sided minimizer by above of the functional $\mathcal{E}_{\rm T}$.

Then, the function u is a viscosity supersolution of the associated Euler-Lagrange equation $\mathcal{L}_{\mathrm{T}}(w) = 0$ in Ω .

Proof. Proceeding as in the proof of Theorem 5.4.6, we slide the touching function φ upwards and take the maximum with u to obtain a weak field. Applying Theorem 5.4.7, we see immediately that φ cannot be a strict subsolution, otherwise, the leaves of the weak field would have smaller energy than the minimizer.

We can finally give the proof of Theorem 5.1.5 in the Introduction:

_	í.
	L
	L
	L

Proof of Theorem 5.1.5. If u is a minimizer, then u and -u are one-sided minimizers by above of the functionals $\mathcal{E}_{N}(\cdot)$ and $\mathcal{E}_{N}(-\cdot)$, respectively. By Theorem 5.4.8, the function u is both a viscosity supersolution and subsolution. In particular, u is a viscosity solution. \Box

Remark 5.4.9. There is a more direct proof of Theorem 5.4.8 which does not use calibrations. In addition to the ellipticity $\partial_{ab}G_N(x, y, a, b) \leq 0$, one further assumes that the functions $(\lambda, q) \mapsto G_L(x, \lambda, q)$ and $(a, b) \mapsto G_N(x, y, a, b)$ are both convex. Proceeding by contradiction as above, we consider the same weak field and pick one leaf φ^{t_0} with $t_0 > 0$. By continuity and the ellipticity assumption, this function will satisfy $\mathcal{L}_T(\varphi^{t_0})(x) < 0$ whenever $\varphi^{t_0}(x) > u(x)$. Then, applying integration by parts in the local term, symmetrizing in the nonlocal one, and using the convexity assumptions, we obtain

$$0 < \int_{\Omega} (u(x) - \varphi^{t_0}(x)) \mathcal{L}_{\mathrm{T}}(\varphi^{t_0})(x) \,\mathrm{d}x$$

$$\leq \int_{\Omega} \left(G_{\mathrm{L}}(x, u(x), \nabla u(x)) - G_{\mathrm{L}}(x, \varphi^{t_0}(x), \nabla \varphi^{t_0}(x)) \right) \,\mathrm{d}x$$

$$+ \frac{1}{2} \iint_{Q(\Omega)} \left(G_{\mathrm{N}}(x, y, u(x), u(y)) - G_{\mathrm{N}}(x, y, \varphi^{t_0}(x), \varphi^{t_0}(y)) \right) \,\mathrm{d}x \,\mathrm{d}y$$

$$= \mathcal{E}_{\mathrm{T}}(u) - \mathcal{E}_{\mathrm{T}}(\varphi^{t_0}).$$

This contradicts the one-sided minimality by below.

Remark 5.4.10. The calibration approach allows us to prove that one-sided minimizers by above are viscosity supersolutions, but says nothing a priori about supersolutions that are not minimizers. Nevertheless, the strategy can be adapted to treat some of these cases. The idea consists on building an auxiliary functional for which the weak supersolution is a one-sided minimizer.

We briefly discuss the semilinear case for the sake of clarity. Let u be a weak supersolution, not necessarily a one-sided minimizer, of the equation $(-\Delta)^s v = f(v)$ in Ω , that is, u satisfies

$$\frac{c_{n,s}}{2} \iint_{Q(\Omega)} \frac{(u(x) - u(y))(\varphi(x) - \varphi(y)))}{|x - y|^{n + 2s}} \,\mathrm{d}x \,\mathrm{d}y \ge \int_{\Omega} f(u(x)) \,\varphi(x) \,\mathrm{d}x$$

for all $\varphi \in C_c^{\infty}(\Omega)$ such that $\varphi \ge 0$. Then, one can check that u is also a weak supersolution of the linear equation $(-\Delta)^s v = g$ in Ω , with g(x) := f(u(x)). In particular, u is a one-sided minimizer of the auxiliary convex energy functional

$$\widetilde{\mathcal{E}}(w) = \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|w(x) - w(y)|^2}{|x - y|^{n+2s}} - \int_{\Omega} g(x) w(x).$$

Applying Theorem 5.4.8, we deduce that u is a viscosity supersolution of the linear equation $(-\Delta)^s w = g$. By definition of g, one clearly concludes that u is a viscosity supersolution of the original semilinear equation.

Appendix G

A calibration for the extension problem of the fractional Laplacian

In this appendix we study minimizers of the energy functional $\mathcal{E}_{s,F}$ by using the extension technique for the fractional Laplacian. The strategy is based on building a calibration for an auxiliary local energy $\widetilde{\mathcal{E}}_{s,F}$ in the extended space $\mathbb{R}^{n+1}_+ = \mathbb{R}^n \times (0, \infty)$. We point out that this construction did not give us, during the conception of our work, any a priori information about the form of a calibration written "downstairs" (i.e., in \mathbb{R}^n) for the original energy functional $\mathcal{E}_{s,F}$. It was only after finding $\mathcal{C}_{s,F}$ by nonlocal arguments that we noticed how to deduce it, at least formally, from the extension problem.

We denote by $(x, z) \in \mathbb{R}^n \times \mathbb{R}$ points in \mathbb{R}^{n+1}_+ . Given a bounded domain $\Omega \subset \mathbb{R}^n$, we say that a bounded set $\widetilde{\Omega} \subset \mathbb{R}^{n+1}_+$ is an extension of Ω if $\partial_0 \widetilde{\Omega} := \partial \widetilde{\Omega} \cap \{z = 0\} \subset \Omega$. It is well known that there is a strong connection between the nonlocal energy functional $\mathcal{E}_{s,F}$ and the local one

$$\widetilde{\mathcal{E}}_{s,F}(W;\widetilde{\Omega}) := \frac{d_s}{2} \iint_{\widetilde{\Omega}} z^{1-2s} |\nabla W(x,z)|^2 \,\mathrm{d}x \,\mathrm{d}z - \int_{\partial_0 \widetilde{\Omega}} F(W(x,0)) \,\mathrm{d}x,$$

where d_s is a positive normalizing constant. For this, given a function u defined in \mathbb{R}^n we consider $U: \mathbb{R}^{n+1}_+ \to \mathbb{R}$ the solution of

$$\begin{cases} \operatorname{div}(z^{1-2s}\nabla U) = 0 & \text{in } \mathbb{R}^{n+1}_+ \\ U = u & \text{on } \partial \mathbb{R}^{n+1}_+ = \mathbb{R}^n \end{cases}$$

Here, U is the so-called *s*-harmonic extension of u. In [32, Lemma 7.2], Caffarelli, Roquejoffre, and Savin showed that u is a minimizer of $\mathcal{E}_{s,F}$ among functions with the same exterior data as u in Ω^c if and only if, for every extension domain $\widetilde{\Omega}$, the *s*-harmonic extension Uof u is a minimizer of $\widetilde{\mathcal{E}}_s(\cdot; \widetilde{\Omega})$ among functions with the same boundary condition as U on $\partial_L \widetilde{\Omega} := \partial \widetilde{\Omega} \cap \{z > 0\}.$

Taking into account this equivalence we can apply the classical theory of calibrations to the mixed Dirichlet-Neumann problem as explained in Remark 4.3.6. To do this, given a field $\{u^t\}_{t\in I}$ in \mathbb{R}^n , for some interval $I \subset \mathbb{R}$, it is clear by the maximum principle that we can define a new field $\{U^t\}_{t\in I}$ in \mathbb{R}^{n+1}_+ where each leaf U^t is the s-harmonic extension of u^t . Then, the functional

$$\widetilde{\mathcal{C}}_{s,F}(W;\widetilde{\Omega}) := d_s \iint_{\widetilde{\Omega}} z^{1-2s} \Big\{ \nabla U^t(x,z) \cdot \nabla W(x,z) - \frac{1}{2} |\nabla U^t(x,z)|^2 \Big\}_{t=t(x,z,W(x,z))} \, \mathrm{d}x \, \mathrm{d}z \\ - \int_{\partial_0 \widetilde{\Omega}} F(W(x,0)) \, \mathrm{d}x$$
(G.1)

can be proved to be a calibration for $\widetilde{\mathcal{E}}_{s,F}$ and U. Therefore, U is a minimizer of $\widetilde{\mathcal{E}}_{s,F}$, and by [32, Lemma 7.2] it follows that u is a minimizer of $\mathcal{E}_{s,F}$.

We point out that although in this way we easily found a calibration for the local energy $\widetilde{\mathcal{E}}_{s,F}(\cdot; \widetilde{\Omega})$, it was not clear at all how it translated into a calibration written "downstairs" for the original energy functional $\mathcal{E}_{s,F}$. It was only after building the calibration $\mathcal{C}_{s,F}$ by using purely nonlocal techniques that we discovered how to pass, at least formally, from $\widetilde{\mathcal{C}}_{s,F}(\cdot; \widetilde{\Omega})$ to $\mathcal{C}_{s,F}$. Let us explain this. First, as in Section 4.3, for $t_0 \in I$, we rewrite (G.1) in the alternative form¹

$$\begin{split} \widetilde{\mathcal{C}}_{s,F}(W;\widetilde{\Omega}) &= -d_s \iint_{\widetilde{\Omega}} \int_{U^{t_0}(x,z)}^{W(x,z)} \operatorname{div} \left(z^{1-2s} \nabla U^t(x,z) \right) \Big|_{t=t(x,z,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}z \, \mathrm{d}x \\ &+ \int_{\partial_0 \widetilde{\Omega}} \int_{U^{t_0}(x,0)}^{W(x,0)} \left\{ \left(-\Delta \right)^s u^t(x) - F'(u^t(x)) \right\} \Big|_{t=t(x,0,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}x \\ &+ d_s \int_{\partial_L \widetilde{\Omega}} \int_{U^{t_0}(x,z)}^{W(x,z)} z^{1-2s} \, \nu_{\partial_L \widetilde{\Omega}} \cdot \nabla U^t(x,z) \Big|_{t=t(x,z,\lambda)} \, \mathrm{d}\lambda \, \mathrm{d}\mathcal{H}^n(x,z) \\ &+ \widetilde{\mathcal{E}}_{s,F}(U^{t_0};\widetilde{\Omega}), \end{split}$$

where $\nu_{\partial_L \tilde{\Omega}}$ is the exterior normal vector to the lateral boundary $\partial_L \tilde{\Omega}$. Notice here that in the second term we have used the identity

$$\lim_{z \neq 0} \left\{ -d_s z^{1-2s} U_z^t(x,z) - F'(U^t(x,z)) \right\} = (-\Delta)^s u^t(x) - F'(u^t(x)),$$

which follows from the Caffarelli-Silvestre extension; see [33]. Finally, taking a sequence of extended domains $\widetilde{\Omega}_i$ converging to the half-space \mathbb{R}^{n+1}_+ , we recover the functional $\mathcal{C}_{s,F}$ (up to an additive constant) as the formal limit of $\widetilde{\mathcal{C}}_{s,F}(\cdot; \widetilde{\Omega}_i)$.

¹Here, the first term is the one associated to the Euler-Lagrange operator of the local energy functional and vanishes by the definition of the field U^t . On the other hand, the second and third terms are the ones involving the Neumann operator for the extended problem.

Appendix H

Other candidates for the fractional calibration

In this section we discuss three other natural candidates to be a calibration for the energy functional

$$\mathcal{E}_{s,F}(w) = \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|w(x) - w(y)|^2}{|x - y|^{n+2s}} \,\mathrm{d}x \,\mathrm{d}y - \int_{\Omega} F(w(x)) \,\mathrm{d}x$$

We will be able to discard two of them since some of the calibration properties fail in these cases. Nevertheless, there is still one candidate for which we cannot determine whether it is a calibration or not.

Let us recall that the local counterpart of $\mathcal{E}_{s,F}$ is the functional

$$\mathcal{E}_{1,F}(w) = \frac{1}{2} \int_{\Omega} |\nabla w(x)|^2 \,\mathrm{d}x - \int_{\Omega} F(w(x)) \,\mathrm{d}x,$$

which admits the calibration

$$\mathcal{C}_{1,F}(w) = \int_{\Omega} \left(\nabla u^{t}(x) \cdot \left(\nabla w(x) - \nabla u^{t}(x) \right) + \frac{1}{2} \left| \nabla u^{t}(x) \right|^{2} \right) \Big|_{t=t(x,w(x))} \, \mathrm{d}x - \int_{\Omega} F(w(x)) \, \mathrm{d}x,$$

a functional that can also be written as

$$\mathcal{C}_{1,F}(w) = \int_{\Omega} \left(\nabla u^{t}(x) \cdot \nabla w(x) - \frac{1}{2} \left| \nabla u^{t}(x) \right|^{2} \right) \Big|_{t=t(x,w(x))} \, \mathrm{d}x - \int_{\Omega} F(w(x)) \, \mathrm{d}x.$$

Inspired by the form of $\mathcal{C}_{1,F}$, the first natural calibration candidate for $\mathcal{E}_{s,F}$ can be built replacing the gradient terms by differences and double integrals. That is, we let

$$\begin{aligned} \mathcal{F}_{s,F}^{1}(w) &:= \frac{c_{n,s}}{2} \iint_{Q(\Omega)} \frac{(u^{t}(x) - u^{t}(y))(w(x) - w(y))}{|x - y|^{n + 2s}} \bigg|_{t = t(x,w(x))} \, \mathrm{d}x \, \mathrm{d}y \\ &- \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|u^{t}(x) - u^{t}(y)|^{2}}{|x - y|^{n + 2s}} \bigg|_{t = t(x,w(x))} \, \mathrm{d}x \, \mathrm{d}y - \int_{\Omega} F(w(x)) \, \mathrm{d}x. \end{aligned}$$

By using Young's inequality and the definition of the leaf-parameter function, one can directly conclude that $\mathcal{F}_{s,F}^1$ satisfies properties (C1) and (C2). It remains to check whether the null-Lagrangian property (C3) is satisfied, but we do not know how to answer this question. For the affirmative answer, the idea would be to use the usual nonlocal integration by parts technique to obtain the Euler-Lagrange equation on the leaves. However, since the leaf-parameter function t depends on the variable x, we get remainder terms that we do not know how to treat. It is then natural to look for a counterexample. We looked at cases where an explicit field is available. For the trivial potential F = 0, for which $u^t(x) = x + t$ are extremals (even if not bounded), property (C3) does not fail. Hence, this case does not discard the candidate $\mathcal{F}^1_{s,F}$. Another interesting example with explicit solutions is the Peierls-Nabarro model, corresponding to the case n = 1, s = 1/2, and $F(u) = 1 - \cos(u)$. Here the equation $(-\Delta)^{1/2}u = \sin(u)$ in \mathbb{R} admits the field of extremals $u^t(x) = 2 \arctan(x+t)$. We do not know if the null-Lagrangian property holds for $\mathcal{F}^1_{s,F}$ in this concrete example.

It is also interesting to compare $\mathcal{F}_{s,F}^1$ with the calibration $\mathcal{C}_{s,F}$ constructed in Section 4.4. There, by the alternative expression for $\mathcal{C}_{s,F}$ derived in Lemma 4.4.4, we see that $\mathcal{F}_{s,F}^1(w)$ and $\mathcal{C}_{s,F}(w)$ would coincide if the following equality were true:

$$-\lim_{\varepsilon \downarrow 0} \iint_{Q(\Omega) \setminus \{|x-y| > \varepsilon\}} \frac{\int_{t(x,w(x))}^{t(y,w(y))} (u^t(x) - u^t(y)) \partial_t u^t(y) \, \mathrm{d}t}{|x-y|^{n+2s}} \, \mathrm{d}x \, \mathrm{d}y$$
$$= \iint_{Q(\Omega)} \frac{(w(x) - u^{t(x,w(x))}(y))(u^{t(x,w(x))}(y) - w(y))}{|x-y|^{n+2s}} \, \mathrm{d}x \, \mathrm{d}y.$$

However, we do not know how to prove or disprove this identity.

The functional $\mathcal{F}_{s,F}^1$ does not capture the symmetry in the variables x and y that has appeared in the two previous works on nonlocal calibrations [18, 80]. Hence, it is also natural to propose the following new candidate:

$$\begin{aligned} \mathcal{F}_{s,F}^{2}(w) &:= \frac{c_{n,s}}{2} \iint_{Q(\Omega)} \frac{(u^{\tau}(x) - u^{t}(y))(w(x) - w(y))}{|x - y|^{n + 2s}} \bigg|_{\substack{t = t(x,w(x)) \\ \tau = t(y,w(y))}} \mathrm{d}x \,\mathrm{d}y \\ &- \frac{c_{n,s}}{4} \iint_{Q(\Omega)} \frac{|u^{r}(x) - u^{t}(y)|^{2}}{|x - y|^{n + 2s}} \bigg|_{\substack{t = t(x,w(x)) \\ \tau = t(y,w(y))}} \mathrm{d}x \,\mathrm{d}y - \int_{\Omega} F(w(x)) \,\mathrm{d}x. \end{aligned}$$

As in the preceding case, we can apply Young's inequality and the definition of the leafparameter function to deduce that $\mathcal{F}_{s,F}^2$ satisfies properties (C1) and (C2). Nevertheless, in this case we can discard it as a calibration since the null-Lagrangian property fails even when F = 0 and $u^t(x) = x + t$.

One could also think of a calibration candidate constructed by replacing the gradient terms in the local theory by fractional ones. That is,

$$\mathcal{F}^3_{s,F}(w) := \int_{\Omega} \left\{ \nabla^s u^t(x) \cdot \nabla^s w(x) \, \mathrm{d}x - \frac{1}{2} \left| \nabla^s u^t(x) \right|^2 \right\} \Big|_{t=t(x,w(x))} \, \mathrm{d}x - \int_{\Omega} F(w(x)) \, \mathrm{d}x.$$

Here, the fractional gradient is defined as

$$\nabla^s w(x) = \widetilde{c}_{n,s} \int_{\mathbb{R}^n} \frac{w(x) - w(y)}{|x - y|^{n+s}} \frac{x - y}{|x - y|} \, dy.$$

This last candidate would be motivated by the identity

$$\int_{\mathbb{R}^n} \nabla^s v(x) \cdot \nabla^s w(x) \, \mathrm{d}x = \frac{c_{n,s}}{2} \iint_{\mathbb{R}^n \times \mathbb{R}^n} \frac{(v(x) - v(y))(w(x) - w(y))}{|x - y|^{n+2s}} \, \mathrm{d}x \, \mathrm{d}y.$$

Nevertheless, a similar equality does not hold when restricting to a domain Ω , i.e.,

$$\int_{\Omega} \nabla^s v(x) \cdot \nabla^s w(x) \, \mathrm{d}x \neq \frac{c_{n,s}}{2} \iint_{Q(\Omega)} \frac{(v(x) - v(y))(w(x) - w(y))}{|x - y|^{n+2s}} \, \mathrm{d}x \, \mathrm{d}y.$$

Hence, $\mathcal{F}^3_{s,F}$ does not satisfy property (C1) and thus it is not a calibration for $\mathcal{E}_{s,F}$.

Appendix I

Minimality via a comparison principle

Here in this appendix, we explain how to prove minimality for a function embedded in a field of extremals via a strong comparison principle. The proof will require an existence and regularity theorem for minimizers.

As in Section 5.2, we let $G_N(x, y, a, b)$ be a nonlocal Lagrangian giving the energy functional

$$\mathcal{E}_{\mathrm{N}}(w) = \frac{1}{2} \iint_{Q(\Omega)} G_{\mathrm{N}}(x, y, w(x), w(y)) \,\mathrm{d}x \,\mathrm{d}y.$$

For the sake of clarity, we assume that G_N is pairwise symmetric, that is,

$$G_{\mathcal{N}}(y, x, b, a) = G_{\mathcal{N}}(x, y, a, b)$$
 for all $(x, y) \in Q(\Omega)$ and $a, b \in \mathbb{R}$,

which can always be done by the symmetry of the domain $Q(\Omega) = (\mathbb{R}^n \times \mathbb{R}^n) \setminus (\Omega^c \times \Omega^c)$, see (5.2.1). The Euler-Lagrange operator associated to \mathcal{E}_N is given in terms of the integral

$$\mathcal{L}_{\mathrm{N}}(w)(x) = \int_{\mathbb{R}^n} \partial_a G_{\mathrm{N}}(x, y, w(x), w(y)) \,\mathrm{d}y,$$

see (5.2.3).

A sufficient condition for the operator \mathcal{L}_{N} to satisfy a strong comparison principle is the strict ellipticity condition $\partial_{ab}^{2}G_{N} < 0$. Indeed, given two regular functions u, $v \colon \mathbb{R}^{n} \to \mathbb{R}$, if u touches v from below at some point x_{0} , then the monotonicity of $\partial_{a}G_{N}$ leads to the inequality $\mathcal{L}_{N}(u)(x_{0}) \geq \mathcal{L}_{N}(v)(x_{0})$. To see this, one must simply integrate $\partial_{a}G_{N}(x, y, u(x_{0}), u(y)) \geq \partial_{a}G_{N}(x, y, u(x_{0}), v(y))$ with respect to y and use that $u(x_{0}) = v(x_{0})$. Moreover, when $u \neq v$ we have the strict inequality $\mathcal{L}_{N}(u)(x_{0}) > \mathcal{L}_{N}(v)(x_{0})$.

Whenever a result on the existence of minimizers for \mathcal{E}_N is available, the comparison principle above can be used to show the minimality of solutions in a field of extremals. This method has been known for a long time for the fractional Laplacian. It does not need the construction of a calibration, but again requires an existence result which in general will not be available for nonlocal energy functionals.

To see this, let $\{u^t\}_{t\in I}$ be a field of extremals in Ω and suppose, for the sake of contradiction, that u^{t_0} with $t_0 \in I$ is not a minimizer. Let v be a minimizer of \mathcal{E}_N in the set of functions with graph $v \subset \mathcal{G} = \{(x, \lambda) \in \mathbb{R}^n \times \mathbb{R} : \lambda = u^t(x) \text{ for some } t \in I\}$ satisfying the exterior condition $v = u^{t_0}$ in Ω^c . In particular, $v \neq u^{t_0}$ and by the monotonicity with respect to the leaf-parameter, there is a first leaf u^{t_1} touching v from (say) above at an interior point $x_0 \in \Omega$. The strong comparison principle now gives $0 = \mathcal{L}_N(v)(x_0) > \mathcal{L}_N(u^{t_1})(x_0) = 0$, which is a contradiction, and thus $u^{t_0} \equiv v$. Note that this argument gives the uniqueness of the minimizer (and even of the extremal) with the given exterior condition. It is also clear that the same argument works for fields made of super and subsolutions, that is, fields such that $\mathcal{L}_N(u^t) \geq 0$ for $t \geq t_0$ and $\mathcal{L}_N(u^t) \leq 0$ for $t \leq t_0$ in Ω .

Appendix J

The calibration for the nonlocal total variation

In this appendix, we relate our functional setting to the geometric calibrations for the nonlocal perimeter appearing in the works of Cabré [18] and Pagliari [80]. This is achieved through the *nonlocal total variation*, which amounts to the integral of the nonlocal perimeters of the levels sets of a function.

Let us recall that, given an even kernel $K : \mathbb{R}^n \setminus \{0\} \to [0, +\infty)$, the K-nonlocal total variation of a function $w : \mathbb{R}^n \to \mathbb{R}$ is defined by

$$\mathcal{E}_{\mathrm{NTV}}(w) := \frac{1}{2} \iint_{Q(\Omega)} |w(x) - w(y)| K(x-y) \, \mathrm{d}x \, \mathrm{d}y,$$

where $\Omega \subset \mathbb{R}^n$ is a bounded domain. In particular, it is an energy functional of the form (5.1.6) with Lagrangian

$$G_{\rm N}(x, y, a, b) = |a - b| K(x - y),$$

which is elliptic and hence covered by our extremal field theory.

There is a strong connection between the nonlocal total variation and nonlocal minimal surfaces. Given a sufficiently regular set $E \subset \mathbb{R}^n$, its K-nonlocal perimeter is

$$\mathcal{P}_{\mathrm{N}}(E) := \frac{1}{2} \iint_{Q(\Omega)} |\mathbb{1}_{E}(x) - \mathbb{1}_{E}(y)| K(x-y) \, \mathrm{d}x \, \mathrm{d}y,$$

and E is called a K-nonlocal minimal surface if the first variation of \mathcal{P}_{N} at E vanishes. Notice that $\mathcal{P}_{N}(E) = \mathcal{E}_{NTV}(\mathbb{1}_{E})$. It is well-known that the sublevel sets of minimizers of \mathcal{E}_{NTV} are K-nonlocal minimal surfaces. Moreover, one can recover the nonlocal total variation \mathcal{E}_{NTV} of any function w in terms of the nonlocal perimeter \mathcal{P}_{N} of its sublevel sets. Namely, we have the following nonlocal coarea formula

$$\mathcal{E}_{\rm NTV}(w) = \int_{\mathbb{R}} \mathcal{P}_{\rm N}(\{w < \lambda\}) \,\mathrm{d}\lambda. \tag{J.1}$$

In [80], Pagliari studied minimality properties of the nonlocal total variation \mathcal{E}_{NTV} when acting on functions taking values in the interval [0, 1]. He showed that characteristic functions of halfspaces minimize \mathcal{E}_{NTV} among those functions by constructing a calibration. On the other hand, Cabré [18] gave a calibration (recalled in (J.2) below) for the *K*nonlocal perimeter \mathcal{P}_{N} and an arbitrary set *E* whenever it is embedded in a family of nonlocal minimal surfaces. Thus, the author extended the classical extremal field theory to nonlocal minimal surfaces. Our thesis is in the spirit of this second result but applied to the nonlocal total variation considered in [80]. In particular, Theorem 5.1.3 provides a calibration for the nonlocal total variation \mathcal{E}_{NTV} and an arbitrary function whenever it is embedded in a field of extremals.

Let us recall the calibration for the perimeter \mathcal{P}_{N} obtained in [18]. Given a smooth function $\phi \colon \mathbb{R}^{n} \to \mathbb{R}$, for each $t \in \mathbb{R}$, we consider the superlevel sets $E^{t} = \{\phi(x) > t\}$. In [18] (see also [23, Section 2]), under the assumption that E^{t} are nonlocal minimal surfaces, it was shown that the functional

$$\mathcal{C}_{\mathcal{P}_{N}}(F) = \frac{1}{2} \iint_{Q(\Omega)} \operatorname{sign} \left(\phi(x) - \phi(y)\right) \left(\mathbb{1}_{F}(x) - \mathbb{1}_{F}(y)\right) K(x - y) \, \mathrm{d}x \, \mathrm{d}y \tag{J.2}$$

is a calibration for \mathcal{P}_{N} and each $E^{t_{0}}, t_{0} \in I$.

Finally, we show that the analogue of the nonlocal coarea formula (J.1) holds for the calibration functional. Namely, the calibration for the nonlocal total variation constructed in the present paper can be written in terms of the calibration for the nonlocal perimeter of each sublevel set. We point out that all identities in Proposition J.1 continue to hold if $\{u^t\}_{t\in I}$ is simply a field in \mathbb{R}^n , that is, if the u^t are not necessarily extremals of \mathcal{E}_{NTV} .

Proposition J.1. Let $\{u^t\}_{t\in I}$ be a field of extremals for \mathcal{E}_{NTV} . Then, the associated calibration functional \mathcal{C}_{NTV} given by Theorem 5.1.3 can be written as

$$\mathcal{C}_{\rm NTV}(w) = \frac{1}{2} \iint_{Q(\Omega)} \int_{w(y)}^{w(x)} \operatorname{sign} \left(u^{t(x,\lambda)}(x) - u^{t(x,\lambda)}(y) \right) \mathrm{d}\lambda \ K(x-y) \ \mathrm{d}x \ \mathrm{d}y.$$

Moreover, the functional $C_{\rm NTV}$ can also be expressed as

$$C_{\mathrm{NTV}}(w) = \int_{\mathbb{R}} C_{\mathcal{P}_{\mathrm{N}},\lambda}(\{w < \lambda\}) \,\mathrm{d}\lambda,$$

where $C_{\mathcal{P}_{N,\lambda}}$ is the calibration for the K-nonlocal perimeter \mathcal{P}_{N} in (J.2) constructed via the foliation given by the sublevel sets $\{u^{t} < \lambda\}_{t \in I}$.

Remark J.2. Before we succeeded in constructing a calibration for general functionals (and even the quadratic one in Chapter 4), we were able to build one for the nonlocal total variation \mathcal{E}_{NTV} . For this, we considered the second identity in Proposition J.1 as our definition of the calibration. This idea was motivated by the coarea formula (J.1). It is quite remarkable that our general construction in Theorem 5.1.3 (found by completely different means) recovers this natural calibration.

Notice that by the properties of the field $\{u^t\}_{t\in I}$, for each $\lambda \in \mathbb{R}$, the level sets $\{u^t = \lambda\}_{t\in I}$ give a foliation of \mathbb{R}^n . Moreover, since each u^t is an extremal of the nonlocal total variation, the sublevel sets

$$E_{\lambda}^{t} := \{ x \in \mathbb{R}^{n} \colon u^{t}(x) < \lambda \}$$

are K-nonlocal minimal surfaces. By monotonicity, we have $u^t(x) < \lambda = u^{t(x,\lambda)}(x)$ if and only if $t < t(x,\lambda)$, and hence $E^t_{\lambda} = \{\phi^{\lambda} > t\}$ with $\phi^{\lambda}(x) := t(x,\lambda)$, consistently with the notation for $\mathcal{C}_{\mathcal{P}_N}$ in (J.2). Proof of Proposition J.1. Let $t_0 \in I$. First, letting $G_N(x, y, a, b) = |a - b| K(x - y)$ in Theorem 5.1.3, the calibration functional associated to C_{NTV} is

$$\mathcal{C}_{\mathrm{NTV}}(w) = \iint_{Q(\Omega)} \mathrm{d}x \,\mathrm{d}y \int_{u^{t_0}(x)}^{w(x)} \mathrm{d}\lambda \,\mathrm{sign}\left(u^{t(x,\lambda)}(x) - u^{t(x,\lambda)}(y)\right) K(x-y) + \mathcal{E}_{\mathrm{NTV}}(u^{t_0}).$$
(J.3)

It is easy to check that the sign term in (J.3) can be written as

$$\operatorname{sign}\left(u^{t(x,\lambda)}(x) - u^{t(x,\lambda)}(y)\right) = \operatorname{sign}\left(t(y,\lambda) - t(x,\lambda)\right)$$

for all $x, y \in \mathbb{R}^n$ and $\lambda \in \mathbb{R}$. Symmetrizing in the variables x and y, and using that $\int_{u^{t_0}(x)}^{w(x)} \cdot d\lambda - \int_{u^{t_0}(y)}^{w(y)} \cdot d\lambda = \int_{w(y)}^{w(x)} \cdot d\lambda - \int_{u^{t_0}(y)}^{u^{t_0}(x)} \cdot d\lambda$, we have

$$\iint_{Q(\Omega)} dx \, dy \int_{u^{t_0}(x)}^{w(x)} d\lambda \, \operatorname{sign} \left(t(y,\lambda) - t(x,\lambda) \right) \ K(x-y)$$

$$= \frac{1}{2} \iint_{Q(\Omega)} dx \, dy \int_{w(y)}^{w(x)} d\lambda \, \operatorname{sign} \left(t(y,\lambda) - t(x,\lambda) \right) \ K(x-y) \qquad (J.4)$$

$$- \frac{1}{2} \iint_{Q(\Omega)} dx \, dy \int_{u^{t_0}(y)}^{u^{t_0}(x)} d\lambda \, \operatorname{sign} \left(t(y,\lambda) - t(x,\lambda) \right) \ K(x-y).$$

On the other hand, by the nonlocal coarea formula (J.1) and the simple identity

$$|1_{\{u^{t_0} < \lambda\}}(x) - 1_{\{u^{t_0} < \lambda\}}(y)| = \operatorname{sign}\left(t(x,\lambda) - t(y,\lambda)\right)\left(1_{\{u^{t_0} < \lambda\}}(x) - 1_{\{u^{t_0} < \lambda\}}(y)\right),$$

it is not hard to show that

$$\mathcal{E}_{\rm NTV}(u^{t_0}) = \frac{1}{2} \iint_{Q(\Omega)} \mathrm{d}x \, \mathrm{d}y \int_{\mathbb{R}} \mathrm{d}\lambda \, |\mathbf{1}_{\{u^{t_0} < \lambda\}}(x) - \mathbf{1}_{\{u^{t_0} < \lambda\}}(y)|K(x-y) = \frac{1}{2} \iint_{Q(\Omega)} \mathrm{d}x \, \mathrm{d}y \int_{u^{t_0}(y)}^{u^{t_0}(x)} \mathrm{d}\lambda \, \mathrm{sign} \left(t(y,\lambda) - t(x,\lambda)\right) K(x-y).$$
(J.5)

Combining (J.4) and (J.5), from (J.3) we deduce

$$\begin{aligned} \mathcal{C}_{\mathrm{NTV}}(w) &= \frac{1}{2} \iint_{Q(\Omega)} \mathrm{d}x \, \mathrm{d}y \int_{w(y)}^{w(x)} \mathrm{d}\lambda \, \mathrm{sign} \left(t(y,\lambda) - t(x,\lambda) \right) \ K(x-y) \\ &= \frac{1}{2} \iint_{Q(\Omega)} \mathrm{d}x \, \mathrm{d}y \int_{w(y)}^{w(x)} \mathrm{d}\lambda \, \mathrm{sign} \left(u^{t(x,\lambda)}(x) - u^{t(x,\lambda)}(y) \right) \ K(x-y), \end{aligned}$$

which was the first claim. Moreover, this expression can also be written as

$$\mathcal{C}_{\mathrm{NTV}}(w) = \frac{1}{2} \iint_{Q(\Omega)} \mathrm{d}x \, \mathrm{d}y \int_{\mathbb{R}} \mathrm{d}\lambda \, \mathrm{sign} \left(t(x,\lambda) - t(y,\lambda) \right) \left(\mathbb{1}_{\{w < \lambda\}}(x) - \mathbb{1}_{\{w < \lambda\}}(y) \right) K(x-y).$$

Changing the order of integration, to finish the proof it remains to show that

$$\mathcal{C}_{\mathcal{P}_{N},\lambda}(\{w < \lambda\}) = \frac{1}{2} \iint_{Q(\Omega)} \operatorname{sign} \left(t(x,\lambda) - t(y,\lambda) \right) \left(\mathbb{1}_{\{w < \lambda\}}(x) - \mathbb{1}_{\{w < \lambda\}}(y) \right) K(x-y) \, \mathrm{d}x \, \mathrm{d}y,$$

but this is precisely the calibration $C_{\mathcal{P}_N}$ in (J.2) with $\phi(x) = \phi^{\lambda}(x) = t(x, \lambda)$, which yields the claim.

Bibliography

- G. Alberti, L. Ambrosio, and X. Cabré, On a long-standing conjecture of E. De Giorgi: symmetry in 3d for general nonlinearities and a local minimality property, Acta Appl. Math. 65 (2001), 9–33.
- [2] W. K. Allard, On the first variation of a varifold, Ann. of Math. 95 (1972), 417–491.
- [3] F. Angrisani, G. Ascione, C. Leone, and C. Mantegazza, Appunti di calcolo delle variazioni, Lecture notes, Dipartimento di Matematica e Applicazioni "Renato Caccioppoli" dell'Università Federico II di Napoli, 2019.
- B. Bakker and A. Scheel, Spatial Hamiltonian identities for nonlocally coupled systems, Forum Math. Sigma 6 (2018), Paper No. e22, 55.
- [5] B. Barrios and M. Medina, Equivalence of weak and viscosity solutions in fractional non-homogeneous problems, Math. Ann. 381 (2021), no. 3-4, 1979–2012.
- [6] E. N. Barron and R. Jensen, Minimizing the L[∞] norm of the gradient with an energy constraint, Comm. Partial Differential Equations 30 (2005), no. 10-12, 1741–1772.
- [7] P. Bénilan and H. Brezis, Nonlinear problems related to the Thomas-Fermi equation, J. Evol. Equ. 3 (2003), 673–770.
- [8] H. Berestycki, A. Kiselev, A. Novikov, and L. Ryzhik, *The explosion problem in a flow*, J. Anal. Math. 110 (2010), 31–65.
- [9] H. Berestycki, L. Nirenberg, and S. R. S. Varadhan, The principal eigenvalue and maximum principle for second-order elliptic operators in general domains, Comm. Pure Appl. Math. 47 (1994), 47–92.
- [10] E. Bombieri, E. De Giorgi, and E. Giusti, *Minimal cones and the Bernstein problem*, Inv. Math. 7 (1969), 243–268.
- [11] H. Brezis, Is there failure of the inverse function theorem? Morse theory, minimax theory and their applications to nonlinear differential equations, New Stud. Adv. Math. 1 (2003), 23–33.
- [12] H. Brezis, T. Cazenave, Y. Martel, and A. Ramiandrisoa, Blow up for $u_t \Delta u = g(u)$ revisited, Adv. Differential Equations 1 (1996), 73–90.
- [13] H. Brézis and T. Kato, Remarks on the Schrödinger operator with singular complex potentials, J. Math. Pures Appl. (9) 58 (1979), 137–151.
- [14] H. Brezis and J. L. Vázquez, Blow-up solutions of some nonlinear elliptic problems, Rev. Mat. Univ. Complut. Madrid 10 (1997), 443–469.
- [15] C. Bucur, A symmetry result in \mathbb{R}^2 for global minimizers of a general type of nonlocal energy, Calc. Var. Partial Differential Equations **59** (2020), Paper No. 52, 26.
- [16] X. Cabré, Regularity of minimizers of semilinear elliptic problems up to dimension 4, Comm. Pure Appl. Math. 63 (2010), 1362–1380.
- [17] _____, A new proof of the boundedness results for stable solutions to semilinear elliptic equations, Discrete Contin. Dyn. Syst. 39 (2019), 7249–7264.
- [18] _____, Calibrations and null-Lagrangians for nonlocal perimeters and an application to the viscosity theory, Ann. Mat. Pura Appl. (4) 199 (2020), 1979–1995.
- [19] _____, Estimates controlling a function by only its radial derivative and applications to stable solutions of elliptic equations, Preprint arXiv 2211.13033 (2022).

- [20] _____, A quantitative proof of the Hölder regularity of stable solutions to semilinear elliptic equations, Preprint arXiv 2205.11352 (2022).
- [21] X. Cabré and A. Capella, Regularity of radial minimizers and extremal solutions of semilinear elliptic equations, J. Funct. Anal. 238 (2006), 709–733.
- [22] X. Cabré, I. U. Erneta, and J.C. Felipe-Navarro, Null-Lagrangians and Calibrations for general nonlocal functionals and an application to the viscosity theory, in preparation.
- [23] _____, A Weierstrass extremal field theory for the fractional Laplacian, Preprint arXiv 2211.16536 (2022).
- [24] X. Cabré, A. Figalli, X. Ros-Oton, and J. Serra, Stable solutions to semilinear elliptic equations are smooth up to dimension 9, Acta Math. 224 (2020), 187–252.
- [25] X. Cabré, P. Miraglio, and M. Sanchón, Optimal regularity of stable solutions to nonlinear equations involving the p-Laplacian, Adv. Calc. Var. 15 (2022), 749–785.
- [26] X. Cabré and G. Poggesi, Stable solutions to some elliptic problems: minimal cones, the Allen-Cahn equation, and blow-up solutions, Geometry of PDEs and related problems, 2018, pp. 1–45.
- [27] X. Cabré and X. Ros-Oton, Regularity of stable solutions up to dimension 7 in domains of double revolution, Comm. Partial Differential Equations 38 (2013), 135–154.
- [28] X. Cabré and M. Sanchón, Geometric-type Sobolev inequalities and applications to the regularity of minimizers, J. Funct. Anal. 264 (2013), 303–325.
- [29] X. Cabré and T. Sanz-Perela, A universal Hölder estimate up to dimension 4 for stable solutions to half-Laplacian semilinear equations, J. Differential Equations 317 (2022), 153–195.
- [30] X. Cabré and Y. Sire, Nonlinear equations for fractional Laplacians II: Existence, uniqueness, and qualitative properties of solutions, Trans. Amer. Math. Soc. 367 (2015), 911–941.
- [31] X. Cabré and J. Solà-Morales, Layer solutions in a half-space for boundary reactions, Comm. Pure Appl. Math. 58 (2005), 1678–1732.
- [32] L. Caffarelli, J. M. Roquejoffre, and O. Savin, Nonlocal minimal surfaces, Comm. Pure Appl. Math. 63 (2010), 1111–1144.
- [33] L. Caffarelli and L. Silvestre, An extension problem related to the fractional Laplacian, Comm. Partial Differential Equations 32 (2007), 1245–1260.
- [34] _____, Regularity theory for fully nonlinear integro-differential equations, Comm. Pure Appl. Math. 62 (2009), 597–638.
- [35] J. A. Carrillo, K. Craig, and Y. Yao, Aggregation-diffusion equations: dynamics, asymptotics, and singular limits, Active particles, Vol. 2, 2019, pp. 65–108.
- [36] H. Chan and J. Wei, On De Giorgi's conjecture: recent progress and open problems, Sci. China Math. 61 (2018), 1925–1946.
- [37] T. H. Colding and W. P. Minicozzi II, A course in minimal surfaces, Graduate Studies in Mathematics, vol. 121, American Mathematical Society, Providence, RI, 2011.
- [38] F. Costa, G. F. de Souza, and M. Montenegro, Extremal solutions of strongly coupled nonlinear elliptic systems and L[∞]-boundedness, J. Math. Anal. Appl. **513** (2022), Paper No. 126225, 28.
- [39] C. Cowan, Optimal Hardy inequalities for general elliptic operators with improvements, Commun. Pure Appl. Anal. 9 (2010), 109–140.
- [40] C. Cowan and N. Ghoussoub, Regularity of the extremal solution in a MEMS model with advection, Methods Appl. Anal. 15 (2008), 355–360.
- [41] M. Cozzi and L. Lombardini, On nonlocal minimal graphs, Calc. Var. Partial Differential Equations 60 (2021), Paper No. 136, 72.
- [42] M. Cozzi and T. Passalacqua, One-dimensional solutions of non-local Allen-Cahn-type equations with rough kernels, J. Differential Equations 260 (2016), 6638–6696.
- [43] M. G. Crandall and P. H. Rabinowitz, Bifurcation, perturbation of simple eigenvalues and linearized stability, Arch. Rational Mech. Anal. 52 (1973), 161–180.

- [44] _____, Some continuation and variational methods for positive solutions of nonlinear elliptic eigenvalue problems, Arch. Rational Mech. Anal. 58 (1975), 207–218.
- [45] A. Davini, On calibrations for Lawson's cones, Rend. Sem. Mat. Univ. Padova 111 (2004), 55–70.
- [46] G. De Philippis and E. Paolini, A short proof of the minimality of Simons cone, Rend. Semin. Mat. Univ. Padova 121 (2009), 233–241.
- [47] E. Di Nezza, G. Palatucci, and E. Valdinoci, *Hitchhiker's guide to the fractional Sobolev spaces*, Bull. Sci. Math. **136** (2012), 521–573.
- [48] L. Dupaigne, Stable solutions of elliptic partial differential equations, Chapman and Hall/CRC, 2011.
- [49] P. Elbau, Sequential lower semi-continuity of non-local functionals, Preprint arXiv:1104.2686 (2011).
- [50] I. U. Erneta, Boundary Hölder continuity of stable solutions to semilinear elliptic problems in C^{1,1} domains, Preprint arXiv 2305.07062 (2023).
- [51] _____, Energy estimate up to the boundary for stable solutions to semilinear elliptic problems, Preprint arXiv 2305.07058 (2023).
- [52] _____, Stable solutions to semilinear elliptic equations for operators with variable coefficients, Commun. Pure Appl. Anal. 22 (2023), 530–571.
- [53] P. Esposito, N. Ghoussoub, and Y. Guo, Mathematical analysis of partial differential equations modeling electrostatic MEMS, Courant Lecture Notes in Mathematics, Courant Institute of Mathematical Sciences, New York; American Mathematical Society, Providence, RI, 2010.
- [54] L. C. Evans, Partial Differential Equations: Second Edition, Graduate Studies in Mathematics, AMS, 2010.
- [55] L. C. Evans and R. F. Gariepy, Measure theory and fine properties of functions, Revised, Textbooks in Mathematics, CRC Press, Boca Raton, FL, 2015.
- [56] A. Farina, Y. Sire, and E. Valdinoci, Stable solutions of elliptic equations on Riemannian manifolds, J. Geom. Anal. 23 (2013), 1158–1172.
- [57] D. A. Frank-Kamenetskii, Diffusion and heat exchange in chemical kinetics, Princeton University Press, 1955.
- [58] E. Gagliardo, Caratterizzazioni delle tracce sulla frontiera relative ad alcune classi di funzioni in n variabili, Rend. Sem. Mat. Univ. Padova 27 (1957), 284–305.
- [59] F. W. Gehring, The L^p-integrability of the partial derivatives of a quasiconformal mapping, Acta Math. 130 (1973), 265–277.
- [60] I. M. Gel'fand, Some problems in the theory of quasilinear equations, Amer. Math. Soc. Transl. (2) 29 (1963), 295–381.
- [61] M. Giaquinta and S. Hildebrandt, Calculus of variations. I. The Lagrangian formalism., Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences], vol. 310, Springer-Verlag, Berlin, 1996.
- [62] M. Giaquinta and L. Martinazzi, An introduction to the regularity theory for elliptic systems, harmonic maps and minimal graphs, Second, Appunti. Scuola Normale Superiore di Pisa (Nuova Serie) [Lecture Notes. Scuola Normale Superiore di Pisa (New Series)], vol. 11, Edizioni della Normale, Pisa, 2012.
- [63] D. Gilbarg and N. S. Trudinger, *Elliptic partial differential equations of second order*, 2nd ed., Springer Berlin, New York, 2001.
- [64] D. D. Joseph and T. S. Lundgren, Quasilinear Dirichlet problems driven by positive sources, Arch. Rational Mech. Anal. 49 (1973), 241–269.
- [65] P. Juutinen, P. Lindqvist, and J. J. Manfredi, On the equivalence of viscosity solutions and weak solutions for a quasi-linear equation, SIAM J. Math. Anal. 33 (2001), 699–717.
- [66] J. Korvenpää, T. Kuusi, and E. Lindgren, Equivalence of solutions to fractional p-Laplace type equations, J. Math. Pures Appl. (9) 132 (2019), 1–26.
- [67] S. Lang, Real and functional analysis, 3rd ed., Graduate Texts in Mathematics, Springer-Verlag, New York, 1993.

- [68] E. H. Lieb, Existence and uniqueness of the minimizing solution of Choquard's nonlinear equation, Studies in Appl. Math. 57 (1976/77), 93–105.
- [69] E. H. Lieb and M. Loss, Analysis, Second, Graduate Studies in Mathematics, vol. 14, American Mathematical Society, Providence, RI, 2001.
- [70] P.-L. Lions, Minimization problems in $L^1(\mathbf{R}^3)$, J. Functional Analysis 41 (1981), 236–275.
- [71] Z. Liu, Stability inequalities for Lawson cones, Asian J. Math. 23 (2019), no. 6, 1001–1012. MR4136487
- [72] A. Maione, D. Mugnai, and E. Vecchi, Variational methods for nonpositive mixed local-nonlocal operators, Preprint arXiv 2207.14008 (2022).
- [73] Y. Martel, Uniqueness of weak extremal solutions of nonlinear elliptic problems, Houston J. Math. 23 (1997), 161–168.
- [74] M. Medina and P. Ochoa, On viscosity and weak solutions for non-homogeneous p-Laplace equations, Adv. Nonlinear Anal. 8 (2019), 468–481.
- [75] A. Mellet and J. Vovelle, Existence and regularity of extremal solutions for a mean-curvature equation, J. Differential Equations 249 (2010), 37–75.
- [76] J. H. Michael and L. M. Simon, Sobolev and mean-value inequalities on generalized submanifolds of Rⁿ, Comm. Pure Appl. Math. 26 (1973), 361–379.
- [77] F. Morgan, Calibrations and new singularities in area-minimizing surfaces: a survey, Variational methods (Paris, 1988), 1990, pp. 329–342.
- [78] G. Nedev, Regularity of the extremal solution of semilinear elliptic equations, C. R. Acad. Sci. Paris Sér. I Math. 330 (2000), 997–1002.
- [79] _____, Extremal solutions of semilinear elliptic equations (2001). Unpublished preprint.
- [80] V. Pagliari, Halfspaces minimise nonlocal perimeter: a proof via calibrations, Ann. Mat. Pura Appl. (4) 199 (2020), 1685–1696.
- [81] F. Peng, Y. R. Zhang, and Y. Zhou, Optimal regularity and Liouville property for stable solutions to semilinear elliptic equations in \mathbb{R}^n with $n \ge 10$, Preprint arXiv 2105.02535 (2021).
- [82] _____, Interior Hölder regularity for stable solutions to semilinear elliptic equations up to dimension 5, Preprint arXiv 2204.06345 (2022).
- [83] X. Ros-Oton, Regularity for the fractional Gelfand problem up to dimension 7, J. Math. Anal. Appl. 419 (2014), 10–19.
- [84] X. Ros-Oton and J. Serra, The Dirichlet problem for the fractional Laplacian: regularity up to the boundary, J. Math. Pures Appl. 101 (2014), 275–302.
- [85] _____, The extremal solution for the fractional Laplacian, Calc. Var. Partial Differential Equations **50** (2014), 723–750.
- [86] T. Sanz-Perela, Regularity of radial stable solutions to semilinear elliptic equations for the fractional Laplacian, Commun. Pure Appl. Anal. 17 (2018), 2547–2575.
- [87] R. Servadei and E. Valdinoci, Variational methods for non-local operators of elliptic type, Discrete Contin. Dyn. Syst. 33 (2013), 2105–2137.
- [88] S. A. Silling, Reformulation of elasticity theory for discontinuities and long-range forces, J. Mech. Phys. Solids 48 (2000), 175–209.
- [89] L. Simon, Schauder estimates by scaling, Calc. Var. Partial Differential Equations 5 (1997), 391–407.
- [90] J. Simons, Minimal varieties in riemannian manifolds, Ann. of Math. (2) 88 (1968), 62–105.
- [91] P. Sternberg and K. Zumbrun, Connectivity of phase boundaries in strictly convex domains, Arch. Rational Mech. Anal. 141 (1998), 375–400.
- [92] _____, A Poincaré inequality with applications to volume-constrained area-minimizing surfaces, J. Reine Angew. Math. 503 (1998), 63–85.
- [93] X. Su, W. Valdinoci, Y. Wei, and J. Zhang, Regularity results for solutions of mixed local and nonlocal elliptic equations, Math. Z. 302 (2022), 1855–1878.

- [94] G. Talenti, Best constant in Sobolev inequality, Ann. Mat. Pura Appl. (4) 110 (1976), 353–372.
- [95] S. Villegas, Boundedness of extremal solutions in dimension 4, Adv. Math. 235 (2013), 126–133.
- [96] P. Winkert, L[∞]-estimates for nonlinear elliptic Neumann boundary value problems, NoDEA Nonlinear Differential Equations Appl. 17 (2010), 289–302.